# The Influence of Contexts on Entrepreneurial Practices: A Qualitative Study

**Tobias Pret** 

Hunter Centre For Entrepreneurship University of Strathclyde

Thesis submitted in accordance with the requirements for the Degree of Doctor of Philosophy

July 2017

## DECLARATION OF AUTHENTICITY AND AUTHOR'S RIGHTS

This thesis is the result of the author's original research. It has been composed by the author and has not been previously submitted for examination which has led to the award of a degree.

The copyright of this thesis belongs to the author under the terms of the United Kingdom Copyright Acts as qualified by University of Strathclyde Regulation 3.50. Due acknowledgement must always be made of the use of any material contained in, or derived from, this thesis.

Signed:

Date: 30<sup>th</sup> July 2017

#### ACKNOWLEDGEMENTS

Many people have provided me with invaluable support throughout the whole process that culminated in this thesis and I would like to express my sincere gratitude to them. I am indebted to my supervisors Professor Eleanor Shaw and Professor Sara Carter. Their enthusiasm, knowledge, confidence and guidance were crucial to my work. I wish also to thank Professor Sarah Drakopoulou Dodd and Professor Erik Monsen for their encouragement and advice. I am grateful to my participants for allowing me such deep insights into their lives and for trusting me to interpret and tell their stories truthfully. Above all, I would like to thank my wife and family for their unwavering support – without them I would not have been able to complete this project.

#### ABSTRACT

While it has long been recognised that access to various resources facilitates entrepreneurial processes, little is known about the influence of diverse contexts on the capital accumulation practices of entrepreneurs. Based on the lived experiences of 10 craft entrepreneurs, this thesis seeks to advance knowledge by developing a theory of practice which explains how embeddedness in multiple contexts affects entrepreneurs' resource management practices. Specifically, this study focuses on the effects of embeddedness in households and communities on the capital conversion and sharing practices of craft entrepreneurs. In doing so, it enhances understanding of entrepreneurs' social, spatial and temporal contexts.

First, this study reveals a range of hitherto unidentified conversion processes, many of which involve the accumulation of multiple forms of capital, highlighting the complex nature of capital transformations. It also identifies a range of inhibitors and facilitators of these processes, which can be conversion-specific and can derive from household and community embeddedness.

Second, the findings show that household members help entrepreneurs not only by providing free resources, but also by developing their psychological capital through boosting self-efficacy, hope, optimism and resilience. Such emotional support is found to be particularly helpful in times of anxiety and stress and reliant upon household members' understanding of the entrepreneur. However, this study also demonstrates that household members' needs, problems and opinions can inhibit entrepreneurial practices, calling attention to the multifaceted relationship between business and family.

Third, the data show the significant impact of embeddedness in communities upon entrepreneurs' actions. Participants are found to employ strategic actions to meet the expectations of community members and to conceal non-conforming traits and behaviours. Furthermore, it is revealed that community norms can lead entrepreneurs to share their various resources and engage in social entrepreneurial practices. Throughout, entrepreneurs' embeddedness within and negotiation between these social and spatial contexts is shown to be dependent upon their temporal contexts, further attesting to the considerable impacts of contexts upon entrepreneurs.

# TABLE OF CONTENTS

DECLARATION OF AUTHENTICITY AND AUTHOR'S RIGHTS	11
ACKNOWLEDGEMENTS	iii
ABSTRACT	iv
TABLE OF CONTENTS	v
LIST OF TABLES	ix
LIST OF FIGURES	x
CHAPTER 1: INTRODUCTION	1
1.1. Introduction	
1.2. Contextualising Entrepreneurship	
1.2.1. Social Contexts	
1.2.2. Spatial Contexts	4
1.2.3. Temporal Contexts	5
1.3. Definitions and Rationale	7
1.4. Research Aim	10
1.4.1. Management of Resources	11
1.4.2. Research Objectives	12
1.5. Research Approach	12
1.6. Structure of the Thesis	14
1.7. Conclusion	16
	. –
CHAPTER 2: RESEARCH CONTEXT	
2.1. Introduction	17
<ul><li>2.1. Introduction</li><li>2.2. Cultural Industries</li></ul>	17 17
<ul><li>2.1. Introduction</li><li>2.2. Cultural Industries</li></ul>	17 17 18
<ul> <li>2.1. Introduction</li></ul>	17 17 18 20
<ul> <li>2.1. Introduction</li></ul>	17 17 18 20 23
<ul> <li>2.1. Introduction</li> <li>2.2. Cultural Industries</li> <li>2.2.1. Cultural Industries Research</li> <li>2.2.2. Creative Communities</li> <li>2.2.3. Craft Field</li> <li>2.3. Craft Entrepreneurs in this Study</li> </ul>	17 17 18 20 23 26
<ul> <li>2.1. Introduction</li> <li>2.2. Cultural Industries</li> <li>2.2.1. Cultural Industries Research</li> <li>2.2.2. Creative Communities</li> <li>2.3. Craft Field</li> <li>2.3. Craft Entrepreneurs in this Study</li> <li>2.4. Rural Entrepreneurship</li> </ul>	17 17 18 20 23 26 29
<ul> <li>2.1. Introduction</li> <li>2.2. Cultural Industries</li> <li>2.2.1. Cultural Industries Research</li> <li>2.2.2. Creative Communities</li> <li>2.2.3. Craft Field</li> <li>2.3. Craft Entrepreneurs in this Study</li> <li>2.4. Rural Entrepreneurship</li> <li>2.4.1. Policy Perspective</li> </ul>	17 17 18 20 23 26 29 30
<ul> <li>2.1. Introduction</li> <li>2.2. Cultural Industries</li> <li>2.2.1. Cultural Industries Research</li> <li>2.2.2. Creative Communities</li> <li>2.2.3. Craft Field</li> <li>2.3. Craft Entrepreneurs in this Study</li> <li>2.4. Rural Entrepreneurship</li> <li>2.4.1. Policy Perspective</li> <li>2.4.2. Academic Perspective</li> </ul>	17 17 18 20 23 26 29 30 32
<ul> <li>2.1. Introduction</li> <li>2.2. Cultural Industries</li> <li>2.2.1. Cultural Industries Research</li> <li>2.2.2. Creative Communities</li> <li>2.2.3. Craft Field</li> <li>2.3. Craft Entrepreneurs in this Study</li> <li>2.4. Rural Entrepreneurship</li> <li>2.4.1. Policy Perspective</li> <li>2.4.2. Academic Perspective</li> <li>2.5. Rural Locations in this Study</li> </ul>	17 17 18 20 23 26 29 30 32 35
<ul> <li>2.1. Introduction</li> <li>2.2. Cultural Industries</li> <li>2.2.1. Cultural Industries Research</li> <li>2.2.2. Creative Communities</li> <li>2.2.3. Craft Field</li> <li>2.3. Craft Entrepreneurs in this Study</li> <li>2.4. Rural Entrepreneurship</li> <li>2.4.1. Policy Perspective</li> <li>2.4.2. Academic Perspective</li> </ul>	17 17 18 20 23 26 29 30 32 35
<ul> <li>2.1. Introduction</li> <li>2.2. Cultural Industries</li> <li>2.2.1. Cultural Industries Research</li> <li>2.2.2. Creative Communities</li> <li>2.2.3. Craft Field</li> <li>2.3. Craft Entrepreneurs in this Study</li> <li>2.4. Rural Entrepreneurship</li> <li>2.4.1. Policy Perspective</li> <li>2.4.2. Academic Perspective</li> <li>2.5. Rural Locations in this Study</li> <li>2.5.1. Artisan Village (Scotland)</li> <li>2.5.2. Craft County (England)</li> </ul>	17 17 18 20 23 26 29 30 35 37 38
<ul> <li>2.1. Introduction</li> <li>2.2. Cultural Industries</li> <li>2.2.1. Cultural Industries Research</li> <li>2.2.2. Creative Communities</li> <li>2.2.3. Craft Field</li> <li>2.3. Craft Entrepreneurs in this Study</li> <li>2.4. Rural Entrepreneurship</li> <li>2.4.1. Policy Perspective</li> <li>2.4.2. Academic Perspective</li> <li>2.5. Rural Locations in this Study</li> <li>2.5.1. Artisan Village (Scotland)</li> </ul>	17 17 18 20 23 26 29 30 35 37 38
<ul> <li>2.1. Introduction</li></ul>	17 17 18 20 23 26 29 30 32 35 37 38 38 38
<ul> <li>2.1. Introduction</li></ul>	17 17 18 20 23 26 26 29 30 30 32 35 37 38 38 38
<ul> <li>2.1. Introduction</li></ul>	17 17 18 20 23 26 29 30 35 35 37 38 38 38 40 40
<ul> <li>2.1. Introduction</li></ul>	17 17 18 20 23 26 29 30 32 35 37 38 38 38 40 41

	. –
3.4. Forms of Capital	
3.4.1. Economic Capital	
3.4.2. Cultural Capital	
3.4.3. Social Capital	
3.4.4. Symbolic Capital	
3.4.5. Critique of Capital Theory	
3.5. Habitus	
3.5.1. Illusio	
3.5.2. Doxa	
3.5.3. Critique of Habitus	
3.6. Fields	
3.6.1. Bourdieu and Families	
3.6.2. Critique of Field Theory	
3.7. Conclusion	64
CHAPTER 4: ENTREPRENEURSHIP RESEARCH ON RESOURCES,	
HOUSEHOLDS AND COMMUNITIES	
4.1. Introduction	
4.2. Resource and Capital Management	
4.2.1. The RBV	
4.2.2. Economic Capital and its Conversions	
4.2.3. Cultural Capital and its Conversions	
4.2.4. Social Capital and its Conversions	
4.2.5. Symbolic Capital and its Conversions	
4.3. Households and Families	
4.3.1. Household Context	
4.3.2. Resources of the Household	
4.3.3. The Role of Emotions	
4.4. Embeddedness in Communities	
4.4.1. Embeddedness Theory	
4.4.2. Communities and Entrepreneurship	
4.4.3. Social Value Creation	
4.5. Conclusion	117
CHAPTER 5: RESEARCH METHODS	
5.1. Introduction	
5.2. Philosophical Assumptions	
5.2.1. Philosophical Research Paradigms	
5.2.2. Phenomenological Approaches	
5.2.3. Paradigmatic Concerns	
5.3. Research Aim and Objectives	
5.4. Research Design	132

5.4.1. Methodological Concerns	
5.4.2. Interpretative Phenomenological Analysis	
5.4.3. Participant Selection	
5.5. Data Collection and Analysis	
5.5.1. Data Collection Approach	
5.5.3. Ethical Concerns	
5.5.3. Data Analysis and Interpretation	145
5.6. Methodological Reflections	
5.7. Conclusion	158
	1 1 7 0
CHAPTER 6: A THEORY OF PRACTICE FOR CRAFT ENTREPRENEURS	
6.1. Introduction	
6.2. Conceptual Model	
6.2.1. Aggregate Stock of Capital	
6.2.2. Inhibitors and Facilitators of Capital Conversion	
6.2.3. Competing Habitus	
6.2.4. Temporal Context	
6.3. Relationship to Bourdieu's Theory of Practice	
6.4. Conclusion	168
CHAPTER 7: CAPITAL CONVERSIONS - MICRO-LEVEL	170
7.1. Introduction	
7.2. Economic Capital Conversions	
7.3. Cultural Capital Conversions	
7.4. Social Capital Conversions	
7.5. Symbolic Capital Conversions	
7.6. Analysis and Interpretation	
7.6.1. The Convertibility of Economic Capital	
7.6.2. The Convertibility of Cultural Capital	
7.6.3. The Convertibility of Social Capital	
7.6.4. The Convertibility of Symbolic Capital	
7.7. Conclusion	
CHAPTER 8: HOUSEHOLD CONTEXT - MESO-LEVEL	202
8.1. Introduction	202
8.2. Household Characteristics	203
8.3. Provision of Capital	207
8.4. Emotional Support	213
8.5. Impediments	218
8.6. Analysis and Interpretation	
8.6.1. Exploring Entrepreneurial Households	223
8.6.2. Extending the Concept of Household Capital	225

8.6.3. Reliance on Psychological Capital	228
8.6.4. Work-Family Conflict	
8.7. Conclusion	
CHAPTER 9: COMMUNITY CONTEXT - MACRO-LEVEL	
9.1. Introduction	
9.2. Embeddedness in Communities	
9.3. The Importance of Fitting-in	
9.4. Social Value Creation	
9.5. Analysis and Interpretation	250
9.5.1. Craft Fields	250
9.5.2. Habitus of Craft Communities	253
9.5.3. Capital Sharing	256
9.5.4. Craft Entrepreneurs as Social Entrepreneurs	258
9.6. Conclusion	259
CHAPTER 10: CONCLUSION	261
10.1. Introduction	
10.2. Key Findings	
10.3. Limitations	
10.4. Theoretical Implications	270
10.5. Implications for Practice and Policy	273
10.6. Contributions of the Research	
REFERENCES	270
NEFENCES	
APPENDIX 1. Interview Topic Guides	328
<b>APPENDIX 2.</b> Interpretive Summary for Interview 1 with Julie	

# LIST OF TABLES

<b>Table 2.1.</b> Research into Creative Communities (excluding Crafts)	
Table 2.2. Craft Research	
<b>Table 2.3.</b> Craft Entrepreneurs in this Study.	27
Table 2.4. Rural Entrepreneurship Research	
Table 3.1. Research utilising Bourdieu's Theory of Practice	
Table 3.2. Research utilising Bourdieu's Master Theories	
Table 3.3. Summary of Bourdieu's Main Concepts	65
Table 5.1. Subjective-Objective Dimension of the Paradigmatic Framework	122
Table 5.2. Phenomenological and Interpretivist Entrepreneurship Research	126
Table 5.3. Overview of Research Participants	139
Table 5.4. Overview of Interviews	141
Table 5.5. Data Analysis Process.	146
Table 5.6. Data Coding Example.	151
Table 5.7. Conference Papers and Journal Articles	152
Table 7.1. Conversions of Economic Capital.	171
Table 7.2. Inhibitors to Economic Capital Conversions	172
Table 7.3. Conversions of Embodied Cultural Capital	173
<b>Table 7.4.</b> Inhibitors to Embodied Cultural Capital Conversions	
Table 7.5. Conversions of Objectified Cultural Capital	
Table 7.6. Inhibitors to Objectified Cultural Capital Conversions.	176
Table 7.7. Conversions of Institutionalised Cultural Capital	177
Table 7.8. Inhibitors to Institutionalised Cultural Capital Conversions	177
Table 7.9. Conversions of Social Capital: Business Networks	178
Table 7.10. Inhibitors to Social Capital Conversions: Business Networks	
<b>Table 7.11.</b> Conversions of Social Capital: Followers.	
<b>Table 7.12.</b> Inhibitors to Social Capital Conversions: Followers	181
Table 7.13. Conversions of Social Capital: Suppliers	
Table 7.14. Conversions of Social Capital: Friends	
Table 7.15. Conversions of Symbolic Capital: Awards	184
<b>Table 7.16.</b> Inhibitors to Symbolic Capital Conversions: Awards	
<b>Table 7.17.</b> Conversions of Symbolic Capital: Exhibitions	
<b>Table 7.18.</b> Inhibitors to Symbolic Capital Conversions: Exhibitions	
Table 7.19. Conversions of Symbolic Capital: Publicity	
<b>Table 7.20.</b> Inhibitors to Symbolic Capital Conversions: Publicity	
<b>Table 7.21.</b> Conversions of Symbolic Capital: Reputation	
Table 7.22. Inhibitors to Symbolic Capital Conversions: Reputation	190

Table 8.1. Cohabitants' Involvement in the Craft Field and Perceived Creativity.	205
Table 8.2. Economic Capital Support.	207
Table 8.3. Cultural Capital Support.	209
Table 8.4. Social Capital Support.	211
Table 8.5. Symbolic Capital Support.	212
<b>Table 8.6.</b> Emotional and Moral Support.	214
Table 8.7. Needs of Household Members.	219
<b>Table 8.8.</b> Problems of Household and Family	220
<b>Table 8.9.</b> Opinions of Family Members	221
<b>Table 9.1.</b> Implications of Embeddedness in Virtual Communities	238
Table 9.2 Craft Community Conventions	240
Table 9.3 Strategic Changes of Practice	243
<b>Table 9.4</b> Motivations for Capital Sharing and Collaboration	246
<b>Table 9.5</b> Reasons for Indifference Concerning Plagiarism	247

# LIST OF FIGURES

Figure 2.1. Cultural Industries Context (for Holly)	18
Figure 2.2. Rural Regions in this Study	
Figure 3.1. Bourdieu's Theory of Practice	40
Figure 3.2. Forms of Capital	47
Figure 4.1. Overview of Relevant Entrepreneurship Theories	118
Figure 5.1. Burrell and Morgan's (1979) Paradigmatic Framework	121
Figure 5.2. Coding Tree for Symbolic Capital	151
Figure 6.1. Theory of Practice for Craft Entrepreneurs	160
Figure 8.1. Household Members of Entrepreneurs in Cultural Industries	224

#### **CHAPTER 1: INTRODUCTION**

#### **1.1. Introduction**

Context matters – it has been shown to both facilitate and restrict entrepreneurial actions (Aldrich and Fiol, 1994; Navis and Ozbek, 2016; Rooks et al., 2016). Accordingly, a growing body of research investigates the influence of embeddedness in different contexts on entrepreneurs (Jack and Anderson, 2002; Korsgaard et al., 2015; McKeever et al., 2015). Despite this increasing scholarly attention, the 'multiplicity of contexts' (Welter, 2011: 168) entails that there remain enduring gaps in understanding of the impacts of diverse contexts on entrepreneurial practices (Wright et al., 2014; Zahra et al., 2014).

Based on the lived experiences of craft entrepreneurs, this study seeks to advance knowledge by developing a theory of practice which explains how embeddedness in multiple contexts affects entrepreneurs' resource management practices. This study provides fine-grained, contextualised accounts of the behaviours of participants to create 'local knowledge' (Steyaert, 1997) and to develop theory inductively, from the 'bottom-up' (Cope, 2005). Thus, the themes discussed in this study, as well as its overall focus, are based on the analysis of what mattered most to participants (Cope, 2011). Specifically, this study focuses on the effects of embeddedness in household and community contexts (Alsos et al., 2014; Martí et al., 2013) on the capital conversion and sharing practices of craft entrepreneurs (Scott, 2012). It finds and argues that entrepreneurs' resource management practices cannot be understood outside of their social, spatial and temporal contexts. Bourdieu's (1977, 1984, 1986) theory of practice was selected as a conceptual lens to frame the data analysis and to interpret the experiences of participants. In adopting a practice-based perspective (Lockett et al., 2014; Terjesen and Elam, 2009), this study seeks to generate practical advice that will be useful for entrepreneurs, scholars, educators and policy makers (Kenworthy and McMullan, 2013).

#### **1.2.** Contextualising Entrepreneurship

Entrepreneurship research has been criticised for 'a scholarly fixation with the entrepreneurial individual' (Chalmers and Shaw, 2017: 21), which has resulted in 'efforts to overgeneralize results across very heterogeneous settings within and across studies' (Wiklund et al., 2011: 4). Consequently, there has been a growing recognition that entrepreneurs need to be studied within their contexts to fully understand their behaviours (Autio et al., 2014; McKeever et al., 2014). Indeed, Welter and Gartner (2016: 156) convincingly argue that

it is not the context, but *contexts* that matter for entrepreneurship ... with the plural of the word, *contexts*, we fully acknowledge the diversity, heterogeneity and multiplicity required to adequately contextualize.

Building on their argument, this study recognises that entrepreneurs are not embedded in a single, homogeneous context, but in multiple, co-existing ones. Extant research demonstrates that embeddedness in social contexts can provide entrepreneurs with opportunities, including access to resources and competitive advantage (Batjargal, 2003; Hite, 2005). Embeddedness has also been shown to restrict entrepreneurs, such as when social or moral obligations negatively affect business relationships or operations (Coleman, 1988; Jack and Anderson, 2002). Prior research has made progress in exploring the impact of certain social (Granovetter, 1985), institutional (Aldrich and Fiol, 1994), spatial (Stevaert and Katz, 2004), industrial contexts (Reuber and Fischer, 2005) and temporal contexts (Lippmann and Aldrich, 2016b) on entrepreneurs. However, despite continued calls for research to further investigate the significance of context in entrepreneurship (Gaddefors and Anderson, 2017; Johannisson et al., 2002; Low and MacMillan, 1988), there remain enduring gaps in understanding concerning the effects of different contexts on entrepreneurial practices (Wright et al., 2014; Zahra et al., 2014). Welter (2011) provides a detailed analysis of developments in this domain and points towards opportunities for future research. This study seeks to follow her recommendation that research should advance understanding of the impacts of social, spatial and temporal contexts on entrepreneurs.

#### 1.2.1. Social Contexts

Much existing research into social contexts employs a network perspective (Hallen and Eisenhardt, 2012; Phillips et al., 2013), which conceives of society as consisting of fluctuating networks of social relationships (Vardaman et al., 2012; Wong and Boh, 2010). As there are several excellent reviews of network-related entrepreneurship studies (cf. Jack, 2010; Slotte-Kock and Coviello, 2010), this section undertakes only a summary of the main arguments in the literature. Among the most frequently studied is the strength of ties connecting network members (Granovetter, 1973). It has traditionally been argued that a central location within a network comprising mainly 'weak ties' (superficial or casual relationships that require limited emotional investment, see Dubini and Aldrich, 1991) is beneficial for entrepreneurs (Vardaman et al., 2012), as it provides access to a wide range of resources (Aldrich and Zimmer, 1986; Birley, 1985). However, more recent studies have challenged this assumption (Antcliff et al., 2007; Patel and Terjesen, 2011), instead demonstrating the importance of 'strong ties' (relationships that entrepreneurs can rely upon, such as family members) for accessing knowledge, maintaining reputation and, thus, facilitating entrepreneurial success (Jack, 2005). There are also ongoing discussions regarding the effects of network density and structural holes (defined as gaps in a social network that are spanned by nonredundant contacts, see Burt, 1992). It has been suggested that effective networks have various structural holes, linking entrepreneurs to many non-overlapping clusters of people (Steier and Greenwood, 2000). Other studies have shown that networks change over time and that different configurations are beneficial at different stages of venture development (Ahuja, 2000; Hite and Hesterly, 2001).

A major shortcoming of extant research into social contexts is its narrow focus on network structures, which results from the overuse of quantitative research designs (i.e. measuring network density, size and activity levels) (Hoang and Antoncic, 2003; Slotte-Kock and Coviello, 2010). Consequently, comparatively little is known about what actually occurs between partners in networks and other social contexts (Jack, 2010). For example, the 'dark side' of embeddedness (i.e. the negative impact of social obligations on economic activities, see Uzzi, 1997) has received scant attention, as studies tend to highlight the positive effects of embedding processes (Zahra et al., 2014). Furthermore, understanding of social contexts outside the business world (Welter, 2011), such as household and family contexts (Carter, 2011; Gras and Nason, 2015), is limited. Aldrich and Cliff (2003) present a convincing argument for adopting a family embeddedness perspective, arguing that family members can affect a range of entrepreneurial processes, including opportunity recognition, venture development and resource management. Alsos et al. (2014) propose that a focus on the immediate family unit (i.e. the household) is particularly revealing, as business and private spheres are often blurred for entrepreneurs. Thus, by exploring the impact of household members on the practices of entrepreneurs, this study makes valuable contributions to knowledge.

#### 1.2.2. Spatial Contexts

In addition to being socially bound, entrepreneurship occurs within spatial contexts (Johannisson et al., 2002). As Zahra et al. (2014: 491) argue, location influences the abilities of entrepreneurs to develop firms, assemble resources and establish links with stakeholders and other partners. Prior studies have shown that certain areas provide entrepreneurs with a range of advantages, such as a rich resource base or a well-crafted business support policy (Parrilli, 2009; Smallbone et al., 2003). In contrast, other regions are depleted or deprived and, as failing economic spaces, are often unable to provide sufficient facilities or resources to promote entrepreneurial efforts (Lee et al., 2011; Parkinson et al., 2017). However, while geography can exacerbate the effects of other factors, it is not usually the sole cause for variations in entrepreneurial activity, as demonstrated by studies in underdeveloped regions (Imas et al., 2012; Mair et al., 2012) and rural areas (Jack and Anderson, 2002; Smallbone et al., 1999). Furthermore, prior research reveals that embeddedness in a region and the resulting 'intimate knowledge of and concern for the place' (Korsgaard et al., 2015: 574) can facilitate entrepreneurship by providing insider information about physical and cultural peculiarities.

Links with local communities appear to particularly affect the practices of entrepreneurs and lead to mutual support activities (Lang et al., 2014; McKeever et al., 2015). Indeed, it has been shown that entire communities can become involved in entrepreneurial efforts (Johannisson, 1990; Johnstone and Lionais, 2004), which can

result in a distinct form of entrepreneurship committed to achieving social change and community development (Haugh, 2007; Vestrum et al., In Press). Consequently, there has been growing interest in investigating the impact of embeddedness in communities on entrepreneurs (Hindle, 2010; Ring et al., 2010), especially in regional clusters (Lechner and Leyronas, 2009; Tan et al., 2013) and local communities (Korsgaard et al., 2015; Parkinson et al., 2017). While communities are typically understood as geographically bound groups of actors (Jennings et al., 2013), Welter (2011) emphasises that they are a result of the convergence of both social and spatial contexts. Departing even further from the traditional understanding, alternative definitions allow for communities to be bound solely by characteristics other than geographic proximity (O'Mahony and Bechky, 2008; von Krogh et al., 2003), such as shared interests (e.g. a love for crafts, see Biraglia and Kadile, 2017). Such communities are growing more prevalent as a result of virtual connections (Reuber and Fischer, 2011; Smith et al., 2017) and human migration (Jones et al., 2014; Patel and Conklin, 2009), and yet virtual and transnational communities have attracted little scholarly attention (Drori et al., 2009b; O'Mahoney, 2007). Accordingly, this study examines not only local communities, but also focuses on the creative communities in which participants are embedded and which connect craft enthusiasts from all over the world. In so doing, this study advances understanding of the effects of embeddedness in communities on entrepreneurial practices.

#### 1.2.3. Temporal Contexts

As well as being socially and spatially embedded, entrepreneurs' actions are conducted within temporal contexts. Time is often recognised as a significant factor impacting entrepreneurs and yet there is a dearth of research into the temporal context of entrepreneurship (Zahra et al., 2014). More importantly, it has been suggested that the small number of dedicated studies on the subject lack direction and cohesion (Welter, 2011), as well as depth of conceptualisation (Lippmann and Aldrich, 2016b).

On one end of the spectrum, time is conceived as fixed and constant. In its simplest form, it is merely considered a tool for measuring periods over which entrepreneurial activities are performed, including entry and exit (Kalnins and Williams, 2014; Naldi and Davidsson, 2014). In other instances, it is further theorised, but still appreciated as immutable and objective: events transpire in the past and constrain future opportunities in perpetuity. These interpretations are most clearly articulated in the life cycle approach (Carter et al., 1996) and the theory of path dependency (Zahra et al., 2006). While studies that focus on the organisational life cycle account for major shifts in a venture's evolution over time (Gersick, 1994; Vohora et al., 2004) and thus accomplish some degree of temporal contextualisation, their conception of temporal context suffers from teleological assumptions about entrepreneurial processes and goals (Lippmann and Aldrich, 2016b). Similarly, though proponents of path dependency ground entrepreneurial decision-making in the context of a firm's past choices (Zahra et al., 2006), they insist upon an overly deterministic understanding of time and history (Dacin et al. 1999), which leaves little room for the dynamic interplay between temporal, social, spatial and other contexts in entrepreneurial activities and decision-making.

In stark contrast with these views, there has been a recent trend towards understanding time as socially constructed, nonlinear and malleable (Garud et al., 2014; Lippmann and Aldrich, 2016b). In this way, time is understood in a historical rather than scientific sense: the past does not exist outside of its interpretation and these interpretations themselves are perpetually changing (Suddaby et al., 2010). Within the context of entrepreneurship, this conception of time enables researchers to perceive how entrepreneurs continuously shift their temporal foci in line with their present activities and reinterpret the past, present and future according to their varying aspirations. It also provides insights into how entrepreneurs learn from the past and manipulate their histories to assert legitimacy (Lippmann and Aldrich, 2016b). In other words, the temporal context is shown to serve not only as a scholarly analytical tool, but as one which entrepreneurs can exploit to gain competitive advantage (Suddaby et al., 2010). For instance, entrepreneurs can draw upon their personal and collective traditions, knowledge and experiences to develop new innovations (De Massis et al., 2016). When the plasticity of time and history is thus recognised, the extent to which this context both impacts entrepreneurial decision-making and is impacted by the actions of entrepreneurs can be appreciated. By acknowledging the significance of temporal embeddedness (Dacin et al., 1999;

Hess 2004) in shaping entrepreneurial actions, this study gives further weight to recent calls for entrepreneurship scholars to take temporal contexts into account within their research (Welter, 2011; Zahra et al., 2014).

#### **1.3. Definitions and Rationale**

This study explores entrepreneurship in the cultural industries, requiring several underlying definitions to be considered. This section first defines entrepreneurship and explains the value in studying everyday entrepreneurs. It then clarifies the concept of cultural entrepreneurship, before defining craft entrepreneurs. In addition, the rationale for exploring this subject is provided.

This study does not focus on 'gazelles' or 'unicorns' (i.e. start-ups that achieve exceptionally high growth or value) (Douglas, 2013), but rather on 'everyday entrepreneurs' (Welter et al., 2017). Interpretations of entrepreneurs are diverse, as the study of entrepreneurship cuts across a range of academic disciplines (Davidsson et al., 2001; Low and MacMillan, 1988). Following Stam et al. (2014: 154), this study defines 'an entrepreneur as the founder, owner, and manager of a small firm'. Prior research highlights various capabilities and activities involved in entrepreneurship, such as dealing with uncertainty (Knight, 1921), innovatively combining resources to produce new goods (Schumpeter, 1942), discovering opportunities (Kirzner, 1973) and creating new ventures (Gartner, 1985). Accordingly, Shane and Venkataraman (2000: 218) define the field of entrepreneurship research broadly as

the scholarly examination of how, by whom, and with what effects opportunities to create future goods and services are discovered, evaluated, and exploited ... Consequently, the field involves the study of sources of opportunity; the processes of discovery, evaluation and exploitation of opportunities; and sets of individuals who discover, evaluate and exploit them.

While entrepreneurship has evolved into an established and vibrant field of research (Shane, 2012; Shepherd, 2015; Venkataraman et al., 2012), existing studies have been criticised for various reasons (Davidsson, 2015; Sorenson and Stuart,

2008; Wiklund et al., 2011), including their overwhelming focus on ventures that generate financial wealth (Rindova et al., 2009), are owned by men (Ahl, 2006) and are backed by venture capital (Winborg and Landström, 2001). Thus, there have been repeated calls for research to explore more 'ordinary people who want to become entrepreneurs' (Sarasvathy et al., 2015: 227). Correspondingly, Welter et al. (2017) present a compelling argument for embracing entrepreneurial diversity, rather than confining scholarly attention to technology-oriented, high-growth ventures. This study heeds this advice and investigates craft entrepreneurs, who operate low-tech, slow-growth businesses and who seek to create not only financial, but also cultural and social wealth.

Research into craft entrepreneurs is situated within the field of cultural entrepreneurship (Jones et al., 2016; Swedberg, 2006). According to Giorgi et al. (2015), the study of culture and its role in business activities continues to gain popularity, resulting in a rich variety of theoretical perspectives. Gehman and Soublière (2017) provide a detailed review of research progress in cultural entrepreneurship and distinguish three streams of research that approach and define the subject in different ways. Rather than conceptualising cultural entrepreneurship as 'a process of storytelling that ... facilitate[s] the crafting of a new venture identity' (Lounsbury and Glynn, 2001: 545) or as 'a process of continual framing and reframing ... future aspirations and past events' (Garud et al., 2014: 1489), this study adopts Swedberg's (2006) perspective. Thus, cultural entrepreneurship is defined here as 'the carrying out of a novel combination that results in something new and appreciated in the cultural sphere' (p. 260).

This 'making culture' approach originates in DiMaggio's (1982) work, which focuses on the production and distribution of cultural products. Cultural products are defined as goods 'directed at a public of consumers, for whom they generally serve an esthetic or expressive, rather than a clearly utilitarian function' (Hirsch, 1972: 641-642). A number of studies have adopted this approach, including Scott's (2012: 241) analysis of the engagement of music producers in 'entrepreneurship sans economic capital' and Johnson's (2007) investigation of the influence of entrepreneurs on organisational imprinting in the cultural industries. These studies are connected by 'a view of culture as a sector or a set of industries, and a focus on entrepreneurship taking place within this domain' (Gehman and Soublière, 2017: 64). This study joins them in adopting this stance.

Within the cultural industries, this study explores entrepreneurship in the craft field. According to Tregear (2005: 2), craft makers 'possess certain technical skills derived from experience or apprenticeship that set them apart from other types of manual worker[s]'. For the purposes of this study, craft makers are defined as individuals who produce and sell products or services which possess a distinct artistic value resulting from a high degree of manual input (Fillis, 2004). This conceptualisation accounts for a broad range of makers, from 'hobbyists' (Biraglia and Kadile, 2017), 'whose household income is only partially supported by selling craft work' (McAuley and Fillis, 2005: 503), to full-time producers, who craft products of consistently high quality, allowing them to establish a reputation for their work. Thus, Fillis (2004) proposes a typology to differentiate between four types of craft makers. First, 'lifestylers' value the quality of life associated with this profession, do not care about business growth and are unwilling to take many risks. Second, 'idealists' primarily view themselves as artists, who take risks with their products and value artistic integrity over meeting customer demands. Third, 'late developers' tend to come from non-creative backgrounds, which provide them with outside skills, and are less likely to accept new ideas or expand their businesses. Finally, 'craft entrepreneurs' are willing to take risks with their businesses, realise the importance of developing customer relationships, are willing to embrace a business philosophy and exhibit ambitions for growth.<sup>1</sup> Consequently, this study drew on the typology of Fillis (2004) to identify and select participants who could be described not only as craft makers, but also as craft entrepreneurs.

There are several reasons why this study investigates craft entrepreneurs. First, craft entrepreneurs are under-studied (Bertacchini and Borrione, 2011; Fillis, 2004), although they have been shown to find creative ways to identify and exploit opportunities (Bhagavatula et al., 2010; Kuhn and Galloway, 2015). This omission leaves a gap in understanding of a sector that is steadily growing and makes a valuable contribution to the UK economy (Schwarz and Yair, 2010). Burns et al. (2012) estimate that there are over 23,000 craft-making businesses in the UK, which

<sup>&</sup>lt;sup>1</sup> This study conceives of the concept of growth as incorporating economic, social and community growth (Korsgaard and Anderson, 2011; Zahra and Wright, 2016).

generate a turnover of around £1.9 billion and employ more than 43,000 people (Bennett et al., 2014). Second, as Bourdieu (1986: 242) argues, human behaviour cannot be fully understood if research focuses solely on examining economic motives; it also needs to account for 'activities of art-for-art's sake'. In addition to crafting products that are both commercially valuable and true to their artistic selfexpression (Eikhof and Haunschild, 2007), craft entrepreneurs have been shown to create social value through work in the community, such as by offering classes and by fostering inclusion and social interaction through shared activities (Schwarz and Yair, 2010). Investigating craft entrepreneurs, who are driven by a number of divergent motivations, including 'goals of lifestyle, community involvement and commercial success' (Tregear, 2005: 11), thus presents an apposite avenue of inquiry for this study. Third, the researcher previously worked as a craft professional, which sparked his interest in exploring this sector and which facilitated building rapport with participants. Gaining the deep insights that result from participants granting open access is of major importance for inductive studies that search 'for the meanings which lie behind actions' (Jack and Anderson, 2002: 473).

#### 1.4. Research Aim

This research set out to investigate the practices of craft entrepreneurs. As understanding of this entrepreneurial sub-culture is limited (Bhagavatula et al., 2010; Kuhn and Galloway, 2015), this study sought to generate new knowledge about the phenomenon. Given this exploratory aim, qualitative methods were deemed appropriate (Edmondson and McManus, 2007; McKeever et al., 2015; Pratt, 2009). More specifically, a phenomenological approach was chosen to gain deep insights into the lives of participating entrepreneurs (Berglund, 2007; Cope, 2005). Accordingly, rather than trying to (dis-)prove hypotheses derived from a review of the literature (Leitch et al., 2010), the researcher entered the field with an open mind to develop theory inductively, based on the perspectives of participants (Kisfalvi, 2002). Consequently, the decision to adopt the overarching aim of *developing a theory of practice for craft entrepreneurs* was a result of the data collection and analysis. The examination of participants' experiences revealed that embeddedness

in households and communities has significant effects on their practices. By comparing the accounts of participants to the extant literature, it became apparent that the empirical data gathered could address enduring gaps in understanding of these domains. Similarly, it was due to participants' repeated descriptions of the importance of various resources and how their connections with household and community members affect the utilisation and sharing of these (mainly nonfinancial) assets that the researcher decided to focus on the capital management practices of craft entrepreneurs.

### 1.4.1. Management of Resources

Prior studies have demonstrated the crucial role that access to resources plays in facilitating entrepreneurial processes (Davidsson and Honig, 2003; Maurer and Ebers, 2006; Mosakowski, 1998). For example, it has been shown that access to funding and specialist knowledge encourages business start-up (Caputo and Dolinsky, 1998; Winborg and Landström, 2001) and that membership in business networks provides entrepreneurs with opportunities to share risks, pool complementary skills and gain technological advantages (Parker, 2008; Ring et al., 2010). Accordingly, scholars have shown a growing interest in investigating the various capitals available to entrepreneurs (Bhagavatula et al., 2010; Jayawarna et al., 2011; Shaw et al., 2008). Extant research has advanced understanding of the impact of financial capital (Chandler and Hanks, 1998; Edelman et al., 2016), human capital (Manolova et al., 2007; Marvel et al., 2016) and social capital (Light and Dana, 2013; McKeever et al., 2014) on entrepreneurs. In comparison, the importance of cultural and symbolic capital has been largely overlooked (De Clercq et al., 2013; Shaw et al., 2017). Prior research has also been criticised for employing myriad definitions of each form of capital (Anderson et al., 2007; Ployhart et al., 2014), which has led to inconsistent results and complicates the comparison of research findings (Adler and Kwon, 2002; Gedajlovic et al., 2013). Furthermore, as most empirical studies focus on individual forms of capital (cf. Batjargal, 2003; Rooks et al., 2016; Stam et al., 2014), there is limited understanding of how entrepreneurs transform the variety of resources available to them (Shaw et al., 2013; Vershinina et al., 2011). While scholars have conceptually considered the convertibility of capital

(Adler and Kwon, 2002; Nahapiet and Ghoshal, 1998; Nicolopoulou, 2014), empirical research is largely absent from the literature (Scott, 2012). Thus, by demonstrating how craft entrepreneurs convert their economic, cultural, social and symbolic capital, this study advances understanding of the resource management practices of entrepreneurs.

## 1.4.2. Research Objectives

In order to *develop a theory of practice for craft entrepreneurs* which explains how embeddedness in multiple contexts affects entrepreneurs' capital management practices, this study investigates three research questions, each of which is addressed in a dedicated findings chapter (Chapters 7-9). These research questions were developed during the data analysis, which involved moving iteratively between empirical evidence and existing literature (Edmondson and McManus, 2007; Miles and Huberman, 1994). The analysis process revealed that this study could make valuable contributions to knowledge by exploring the following questions:

How do entrepreneurs convert their economic, cultural, social and symbolic capital?
 How do household members impact the practices of entrepreneurs?

- 2) 110w up household members impact the practices of entrepreneurs:
- 3) How and why does embeddedness in communities affect the practices of entrepreneurs?

It is important to note that the objectives of this research derive from the interest in and commitment to understanding 'what it feels like to be an entrepreneur' (Drakopoulou Dodd, 2002: 532) and develop a craft enterprise.

#### 1.5. Research Approach

Addressing calls for more methodological and paradigmatic diversity in entrepreneurship research (Karataş-Özkan et al., 2014; Shepherd, 2015), this study embraces an interpretivist perspective (Burrell and Morgan, 1979) and conducts an interpretative phenomenological analysis (Cope, 2011; Smith et al., 2009). A growing number of entrepreneurship studies adopt phenomenological approaches (McKeever et al., 2014; Shaw et al., 2017; Zou et al., 2016), because they are well-

suited for 'developing both new theoretical constructs and enhancing the potency of existing ones, bridging the gap between real-life occurrences and theoretical concepts' (Cope, 2011: 610). Phenomenological research explores the lived experiences of participants to gain comprehensive knowledge of their personal and social worlds (Berglund, 2007; Sanders, 1982). While phenomenological research approaches and small sample sizes constrain generalisation of results to wider populations (Leitch et al., 2010; Lewis, 2015), they facilitate inductive theory development by providing deep insights into the experiences of participants (Geertz, 1973).

Heeding the advice of Smith (2004), 10 participants were selected for this study using a purposeful sampling approach to account for a breadth of experience (Patton, 2002). This study adopts a longitudinal design, which is under-represented in entrepreneurship research, despite its capacity to generate valuable knowledge of entrepreneurial processes (McMullen and Dimov, 2013). Data collection was conducted primarily through four rounds of interviews, which were carried out in intervals of approximately six months, starting in March 2013. Initial interviews were unstructured to gain deep insights into the perspectives of participants (Cope, 2005) and subsequent interviews followed a semi-structured format to explore emerging themes (Al-Dajani et al., 2015). In addition, household members were interviewed, participant observations were conducted and other sources of information were accessed (e.g. participants' websites) to fill potential gaps in understanding (Lewis, 2015; McKeever et al., 2015).

This study adhered to the principles of interpretative phenomenological analysis (Cope, 2011; Smith et al., 2009). After each round of interviews, the researcher (re-)read available data to familiarise himself with the stories of participants (Kempster and Cope, 2010), before highlighting potentially significant passages in the transcripts to enable the discovery and coding of emerging themes (Hycner, 1985). This procedure was completed for all interviews prior to comparing themes between participants and searching for patterns to develop superordinate categories (Leitch et al., 2013). Interpretive summaries were then developed and shared with participants to gain their feedback (Shaw et al., 2017). Finally, emerging findings were compared to arguments in the literature to gain a higher level of

conceptualisation and to establish how the data could contribute to contemporary academic debates (Marlow and McAdam, 2015).

Since the first round of interviews were unstructured, resulting findings were based on the issues that mattered most to participating entrepreneurs. As mentioned above, the empirical data gathered provided insights into participants' management of various resources and the influence of household and community members on their practices. As phenomenological research seeks to portray the lived-worlds of participants (Berglund, 2015; Sandberg, 2000), these themes determined the focus of this study and influenced the choice of a theoretical framework for the analysis and presentation of data. Specifically, Bourdieu's (1977, 1984, 1986) theory of practice was selected for this purpose because of its conceptual clarity concerning the different resources entrepreneurs access and share with others (Shaw et al., 2013; Vershinina et al., 2011), as well as for its appreciation of the influence of norms and conventions (habitus) within different contexts (Anderson et al., 2010; Stringfellow et al., 2014). Essentially, Bourdieu's theory of practice argues that individuals compete for economic, cultural, social and symbolic *capital* to gain dominant positions within social *fields* and that their actions are directed by a socially shaped habitus (DiMaggio, 1979; Özbilgin and Tatli, 2005). Drakopoulou Dodd et al. (2016) therefore propose that Bourdieu's framework is appropriate for investigating the effects of embeddedness in different contexts on entrepreneurial practices.

#### **1.6. Structure of the Thesis**

There is an expectation that business and management theses will adopt a certain structure (Easterby-Smith et al., 2012), which progresses from an introduction to a literature review, context and methods, findings, discussion and then conclusion. However, it could be argued that, similar to the entrepreneurs under investigation, entrepreneurship scholars are tasked with finding a balance between 'fitting in' with existing norms and 'standing out' as acceptably deviant innovators (De Clercq and Voronov, 2009a; Drakopoulou Dodd et al., 2014). Thus, this thesis is structured as follows.

After this introduction, Chapter 2 discusses the research context, specifically the industry sector in which participants are embedded, as well as their local environments. As this study emphasises the influence of contexts on entrepreneurial practices, it is appropriate to begin this investigation by situating readers within these contexts.

Chapter 3 introduces the analytical framework, Bourdieu's theory of practice, which was used to interpret the experiences of participants. While this thesis is not a study of Bourdieu's theories, it employs the concepts of capital, habitus and field to analyse and explain the practices of participating entrepreneurs. Given that the literature review draws upon a number of definitions provided in this chapter (e.g. Bourdieu's perspective on the different forms of capital), it appears sensible to first familiarise the reader with this framework.

Chapter 4 reviews the entrepreneurship literature on resource and capital management, household and families, and embeddedness in communities. This chapter reveals research gaps and explains how this study seeks to advance understanding of these domains.

Chapter 5 discusses the research methods employed by this study. It first discusses phenomenological research and its underlying assumptions and then presents the specific research approach adopted. This chapter demonstrates the rigorous process this study employed to gather and analyse the empirical data.

Chapter 6 proposes a theory of practice for craft entrepreneurs. Based on the analysis of participants' perspectives, this conceptual model explains how embeddedness in multiple contexts affects the capital management practices of entrepreneurs. It highlights the links between the subsequent findings chapters and enables future research to compare the experiences of craft entrepreneurs to those who operate in different industries and contexts.

Chapter 7 examines the capital conversion practices of craft entrepreneurs (micro-level). It systematically explores the various capital transformation processes that participants adopt and, in so doing, reveals a range of previously unidentified capital conversions. Heeding the advice of Cope (2005), the presentation of empirical data is separated from its analysis and enriched by numerous quotations to convey the voices of participants and to facilitate the understanding of phenomena

'from the perspective of those who experience it' (p. 180). However, rather than entirely isolating the findings from their discussion (by moving the discussion into a dedicated chapter), each findings chapter features an 'Analysis and Interpretation' section, which links results to contemporary debates in the literature (Cope, 2011).

Chapter 8 explores the meso-level by considering the household context. This chapter reveals ways in which household members support participants, such as by providing various forms of capital which craft entrepreneurs employ to exploit opportunities. However, it also highlights the negative impacts of embeddedness in households.

Chapter 9 is concerned with the macro-level and investigates community contexts. It demonstrates that embeddedness in communities can lead entrepreneurs to engage in virtuous behaviours, such as collaboration and social value creation. It also shows that expectations of camaraderie can restrict entrepreneurial actions by demanding conformity, causing participants to conceal unconventional practices.

Finally, Chapter 10 presents conclusions, implications and recommendations. It summarises the findings of this study, considers limitations and makes suggestions for future research. It seeks to provide advice not only for scholars and educators, but also practitioners and policy makers.

#### **1.7. Conclusion**

This chapter discussed how this thesis will explore the topic and questions under investigation. It explained the importance of studying the role of context in shaping entrepreneurial practices and examined underlying definitions and the rationale for conducting this study. It also introduced the research goal and the approach adopted to collect and analyse the data. By providing an overview of this study, this chapter sought to spark the interest of the reader in learning about the lives and practices of participating craft entrepreneurs.

#### **CHAPTER 2: RESEARCH CONTEXT**

#### **2.1. Introduction**

Given the influence of contextual factors on entrepreneurial processes (Aldrich and Fiol, 1994; Mair et al., 2012; Welter, 2011) and its centrality to the arguments of this study, it is essential to discuss the context of this empirical research. Thus, this chapter explores the industry sector in which participants are embedded and describes their local environments, which are rural in nature. The objectives of this chapter are twofold: (1) to examine contemporary debates regarding cultural industries and rural entrepreneurship and (2) to introduce the research participants and the geographic locations in which they reside. In contrast to Chapter 9, which discusses the impacts of these contexts on participating entrepreneurs, this chapter provides an introduction to the people whose stories are portrayed and to the communities in which they are embedded. Accordingly, relevant quotations of participants are included to demonstrate their perspectives. Details about participant selection and data collection are presented in Chapter 5. This chapter is structured as follows. First, extant literature on cultural industries is reviewed, followed by a description of the craft entrepreneurs in this study. Second, rural entrepreneurship is discussed, before the rural locations in this study are presented. Finally, conclusions are drawn for this research.

#### **2.2. Cultural Industries**

'Cultural industries' is an umbrella term used to describe a variety of industries (Hirsch, 1972; Lampel et al., 2000), which consist of a large range of smaller communities. Similar to the field of entrepreneurship, which has been described as a 'hodgepodge' of research driven by scholars from various disciplines (Davidsson et al., 2001; Gartner, 2001; Shane and Venkataraman, 2000), cultural industries unite a 'motley crew' of creative individuals (McLeod et al., 2009; Townley et al., 2009). The interpretation of the cultural industries used in this study is comparable to the concept of an onion: the entrepreneur in the centre is surrounded by a multitude of layers (i.e. communities or social fields), which influence her/his practices.

Figure 2.1 illustrates this conceptualisation based on the example of Holly, a teddy bear maker. Importantly, several fields coexist at each level (e.g. 'Artist Bear Makers' and 'Traditional Bear Makers') and individuals can belong to more than one of these communities at the same time.





#### 2.2.1. Cultural Industries Research

Cultural industries are gaining popularity as the context and subject of research (Hirsch, 2000; Lawrence and Phillips, 2002), mainly due to the increasing economic importance of the sector (Nathan et al., 2015, 2016) and the introduction of national policies aimed at their development (Bell and Jayne, 2010; O'Connor, 2015). However, as Peltoniemi (2015: 41) points out, 'it remains unclear what characteristics make industries cultural, which industries belong under the cultural umbrella, and what phenomena are specific to them'. In her review of the literature, Peltoniemi compares various conceptualisations of the cultural industries. Based on her findings, this study broadly defines cultural industries as *the branches of economic activity dedicated to developing products or services that provide customers with an experience and incorporate a substantial creative element.* As cultural goods are experiential, 'they are judged on their ability to offer fun, enjoyment and pleasure' (p. 42). This definition highlights the importance of creativity for cultural production (DeFillippi et al., 2007; Jones et al., 2016). In fact, there are many studies that label the cultural industries as creative industries or use

the terms interchangeably (cf. Jones, 2010; Suddaby and Young, 2015; Thompson et al., 2007), making it difficult to differentiate between them. In an effort to distinguish between their outputs, cultural industries have been described as copyrightable and creative industries as serving a utilitarian function (Townley et al., 2009). This study does not subscribe to this view, as creative works can be both useful and also protected by copyright. Thus, while the term cultural industries is adopted as an overarching label, this study simultaneously draws on and advances understanding of creative industries.

There are several international institutions, including the Organisation for Economic Co-operation and Development (OECD, 2014b) and the United Nations Educational, Scientific and Cultural Organization (UNESCO, 2013), which track the development of the cultural industries. They estimate that global, annual trade of creative goods and services exceeds \$600 billion and grows at an average rate of 8.8% per year, making it one of the fastest growing sectors in the world (OECD, 2014b). The significance of the cultural industries is further illustrated by the large number of jobs they create in regions such as Canada (2.2 million jobs / 12.9% of total employment), the USA (14.2 million /9.75%) and the EU (11.4 million /5.2%) (Nathan et al., 2015, 2016). Within the UK, the Department for Culture, Media and Sport (DCMS, 2016) estimates that the creative economy sustains over 2.8 million jobs (8.8%). While it is more difficult to obtain exact figures for developing nations, UNESCO (2013) estimates that cultural industries contribute similarly to the total employment in countries such as the Philippines (over 10%), Russia (over 7%) and China (over 6%). Considering that cultural industries are regarded as under-studied (Chaston and Sadler-Smith, 2012; Shaw, 2006), dedicated research is therefore both timely and warranted.

Extant studies have identified certain characteristics that are common across cultural industries. First, creative labour is usually in oversupply, resulting in a competitive environment for aspiring entrepreneurs (Coulson, 2012; McLeod et al., 2011). Second, the prospects of success of newly developed cultural products and services are hard to predict (Berg, 2016; Hirsch, 1972) as fashions and consumer tastes are highly mutable (Bourdieu, 1984; Simmel, 1957). In addition, cultural artefacts are difficult to compare and there are no definite, objective measures to

judge their financial or aesthetic value (Ertug et al., 2016). Finally, gatekeepers often play an important role in deciding which products or services reach consumers (Foster et al., 2011; Wright and Zammuto, 2013), requiring entrepreneurs to take gatekeepers' needs and expectations into consideration (Newman et al., 2013). Thus, continuous innovation is normally essential for satisfying the evolving demands of customers and key players in this sector (Scott, 2006).

As Lampel et al. (2000) argue, there are several ambiguities that creative individuals need to resolve when seeking to achieve success in cultural industries. Among others, entrepreneurs need to reconcile (1) artistic goals with economic objectives (Eikhof and Haunschild, 2007; Glynn and Lounsbury, 2005); (2) personal style with market trends (Lawrence and Phillips, 2002); (3) novelty with product familiarity (Islam et al., 2016; Slavich and Castellucci, 2016); and (4) individual with collective creativity (Harvey, 2014; Perry-Smith and Mannucci, 2017). Having to face these enduring tensions can place psychological and emotional strain on entrepreneurs (Scott, 2012). Furthermore, low income, tough competition and a high failure rate imply that pursuing a creative career can be a demanding endeavour (Chaston and Sadler-Smith, 2012). However, despite blurred lines between work and home life, which can lead to both self-exploitation and work-family conflict (Harrison and Wagner, 2016), entrepreneurs continue to start and develop cultural businesses. This is likely due to a deep passion for and commitment to creative activities (Simpson et al., 2015; Svejenova, 2005). Among others, it is this dedication in the face of adversity which makes entrepreneurship in the cultural industries such a fascinating subject to study.

#### 2.2.2. Creative Communities

The cultural industries consist of various smaller, interlinked creative communities (Jones et al., 2016). As definitions and individual classifications vary among countries (OECD, 2014b; UNESCO, 2013), this study focuses on the approach used in the UK. On a national level, the sector is regulated and supported by the DCMS (2016), which differentiates between computer/software design (responsible for

43.5% of the cultural industries' GVA),<sup>2</sup> advertising/marketing (15.8%), film/radio (12.9%), publishing (12.1%), music/performing arts (6.5%), architecture (5.1%), design/fashion (3.8%) and crafts (0.3%).

It may appear counter-intuitive to select the craft field as a research context, given that it makes the smallest contribution to the GVA of the cultural industries. However, the DCMS vastly underestimates the actual size of the craft sector (Bennett et al., 2014), as it does not account for micro businesses below the VAT threshold (£85,000 annual turnover). Thus, those creative communities that consist of large numbers of non-VAT registered micro firms are not adequately captured by the DCMS. While exact figures are not available, it is likely that there are many such ventures, seeing that over 5 million businesses (96% of all enterprises in the UK) have 0-9 employees and are considered micro firms (Ward and Rhodes, 2014). This highlights a limitation of the classification approach of the DCMS. Other deficiencies are discussed by Roodhouse (2011), who reviews the creative industries concept and its application in the UK. He concludes that the repeated redefinitions of what constitutes the cultural industries have been exceedingly challenging, especially for creative communities that have been excluded as a result (e.g. the visual arts).

Theoretical understanding of creative communities has been advanced by a number of empirical studies. Table 2.1 presents an overview of leading research published in 'top journals' (i.e. Grade 4 and 4\*), as defined by the Association of Business Schools (ABS) (Wood and Peel, 2015: 7). While this table is not exhaustive (e.g. craft research is deliberately excluded, as it is presented in the next section), it provides several valuable insights. First, it highlights that some creative communities, especially film, television, music and performing arts, receive more attention than others, such as architecture and software design. Second, it shows that scholars adopt a range of perspectives, including occupational, institutional and networking perspectives, to examine creative communities. At the same time, it reveals that entrepreneurship approaches are comparatively uncommon and should therefore be encouraged. The findings of prior studies improve knowledge of various subjects, such as the effects of norms and conventions, of social and cultural backgrounds and

<sup>&</sup>lt;sup>2</sup> Gross Value Added (GVA) is a measure of the value of goods and services produced by a sector of the economy. The DCMS (2016: 3) estimates 'the contribution of the Creative Industries to UK Gross Value Added (GVA) ... using approximate GVA (aGVA) from the Annual Business Survey'.

Context	Article	Perspective	Key Finding	Method	Country
Advertising	Koppman (2016)*	Occupational	Shared culture signals creative potential to employers.	Mixed	USA
	McLeod et al. (2009)*	Occupational	Social class is less significant once accepted in the industry.	Qualitative	UK
Architecture	Jones (2010)	Occupational	Symbolic networks are key to gaining career recognition.	Archival	Multiple
Design/Fashion	Dalpiaz et al. (2016)*	Institutional	Combining different logics facilitates opportunity recognition.	Qualitative	Italy
	Elsbach (2009)	Creativity	Signature styles allow designers to affirm their identities.	Qualitative	USA
	Khaire & Hall (2016)*	Innovation	International norms can co-exist with local conventions.	Archival	India
Film/TV	Anand & Watson (2004)*	Institutional	Award ceremonies construct prestige hierarchies within fields.	. Mixed	<b>USA</b>
	Patriotta & Hirsch (2016)*	Innovation	New conventions develop through cooperative interactions.	Archival	<b>USA</b>
	Randle et al. (2015)*	Occupational	Actors employ various forms of capital to enter the industry.	Qualitative	UK
	Svejenova (2005)	Occupational	Authenticity shapes meaningful careers in creative industries.	Qualitative	Spain
	Zaheer & Soda (2009)*	Networking	Spanning structural holes results in superior team performance. Quantitative	e. Quantitative	Italy
Music	Coulson (2012)	Entrepreneurship	Musicians are enterprising when pursuing a musical calling.	Qualitative	UK
	Glynn & Lounsbury (2005)* Institutional	Institutional	Critics' stories are embedded in broader discursive fields.	Qualitative	USA
	Johnson (2007)	Entrepreneurship	Entrepreneurship Imprinting is an outcome of cultural entrepreneurship.	Archival	France
	Lingo & O'Mahony (2010)	Creativity	Ambiguity is inherent in collective creative processes.	Qualitative	NSA
Performing Arts	Berg (2016)	Creativity	Divergent thinking facilitates forecasting creative successes.	Quantitative Multiple	Multiple
	Eikhof & Haunschild (2007)* Institutional	Institutional	Economic logics tend to prevail over artistic logics in theatres.	Qualitative	Germany
	Glynn (2000)	Institutional	Organisational identity affects beliefs about core competencies. Qualitative	s. Qualitative	USA
	Harrison & Rouse (2015)	Creativity	Prototypes are co-created by creatives and feedback providers.	Qualitative	USA
	Uzzi & Spiro (2005)*	Networking	Small world networks influence performance of musicals.	Quantitative	USA
	Vera & Crossan (2005)	Innovation	Improvisational skills can be taught through training.	Quantitative Canada	Canada
Publishing	Thornton (2002)	Institutional	Publishers shifted from an editorial to a market logic in the 70s. Mixed	s. Mixed	USA
oftware Design	Software Design Batjargal (2010)	Networking	Institutional contexts influence the effects of structural holes.	Quantitative China/Russia	China/Rus
	Metiu (2006)*	Innovation	Status can be both an input and output of intergroup relations.	Qualitative	India/USA

Table 2.1. Research into Creative Communities (excluding Crafts)

of reputation within the industry. Thus, this study will be able to contribute to these existing debates. The table also demonstrates that qualitative methods are popular in this field and that most studies explore creative communities in the EU or the USA. Finally, it illustrates that Bourdieu's work is frequently cited by leading research. While this is not surprising, given Bourdieu's (1984) focus on culture and society, it highlights the acclaim his work has achieved and the relevance of his theories to this context.

## 2.2.3. Craft Field

As mentioned above, the significance of the craft sector is often underestimated (Bertacchini and Borrione, 2011). A number of studies therefore attempt to measure the contribution of the sector to the economy and society (cf. Burns et al., 2012; Schwarz and Yair, 2010). Bennett et al. (2014) estimate that there are over 11,500 craft businesses in the UK (half of which are not VAT registered), which employ more than 43,000 people and generate around £1.9 billion turnover. Other reports propose even higher figures (e.g. over 23,000 craft businesses, see Burns et al., 2012), highlighting the impact of the sector on the economy. In addition, craft makers create social value through work in the community, such as by offering classes to help students develop their imagination, concentration and confidence and by fostering inclusion and social interaction through shared activities (Schwarz and Yair, 2010).

Many makers choose to undertake secondary activities, such as teaching, to supplement their income, which is approximately 12% below the national average in the UK (Bennett et al., 2014). Thus, while craft making is often depicted as a rewarding experience (Sennett, 2008), this is not primarily due to prospects of profit and growth, but rather a mixture of benefits including personal satisfaction and community engagement (Paige and Littrell, 2002; Tregear, 2005). Given the considerable amount of skill and experience that is required to practise a craft (Bhagavatula et al., 2010), it is not surprising that 85% of makers in the UK hold a craft-related qualification (Burns et al., 2012). Another peculiarity of the craft sector is the high number of sole traders (88%), which is attributed to the desire of makers 'to be free' (Schwarz and Yair, 2010: 96). However, working by themselves

also causes a feeling of 'aloneness' for many practitioners. Furthermore, it should be noted that craft makers in the UK are approximately '69% female, 31% male' (Burns et al., 2012: 18).

The craft sector is supported by national development agencies, including the Craft Council, the Arts Council of Wales, Creative Scotland and Craft Northern Ireland (Morris et al., 2010a). Interestingly, the spheres of responsibility held by these agencies vary. A representative example is the Craft Council: though it profiles makers from across the UK, its mission is restricted to advocating crafts in England and Wales and its development programmes are only available to makers in England (Schwarz and Yair, 2010). Thus, while practitioners benefit from region-specific developmental activities, the sector as a whole is hindered by the disparate approach entailed by the government's devolvement of responsibilities within the cultural industries (Roodhouse, 2011; Thomas et al., 2013). An underlying cause for existing difficulties is the industry-focused approach that the DCMS adopts to define crafts (Bennett et al., 2014). They employ an occupational classification list, which features smiths, weavers, knitters, potters and furniture makers and an industry group classification scheme that incorporates only the manufactory of jewellery (DCMS, 2016: 21-22). Evidently, this approach excludes a number of crafts, especially contemporary ones. Accordingly, this study employs a conceptualisation of the craft sector which is more inclusive and based on alternative definitions provided by Fillis (2004) and Tregear (2005). This study defines craft makers as individuals who produce and sell products or services which possess a distinct artistic value resulting from a high degree of manual input.

A review of the craft business literature reveals several valuable insights (see Table 2.2). First, it shows that extant studies examine a number of craft fields, which can broadly be classified into three categories: (1) textile-related; (2) food-related; and (3) multiple craft-focused (though there are exceptions, such as pottery-focused studies). These studies are primarily devoted to product-based crafts and rarely consider the craftsmanship involved in service-based activities, such as hairdressing (Holmes, 2015) or Haute Cuisine (Slavich and Castellucci, 2016; Svejenova et al., 2007). Second, tourism and marketing-related issues are frequently explored by craft research, whereas they are not prevalent in studies of other creative

<b>Craft Field</b>	Article	Perspective	Key Finding	Main Method Country	I Country
Artisan food	Blundel (2002)	Networking	Networks are dynamic, idiosyncratic and living phenomena.	Archival	UK
	Tregear (2005)	Entrepreneurship	Food producers pursue commercial and lifestyle-oriented goals.	Qualitative	UK
Cleaning	Tweedie & Holley (2016)	Occupational	Craft concepts help workers feel satisfied, even in menial tasks.	Qualitative	Australia
Clothing	Porac et al. (1995)	Networking	Rivalry networks are localised into clique-like subgroups.	Mixed	UK
	Rantisi (2014)	Gender	Guild systems are used to devaluate female makers' work.	Qualitative	Canada
	Thomas et al. (2013)	Networking	Guilds are fragmented, complex and shaped by various actors.	Qualitative	UK
Hairdressing	Holmes (2015)	Occupational	Craft of service work is often ignored and undervalued.	Qualitative	UK
Haute Cuisine	Slavich & Castellucci (2016)	Occupational	Apprentices can gain esteem through association with masters.	Quantitative	Multiple
	Svejenova et al. (2007)	Institutional	Reputation plays a critical role in the dissemination of ideas.	Qualitative	Spain
Luxury wines	Beverland (2005)	Marketing	Advertising handcrafted techniques helps establish authenticity.	. Qualitative	Multiple
Aicro brewing	Micro brewing Carroll & Swaminathan (2000) Institutional	Institutional	Craft identity is crucial for the success of specialist breweries.	Archival	NSA
	Clemons et al. (2006)	Marketing	Hyperdifferentiation helps firms meet customers' specific needs.	. Quantitative	<b>NSA</b>
	Verhaal et al. (2015)	Marketing	Highlighting craft production in beer names increases success.	Quantitative	NSA
Multiple	Fillis (2004)	Marketing	Craft makers can be classified into four distinct types.	Mixed	UK
	Kuhn & Galloway (2015)	Networking	Craft makers' motivations influence what advice they share.	Quantitative	Multiple
	Paige & Littrell (2002)	Entrepreneurship	Craft retailers judge success by financial and personal criteria.	Quantitative	NSA
	Parry (2010)	Entrepreneurship	Artisan identity and management styles are barriers to growth.	Qualitative	UK
	Shultz (2015)*	Networking	Craft makers form networks to share intermediary duties.	Qualitative	NSA
Pottery	Revilla & Dodd (2003)	Tourism	Producing authentic, local crafts encourages incoming tourism.	Quantitative	Mexico
	Sánchez-Medina et al. (2015)	Entrepreneurship	Environmental compliance increases craft firm performance.	Quantitative	Mexico
Weaving	Bhagavatula et al. (2011)	Networking	Structural holes have an adverse effect on resource acquisition.	Quantitative	India
	Cohen (2001)	Tourism	Individual makers' successes increase socioeconomic inequality. Quantitative	. Quantitative	Mexico
	Ramachandran et al. (2012)	Institutional	Bridging enterprises reconcile interests of makers and markets.	Qualitative	India
	Sayce et al. (2007)	Gender	Modernising craft production can cause male identity crisis.	Qualitative	UK

Selected articles significantly advance knowledge of the craft field and were published in journals ranked at ABS Grade 3 or higher (Wood & Peel, 2015).

Table 2.2. Craft Research

communities (see Table 2.1). Third, institutional and entrepreneurship perspectives are comparatively rare within craft research. Concerning preferred methodologies, both qualitative and quantitative approaches are employed in this context. While prior research explores craft businesses in a number of countries, the UK appears to be a particularly popular context. It may be, as Shultz (2015: 458) suggests, that craft makers have earned a higher reputation in Europe, where they are seen as providing 'diversity and competition [to] the largest companies', than in North America, where both academics and the general public are more interested in cultural elites (e.g. celebrities). The findings of extant studies advance knowledge of various subjects, including guilds, networking, craft motivations and authenticity. In contrast, the influence of different communities on entrepreneurs or their resource management practices have not received adequate attention. Finally, Table 2.2 shows that Bourdieu's work is almost never cited in craft research, which is surprising given its popularity among scholars of cultural industries.

#### 2.3. Craft Entrepreneurs in this Study

Following the review of cultural industries research, this section introduces the craft entrepreneurs in this study. While further justification for selecting these participants is provided in Chapter 5, along with a tabular comparison of their main characteristics, it is important to first explain why their stories are interesting and worth studying.

As shown in Table 2.3, the 10 participating craft entrepreneurs contribute a variety of perspectives to this study. They belong to a range of fields covering both contemporary crafts, such as souvenir and teddy bear making, and more traditional crafts, including silversmithing and basket weaving. Before establishing their ventures, all participants gained valuable skills either by studying their crafts at college/university or by pursuing a degree in a different cultural discipline. In addition, they acquired work experience through internships, employment and, in some cases, running a business, which prepared them for launching their craft firms. As all participants continue to operate their ventures over two years past the end of the data collection, their businesses are considered to be sustainable. Their work is regularly on display at craft shows and exhibitions, in some cases nationally or even
# Table 2.3. Craft Entrepreneurs in this Study

Participant	Profile
Amy* (late-20s)	Following a degree in fashion design, Amy set up a wallet making business with two friends who taught her valuable business skills. In 2011, she left to start her own firm, crafting individually designed accessories. During the course of this study, Amy lost her mother, who had taught her how to sew, and broke up with her partner, Adam. Building on her past business successes, she also decided to establish a fashion label.
Betty* (early-30s)	After studying human geography at university, Betty worked for a library where she ran children's reading groups. Only weeks before the first interview, she decided to capitalise upon her creativity to produce individualised children's playbooks, for which she perceived a growing demand. A few months into this study, her partner Brenda moved away for further education, and Betty decided to relocate with her.
Carol (mid-30s)	Born and raised in North America, Carol became aware of the internationally growing demand for hand-dyed yarn earlier than most craftspeople in the UK. She seized the opportunity and opened her studio in 2007. Carol had studied English literature and gained business experience from running an instrument repair firm for two years. She lives with her partner Clara, who is employed by an insurance company.
Diana (late-30s)	When Diana started studying silversmithing in 1995, the profession was dominated by men. She managed to succeed in this craft by gaining international acclaim for her work. Since moving into her studio in 2004, Diana has won several prestigious awards and a number of her pieces were selected for display in permanent exhibitions of national museums. She met her partner Daniel, who works as a sailor, at university.
Emily (mid-40s)	After earning a degree in printed textiles and working as an independent textile designer for over 15 years, Emily was seeking a new challenge. She opened her own hand-stitching studio in 2007 but, three months into this study, decided to work from home again to gain more time for her design work. Emily is mother to one child and has been with her husband, Evan, who works for the government, for over 20 years.
Frank* (mid-40s)	While studying medicine, Frank discovered his passion for ceramics. He decided to pursue this interest by training in pottery at college. Frank moved into his workshop shortly before the start of this study. After breaking up with Fred, his partner, he invited his friend Felix, a fellow caving club member, to move into his spare room.
George (early-50s)	Raised in the Baltics, George was trained in the European approach to basket weaving before learning the British technique. As one of the only makers with such training, he found work with several employers before opening his studio in 2004. He is a father of two children and receives help from his wife, Grace, who is a trained accountant.
Holly (early-50s)	As a teddy bear maker, Holly engages in a contemporary craft. Originally trained in costume design, she decided to change professions when her sister's teddy bear business became successful. Holly has practised this craft independently for over 15 years and has won prestigious industry awards. Holly is a mother of two children and has been with her partner, Henry, who works in set design, for over 20 years.
Isla* (early-60s)	After being trained in ceramics at university, Isla worked for the same employer for most of her life. In 2010 she discovered a lack of individualised souvenirs for her local area, so she decided to set up a business producing such memorabilia. Isla is a mother of one and has been with her husband, Iain, who is an architect, for almost 40 years.
Julie* (mid-60s)	Following degrees in English, Fine Art and Education and a career as a lecturer of arts at a university, Julie possessed a wealth of relevant experience which prepared her for opening a print-making business in 2006. Presenting her work in solo exhibitions raised her profile nationally. Julie receives help from her husband, Jack, and her two children, as she is now the sole earner of the household.

\* Denotes the five participants who are based in England. The other five are located in Scotland.

internationally, which highlights the quality of their work. Furthermore, several of the more experienced craft entrepreneurs have gained acclaim through winning prestigious awards. As Carol explains, 'the makers and the designers and the dyers ... are sort of famous ... people know who you are and they are excited to meet you'.

Within the academic literature, scholars generally differentiate between necessity-driven and opportunity-driven entrepreneurs (Bird and Wennberg, 2016; Jennings and Brush, 2013). In other words, they distinguish between those individuals who are pushed into entrepreneurship because 'all other options for work are absent or unsatisfactory' (Vorley and Rodgers, 2014: 429) and those who are pulled into it by the choice to exploit a perceived opportunity. Participants in this study are a mixture of both. On the one hand, they are passionate about their crafts and most feel like Emily, who describes herself as 'a bit of a control freak' when it comes to her working life. As Diana puts it, 'being self employed is about having control over everything'. Control is essential to participants because 'what I do is quite individual, my standards are quite high' (Julie) and 'I do get quite caught up in doing things over and over again until I feel that they are perfect' (Betty). Since all participants need to earn an income to help sustain their households, starting their own firms was therefore the only way for them to stay in complete control of their creative outputs. On the other hand, Table 2.3 also illustrates that participants started their firms to seize specific opportunities that they identified in the marketplace. They were either trying to satisfy a perceived customer need, for example, for handdved varn or individualised children's playbooks, or developing a novel product range, including caving mugs and square willow baskets, which stimulated demand with a target audience. Thus, it could be argued that participants engaged in both discovery and creation strategies of entrepreneurial action (Alvarez and Barney, 2007; Suddaby et al., 2015).

Concerning their personal lives, Table 2.3 shows that all participants live in households that are shared with other people, mainly consisting of nuclear family members. Most participants are in stable relationships, which provide them with a level of consistency and support. In addition, though they no longer require constant care, some participants still have children living at home. Thus, participants must manage the balancing act of operating a business and engaging in their family lives

(Eddleston and Powell, 2012; Jennings and McDougald, 2007). In some cases, the resulting tensions have led to difficulties and the end of relationships. Furthermore, due to the longitudinal nature of this study the researcher was witness to several critical events, such as the loss of a family member and the relocation of home or business. These experiences influenced the practices of the entrepreneurs, highlighting the crucial role that household and family members play in their working lives.

### 2.4. Rural Entrepreneurship

Following the exploration of the industry context and the introduction of the craft entrepreneurs, this section focuses on the regional contexts in which participants reside. Accordingly, it reviews relevant literature on rural entrepreneurship, examining both the policy perspective and the academic perspective, before presenting the rural locations in this study.

It is important to first define rural entrepreneurship. Extant studies mainly refer to spatial characteristics to differentiate between rural and other types of entrepreneurship (Galloway et al., 2011; Korsgaard et al., 2015; Lang et al., 2014). Rurality is conceived 'as a spatial category dominated by large open spaces and, relative to the national context, small settlements' (Kalantaridis and Bika, 2006: 112). To operationalise the concept, research frequently employs the OECD's (2006) definition, which classifies areas as rural if their population density is low (less than 150 inhabitants per square kilometre) and they do not comprise a main urban centre. The UK government adopts a similar approach; areas are defined as rural 'if they fall outside of settlements with more than 10,000 resident population' (Defra, 2016: 1). This study employs the latter conceptualisation, as it is more inclusive and applicable in more densely populated countries, such as England (OECD, 2011).

Furthermore, a distinction needs to be made between spaces and places, as they represent different facets of spatial contexts (Steyaert and Katz, 2004). Space denotes 'an economic (capitalistic) evaluation of location based on its capacity for profit' (Johnstone and Lionais, 2004: 218). Thus, it is primarily concerned with the various resources available in an area, which are used to maximise economic gains. In comparison, place denotes 'an evaluation of a location in relation to lived life and experience' (Korsgaard et al., 2015: 579). In other words, places are created through the meanings and experiences that individuals connect with a location. While a number of studies have explored space-related issues (cf. Kloosterman, 2010; Lee et al., 2011), there is a need for further research into the role of place in entrepreneurship (McKeever et al., 2015). Consequently, this study seeks to account for place-related aspects when analysing the spatial contexts in which participants reside.

# 2.4.1. Policy Perspective

As a result of technical advances, the need for manual labour in traditional rural industries, such as agriculture, forestry and mining, has reduced significantly (Lang et al., 2014), triggering a shift in economic activity in rural regions around the world (OECD, 2014a). Particularly in Western nations, rural areas have transformed from production to consumption spaces, serving a multitude of purposes, including tourism, leisure, living and working (Markantoni and van Hoven, 2012; Mitchell, 2013). Such shifts have been facilitated by increased connectivity, achieved largely through improved Internet access (Anderson et al., 2016; Cumming and Johan, 2010) and better-developed transportation systems (Kalantaridis and Bika, 2006). Within the resulting 'global village', interpretations of rurality have become more fluid and subjective (Anderson, 2000). In the past, peripheral areas have often been considered as separated by space and time, earning them deprecating labels, such as hinterland or backwater. Over time, perceptions have changed and rurality is now seen to provide a higher quality of life (Akgün et al., 2011) and an escape from the 'rat race' of cities (Bell and Jayne, 2010).

Nevertheless, many rural regions still face significant challenges (OECD, 2006). First, they are burdened by out-migration, which is caused by an oversupply of labour and a lack of social attractions, and which in turn results in ageing resident populations (Smallbone, 2009). Second, public services are less developed due to higher delivery costs, limiting access to essential local support services, such as medical care (Phillipson et al., 2004). Third, average labour productivity is lower, particularly in sparsely populated areas, resulting in lower outputs and less competitive wages (Webber et al., 2009). Fourth, access to education and specialised training is limited, leading to a comparatively small proportion of inhabitants

pursuing tertiary education (Anthopoulou, 2010). As a consequence, the average Gross Domestic Product (GDP) per person is often lower in rural regions than in urban areas (Lang et al., 2014). However, this does not mean that rural areas are in economic decline. In fact, OECD (2014a) reports that rural regions have, on average, grown faster than their urban counterparts.

It is also important to differentiate between 'remote' regions, which are more isolated and subject to distance penalties (e.g. longer transportation times) and 'accessible' regions, which are comparatively close to urban areas and benefit from their infrastructure and services (Galloway et al., 2011; Smallbone et al., 1999). In Scotland, there is a significant divide between these regions. Many remote rural areas (i.e. localities that are more than 30 minutes drive from a major settlement) are 'fragile' and 'deprived', such as the 'Western Isles which has a GDP per capita of 60% of the EU average' (OECD, 2008: 5). In contrast, accessible rural regions experience a positive net migration and have an average household income that is higher than any other area in Scotland, including its main cities (Scottish Government, 2015a). However, the growing popularity of these regions causes problems, including increasing commuter-related congestion, higher land prices and housing shortages (OECD, 2008).

While 70% of the landmass in Scotland is considered remote rural and 27% accessible rural (Scottish Government, 2015a), 'there is no region in England that can be considered predominantly [remote] rural' (OECD, 2011: 14). There are, of course, sparsely populated areas which have struggled with poverty and social exclusion and have therefore been described as remote (Kalantaridis and Bika, 2006; Smallbone et al., 1999). Nevertheless, there is significantly more interaction between rural and urban regions in England than in most other countries and those who live in accessible rural areas (about 20% of the population) have comparatively good access to public services (OECD, 2011). Furthermore, rural areas enjoy a largely positive image, as the English countryside is usually associated with pastoral settings and rustic cottages, especially by international visitors and residents of cities (Bell and Jayne, 2010).

From an industry perspective, there are many similarities between Scotland and England. As the role of agriculture becomes less significant, rural areas in both countries draw on their natural and cultural resources to develop other sectors, particularly tourism and renewable energy (OECD, 2008). However, it has been argued that 'there is untapped potential in all the rural ... regions' (OECD, 2011: 97) and fostering entrepreneurship is seen as an effective means for stimulating economic activity and revitalising rural communities (Ring et al., 2010). Accordingly, the UK government introduced a range of support schemes to encourage business start-up, especially of high-growth firms (Lee and Shaw, 2016; Smallbone et al., 2003). Similar to the cultural sector, rural development responsibilities have been devolved within the UK, leading to separate programmes for Scotland (Scottish Government, 2015b) and England (Defra, 2015), despite their mutual objectives of (1) growing the rural economy, (2) providing high-quality jobs and (3) boosting productivity. To achieve these goals, a range of measures have been instituted, including funding schemes, rural growth networks (which provide business support and networking opportunities) and skills development (Lee and Cowling, 2015).

It should be noted that prior studies have questioned the utility of some such development programmes (Phillipson et al., 2004; Smallbone et al., 2003). For example, Webber et al. (2009) argue that investments should be focused on attracting in-migrants, given their comparatively higher levels of human and social capital and their success in developing Small and Medium-sized Enterprises (SMEs) (Akgün et al., 2011; Kalantaridis and Bika, 2006), rather than increasing spending on retraining the local population. Furthermore, Lee et al. (2011) highlight the need to provide more networking advice, as existing training programmes do not adequately explain how to effectively acquire resources through networks. Most importantly, evidence suggests that the majority of rural firms have either not made 'use of these various instruments of assistance ... [or] not taken any effective advantage from their availability' (Meccheri and Pelloni, 2006: 372).

## 2.4.2. Academic Perspective

As Korsgaard et al. (2015) note, only a scant body of research examines the impact of local and spatial contexts on entrepreneurial activities, particularly in peripheral areas. Consequently, there is limited understanding of how entrepreneurs can overcome inherent liabilities (e.g. resource-scarcity) which are common in rural contexts (Alsos et al., 2014; Lee et al., 2011). Furthermore, little is known about the ways in which context-related factors influence the abilities of rural entrepreneurs to identify and exploit opportunities (Meccheri and Pelloni, 2006: 372). However, it has been argued that knowledge of the local environment and its potential can have a significant effect on opportunity recognition (Anderson, 2000). Accordingly, calls have been made for research to employ a 'milieu' perspective (Baumgartner et al., 2013) and to explore the embedded nature of entrepreneurship in communities and places (Jack and Anderson, 2002; McKeever et al., 2015). This study heeds this advice and examines the influence of embeddedness in communities on entrepreneurs.

As illustrated in Table 2.4, there are several studies that develop understanding of rural entrepreneurship. This table reveals that the majority of studies focus on the individual rural entrepreneur or on SMEs. Nevertheless, rural communities are beginning to gain popularity as subjects of research in their own right. The table also shows that most prior studies adopt a developmental stance, exploring how rural development can be fostered through entrepreneurial activities (or vice versa). Other perspectives, such as gender or marketing, are much less common in this context. Furthermore, quantitative methods are predominant, which is in line with wider trends in the entrepreneurship literature (McDonald et al., 2015; Neergaard and Ulhøi, 2007). Interestingly, the majority of studies examine rural entrepreneurship in Europe, particularly within the UK. Despite the large number of economically depressed rural communities in the USA (Ring et al., 2010), scholarly interest in these regions is limited. The findings of prior studies are broad and advance knowledge of such subjects as the effects of regulative institutions on rural entrepreneurship and the value of farm versus non-farm activities. Also frequently investigated are the impacts of Internet access and the influence of newcomers (both in-migrants and immigrants) on entrepreneurial activity in rural communities. In contrast, the link between cultural industries and rural locations has not received sufficient attention (Bell and Jayne, 2010).

Baumgartner et al. (2013)DevelopmentalEntrepreneurship has a moderate short-term effect on rural growth. QuantitativeBrinjes & Diez (2013)DevelopmentalA developed non-farm economy increases opportunity emergence. QuantitativeLang et al. (2014)EmbeddednessRegulative institutions constrain rural entrepreneurial activity.QuantitativeRing et al. (2010)NetworkingSocial structures of rural communities affect network formation.TheoreticalRooks et al. (2014)NetworkingSocial structures of rural communities affect network formation.TheoreticalRooks et al. (2011)NetworkingDevelopmentalIn-migrants are predominantly attracted by a rural lifestyle.QuantitativeAnderson (2000)*EmbeddednessRural entrepreneurs interpret their own version of the environment.QuantitativeJack & Anderson (2000)*EmbeddednessRural entrepreneurs interpret their own version of the environment.QuantitativeLafuente et al. (2011)DevelopmentalIn-migrants are predominantly attracted by a rural lifestyle.QuantitativeJack & Anderson (2000)*EmbeddednessRural entrepreneurs interpret their own version of the environment.QuantitativeIafuente et al. (2013)DevelopmentalIn-migrants are predominantly attracted by a rural lifestyle.QuantitativeIafuente et al. (2013)DevelopmentalIn-migrants are prodominantly attracted by a rural lifestyle.QuantitativeIafuente et al. (2013)DevelopmentalIn-migrants are predominantly attracted by a rural lifestyle.Quantitative <tr <td="">Nattracted by a rural entrepr</tr>	Quantitative Switzerland Quantitative Vietnam Qualitative Vietnam Theoretical USA Mixed USA Mixed USA Mixed USA Qualitative Norway Quantitative Norway Quantitative UK Quantitative UK Quantitative Spain Quantitative China Quantitative UK
DevelopmentalA developed non-farm economy increases opportunity emergence.EmbeddednessRegulative institutions constrain rural entrepreneurial activity.NetworkingSocial structures of rural communities affect network formation.NetworkingSocial structures of rural communities affect network formation.NetworkingSocial structures of rural communities affect network formation.NetworkingSocial structures of rural communities affect network formation.NetworkingDense networks foster resource sharing in individualistic contexts.EmbeddednessIn-migrants are predominantly attracted by a rural lifestyle.RegionalEntrepreneurs interpret their own version of the environment.RegionalEmbeddedness is key to shaping and sustaining business.In-migrants help integrate rural economics into global markets.RegionalRole models increase rural entrepreneurial activity levels.InstitutionalWeak regulatory protection discourages external collaboration.DevelopmentalPortfolio entrepreneurship is popular among farm households.DevelopmentalThe farm sector's legacy of separation impedes entrepreneurship.CoupationalFarmers consider themselves entrepreneurs and producers.InstitutionalOccupationalInstitutionalInternet access stimulates entrepreneurship in larger communities.InstitutionalGenderMarketingInternet portals are source of market exclusion.InstitutionalKome entrepreneurs start dynamic and often innovative firms.MarketingInternet portals are predominantly used for local	
<ul> <li>Embeddedness Embeddedness Regulative institutions constrain rural entrepreneurial activity. Networking Social structures of rural communities affect network formation. Networking Embeddedness Social structures of rural communities affect network formation. Networking Embeddedness Embeddedness Embeddedness is key to shaping and sustaining business. In-migrants help integrate rural economies into global markets. Regional Entrepreneurs interpret their own version of the environment. Embeddedness In-migrants help integrate rural economies into global markets. Regional Entrepreneurs interpret their own version of the environment. Embeddedness In-migrants help integrate rural economies into global markets. Regional Institutional Weak regulatory protection discourages external collaboration. Developmental Portfolio entrepreneurship is popular among farm households. Developmental Portfolio entrepreneurship is popular among farm households. The farm sector's legacy of separation impedes entrepreneurship. Farmers consider themselves entrepreneurs and producers. Developmental Institutional Developmental References stimulates entrepreneurs and producers. Institutional Internet access stimulates entrepreneurship in larger communities. Institutional Notes are a source of market exclusion. Weak the sector's legacy of separation impedes entrepreneurship. Institutional References stimulates entrepreneurs and producers. Institutional References stimulates entrepreneurs and producers. Institutional Notes are a source of market exclusion. Women entrepreneurs start dynamic and often innovative firms. Institutional such are a source of ranket exclusion.</li> <li>Marketing Remet portals are predominantly used for local trade. Institutional Rutepreneurs prefer local resource bases to non-local ones.</li> </ul>	ative stical ative ative ative tative ative ative ative
NetworkingSocial structures of rural communities affect network formation.NetworkingBoense networks foster resource sharing in individualistic contexts.EmbeddednessDevelopmentalEmbeddednessIn-migrants are predominantly attracted by a rural lifestyle.RegionalIn-migrants are predominantly attracted by a rural lifestyle.RegionalEntrepreneurs interpret their own version of the environment.RegionalIn-migrants are predominantly attracted by a rural lifestyle.RegionalIn-migrants are predominantly attracted by a rural lifestyle.RegionalEntrepreneurs interpret their own version of the environment.RegionalRural embeddedness is key to shaping and sustaining business.InstitutionalIn-migrants help integrate rural economies into global markets.RegionalIn-migrants help integrate rural economies into global markets.RegionalPowelopmentalInstitutionalVeak regulatory protection discourages external collaboration.DevelopmentalPortfolio entrepreneurship is popular among farm households.DevelopmentalPortfolio entrepreneurship is popular among farm households.DevelopmentalPortfolio entrepreneurship in larger communities.InstitutionalDevelopmentalDevelopmentalInternet access stimulates entrepreneurs and producers.DevelopmentalInternet access stimulates entrepreneurs and producers.InstitutionalMomen entrepreneurs start dynamic and often innovative firms.MarketingRural entrepreneurs start dynamic and often innovative firms.Marketing<	stical ative ative ative tative tative ative
NetworkingDense networks foster resource sharing in individualistic contexts.EmbeddednessEmbeddednessDevelopmentalIn-migrants are predominantly attracted by a rural lifestyle.RegionalEntrepreneurs interpret their own version of the environment.RegionalEntrepreneurs interpret their own version of the environment.RegionalRural embeddedness is key to shaping and sustaining business.InstitutionalRole models increase rural entrepreneurial activity levels.InstitutionalWeak regulatory protection discourages external collaboration.DevelopmentalPortfolio entrepreneurship is popular among farm households.DevelopmentalPortfolio entrepreneurship is popular among farm households.DevelopmentalPortfolio entrepreneurship in larger communities.InstitutionalInternet access stimulates entrepreneurs and producers.DevelopmentalInternet access stimulates entrepreneurs and producers.MarketingInternet portals are pred	ative ative ative ative tative ative ative
EmbeddednessCommunity entrepreneurs connect villagers with external actors.DevelopmentalIn-migrants are predominantly attracted by a rural lifestyle.RegionalEntrepreneurs interpret their own version of the environment.RegionalEntrepreneurs interpret their own version of the environment.FembeddednessRural embeddedness is key to shaping and sustaining business.In-migrants help integrate rural economies into global markets.RegionalRole models increase rural entrepreneurial activity levels.InstitutionalWeak regulatory protection discourages external collaboration.DevelopmentalPortfolio entrepreneurship is popular among farm households.DevelopmentalPortfolio entrepreneurship is popular among farm households.DevelopmentalFarmes consider themselves entrepreneurs and producers.DevelopmentalInternet access stimulates entrepreneurs and producers.DevelopmentalInternet access stimulates entrepreneurs and producers.DevelopmentalInternet access stimulates entrepreneurs and producers.MarketingKomen entrepreneurs start dynamic and often innovative firms.MarketingInternet portals are predominantly used for local trade.EmbeddednessRural entrepreneurs prefer local resource bases to non-local ones.	
DevelopmentalIn-migrants are predominantly attracted by a rural lifestyle.RegionalEntrepreneurs interpret their own version of the environment.EmbeddednessRural embeddedness is key to shaping and sustaining business.)* EmbeddednessIn-migrants help integrate rural economics into global markets.RegionalIn-migrants help integrate rural economics into global markets.InstitutionalNeak regulatory protection discourages external collaboration.DevelopmentalPortfolio entrepreneurship is popular among farm households.DevelopmentalPortfolio entrepreneurship is popular among farm households.DevelopmentalThe farm sector's legacy of separation impedes entrepreneurship.DevelopmentalInternet access stimulates entrepreneurs and producers.DevelopmentalInternet access stimulates entrepreneurship in larger communities.InstitutionalWomen entrepreneurs start dynamic and often innovative firms.MarketingInternet portals are predominantly used for local trade.EmbeddednessRural entrepreneurs prefer local resource bases to non-local ones.	
RegionalEntrepreneurs interpret their own version of the environment.EmbeddednessRural embeddedness is key to shaping and sustaining business.ImbeddednessIn-migrants help integrate rural economics into global markets.RegionalRole models increase rural entrepreneurial activity levels.InstitutionalWeak regulatory protection discourages external collaboration.DevelopmentalPortfolio entrepreneurship is popular among farm households.DevelopmentalPortfolio entrepreneurship is popular among farm households.DevelopmentalThe farm sector's legacy of separation impedes entrepreneurship.DevelopmentalInternet access stimulates entrepreneurs and producers.DevelopmentalInternet access stimulates entrepreneurs and producers.DevelopmentalInternet access stimulates entrepreneurship in larger communities.InstitutionalKomen entrepreneurs start dynamic and often innovative firms.MarketingRural entrepreneurs prefer local resource bases to non-local ones.	
<ul> <li>Embeddedness</li> <li>Rural embeddedness is key to shaping and sustaining business.</li> <li>1* Embeddedness</li> <li>In-migrants help integrate rural economies into global markets.</li> <li>Regional</li> <li>Role models increase rural entrepreneurial activity levels.</li> <li>Institutional</li> <li>Weak regulatory protection discourages external collaboration.</li> <li>Developmental</li> <li>Portfolio entrepreneurship is popular among farm households.</li> <li>Developmental</li> <li>Portfolio entrepreneurship is popular among farm households.</li> <li>Developmental</li> <li>Portfolio entrepreneurship is popular among farm households.</li> <li>Developmental</li> <li>Internet access stimulates entrepreneurs and producers.</li> <li>Developmental</li> <li>Internet access stimulates entrepreneurship in larger communities.</li> <li>Institutional</li> <li>Gender</li> <li>Women entrepreneurs start dynamic and often innovative firms.</li> <li>Marketing</li> <li>Rural entrepreneurs prefer local resource bases to non-local ones.</li> </ul>	
<ul> <li>)* Embeddedness In-migrants help integrate rural economies into global markets.</li> <li>Regional Role models increase rural entrepreneurial activity levels.</li> <li>Institutional Weak regulatory protection discourages external collaboration.</li> <li>Developmental Portfolio entrepreneurship is popular among farm households.</li> <li>Developmental The farm sector's legacy of separation impedes entrepreneurship.</li> <li>Developmental The farm sector's legacy of separation impedes entrepreneurship.</li> <li>Developmental Internet access stimulates entrepreneurs and producers.</li> <li>Developmental Internet access stimulates entrepreneurship in larger communities.</li> <li>Institutional Koids are a source of market exclusion.</li> <li>Marketing Internet portals are predominantly used for local trade.</li> <li>Embeddedness Rural entrepreneurs prefer local resource bases to non-local ones.</li> </ul>	
RegionalRole models increase rural entrepreneurial activity levels.InstitutionalWeak regulatory protection discourages external collaboration.DevelopmentalPortfolio entrepreneurship is popular among farm households.DevelopmentalPortfolio entrepreneurship is popular among farm households.DevelopmentalPortfolio entrepreneurship is popular among farm households.DevelopmentalThe farm sector's legacy of separation impedes entrepreneurship.OccupationalFarmers consider themselves entrepreneurs and producers.DevelopmentalInternet access stimulates entrepreneurship in larger communities.InstitutionalInternet access stimulates entrepreneurship in larger communities.MarketingWomen entrepreneurs start dynamic and often innovative firms.MarketingInternet portals are predominantly used for local trade.EmbeddednessRural entrepreneurs prefer local resource bases to non-local ones.	
InstitutionalWeak regulatory protection discourages external collaboration.DevelopmentalPortfolio entrepreneurship is popular among farm households.DevelopmentalPortfolio entrepreneurship is popular among farm households.DevelopmentalThe farm sector's legacy of separation impedes entrepreneurship.OccupationalFarmers consider themselves entrepreneurs and producers.DevelopmentalInternet access stimulates entrepreneurship in larger communities.InstitutionalInstitutional voids are a source of market exclusion.MarketingWomen entrepreneurs start dynamic and often innovative firms.EmbeddednessRural entrepreneurs prefer local resource bases to non-local ones.	
DevelopmentalPortfolio entrepreneurship is popular among farm households.DevelopmentalThe farm sector's legacy of separation impedes entrepreneurship.OccupationalFarmers consider themselves entrepreneurs and producers.DevelopmentalInternet access stimulates entrepreneurship in larger communities.InstitutionalInstitutional voids are a source of market exclusion.GenderWomen entrepreneurs start dynamic and often innovative firms.MarketingInternet portals are predominantly used for local trade.EmbeddednessRural entrepreneurs prefer local resource bases to non-local ones.	
DevelopmentalThe farm sector's legacy of separation impedes entrepreneurship.OccupationalFarmers consider themselves entrepreneurs and producers.DevelopmentalInternet access stimulates entrepreneurship in larger communities.InstitutionalInstitutional voids are a source of market exclusion.GenderWomen entrepreneurs start dynamic and often innovative firms.MarketingInternet portals are predominantly used for local trade.EmbeddednessRural entrepreneurs prefer local resource bases to non-local ones.	
OccupationalFarmers consider themselves entrepreneurs and producers.DevelopmentalInternet access stimulates entrepreneurship in larger communities.InstitutionalInstitutional voids are a source of market exclusion.GenderWomen entrepreneurs start dynamic and often innovative firms.MarketingInternet portals are predominantly used for local trade.EmbeddednessRural entrepreneurs prefer local resource bases to non-local ones.	
DevelopmentalInternet access stimulates entrepreneurship in larger communities.InstitutionalInstitutional voids are a source of market exclusion.GenderWomen entrepreneurs start dynamic and often innovative firms.MarketingInternet portals are predominantly used for local trade.EmbeddednessRural entrepreneurs prefer local resource bases to non-local ones.	Quantitative Finland
InstitutionalInstitutional voids are a source of market exclusion.GenderWomen entrepreneurs start dynamic and often innovative firms.MarketingInternet portals are predominantly used for local trade.EmbeddednessRural entrepreneurs prefer local resource bases to non-local ones.	Quantitative Canada
GenderWomen entrepreneurs start dynamic and often innovative firms.MarketingInternet portals are predominantly used for local trade.EmbeddednessRural entrepreneurs prefer local resource bases to non-local ones.	Qualitative Bangladesh
MarketingInternet portals are predominantly used for local trade.EmbeddednessRural entrepreneurs prefer local resource bases to non-local ones.	Quantitative Greece
Embeddedness Rural entrepreneurs prefer local resource bases to non-local ones.	ixed UK
	Qualitative Denmark
Lee & Cowling (2015) Regional Rural firms are likely to perceive regulation as a problem. Quanti	Quantitative UK
Mason et al. (2010) Regional Rural areas have the highest proportion of home-based businesses. Quantitative	antitative UK
Meccheri & Pelloni (2006)* Institutional Institutional rural policies overestimate formal training's value. Quanti	Quantitative Italy
Smallbone et al. (1999) Developmental Proactive product development helps overcome limited markets. Quanti	Quantitative UK
owledge of rural entrepr	nked at ABS Grade 3 or high

Table 2.4. Rural Entrepreneurship Research

As discussed by Baumgartner et al. (2013), several attributes typical to rural settings have been shown to affect entrepreneurial activities. First, prior research emphasises the effects of dense social networks, which facilitate access to and sharing of resources (Ring et al., 2010; Rooks et al., 2016). Second, industry specialisation and clustering is a prominent feature (Johannisson, 1998; Lechner and Leyronas, 2012), which results in regions varying highly in their competitiveness (Brünjes and Diez, 2013). While some areas benefit from such developments (e.g. Shropshire's evolution into a creative 'hub', see Bell and Jayne, 2010) other places 'experience a spiral of economic decline' (McKeever et al., 2015: 51), which can lead to the depletion of local resources and a loss of purpose and identity (Johnstone and Lionais, 2004). At the same time, extant studies have shown that rural entrepreneurs usually benefit from (1) comparatively sheltered regional markets, (2) greater employee loyalty (mainly a result of limited alternatives for work) and (3) lower costs for acquiring land and premises (Kalantaridis and Bika, 2006). Thus, rural areas do not necessarily limit business activities; they can provide a host of opportunities for entrepreneurship (Alsos et al., 2014).

#### 2.5. Rural Locations in this Study

This study investigates two rural locations, which are given the pseudonyms 'Artisan Village' (Scotland) and 'Craft County' (England). As shown in the regional map of the UK (Figure 2.2), the research sites are based in South West Scotland and North East England. While investigating entrepreneurs in two countries may suggest a comparative study, it is important to highlight that this research does not seek to contrast rural entrepreneurship in England and Scotland for several reasons. First, phenomenological research explores the lived experiences of individuals and as such, does not lend itself to generalisations (see Chapter 5). Second, although participants live in different countries, both are part of the United Kingdom and the research sites are less than 200 miles apart. Third, as described below, the stories of both places are similar, which indicates that the economic conditions for developing a business in the two regions are comparable.

Figure 2.2. Rural Regions in this Study



Artisan Village and Craft County are both located in former mining areas and consequently suffered severely when the reserves of the mines were depleted. The situation of both regions deteriorated when the demand for labour decreased in other main industries, such as agriculture. Out-migration, rising unemployment and economic downturn created various social problems, including increasing crime and poverty levels. In other words, these areas could be described as 'depleted' or 'deprived' (Lee et al., 2011; McKeever et al., 2015). To support the regeneration process, both regions introduced support initiatives designed to stimulate local development by increasing tourism to the areas. Furthermore, as the research sites are within 15-20 minutes drive from a bigger settlement (with over 10,000 inhabitants) and only 30-60 minutes from the nearest city, they are located in accessible rural areas and thus benefit from comparatively good public services and

infrastructure. Transportation options comprise bus and/or train links to nearby towns and local facilities include at least one grocery store, bank, pub, primary school and post office. A main attraction of both regions is the natural beauty of the countryside, which is why tourism is fostered.

#### 2.5.1. Artisan Village (Scotland)

The main difference between the research sites is that participants in Scotland are all located in the same village (which is home to less than 5,000 inhabitants), while participants in England are spread across the county and reside in different villages. Thus, the residents of Artisan Village know each other (although the identities of participants were kept confidential), whereas participants who live in Craft County do not. In presenting these locations, this section draws on the experiences of participants to convey the place-related meanings that they associate with these spatial contexts (Korsgaard et al., 2015; Steyaert and Katz, 2004). For details about participant selection and data collection, see Chapter 5.

In Diana's words, '[Artisan Village] is a charming little place with lots of independent shops, with many creative people ... There's a rich history in this area of arts and crafts'. George elaborates that 'it just feels very, very peaceful, very calm ... the atmosphere in the town'. In addition, 'it's quite a stimulating place to be ... I like the fact that there are like-minded people here, I like the fact that it's very beautiful and quiet. I'm very lucky to live here' (Emily). Carol agrees that 'It's a nice community ... it seems to be very, very accepting'. However, as Holly points out 'it has its element, you know, its dodgy element, as any town does'. Especially 'the local youth ... can cause trouble' (George). Furthermore, as 'it's a small village, people make up stories about what's going on and what's happened and all the rest of it' (Emily). Nevertheless, the comments of participants are largely positive and they appear to enjoy living in the area. Participants believe that the recent resurgence of craft businesses also helped improve the situation in the area. While, in the past, 'there were so many places boarded up, so many places for sale' (Diana), nowadays 'there are people who sell craft work' (Emily) and 'they do free workshops' (George), which 'has helped revive the town centre' (Carol).

# 2.5.2. Craft County (England)

In many ways, Craft County is similar to Artisan Village and its surroundings. As Julie describes,

at one time [Craft County] was like an oasis of cultural nothing really, there was a cathedral that everybody loves ... but there was very little else that people really knew about... and I think that's improving all the time and people are beginning to realise that there is a lot of potential in this area.

Betty elaborates that 'there are some really lovely villages ... and they've got like lots of little craft shops and ... they do embrace their traditions and their traditional craft'. According to Frank, 'it's really nice [and] quiet ... They're beautiful surroundings, really calming. It's nice to be in with nature'. Isla agrees that 'it's a great quality of life ... very pleasant, very green'. However, as Amy points out, 'everyone up here's a bit ... sort of stuck in the past, it's not very forward-thinking ... my client market isn't up here'. Similarly, Frank explains that 'there are nice people here, don't get me wrong, but it's very, very working class'. In addition, 'there are a lot of unemployed people, as well, in the area' (Julie). Nevertheless the region seems to be developing and Isla believes that

a lot of people would say the North East is deprived ... Yes, there are aspects of deprivation ... but there is also a huge amount of support if you want it ... there are many ... very forward thinking people and they are there helping other creative people.

As Frank notes, 'they [the council] set up these creative courses', which Amy describes as 'very good for ... connecting people and sort of leading you on to different things'. In Betty's opinion, 'they just make you want to go out there and set something up and they ... help you understand how you can do that'.

# 2.6. Conclusion

This chapter examined contemporary debates regarding cultural industries and rural entrepreneurship. It also introduced the research participants and the rural areas in which they reside. There are a number of conclusions that can be drawn from this discussion. First, cultural industries consist of a large range of interconnected communities, which make a significant contribution to the global economy. As the sector is under-studied, dedicated research is both timely and warranted. Second, certain characteristics are common across cultural industries, including the oversupply of skilled labour and the lack of objective measures to judge the value of cultural goods, which can affect entrepreneurial efforts. Furthermore, entrepreneurs in the cultural industries need to reconcile various tensions, such as those between artistic and economic objectives, and they are faced with low incomes and high failure rates. Thus, pursuing a creative career is often a demanding, risky endeavour.

Given the limited amount of research into cultural industries and rural entrepreneurship, there are a number of enduring gaps in understanding. First, this chapter revealed that scant research has adopted an entrepreneurship perspective to investigate the cultural industries. Second, extant studies concentrate on a limited number of subjects, including networking, legitimacy and sociocultural differences. Craft research in particular tends to focus on craft guilds, motivations and authenticity. In comparison, the influence of embeddedness in communities on entrepreneurs and their resource management practices have not received adequate consideration. Within the rural entrepreneurship literature, little attention has been afforded to exploring the link between cultural industries and rural locations. This study therefore advances understanding by investigating the lived experiences of entrepreneurs who practise crafts in rural locations. By examining the influence of different communities on the practices of participants, this study seeks to make valuable contributions to knowledge.

#### **CHAPTER 3: BOURDIEU'S THEORY OF PRACTICE**

#### **3.1. Introduction**

Pierre Bourdieu (1930-2002) was a French philosopher, sociologist and anthropologist whose works have had a major impact on the study of culture, social class and education, among other significant arenas of social science (Golsorkhi et al., 2009; Swartz, 2008). Influenced by the writings of Marx (1887), Durkheim (1964) and Weber (1968), Bourdieu developed one of the most sophisticated 'grand theories' (defined as 'theoretical concepts that describe and explain the overall functioning and dynamics of a given social order', see Chudzikowski and Mayrhofer, 2011: 20). His conceptual framework, called the theory of practice (Bourdieu, 1977), consists of the three interrelated master concepts of capital, habitus and field. As shown in Figure 3.1, Bourdieu proposes that individuals compete for economic, cultural, social and symbolic *capital* to gain dominant positions within social *fields* and that their actions are directed by a socially shaped *habitus*.

Figure 3.1. Bourdieu's Theory of Practice



Adapted from Pret and Carter (2016: 37)

The objective of this chapter is to introduce and discuss Bourdieu's theories and to consider their implications for entrepreneurship research. This chapter first critically analyses Bourdieu's impact on management research, before discussing this study's approach to his work. It then explores each of his master concepts in turn, relating them to the context of craft entrepreneurship and discussing critiques of their explanatory capacity.

#### 3.2. Bourdieu's Impact on Management Research

Over time, there have been repeated calls for business and management research to employ Bourdieu's theories (cf. Özbilgin and Tatli, 2005; Tatli et al., 2014). Using a citation context analysis of leading management journals, Sieweke (2014) provides a comprehensive overview of the use of Bourdieu's concepts by business scholars. It reveals a steady increase of citations of Bourdieu's work, highlighting that, since the early 2000s, scholars have begun engaging much deeper with Bourdieu's theories.

Growing scholarly interest in Bourdieu's work can be attributed to the burgeoning popularity of practice theories (Feldman and Worline, 2016), which has resulted in the development of several interrelated concepts and approaches, including the 'practice perspective' (Keating et al., 2014), 'strategy-as-practice' (Chia and Holt, 2006) and the 'practice turn' in management and organisational research (Miettinen et al., 2009). As Sandberg and Dall'Alba (2009: 1349) explain, 'Pierre Bourdieu (1977), Anthony Giddens (1984) and, more recently, Theodore Schatzki (2002) have had considerable impact ... through developing extensive theories of practice'. While these scholars developed distinct theories, they share a set of underlying assumptions (Feldman and Orlikowski, 2011): (1) everyday actions are consequential in producing social life, (2) dualisms should be rejected (e.g. between body/mind and objective/subjective) and (3) phenomena always exist in relation to one another and are mutually constituted.

A main theme connecting these ideas is the tenet that agency and structure are inextricably intertwined. In contrast to prior research, which has emphasised the role of structure over agency (cf. Ahuja, 2000; Zaheer and Soda, 2009), practice theories contend that agency is key to (re-)producing social orders (e.g. routines or institutions), while simultaneously acknowledging that 'agency cannot be understood "simply" as

human action, but rather must be understood as always already configured by structural conditions' (Feldman and Worline, 2016: 1243). In other words, practice theorists argue that social norms and beliefs not only influence individual practices, they themselves are constantly (re-)created through everyday (inter-)actions.

An underlying cause for the increasing interest in practice theories in general, and Bourdieu's theory of practice in particular, is their ability to facilitate multilevel research, connecting individuals and their actions to social orders and wider social fields, such as organisations and institutions (Özbilgin and Tatli, 2005). However, a major criticism of empirical research that employs Bourdieu's framework is its tendency to focus on only one of his concepts rather than apply the full complement of his theories (Golsorkhi et al., 2009; Swartz, 2008). For example, Emirbayer and Johnson (2008) criticise management research's 'almost total inattention to habitus ... without which the concepts of field and capital ... make no sense, [which] further attests to the misappropriation of his ideas'. Furthermore, Sieweke's (2014) systematic review of the literature reveals that capital theory, while frequently cited, is rarely comprehensively used.

Despite these criticisms, this research finds that an increasing number of studies adopt Bourdieu's full theory of practice. As Table 3.1 shows, apart from some exceptions (e.g. DiMaggio, 1979; Özbilgin and Tatli, 2005), most such studies have been published in the past decade. Thus, it appears that the repeated calls for research to employ the full theory of practice are being heeded. This table also shows that entrepreneurship and organisational perspectives are frequently adopted. Bourdieu's theory of practice is used to explore a broad range of contexts, from consulting and accounting to academia and creative industries. Interestingly, 10 studies were found to include the concepts of doxa and/or illusio (explained below), which are otherwise rarely cited (Sieweke, 2014). Regarding methods, about half of the studies are theoretical. Empirical research is primarily qualitative, which is likely due to the significant insight into participants' personal lives that is required to employ Bourdieu's theories in depth. The findings of these studies are broad, highlighting the utility of his concepts for exploring human behaviours and advancing understanding of various subjects, including entrepreneurial legitimacy and transnational entrepreneurship.

Context	Article	Perspective	Key Finding	Main Method
Academia	Drakopoulou Dodd et al. (2014)*	Entrepreneurship	Drakopoulou Dodd et al. (2014)* Entrepreneurship Qualitative scholars passionately engage with subjects, peers and text.	Quantitative
	Golsorkhi et al. (2009)*	Organisational	Scholars have a moral responsibility to reflect upon their practices.	Theoretical
	Karataş-Özkan & Chell (2015) Gender	Gender	Gender inequalities affect academics' experiences of entrepreneurship. Qualitative	Qualitative
Accounting	Spence et al. (2016)	Occupational	Habitus can elucidate the informal norms of national business systems. Qualitative	Qualitative
	Stringfellow et al. (2014)*	Entrepreneurship	Fitting in with prevailing logics can increase access to resources.	Qualitative
Consulting	Vincent (2016)*	Gender	Structuring lives according to field routines improves resource access.	Qualitative
Creative industries	Drakopoulou Dodd (2014)	Entrepreneurship	Places lacking in economic resources can still enable entrepreneurship. Qualitative	Qualitative
	Karataş-Özkan (2011)	Entrepreneurship	Learning is achieved through interactions and dealing with uncertainty. Qualitative	Qualitative
Culture	Spence et al. (2017)*	Occupational	Cultural capital is a salient source of distinction between elite factions.	Qualitative
	Spigel (2013)*	Entrepreneurship	Cultural outlooks influence entrepreneurial actors and vice versa.	Theoretical
Health care	Lockett et al. (2014)	Organisational	Individuals' endowments of cultural capital shape their sensemaking.	Qualitative
	Witman et al. (2010)	Occupational	Managers have to be renowned doctors to influence clinical activities.	Qualitative
Legitimacy	De Clercq & Voronov (2009a)	Entrepreneurship	Cultural and symbolic capital shape one's ability to attain legitimacy.	Theoretical
	De Clercq & Voronov (2009b)*	Entrepreneurship	Legitimisation is a reciprocal process that perpetuates power relations.	Theoretical
	De Clercq & Voronov (2009c)	Entrepreneurship	Entrepreneurs have to fit in with extant norms while also standing out.	Theoretical
Literature Review	DiMaggio (1979)	Organisational	Fields signify both common purpose and arenas of strategy or conflict.	Theoretical
	Emirbayer & Johnson (2008)*	Organisational	Successful organisational field analysis requires the concept of habitus.	Theoretical
	Friedland (2009)*	Organisational	Field theory over-simplifies human behaviour as contest for power.	Theoretical
	Özbilgin & Tatli (2005)	Organisational	Practice theory overcomes the dualities between agency and structure.	Theoretical
	Tatli et al. (2014)*	Entrepreneurship	A relational perspective allows overcoming dominant dichotomies.	Theoretical
Minority groups	De Clercq & Honig (2011)	Entrepreneurship	Disadvantaged persons can modify perceptions of entrepreneurship.	Theoretical
Transnational firms	Transnational firms Patel & Conklin (2009)	Entrepreneurship	Higher levels of biofocality facilitate transnational venture activity.	Quantitative
	Patel & Terjesen (2011)	Entrepreneurship	Entrepreneurship Network range facilitates resource access in the presence of strong ties. Quantitative	Quantitative
	Terjesen & Elam (2009)	Entrepreneurship	Entrepreneurship Diverse resources help navigate multiple institutional environments.	Qualitative

\* Article refers to Bourdieu's concepts of *illusio* and/or *doxa* Selected articles employ Bourdieu's theory of practice in a meaningful way and were published in journals ranked at ABS Grade 3 or higher (Wood & Peel, 2015).

Table 3.1. Research utilising Bourdieu's Theory of Practice

Studies that draw primarily on only one of Bourdieu's theories are presented in Table 3.2. While the majority of these studies focus on capital theory, several were also found to utilise habitus or field theory. Unlike the research displayed in Table 3.1, only a small number of these studies refer to the concepts of doxa or illusio. However, similar to those utilising the full theory of practice, most of these studies were published during the past decade (a notable exception is Anheier et al., 1995), which demonstrates the mostly contemporary nature of interest in Bourdieu's approach. While the studies examined were purposefully selected for their exploration of entrepreneurship-related topics and are therefore not representative of all research that employs Bourdieu's theories, they also incorporate discussions of cultural, organisational and occupational issues. The findings of these studies advance knowledge of a range of subjects, including ethnic communities, social positioning and capital development. As such, this table highlights the broad range of topics that can be explored with Bourdieu's theories. While several quantitative studies are featured in this table, these almost exclusively adopt a capital lens (a notable exception is Maclean et al., 2014). This indicates that explorations of field and habitus usually require qualitative or theoretical approaches. With regards to the countries in which empirical studies are conducted, European contexts are the most common. This chimes with observations of other scholars who note that Bourdieu's influence is concentrated within Europe, rather than extending into North America (DiMaggio, 1979; Drakopoulou Dodd, 2014).

# 3.3. This Study's Approach to Bourdieu's Work

As might be expected for a major social theory of this scale (Chudzikowski and Mayrhofer, 2011), Bourdieu's work has been criticised for a number of reasons (DiMaggio, 1979). While relevant limitations of the theory of practice are discussed throughout this chapter, it should be noted that this is not an in-depth study of Bourdieu's work. The British Sociological Association (BSA) provides a comprehensive list of publications (over 375 recent articles, books and PhD theses) that analyse and critique Bourdieu's work (BSA, 2017). In this study, Bourdieu's conceptual framework is employed as a means of exploring the perspectives of

Main Theory Article	ry Article	<b>Perspective</b>	Key Finding	Main Method Country	Country
Capital	Al Ariss & Syed (2011)	Occupational	Migrants rely on several sources of capital to enjoy career success.	Qualitative	France
	Anheiher et al. (1995)	Cultural	Differences in cultural capital separate high and low culture.	Quantitative	Germany
	Fuller & Tian (2006)	Entrepreneurship	The power inherent in social relations can foster ethical behaviours.	. Qualitative	UK
	Harvey & Maclean (2008)	Cultural	Serving on boards facilitates business elites' cultural reproduction.	Multiple	Multiple
	Harvey et al. (2011a)	Entrepreneurship	Philanthropic investments help develop non-material capitals.	Archival	NSA
	Honig et al. (2006)	Entrepreneurship	Producers for the defence industry have greater access to capital.	Quantitative	Israel
	Jayawarna et al. (2014)	Entrepreneurship	Human capital development is a lifelong journey that starts early.	Quantitative	UK
	Lee & Shaw (2016)	Entrepreneurship	Professional entrepreneurs possess valuable non-material capitals.	Qualitative	UK
	Lehner (2014)	Entrepreneurship	Cultural and symbolic capital can moderate resource exchanges.	Qualitative	Multiple
	Nicolopoulou (2014)	Entrepreneurship	Multiple capitals should be explored within social entrepreneurship.	. Theoretical	Multiple
	Prieur et al. (2008)	Cultural	Cultural capital is not always an outcome of cultural consumption.	Quantitative	Denmark
	Scott (2012)*	Entrepreneurship	Capital conversion can facilitate the creation of cultural goods.	Qualitative	NZ
	Shaw et al. (2013)	Entrepreneurship	Blending forms of capital is a defining feature of philanthropy.	Mixed	UK
	Townley et al. (2009)	Cultural	Each creative industry relies on different forms of capitals.	Theoretical	Multiple
	Vershinina et al. (2011)	Entrepreneurship	Within ethnic communities, access to forms of capital varies widely. Qualitative	. Qualitative	UK
Field	Battilana (2006)	Entrepreneurship	Social position enables one to act as an institutional entrepreneur.	Theoretical	Multiple
	De Clercq & Voronov (2011)* Entrepreneu	* Entrepreneurship	Strategic actions can reduce impacts of field-specific expectations.	Theoretical	Multiple
	Maclean et al. (2014)	Cultural	Social origin remains a powerful driver in determining success.	Quantitative	France
Habitus	Barrett & Vershinina (In Press) Entrepreneu	) Entrepreneurship	Entrepreneurs' ethnicity is pertinent, but does not define activities.	Qualitative	UK
	McDonough (2006)*	Occupational	Public service creates civil servants through socialization in habitus.	. Qualitative	Canada
	McKeever et al. (2014)*	Entrepreneurship	Social capital is an enabler for building a wider community capital.	Qualitative	Ireland
	Newman et al. (2013)	Cultural	Life course experiences affect cultural capital, habitus and class.	Qualitative	UK
	O'Mahoney (2007)	Organisational	Organisations act as sites of negotiation for ethical forms of habitus. Qualitative	. Qualitative	UK
	Spence & Carter (2014)	Organisational	Habitus allows disembodiment of constraining institutional logics.	Qualitative	Multiple

Table 3.2. Research utilising Bourdieu's Master Theories

Selected articles employ Bourdieu's field, habitus or capital theory and were published in journals ranked at ABS Grade 3 or higher (Wood & Peel, 2015).

participating entrepreneurs. As explained in Chapter 5, the theory of practice was not selected a priori, but as a result of the data analysis process.

Furthermore, rather than trying to incorporate the vast body of literature published by Bourdieu in over 40 years (several hundred articles and over 30 books), this study primarily draws on three core publications (Bourdieu, 1977, 1984, 1986). As the researcher's command of the French language is limited, the selected publications are not originals, but rather translations by Richard Nice, whose work has been described as 'excellent' (DiMaggio, 1979: 1466). These versions have been updated, as indicated below (Bourdieu, 1977; vii):

Outline of a Theory of Practice was first published in French in 1972 ... However, this English text incorporates most of the changes which Pierre Bourdieu has made since then. The argument is carried further, particularly as regards the concepts of practical logic and symbolic capital.

Thus, it could be argued that the selected versions are in some ways superior to the originals. By drawing on his three core publications only, this study is also able to employ a largely consistent set of definitions of key concepts. While there is merit in analysing the changes in Bourdieu's concepts over time (cf. Desan, 2013; DiMaggio, 1979), this study does not seek to engage in these discussions.

Another endeavour this study does not undertake is to provide an in-depth comparison of the theory of practice and alternative conceptual models that are used to analyse and explain entrepreneurial behaviours (for detailed discussions, see Emirbayer and Mische, 1998; Feldman and Orlikowski, 2011; Schatzki, 1997). While other theories have been adopted to explore embeddedness in social contexts, including structuration theory (Jack and Anderson, 2002) and social network theory (Ozcan and Eisenhardt, 2009), Bourdieu's framework was selected because of its conceptual clarity concerning the various resources entrepreneurs can access and share with others (Karataş-Özkan, 2011; Shaw et al., 2013), as well as for its appreciation of the influence of norms and conventions (habitus) within social fields (De Clercq and Honig, 2011; Stringfellow et al., 2014). Since all 'grand social theories' have both their advocators and critics (Chudzikowski and Mayrhofer, 2011), selecting an analytical framework should be based on its applicability to a research

setting and on the preferences of the researcher. The findings chapters (Chapters 7-9) demonstrate that Bourdieu's theory of practice is appropriate for this context.

# 3.4. Forms of Capital

Bourdieu argues throughout his works that the accumulation of various forms of capital is a primary goal of human activity. Importantly, Bourdieu does not propose that capital exists only in economic form (DiMaggio, 1979), but differentiates between economic, cultural, social and symbolic capital (see Figure 3.2). He argues that these forms of capital can be exchanged and transformed into various goods, services and other resources (Karataş-Özkan, 2011). Most capital accumulation and conversion processes are time-consuming but, once acquired, Bourdieu (1986: 241) proposes that capital has 'a tendency to persist in its being' and 'reproduce itself in identical or expanded form'. As the conversion of capital is a major aspect of the theory of practice (Anheier et al., 1995), the following discussion outlines the transformation processes proposed by Bourdieu. Importantly, the term conversion is used to denote the creation of new capital from existing capital, but does not entail the loss of the original form of capital. For example, craft entrepreneurs can convert their cultural capital into various other capitals by drawing on their experience and training to create valuable products, but they do not lose this capital in the process. Only financial assets are usually lost to the agent during the conversion process.





Importantly, every social context (field) develops its own forms, volumes and distributions of capital (Drakopoulou Dodd, 2014). While some forms, such as economic and cultural capital, are more likely to be useful and convertible between different contexts, others are field-specific and lose their value outside of a particular social sphere (e.g. symbolic capital). As capital is generally linked to individual agents and unequally distributed among different field members, those with larger amounts of capital are able to acquire dominant positions, better connections and higher levels of distinction within a field (Golsorkhi et al., 2009).<sup>3</sup> Thus, Bourdieu argues that individuals act strategically to increase the various forms of capital at their disposal in order to gain power over others (Maclean et al., 2010).

# 3.4.1. Economic Capital

Economic capital presents itself in the form of financial assets and other tangible and intangible business assets (Anheier et al., 1995). For craft entrepreneurs, tangible business assets include workshops, tools, materials and stock. Intangible assets generally comprise intellectual property rights, such as patents and trademarks. Furthermore, distinctive patterns and designs are of particular importance to craft makers (Becker, 1978). It has been argued that economic capital is perhaps the most critical asset for entrepreneurs, as the availability of and access to financial resources can directly influence entrepreneurial success (Allison et al., 2015; De Bettignies, 2008). In contrast, Bourdieu (1986) argues that economic theory has overemphasised the role of economic capital. He proposes that human behaviour can only be fully understood if the 'universe of the artist and the intellectual and the gratuitous activities of art-for-art's sake' (p. 242) are also considered.

Accordingly, Bourdieu highlights the importance of non-economic forms of capital (Lee and Shaw, 2016). Nevertheless, he maintains that economic capital is the most mutable form, as it can be used to purchase training and education (cultural capital) as well as memberships in networks (social capital). Concerning conversions of other forms of capital into economic capital, Bourdieu proposes that these

<sup>&</sup>lt;sup>3</sup> Bourdieu refers to agents rather than subjects (associated with structuralism) or actors (related to individualism) (Golsorkhi et al., 2009). Incidentally, this designation is also suitable for entrepreneurs, who are often described as 'agents of change' (cf. Anderson, 2000; Steyaert and Katz, 2004).

processes are more complex (Anheier et al., 1995). As explained in the subsequent sections, however, such transformations are still carried out.

# 3.4.2. Cultural Capital

Bourdieu (1986) conceptualises three forms of cultural capital. First, in its embodied state, cultural capital manifests in the form of 'long lasting personal dispositions of the mind and the body' (p. 243). Transmission of this type of capital is time-consuming and requires socialisation (Dana and Light, 2011), with family background and status playing an important role in the amount and value of cultural capital imparted to individuals (Anderson and Miller, 2003; Harvey et al., 2011b). It is particularly important for businesses in the cultural industries that their customers possess this type of capital, as cultural understanding influences the abilities of consumers to appreciate the value of creative products and to make purchasing decisions (McLeod et al., 2009; Townley et al., 2009).

Second, in its objectified state, cultural capital materialises as cultural goods, such as books or paintings (Bourdieu, 1986). As legal ownership of such goods can be transferred, it is comparably easy to pass on this form of cultural capital or exchange it with other agents (Vershinina et al., 2011). However, the value of objectified cultural capital will only be realised if the receiving party is able to consume the good (i.e. possess the required embodied cultural capital to appreciate its value). For craft entrepreneurs, this objectified state is important: craft items hold a cultural meaning, imbuing their products with more than purely economic worth.

Third, in its institutionalised state, cultural capital refers mainly to educational qualifications. As Bourdieu (1977: 187) describes it, 'academic qualifications are to cultural capital what money is to economic capital'. Similar to money, academic qualifications hold a value which is not tied to a specific location, which is guaranteed by law, and which endures over time. Once achieved, such credentials do not continuously need to be proven, but remain linked to an individual, making it less mutable than other forms.

Given its focus on education, cultural capital is markedly similar to the concept of human capital (Becker, 1964). As human capital is more easily measured and has received more attention within the literature (Marvel et al., 2016; Unger et

al., 2011), there are a number of studies that employ the concept of 'human-cultural capital' instead of differentiating between the various forms of cultural capital (cf. Ram et al., 2008; Shaw et al., 2008). Furthermore, it should be mentioned that most business research includes the training and experience required to attain educational qualifications in their definitions of cultural capital (cf. Harvey et al., 2011a; Lockett et al., 2014; Terjesen and Elam, 2009). Within a cultural industries context, Karataş-Özkan (2011) argues that creativity, artistic ability and sector knowledge are also relevant. This study follows these examples and broadens the concept of cultural capital to incorporate these crucial aspects of experience and competence.

#### 3.4.3. Social Capital

Bourdieu (1986) defines social capital as the sum of all actual and potential resources that can be accessed through a durable network of relationships. Such relationships can either be casual or socially instituted. Bourdieu (1986) argues that while physical proximity is not a requirement, it usually facilitates exchanges between partners. Furthermore, the more people linked to an individual and the more resources these connections possess, the higher the amount of social capital available (Honig et al., 2006). Although Bourdieu acknowledges that access to resources is not necessarily a conscious goal of agents when they develop connections, the profits that result from capital exchanges are a main reason for the solidarity among network members (Vincent, 2016). Within a business context, the acquisition of social capital is often a deliberate endeavour, as access to collectively owned and shared network resources can provide major advantages (Adler and Kwon, 2002; Nahapiet and Ghoshal, 1998), especially for entrepreneurs (Gedajlovic et al., 2013; Stam et al., 2014).

Bourdieu (1986: 249-250) argues that maintaining connections requires sustained efforts and employment of 'investment strategies ... aimed at establishing or reproducing social relationships that are directly usable ... or institutionally guaranteed'. Business owners, such as craft entrepreneurs, must therefore make conscious efforts not only to build relationships, but also to maintain them over time (Lechner and Dowling, 2003). Similar to cultural capital, social capital can be inherited from family members, though continuous investments are required to

preserve such relationships (Vershinina et al., 2011). Furthermore, relocations can complicate the storage of social capital (Ram et al., 2008).

Social capital plays an important role in the development of cultural capital, as access to network members can help agents become socialised into norms of behaviour (Lockett et al., 2014). It can also be used to substitute for a lack of cultural capital, for example when connections provide access to employment opportunities that are otherwise reserved for those with a requisite qualification (Bourdieu, 1984). Social capital is converted into economic capital when agents receive gifts, services or other physical artefacts from their connections (Bourdieu, 1986). However, there is a risk of non-equivalent returns on investment, as network members may chose not to reciprocate services or favours. However, as De Carolis and Saparito (2006) note, such risks are a price most entrepreneurs are willing to pay to enjoy the benefits social capital provides.

# 3.4.4. Symbolic Capital

Once any of the aforementioned types of capital are recognised by others in the same field, they can generate symbolic capital (Bourdieu, 1986). Symbolic capital is usually associated with the possession of prestige, status and reputation (Wilson et al., 2017). It is objectified in awards and recognitions, but its actual value depends on the importance others attribute to such capital (Fuller and Tian, 2006). Consequently, the value of symbolic capital is more subjective than other forms; it is connected to a specific field and reliant upon the expectations of its members (Shaw et al., 2017). Symbolic capital is particularly important for entrepreneurs 'in producing "symbolic effects"; attracting explicit or practical recognition, capturing legitimacy and securing profits' (Stringfellow et al., 2014: 572). Securing belief in the quality of new products and services is a significant outcome of possessing symbolic capital, along with the ability to reduce uncertainty among stakeholders and potential investors (De Clercq and Voronov, 2009c). For craft entrepreneurs, acquisition of status and reputation has been shown to matter greatly (Lamertz et al., 2005; Sayce et al., 2007; Svejenova et al., 2007).

Emirbayer and Johnson (2008) suggest that symbolic capital can be selfgenerating. If a firm continuously maintains its authority and reputation, others will gradually come to expect a high level of performance that does not necessarily need to be verified through the acquisition of new symbolic capital. As Kerr and Robinson (2016: 703) note, symbolic capital has been differently interpreted by scholars, either as a separate form of capital or as a 'sort of meta-capital' which represents the power that is accrued from possession of the other forms of capital. In this study, in line with most entrepreneurship research (De Clercq and Voronov, 2009a; Drakopoulou Dodd et al., 2014; Harvey et al., 2011a), the former approach is adopted.

Unlike economic capital, which is comparably stable and suffers only from inflation and depreciation, symbolic capital is relatively volatile. In his writings, Bourdieu (1977) focuses on the negative effects of the deaths of prominent family members on their relatives, but this issue is similarly relevant to businesses whose reputation can be affected or even lost through unexpected incidents (Ertug et al., 2016; Reuber and Fischer, 2007). Concerning its transmutability, Bourdieu (1977: 179) argues that symbolic capital can be used to gain access to reputable schools (to create cultural capital) and exclusive networks (social capital) while, vice versa, membership in either can also help develop symbolic capital. However, most conversions into symbolic capital come 'at the cost of a wastage of social energy' (Bourdieu, 1977: 195) and require recognition by associated group members.

# 3.4.5. Critique of Capital Theory

DiMaggio (1979) discusses several criticisms of Bourdieu's concept of capital. First, capitals proliferate – Bourdieu employs several names for the different forms of capital. For example, cultural capital is also referred to as 'scholastic capital', 'educational capital' and 'academic capital' (Bourdieu, 1984) and symbolic capital as 'capital of prestige' and 'capital of authority' (Bourdieu, 1977). As DiMaggio (1979) points out, Bourdieu does not make clear whether these alternative designations are meant to denote sub-forms of capital and, 'as the number of capitals increases, the metaphorical currency undergoes inflation and its value declines accordingly' (p. 1469). This study therefore refers only to economic, cultural, social and symbolic capital.

Another criticism relates to Bourdieu's understanding of the value of economic capital. While Bourdieu emphasises the significance of non-financial forms of capital (Lee and Shaw, 2016), he still considers economic capital the underlying 'currency of exchange', which is 'at the root of all the other types of capital' (Bourdieu, 1986: 252) and which facilitates the transactions of most capital conversions. This focus on the importance of economic capital is limiting, as it obscures other capital transformation processes that do not involve economic exchanges in any way. Accordingly, this study explores a broader range of capital conversions than those proposed by Bourdieu.

A further weakness lies in the fact that Bourdieu's work is primarily based on observations of French society and its colonies in the 20th century. As a result, some of his conceptualisations are less applicable to other, more contemporary contexts (DiMaggio, 1979). In addition, it has been argued that the concept of capital is under-theorised (Desan, 2013; DiMaggio, 1979). While cultural and social capital have been defined in great depth (Sieweke, 2014), economic and symbolic capital have received much less attention by Bourdieu. However, this issue is partly ameliorated by the large amount of recent studies which employ capital theory and have further conceptualised the various forms (see Table 3.2).

# 3.5. Habitus

While the different forms of capital represent 'the stakes of the game' (Drakopoulou Dodd et al., 2014: 636), habitus provides the principles, or 'rules of the game', that guide the behaviours of individual agents within fields. Thus, the concept of habitus acts as a linking mechanism connecting individual agency, including the strategies aimed at accumulating and transforming various capitals, with wider social structures (fields) (Emirbayer and Johnson, 2008). It has been argued that without an analysis of habitus, human behaviours cannot be fully understood (Chudzikowski and Mayrhofer, 2011) and business research has been criticised for a lack of attention towards the concept (Swartz, 2008). Accordingly, the following sections explore Bourdieu's concept of habitus and the related concepts of illusio and doxa, before discussing their associated criticisms.

Bourdieu (1977: 72) defines habitus (which is used both in the singular and plural) as:

systems of durable, transposable dispositions, structured structures predisposed to function as structuring structures, that is, as principles of the generation and structuring of practices and representations which can be objectively 'regulated' and 'regular' without in any way being the product of obedience to rules, objectively adapted to their goals without presupposing a conscious aiming at ends or an express mastery of the operations necessary to attain them and, being all this, collectively orchestrated without being the product of the orchestrating action of a conductor.

This definition is not only exemplary of Bourdieu's complex (and sometimes convoluted) writing style; it also illustrates the breadth of his concept of habitus. Thus, further examination of the concept is required.

First, Bourdieu proposes that habitus refers to durable principles of practice and taste that are developed during an agent's early life and upbringing (Emirbayer and Johnson, 2008). In the form of personal dispositions, habitus is directly related to embodied cultural capital and is a product of the socio-economic conditions under which it is acquired. Bourdieu (1984) argues that habitus affects individuals mainly on a subconscious level and serves as a modus operandi (Anderson et al., 2010). Habitus has been described as an 'inner strategic orientation' (McKeever et al., 2014: 457), which provides agents with a sense of how to behave in different contexts. It equips agents with a 'feel for the game' (Tatli et al., 2014: 625) and an understanding of what choices are appropriate and how to cope with unforeseen situations. Importantly, Bourdieu (1984) does not conceive of habitus as deterministic, but rather as 'a generative grammar, providing agents with [certain] creative capacities' (Golsorkhi et al., 2009: 783).

In addition to being embodied within individuals, habitus is also a constraining social structure (Friedland, 2009). Through its various members, each field develops its own overarching habitus, which affects perceptions within the field at large (Golsorkhi et al., 2009). While not imposing strict rules, habitus shapes the direction of the field (Drakopoulou Dodd et al., 2014) and influences the valuation of different forms of capital as well as the conduct of their exchanges (Patel and Conklin, 2009). Friedland (2009: 889) proposes that 'common habitus assures that practices always have a common sense, a reasonableness as opposed to a rationality'.

It is through routines of socialisation that fields, such as rural and craft communities, (re-)create habitus as 'a shared understanding of how their shared context operates' (McKeever et al., 2014: 457). For a field to function harmoniously, newcomers, such as entrepreneurs, are expected to comply with its habitus (De Clercq and Voronov, 2009b, 2009c), which requires a process of learning (Karataş-Özkan, 2011) and thus benefits long-standing field members (Martí et al., 2013). In other words, habitus predisposes agents to act in certain ways in response to pressure from the social environment (Light and Dana, 2013).

As Emirbayer and Johnson (2008) note, habitus connects past fields with present fields through the agents who move through them. Each agent contributes to a field, not only through exchanges of capital, but also through adding to and sharing their interpretations of its habitus. Habitus therefore provides a means of examining commonalities within groups as outcomes of shared experiences, lifestyles (even across generations) and historical understandings (McKeever et al., 2014). Indeed, temporality is an important aspect of habitus (Bourdieu, 1984). Early experiences are crucial, as they direct the future responses of agents and the ways they assimilate new experiences (Mutch, 2003). However, habitus changes over time as later experiences create new dispositions and modify preferences and tastes (Harvey et al., 2011b; O'Mahoney, 2007). Thus, prevailing norms and understandings within fields can change as members become accustomed to new practices. It has been argued that entrepreneurs in particular are able to effect such change through introducing innovations (De Clercq and Voronov, 2009a, 2009b).

Within the entrepreneurship literature, the concept of habitus has been applied by some studies (see Table 3.2), which have recently been reviewed by Tatli et al. (2014) and Drakopoulou Dodd et al. (2014). Among the most prominent discussions in the literature are those concerning transnational entrepreneurs, who require an understanding of habitus in both their home countries and host countries in order to be successful (Drori et al., 2006; Patel and Conklin, 2009; Terjesen and Elam, 2009). Furthermore, within a networking context, habitus has been used to identify various modi operandi employed by entrepreneurs to co-create growth (Anderson et al., 2010). Other studies have used the concept to investigate entrepreneurial identities in relation to cultural identities (Barrett and Vershinina, 2017) and to theoretically explore entrepreneurial legitimacy as an enactment of habitus (De Clercq and Voronov, 2009c). Importantly, while habitus has been examined within a range of contexts, its effects have rarely been analysed within the cultural industries (a notable exception is Randle et al., 2015).

### 3.5.1. Illusio

While habitus provides the 'rules of the game' and guides the behaviours of individuals, their interest in belonging to a field is maintained through illusio. Illusio represents 'an investment in the game' (Bourdieu and Wacquant, 1992: 98) and implies 'tacit recognition of the value of the stakes of the game' (p. 117). In other words, illusio refers to an agent's conviction that the stakes at play are meaningful and worth pursuing. By internalising the field-specific illusio, agents obtain the right to enter a field, which requires them to recognise existing power structures, rules and logics (Tatli et al., 2014). Membership in a field generates competition for power within it, which helps (re-)create the ends at stake as well as their value (Friedland, 2009). Illusio is therefore a collective understanding (or self-deception), which leads field members to pursue a shared 'dream' (Drakopoulou Dodd et al., 2014).

The stakes at play are the different forms of capital, which can be of more or less value depending on the context. While entrepreneurship research has traditionally suggested that economic and social capital are the most useful and sought after resources (Huang and Knight, 2017; Jonsson and Lindbergh, 2013), this may not always be the case. Indeed, within the context of the cultural industries, symbolic and cultural capital may be of similar importance (Jones et al., 2016; Townley et al., 2009). The differences between fields can be so significant that the struggle for certain resources, such as awards and qualifications, can seem meaningless for agents external to the field (Golsorkhi et al., 2009). Since illusio acts primarily on a subconscious level, agents may not even realise that competition for specific resources the shared illusion.<sup>4</sup>

<sup>&</sup>lt;sup>4</sup> It should be noted that Bourdieu (1977, 1984) initially refers to 'illusion' rather than 'illusio'. It is only in his later works that he employs the term 'illusio' and defines its meaning. However, the concept is still included in this study, as it helps connect capital theory with habitus and field.

Maintaining belief in the stakes of the game is particularly important for dominant agents, as it helps them consolidate power (De Clercq and Voronov, 2009b). For entrepreneurs, who often hold subordinate positions due to their initial lack of capital, submission to the illusio reinforces their lower status (Tatli et al., 2014). In other words, by seeking to fit in, entrepreneurs not only help maintain the status quo, but also submit to symbolic violence from dominating agents (De Clercq and Voronov, 2009b). The status quo can, of course, be challenged, as neither fields nor habitus are permanent. Standing out from norms may, in fact, be expected from entrepreneurs in some contexts (De Clercq and Voronov, 2009a). However, it is usually difficult to expose the true nature of a field to its members, who are ensnared by its illusio. Newcomers who are equipped with significant capital and connections from other contexts may be best positioned to achieve change, which can eventually lead to new power structures, stakes and rules of the game (Golsorkhi et al., 2009).

# 3.5.2. Doxa

Related to the 'rules of the game' (habitus) and the acceptance of its stakes (illusio) is another concept, called doxa, which is involved in the reproduction of existing power structures within a field. Bourdieu (1984: 424) defines doxa as 'the ordinary acceptance of the usual order which goes without saying and therefore usually goes unsaid'. Thus, doxa represents the adherence to existing social orders.

Doxa proscribes the explicit questioning of taken-for-granted assumptions within a field (Bourdieu, 1977). By naturalising these assumptions, the reality of field members is formatted and presuppositions are formed which set boundaries to their thinking and limit their critical faculties (Golsorkhi et al., 2009). Among others, doxa helps reproduce divisions by sex, age and class in most fields (Vincent, 2016). It is therefore in the interest of those with comparably little or no power to push at the boundaries of doxa and to point out unchallenged assumptions in order to change the perceptions of field members (Drakopoulou Dodd et al., 2014).

The development of these assumptions is usually an outcome of prolonged struggles among competing agents who have different ideals and goals (De Clercq and Voronov, 2011). When entering a new field, entrepreneurs face dominant forces and fundamental truths, such as 'art for art's sake' in the cultural industries (Bourdieu, 1984: 49), and they must find ways to reconcile their own views with those of other field members (Eikhof and Haunschild, 2007). As De Clercq and Voronov (2009b: 807) argue, by submitting to existing power relations, entrepreneurs 'become complicit in their own domination'. A relevant example would be the expectation of an initial public offering (IPO), which may be required by potential investors but may not be in the best interest of a young firm (Huang and Pearce, 2015).

Similar to habitus and illusio, doxa is field-specific. As such, certain cardinal truths, such as the importance of sustainability, may only be relevant within some contexts (De Clercq and Voronov, 2011). Thus, doxa is linked to an individual field and affects the practices of its members, who need to signal compliance with unchallenged assumptions to avoid punitive measures (De Clercq and Voronov, 2009b). Since the 'rules of the game' (habitus), the commitment to the stakes at play (illusio) and the acceptance of taken-for-granted assumptions (doxa) are all inextricably linked, Bourdieu (1977, 1984) subsumes aspects related to doxa and illusio in his discussion of habitus. Accordingly, while accounting for all three concepts, this study also focuses on habitus in its analysis of findings, which is in line with much of the management literature (see Sieweke, 2014).

# 3.5.3. Critique of Habitus

Habitus and its related concepts have been criticised by a number of scholars (DiMaggio, 1979; Mutch, 2003). Among other issues, it has been argued that habitus is too deterministic (King, 2000) and does not adequately account for clashes between individuals and the constraining social structures acting upon them (Özbilgin and Tatli, 2005). Regarding its embodied form of durable dispositions, the concept has also been described as weak at explaining change (Mutch, 2003). However, while the label 'rules of the game' suggests that habitus is a stagnant and limiting force, rather than one which allows for adaptation and experimentation (Maclean et al., 2010), it does incorporate the evolution of norms over time (De Clercq and Voronov, 2009b). Furthermore, although much of Bourdieu's theorising explains why agents may choose to follow extant dominant practices, individual actions and choices are considered a central element of habitus (Bourdieu, 1984). As

Özbilgin and Tatli (2005) point out, Bourdieu's (1977) notion of strategic behaviours highlights the voluntaristic nature of his theory.

DiMaggio (1979: 1468) argues that, in emphasising the role of subconscious elements of human activity, Bourdieu 'minimizes the role of conscious (rational) thought on inconclusive grounds'. In other words, Bourdieu's theory does not adequately account for the deliberate motives of agents, unknown or unknowable to an outside observer. Concerning the durability of habitus, Bourdieu has been criticised for not being clear enough about the impact of experiences post-childhood on transforming personal dispositions (DiMaggio, 1979). This study therefore pays particular attention to the influence of household and community members on the development of entrepreneurs during their adult lives.

When applied within the entrepreneurship literature, habitus and its related concepts have been interpreted in different ways. For example, De Clercq and Voronov (2009c) conceive of an entrepreneurial habitus, which they equate with struggles for legitimacy. In contrast, Tatli et al. (2014: 623) consider illusio the legitimacy-conferring mechanism. This study favours the approach of Stringfellow et al. (2014: 575), who view 'legitimacy as determined by field members who judge legitimacy based on ... the volume and configuration of capital that they [entrepreneurs] possess relative to the field'.

### 3.6. Fields

Bourdieu (1984) proposes that the world is composed of a multitude of social fields, consisting of various agents who strive for power and better positions through the acquisition of different forms of capital. Interestingly, although Bourdieu sought to transcend the divide between the objective and subjective, Özbilgin and Tatli (2005) argue that habitus and capital focus more on the subjective, whereas the concept of field lies at the objective end of the continuum, as fields are linked to structural arrangements which are more readily observed and measured. While this line of thinking is, of course, a simplification, it does facilitate comprehension of the theory of practice. The discussion below examines the concept of field, which is followed by a review of Bourdieu's stance on families as fields and criticisms of the concept.

Fields are arenas of conflict and struggle for positions, which agents either consciously or subconsciously pursue (DiMaggio, 1979). According to Bourdieu and Wacquant (1992: 97):

a field may be defined as ... a configuration of objective relations between positions. These positions are objectively defined, in their existence and in the determinations they impose upon their occupants, agents or institutions, by their present and potential situation (situs) in the structure of the distribution of species of power (or capital) whose possession commands access to the specific profits that are at stake in the field, as well as by their objective relation to other positions.

This definition highlights that the concept of field is inseparable from the forms of capital, which represent both the stakes of the game and a basic requirement for admission to a field (Özbilgin and Tatli, 2005). Agents who acquire large amounts of desired capital can gain dominant positions over others (Golsorkhi et al., 2009). The forms of capital that are considered worth pursuing are determined by the field's habitus and illusio, which are influenced by powerful agents (Drakopoulou Dodd et al., 2014). As DiMaggio (1979) notes, the provision of embodied cultural capital by parents to their children helps dominant agents preserve their authority and control. Bourdieu (1977, 1984) argues that the logic of domination affects most fields, which suggests that conflict and struggles for power are at the core of even the most neutral organisations. As a result, Bourdieu's approach facilitates comparisons between a broad range of fields, such as academia, art and religion (DiMaggio, 1979).

Fields are conceived to be relatively autonomous (Drakopoulou Dodd et al., 2014), each a product of a continuous struggle for power (Golsorkhi et al., 2009). Fields can range in form and size from families (see below), to firms (Desan, 2013) and entire industries (De Clercq and Honig, 2011). Consequently, the concept of field is a popular conceptual tool within organisational analysis for exploring not only individual organisations, but also related agents and institutions involved in a process or activity, such as cultural production (Emirbayer and Johnson, 2008). Bourdieu proposes that different fields (e.g. business and education) overlap (Randle et al., 2015). Thus, individuals can simultaneously be part of several fields (Kislov et al., In Press). Within the entrepreneurship literature, scholars have explored the effects of membership in multiple fields experienced by immigrant (Barrett and

Vershinina, 2017; Vershinina et al., 2011) and transnational entrepreneurs (Patel and Conklin, 2009; Terjesen and Elam, 2009).

Emirbayer and Johnson (2008) argue that the construction of a field for research purposes is a challenging process and is usually partly based on the researcher's personal knowledge and intuition. Within the cultural industries in particular, it can be difficult to determine the boundaries of a field: which producers, suppliers, customers and other agents may be part of the field and guided by its principles? This issue is compounded by the fact that fields can consist of subfields (Elam, 2008; Kislov et al., In Press). Bourdieu proposes that fields are in a hierarchy to one another, organised within an overarching field of power (i.e. 'a metafield of contestation for dominant agents', see Maclean et al., 2014: 826). Business fields, for example, are thought to dominate artistic and educational fields (De Clercq and Voronov, 2011). While it has been suggested that these hierarchal relationships between fields should be further investigated (Emirbayer and Johnson, 2008), such discussions are not the focus of this study, which is based on a small sample of entrepreneurs and their experiences.

As a consequence of the overlapping nature of different fields, not all agents are equally well integrated into each field, which 'leaves open the possibility that field participants are torn by contradiction and internal division' (De Clercq and Voronov, 2009b: 805). Consequently, they may favour one field (often referred to as a 'game', see Kislov et al., In Press) over another. As Tatli et al. (2014) argue, the metaphor of a playing a game is effective in the context of entrepreneurship: entrepreneurs typically seek to gain influence through the acquisition of various capitals, which requires them to comply with certain rules, even if they ultimately wish to overturn these rules and challenge other players for domination (De Clercq and Voronov, 2009a, 2009c). After all, fields are not dead structures, but dynamic spheres of action and strategy (Bourdieu and Wacquant, 1992).

# 3.6.1. Bourdieu and Families

Families are an important aspect of Bourdieu's theory of practice and they require further discussion, especially since this thesis explores the household context of participants (see Chapter 8). Bourdieu's understanding of families is multifaceted and has developed over time. Thus, it should be noted that this study draws primarily on Bourdieu's (1996) article, which provides a dedicated analysis of families. It proposes that the family functions as a field in and of itself:

with its physical, economic and, above all, symbolic power relations (linked, for example, to the volume and structure of the capital possessed by each member) [and] its struggles for conservation and transformation of these power relations. (p. 22)

As family members struggle for power, Bourdieu (1996) argues that they must engage in countless acts of ordinary exchanges, such as providing support and attention, and celebrate events, including birthdays and holidays, to generate a 'family feeling'. These practical and symbolic acts of creation 'transform the obligation to love into a loving disposition' (p. 22) and promote solidarity, generosity and loyalty among family members, despite their contention for dominant positions. A family is commonly associated with a separate social universe, whose members share a joint residence. Consequently, families often make collective decisions, requiring members to act in a unified way (Glover, 2010). However, similar to other fields, not all members have the same propensity to become part of the unified body and accept a common vision, preferring instead to follow their own points of view (Bourdieu, 1996).

Bourdieu (1986) argues that families are also an important source of social capital, as they enable their members to benefit from collectively owned and shared resources (Al Ariss and Syed, 2011; Vershinina et al., 2011). Family members gain access to shared economic capital, such as financial assets, cultural capital, in the form of personal dispositions imparted over time and objectified as heirlooms, social capital, through connections to relatives and their diverse networks, and symbolic capital, including the family name and reputation (Jayawarna et al., 2014; Maclean et al., 2014). Wealthy families who can afford to send their children to elite schools enable them to acquire prized academic credentials (institutionalised cultural capital) and connections with powerful field members (social capital) (Desan, 2013; Maclean et al., 2010). Bourdieu (1996) argues that this transmission of privileges plays an important role in maintaining the family as a social order.
Within the entrepreneurship literature, Bourdieu's perspective on families has received scant attention. Nordqvist and Melin (2010: 222) provide the most detailed discussion, but they interpret 'the family as an entrepreneurial actor' and advocate for its use as a unit of analysis in empirical studies, because Bourdieu (1996: 20) describes the family as 'an active agent'. While this view is consistent with Bourdieu's earlier studies (for more details see DiMaggio, 1979), it disregards certain tensions which may exist among family members. Thus, this study conceives of families as fields.

# 3.6.2. Critique of Field Theory

Similar to habitus and capital, the concept of field has been criticised for a number of reasons (Friedland, 2009; Özbilgin and Tatli, 2005). First, DiMaggio (1979) points out that Bourdieu's notion of relatively autonomous fields is not sufficiently developed, especially in relation to the limits to that autonomy. This is a relevant concern, as Bourdieu argues that fields overlap and also belong to an overarching field of power (metafield). This study therefore draws on the perspectives of participants to define the fields in which they are embedded. Second, Özbilgin and Tatli (2005) propose that Bourdieu's approach to position-taking within fields, which entails generic struggles for domination, is undifferentiated and does not adequately account for alternative forms of contestation and competition. The strategies that agents employ may be more complex than Bourdieu suggests, as demonstrated by the emerging literature on social entrepreneurship (Karataş-Özkan, 2011; Nicolopoulou, 2014) and cultural entrepreneurship (Drakopoulou Dodd, 2014; Hjorth and Holt, 2016).

A similar criticism is that fields are conceived too homogeneously, disregarding the cultural specificity of diverse contexts (Friedland, 2009). For example, it may be assumed that fields in collectivist societies (e.g. China) function differently than those in individualist societies (e.g. USA). There is also a question concerning the role of morals. Evens (1999) argues that if gaining dominance is a primary objective of most field members, moral considerations should play a bigger role in Bourdieu's theorising. Finally, concerning the application of field theory, Emirbayer and Johnson (2008) contend that too many studies focus only on an

individual field (e.g. organisations), overlooking the effects of embeddedness in diverse contexts. Multi-field configurations in particular and how agents navigate embeddedness in co-existing and intersecting fields are considered to be underresearched (Kislov et al., In Press). Thus, this study explores embeddedness in multiple fields and its effects on craft entrepreneurs.

#### 3.7. Conclusion

This chapter has critically reviewed Bourdieu's theory of practice and its application within the entrepreneurship literature. As shown, Bourdieu developed a comprehensive social theory, which encompasses forms of capital, habitus and fields and employs these concepts to explain how contextual influences affect individual practices (see Table 3.3 for a summary of the main concepts). Bourdieu proposes that practices are generally aimed towards acquiring types of capital which are considered valuable within specific fields. By acquiring such capital, agents can gain and maintain power over others. Since businesses in most capitalist societies strive to maximise profits by outperforming their competition, this theory links well with the prevailing understanding of the business world. Accordingly, Bourdieu's theory of practice is gaining popularity within management research. In order to take advantage of the full complement of Bourdieu's theories, empirical studies have primarily adopted qualitative approaches (Karataş-Özkan, 2011; Stringfellow et al., 2014; Terjesen and Elam, 2009). This study follows their example and provides a detailed analysis of the experiences of a small sample of craft entrepreneurs.

Based on the critiques of Bourdieu's theories, this chapter highlights a number of gaps in knowledge. First, while Bourdieu discusses the conversion of various forms of capital, his conceptualisations are constrained by a focus on the importance of economic capital as the 'currency of exchange'. Thus, this study seeks to advance understanding by exploring a broader range of capital conversions which craft entrepreneurs carry out, particularly those related to non-economic forms of capital. Second, given the scarcity of empirical investigations into the role of habitus in cultural entrepreneurship, this study also strives to extend knowledge by exploring the impact of the 'rules of the game' in craft communities. Furthermore, there is lack of understanding concerning the effects of embeddedness in diverse fields on the behaviours of individuals. Accordingly, this study investigates the influence of embeddedness in co-existing fields on craft entrepreneurs. Finally, it challenges Bourdieu's premise that a generic struggle for domination is the guiding principle of all fields. In filling these gaps, this study builds upon the strengths and ameliorates several weaknesses of Bourdieu's comprehensive social theory in order to more fully understand the actions of craft entrepreneurs through its explanatory framework.

# Table 3.3. Summary of Bourdieu's Main Concepts

<b>Orientation*</b>	Concept	Profile
Agentic + Subjective	Capitals	Economic, cultural, social and symbolic resources available to individuals; convertible and facilitate the acquisition of power (stakes of the game)
Ť	Habitus	Personal dispositions which subconsciously affect agent's practices (modus operandi); a constraining social structure representing the rules of the game
	Doxa	Naturalised, taken-for-granted assumptions, which remain unchallenged, limit agents' thinking and facilitate the reproduction of social orders
Ļ	Illusio	Commitment to shared stakes, causing agents to be caught up in the game; a shared belief in the status quo and thus a collective self-deception
Structural + Objective	Fields	Social spheres that are structured according to their members' distribution of various capitals; outcomes of ongoing struggles for power and positions

\* The idea to place Bourdieu's concepts on this continuum was inspired by Tatli et al. (2014).

# CHAPTER 4: ENTREPRENEURSHIP RESEARCH ON RESOURCES, HOUSEHOLDS AND COMMUNITIES

#### 4.1. Introduction

This chapter reviews entrepreneurship research related to the main themes explored in this study. As explained in Chapter 5, this study adopts a phenomenological approach and generates theory inductively, based on the analysis of participants' experiences (Berglund, 2007; Cope, 2005). Thus, the decision to review entrepreneurship literature on resource and capital management, households and families, and embeddedness in communities is an outcome of the analysis of empirical data. Accordingly, when discussing contemporary debates regarding these main themes under investigation, this review incorporates concepts and theories which do not traditionally fall under these three categories, but were found to impact the practices of craft entrepreneurs. For example, psychological capital (Baron et al., 2013; Zou et al., 2016) was discovered to be relevant to the household and family context and is therefore addressed in the discussion of research in this arena. Similarly, virtual embeddedness (Morse et al., 2007; Smith et al., 2017) was revealed as a significant aspect of embeddedness in craft communities and thus appears in the analysis of that literature. While both of these concepts have received scant attention in the entrepreneurship literature, they feature in this literature review as a result of the adopted *a posteriori* approach.

This chapter first examines academic debates concerning the resource and capital management practices of entrepreneurs, introducing the Resource Based View (RBV) and reviewing entrepreneurship research on economic, cultural, social and symbolic capital and their conversions. Next, it discusses extant theories on the effects of households and families on entrepreneurs, exploring research on household contexts, resources of the household and the role of emotions. Finally, this chapter reviews the academic literature on embeddedness in communities, examining research progress on embeddedness theory, embeddedness in communities and social value creation. As such, this chapter seeks to connect a diverse range of literatures, which collectively help to explain the practices of craft entrepreneurs.

#### 4.2. Resource and Capital Management

Entrepreneurial firms tend to be small in size and have a limited supply of resources, especially at start-up (George, 2005; Kiss et al., In Press; Penrose, 1959). This can present them with challenges compared to larger, more established organisations, as some resources can be expensive to obtain for firms lacking economies of scale (Carroll and Swaminathan, 2000; Scott, 2006) or even be unavailable through conventional market channels (Starr and MacMillan, 1990; Witt, 2004). Prior research has shown that entrepreneurs therefore employ a range of strategies to locate and obtain required resources (Fisher, 2012; Ozdemir et al., 2016; Stinchfield et al., 2013), including making do with what is at hand (i.e. 'bricolage', see Baker and Nelson, 2005; Senyard et al., 2014; Stenholm and Renko, 2016) and avoiding external debt by utilising supplier credits, personal funds and shared investments (i.e. 'bootstrapping', see Grichnik et al., 2014; Jonsson and Lindbergh, 2013; Winborg and Landström, 2001).

To date, entrepreneurship research has focused on the importance of access to financial (De Bettignies, 2008; Uzzi, 1999), human (Marvel et al., 2016; Rauch et al., 2005) and social capital (Anderson et al., 2007; Stam et al., 2014). Consequently, considerable knowledge of the impacts of these resources on entrepreneurial activity has been developed (Anderson and Miller, 2003; Manolova et al., 2007; Semrau and Hopp, 2016). However, as most studies explore only one or two forms of capital (cf. Bhagavatula et al., 2010; Chandler and Hanks, 1998; Davidsson and Honig, 2003), little is known about the ways in which entrepreneurs combine and transform various capitals (Klyver and Schenkel, 2013; Scott, 2012). There have also been calls for entrepreneurship research to explore a wider range of resources (De Clercq et al., 2013; Lee and Shaw, 2016; Nicolopoulou, 2014), such as cultural (Jayawarna et al., 2014; Kim et al., 2006) and symbolic capital (De Clercq and Voronov, 2009a; Fuller and Tian, 2006). Thus, this study seeks to advance understanding by exploring how entrepreneurs convert their economic, cultural, social and symbolic capital. By considering these micro-level activities, this study seeks to shed light onto the individual context of craft entrepreneurs. This section first introduces the RBV, followed by a review of the literature on economic, cultural, social and symbolic capital and their conversions.

#### 4.2.1. The RBV

There is a long tradition within the field of entrepreneurship of importing conceptual models and perspectives from other, more established disciplines to explore the experiences of entrepreneurs (Busenitz et al., 2003; Shepherd, 2015; Zahra, 2007). One of these models is the RBV (Barney, 1991; Wernerfelt, 1984), which has been employed to examine entrepreneurial actions related to 'the creation and combining of resources that create new heterogeneous resources ... that are potentially valuable and contribute to a firm's competitive advantage' (Alvarez and Busenitz, 2001: 756). A range of entrepreneurship studies have adopted the RBV over the past two decades (Chandler and Hanks, 1994; Kellermanns et al., 2016; Yli-Renko et al., 2001) to investigate such diverse subjects as alliance formation (Eisenhardt and Schoonhoven, 1996), organisational culture (Zahra et al., 2004) and portfolio entrepreneurship (Sieger et al., 2011).

The RBV originates in Penrose's (1959) work and was developed by strategic management scholars (Barney, 1991; Wernerfelt, 1984) to facilitate explorations of how firms achieve and maintain competitive advantage through the acquisition and ownership of certain sets of resources (Markman et al., 2004; Mosakowski, 1998). It considers firms to consist of heterogeneous bundles of resources, which are created and combined in distinct ways and thus make each business a unique product of the skills, experience and vision of its founder(s) and/or manager(s) (Alsos and Carter, 2006; Wiklund and Shepherd, 2009). According to Barney's (1991: 105) definition, a resource must possess four attributes to generate sustained competitive advantage:

(a) it must be valuable, in the sense that it exploits opportunities and/or neutralizes threats in a firm's environment, (b) it must be rare among a firm's current and potential competition, (c) it must be imperfectly imitable, and (d) there cannot be strategically equivalent substitutes for this resource.

As Newbert (2007) notes, competitive advantage can, of course, be attained with resources that are solely valuable and/or rare, but in order for this advantage to sustain over time, they have to be inimitable and non-substitutable, as well. Various resources have been identified which meet the above criteria, including organisational identity (Fiol, 1991) and patents held by businesses (Markman et al., 2004).

While several meta-analyses have found support for the RBV (cf. Crook et al., 2008; Nason and Wiklund, In Press), it has also been criticised for a number of reasons (for a detailed review, see Priem and Butler, 2001), including its rather static nature (Newbert, 2007: 123) and the lack of conceptual clarity regarding "how" managers/firms transform resources to create value' (Sirmon et al., 2007: 273). Thus, the RBV has evolved over time and scholars have expanded upon and improved the concept (Dyer and Singh, 1998; Kellermanns et al., 2016). For example, it has been proposed that not only resources, but also capabilities can create sustained competitive advantage and generate 'economic rents' (i.e. payments made in excess of the amounts required for the products/services, see Amit and Schoemaker, 1993). In addition, a number of 'spin-off' approaches have been developed from the RBV (Barney et al., 2011), including the relational view, which argues that the capabilities embedded in network relationships can provide competitive advantage (Dyer and Singh, 1998; Yli-Renko et al., 2001), and the knowledge-based view, which proposes that business-relevant knowledge is strategically important and can help firms maintain a competitive edge (Grant, 1996; Sullivan and Marvel, 2011).

Given that entrepreneurial firms differ from larger organisations (e.g. in size, reputation and reliance on relationships, see Sorenson and Stuart, 2008; Wiklund and Shepherd, 2009), it can be assumed that the resources needed for an emerging venture to succeed are distinctive from those required by an established business (Alvarez and Busenitz, 2001; Kiss et al., In Press). Based on Barney's (1991: 105) definition that resources are 'all assets, capabilities, organizational processes, firm dimensions, information, knowledge, etc. controlled by a firm that enable the firm to conceive of and implement strategies that improve its efficiency and effectiveness', it can be surmised that a multitude of different resources can provide entrepreneurs with competitive advantage depending on their particular circumstances (Kellermanns et al., 2016). Thus, there have been calls for research to contextualise the RBV (Lado et al., 2006), especially within the field of entrepreneurship (Edwards et al., 2010; Perry-Rivers, 2016). As Foss et al. (2008: 74) argue, scholars should investigate the subjective views of entrepreneurs regarding their strategic use of resources and account for the fact that 'individuals hold different preferences, knowledge, and expectations'.

Drawing on the work of Brush and colleagues (cf. Brush et al., 2001; Lichtenstein and Brush, 2001), Kellermanns et al. (2016: 36) suggest exploring both tangible and intangible assets, such as 'human, financial, physical, and relationship capital', which can provide entrepreneurs with competitive advantage. As Bourdieu's (1986) perspective accounts for all of these resources and provides a consistent set of definitions of key concepts (see Chapter 3), it represents an appropriate framework for examining the resources available to entrepreneurs (Drori et al., 2006; Shaw et al., 2009). Indeed, while prior research has employed different labels, it has already demonstrated the importance of access to economic capital (e.g. wealth, see Perry-Rivers, 2016), cultural capital (e.g. knowledge, see Alsos and Carter, 2006), social capital (e.g. alliances, see Eisenhardt and Schoonhoven, 1996) and symbolic capital (e.g. reputation, see Reuber and Fischer, 2007) for achieving sustained competitive advantage. In comparison, scholars 'have given scant attention to the process by which these resources are discovered ... and exploited' (Alvarez and Busenitz, 2001: 757), leading to 'the processes and mechanisms by which new resources or resource combinations are created [to] remain a black box' (Foss et al., 2008: 88). Thus, there is a need for research to explore the ways in which entrepreneurs accumulate (Lee and Shaw, 2016), combine (Klyver and Schenkel, 2013) and convert (Scott, 2012) various forms of capital to achieve commercial sustainability and success. The following sections therefore discuss progress in the entrepreneurship literature related to economic, cultural, social and symbolic capital.

## 4.2.2. Economic Capital and its Conversions

Bourdieu (1986) proposes that economic capital presents itself in the form of monetary assets as well as tangible and intangible business assets (Anheier et al., 1995; DiMaggio, 1979; Glover, 2010). In contrast, entrepreneurship research has mainly focused on the impact of financial resources on entrepreneurial processes (De Bettignies, 2008; Winborg and Landström, 2001). Therefore, while this review of the literature encompasses the many manifestations of economic capital discussed by Bourdieu, it focuses on how financial resources have been addressed by current research.

As Shane and Cable (2002: 364) argue, 'entrepreneurs are often wealth constrained, and need to obtain external financing to pursue their opportunities, making financing central to the process of entrepreneurship'. Consequently, a large body of research has explored how entrepreneurs can gain financial capital to develop their ventures (De Bettignies, 2008; Huang and Knight, 2017; Uzzi, 1999). Given that access to traditional funding sources, such as bank loans (Carter et al., 2007; Le and Nguyen, 2009), tends to be restricted for small and emerging firms (Ebben and Johnson, 2006), entrepreneurs must frequently find alternative ways of financing their ventures (Grichnik et al., 2014; Jonsson and Lindbergh, 2013; Winborg and Landström, 2001).

One means of doing so is through government and business grants (Jayawarna et al., 2011; Rouse and Kitching, 2006). In other cases, external investors, such as venture capital firms (Cox Pahnke et al., 2015; Rosenbusch et al., 2013) and business angels (Harrison and Mason, 2007; Steier and Greenwood, 2000), may be willing to back new ventures. Furthermore, there is growing recognition of the financing available through online channels (Morse et al., 2007; Parhankangas and Renko, 2017; Short et al., 2017), such as rewards-based (Josefy et al., 2017), donation-based (Allison et al., 2015) and equity-based crowdfunding (Cholakova and Clarysse, 2015). However, most of these funding sources require entrepreneurs to pitch their ventures and business ideas, which often necessitates the development of a business plan (Chen et al., 2009; Foo et al., 2005) and can deter some from pursuing such opportunities (especially those entrepreneurs who are not growth-oriented, see Brinckmann et al., 2011). Consequently, entrepreneurs often draw on their social networks, such as family and friends (Bird and Wennberg, 2016; Edelman et al., 2016), and their personal savings (Harrison et al., 2004; Klyver and Schenkel, 2013) to acquire the financial capital needed to develop their businesses.

While prior research has described economic capital as a crucial 'ingredient for entrepreneurial survival and growth' (Allison et al., 2015: 53) and has paid considerable attention to the various funding options pursued by entrepreneurs (De Bettignies, 2008; Shane and Cable, 2002), other financial activities, such as strategic financial management and planning, have largely been overlooked (Brinckmann et al., 2011). Given that entrepreneurs can grow their businesses solely with the profits from their operations (Alsos et al., 2006), it is important to gain further understanding of the ways in which they manage available financial resources (Glover, 2010). Gaining deeper insights into such practices is essential, as entrepreneurial firms usually lack financial reserves (Ebben and Johnson, 2006; Vanacker et al., 2011), which makes them more vulnerable to financial stresses (e.g. economic downturns and increased competition, see Morse et al., 2007). Thus, the abilities of entrepreneurs to transform other available resources into economic capital can have significant impacts on the chances of business survival and success (Stenholm and Renko, 2016; Stinchfield et al., 2013).

In contrast to their abundant considerations of monetary assets, scholars within the field of entrepreneurship have paid little attention to other forms of economic capital, such as non-financial tangible business assets (Anheier et al., 1995; DiMaggio, 1979). Research that does investigate such elements of economic capital has mainly been conducted within the context of farm-based entrepreneurship and has highlighted the value of 'land and natural resources, buildings and installations' (Alsos and Carter, 2006: 313). In addition, it has been shown that material, stock, reserves (Jonsson and Lindbergh, 2013; Powell and Baker, 2014), technologies and machinery (Haber and Reichel, 2007; Perry-Rivers, 2016) can support entrepreneurial efforts. Scholars have also noted the value of intangible resources, such as licences (Dushnitsky, 2010), trademarks (Block et al., 2014) and patents (Sleuwaegen and Boiardi, 2014), for acquiring competitive advantage. However, it has been argued that entrepreneurs rarely pursue such legal protections (Bradley et al., 2012), because acquiring property rights can be costly (Somaya, 2012) and does not necessarily prevent counterfeiting (Qian, 2014). Given the limited amount of empirical research on the subject, there is a need for further investigations into the roles that these alternative types of economic capital play in entrepreneurial processes.

Conversions of economic capital into other forms of capital are rarely the subject of discussion in entrepreneurship research. Instead, they are largely taken-for-granted, which implies that they are part of the doxa of the field (Drakopoulou Dodd et al., 2014). Given that such naturalised assumptions can limit scholars' abilities to critically evaluate common practices and processes (Golsorkhi et al.,

2009; Vincent, 2016), it is important to explore and question them. Thus far, it has been shown that entrepreneurs can transform economic capital by investing in an elite education to develop their skills (cultural capital), networks (social capital) and standing within a field (symbolic capital) (Harvey et al., 2011a; Shaw et al., 2013). However, it has been noted that such conversions tend to be complex and time-consuming (Greve and Salaff, 2003; Jayawarna et al., 2014), which can inhibit their performance. As might be expected, prior research also illustrates that financial assets can be used to acquire materials (economic capital) (Hanna and Walsh, 2008; Hoegl et al., 2008) and to signal that a firm is viable, which helps create symbolic capital (Eisenhardt and Schoonhoven, 1996). Furthermore, charitable giving has been shown to raise the reputation of entrepreneurs (Harvey et al., 2011a). While these conversions of economic capital are acknowledged by entrepreneurship scholars, they are rarely empirically investigated, necessitating further research into such transformation processes.

#### 4.2.3. Cultural Capital and its Conversions

Bourdieu (1986) differentiates between three forms of cultural capital: embodied (knowledge, skills and experience, see Vershinina et al., 2011; Vincent, 2016), objectified (cultural goods, see Lee and Shaw, 2016; Scott, 2012) and institutionalised (educational qualifications, see Spence et al., 2016; Stringfellow et al., 2014). Instead of utilising this multifaceted concept, which originated in the field of sociology, entrepreneurship research generally adopts an economics-based approach and employs Becker's (1964) concept of human capital (Davidsson and Honig, 2003; Marvel, 2013; Rauch et al., 2005). Human capital is defined as the 'skills and knowledge that individuals acquire through investments in schooling, on-the-job training, and other types of experience' (Unger et al., 2011: 343). As the concepts of human and embodied cultural capital are thus closely related (see also Chapter 3), this section combines them in its discussion of research progress in this domain.

There has been sustained academic interest in the impact of knowledge, skills and experience on entrepreneurial processes (Anderson and Miller, 2003; Chandler and Hanks, 1998; Coleman, 2007). Initially, the concept of human capital was developed to examine the value of education (Becker, 1964), with early studies highlighting that investments into education could lead to higher outputs and increased wages within the population (Mincer, 1958; Schultz, 1961). As Unger et al. (2011) note, entrepreneurship research first adopted the concept to link the possession of human capital to entrepreneurial success (Cooper et al., 1994; Sanders and Nee, 1996). Specifically, human capital has been shown to facilitate entrepreneurship by supporting opportunity discovery, creation (Alvarez and Barney, 2007; Fiet et al., 2013), evaluation and exploitation (Dimov, 2010; Yli-Renko et al., 2001) and by enabling the development of competitive advantage (Bradley et al., 2012; Sullivan and Marvel, 2011). Notably, entrepreneurship research has expanded the concept of human capital to include capabilities, such as judgement and decision-making (Jansen et al., 2013; Marvel et al., 2016). While entrepreneurship studies have mainly focused on the human capital of individuals (Unger et al., 2011), it has been shown that it can also exist at the firm level, comprising all knowledge and skills embedded within an organisation (Onkelinx et al., 2016; Ployhart et al., 2011).

Another distinguishing feature of the concept of human capital is the differentiation between investments and outcomes (Becker, 1964). According to Marvel et al. (2016), education and training/experience are considered investments into human capital which can create favourable outcomes, such as knowledge, skills and abilities. In firms that grow beyond the founding entrepreneur(s), recruitment is another investment strategy that can produce desired human capital outcomes. Unger et al. (2011) highlight that knowledge, skills and abilities (outcomes) are more likely to lead to entrepreneurial success than human capital investments alone, as education and training/experience do not guarantee the acquisition of relevant outcomes (Haber and Reichel, 2007). Furthermore, prior research shows that knowledge tends to be most useful when it is related to entrepreneurial actives in a specific domain (e.g. understanding of markets and technologies, see Markman and Baron, 2003; Shane, 2000). Skills, such as accounting, marketing and coding proficiency, were also found to support entrepreneurship, as they facilitate the application of knowledge to specific tasks (Baron and Markman, 2003; Chell, 2013). However, it has been noted that they require sustained practice or they can be forgotten over time (Ployhart et al., 2014). Cognitive abilities, including listening, learning and reasoning, have been shown to be more enduring and broadly applicable than technical skills, though they cannot be developed in the same fashion (Dimov, 2010; Jayawarna et al., 2014). Thus, adopting Becker's (1964) terminology, skills (especially those related to a particular industry or setting) represent a form of specific human capital, whereas abilities denote general human capital (Marvel, 2013; Rauch and Rijsdijk, 2013).

While human capital research has advanced understanding of entrepreneurial processes (Davidsson and Honig, 2003; Estrin et al., 2016; Rauch et al., 2005), it has been criticised for its disjointed nature and for applying inconsistent definitions (Ployhart et al., 2014), which have led to some divergent and even conflicting findings (Marvel et al., 2016; Unger et al., 2011). For example, Kacperczyk and Younkin (In Press: 2) propose that 'entrepreneurial skill [represents] broad functional experience', which allows entrepreneurs to perform various tasks independently and master different jobs successfully (Åstebro et al., 2011). However, this definition of skill contrasts with the prevailing conceptualisation discussed above. In addition, the 'jack-of-all-trades' argument (Lazear, 2004: 208), upon which this definition is based, has been challenged (Estrin et al., 2016; Marvel, 2013). Marvel et al. (2016) provide a detailed analysis of other criticisms of human capital research. Their systematic literature review reveals that the vast majority of studies adopt quantitative approaches (only 2.2% of the studies in their review are qualitative) and that extant research focuses on human capital investments rather than outcomes. Furthermore, they show that there is a clear preference for exploring high-technology contexts (32% of studies) over 'manufacturing or retail contexts, [which] each represented less than 4% of the research stream' (p. 804). Jayawarna et al. (2014: 921) also note that 'most research in entrepreneurship treats human capital as an asset fixed in time', instead of accounting for its variable nature (Ployhart et al., 2011). Thus, there is a need for more qualitative research into the role of knowledge, skills and abilities in entrepreneurship and how these human capital outcomes (i.e. embodied cultural capital) change over time.

In line with its focus on human capital, the entrepreneurship literature has paid scant attention to the impact of objectified and institutionalised cultural capital on entrepreneurial processes. Indeed, the concept of cultural capital is almost exclusively employed by the comparatively few studies that adopt Bourdieu's theory of practice (De Clercq and Voronov, 2009a; Karataş-Özkan, 2011; Patel and Conklin, 2009) and/or employ both human and cultural capital (Anderson and Miller, 2003; Jayawarna et al., 2014; Kim et al., 2006). These studies acknowledge the value of objectified cultural capital in the form of 'sophisticated and novel products, goods and materials' (Lee and Shaw, 2016: 1742) which can be used by entrepreneurial firms to signal their high standards and novelty to customers. They also show that educational qualifications and credentials (institutionalised cultural capital) can help entrepreneurs meet the expectations of stakeholders and clients (Nagy et al., 2012; Stringfellow et al., 2014). Furthermore, by gaining understanding of the 'customs, habits, interests, lifestyle[s] and ... traditions' (Dana and Light, 2011: 342) that prevail in different contexts, entrepreneurs can gain valuable cultural knowledge, which is an aspect of embodied cultural capital (De Clercq and Voronov, 2009b), but not considered part of the concept of human capital (Vershinina et al., 2011). Thus, entrepreneurship research can benefit from adopting the concept of cultural capital in its empirical analyses (Jayawarna et al., 2014; Kim et al., 2006).

With regard to its convertibility, embodied cultural capital has been shown to facilitate the development of economic capital, as knowledge of alternative funding sources can improve financial decision-making (Coleman, 2007; Seghers et al., 2012) and management competence can increase profits and reduce business expenses (Brinckmann et al., 2011; Haber and Reichel, 2007). In addition, market knowledge can provide access to supplier and customer contacts (social capital) and facilitate meeting the needs and expectations of these network members (Bhagavatula et al., 2010; Mosey and Wright, 2007). Experience within an industry can also promote reputation building (symbolic capital) (Beverland, 2005; Clarke, 2011) and storytelling skills can further support this process (Lounsbury and Glynn, 2001; Martens et al., 2007). In fact, it has been shown that it is only through competence at managing symbolic resources (Überbacher et al., 2015; Zott and Huy, 2007) and networks (Lans et al., 2015; Rasmussen et al., 2015) that entrepreneurs can attain their benefits. Similarly, technical skills can be invaluable in certain contexts, such as for entrepreneurs who use their websites (economic capital) to sell the majority of their products or services (i.e. digital entrepreneurs, see Martinez Dy et al., 2017; Nambisan, In Press). Prior research also shows that high socio-economic position (class) and cultural distinction can provide access to wider networks (Anderson and Miller, 2003; Jayawarna et al., 2014).

At the same time, a lack of cultural understanding and sophistication can inhibit the abilities of entrepreneurs to fit in with extant norms (habitus), which can limit their abilities to join networks (social capital) and affect their reputation building efforts (symbolic capital) (De Clercq and Voronov, 2009c; Perry-Rivers, 2016). Furthermore, a lack of expertise may require entrepreneurs to rely on external agents to complete necessary tasks, which can impede their ability to generate economic capital (Martinez Dy et al., 2017). With regards to institutionalised cultural capital, Vershinina et al. (2011: 104) show that the possession of academic qualifications may be irrelevant in certain fields or even disparaged by field members and therefore negatively impact development of economic capital. It has also been argued that displaying a breadth of experience can inhibit the acquisition of symbolic capital (Kacperczyk and Younkin, In Press), as stakeholders and customers may prefer the appearance of specialisation (Navis and Glynn, 2010; 2011). Thus, prior research shows that cultural capital can be converted into various other capitals and that there are inhibitors to this process.

#### 4.2.4. Social Capital and its Conversions

According to Bourdieu (1986: 248), social capital represents 'the aggregate of the actual or potential resources which are linked to possession of a durable network of more or less institutionalized relationships'. Within the entrepreneurship literature, research mainly adopts Nahapiet and Goshal's (1998: 243) definition, which conceptualises social capital as 'the sum of actual and potential resources embedded within, available through, and derived from the network of relationships possessed by individuals or social units'. As these definitions are largely consistent, Bourdieu's approach to social capital is relevant to and drawn upon by numerous entrepreneurship studies (cf. Anderson and Jack, 2002; Batjargal, 2003; Hernández-Carrión et al., 2017; Maurer and Ebers, 2006).

Scholars have shown an enduring interest in exploring the impact of social capital on entrepreneurs (Cope et al., 2007; Gedajlovic et al., 2013; Yli-Renko et al., 2001). Extant studies reveal that possession of social capital facilitates a range of

activities involved in entrepreneurship, such as opportunity recognition (Bhagavatula et al., 2010; Hmieleski et al., 2015), business start-up (Liao and Welsch, 2005; Semrau and Hopp, 2016) and venture growth (Davidsson and Honig, 2003; Hernández-Carrión et al., 2017). Furthermore, prior research demonstrates that adopting a social capital perspective enables investigations of various constructs and concepts related to entrepreneurship, such as entrepreneurial intention (Edelman et al., 2016), innovation (Dakhli and De Clercq, 2004), creativity (Shalley and Perry-Smith, 2008), finance (Shane and Stuart, 2002) and performance (Stam et al., 2014). These studies highlight that social capital plays an important role in achieving entrepreneurial success (Aarstad et al., 2010; Steier and Greenwood, 2000).

Despite the prevalent acknowledgment of its significance, the concept of social capital has been inconsistently applied (Adler and Kwon, 2002; Payne et al., 2011), which is largely due to the multitude of definitions employed within the literature (for detailed reviews, see Bowey and Easton, 2007; Portes, 1998). It has also been argued that its designation as a single 'capital' is in itself problematic, given the range of phenomena described by the concept (Anderson et al., 2007). Nevertheless, there is growing consensus that social capital resides within networks (Liao and Welsch, 2005; McKeever et al., 2014) and provides entrepreneurs with access to resources that can be utilised to achieve desired outcomes (Batjargal, 2003; Smith et al., 2017). Indeed, it has been suggested that the concepts of social capital and networks are inextricably intertwined (Casson and Della Giusta, 2007; Kwon and Adler, 2014). Accordingly, in its analysis of social capital conversions, this study focuses primarily on the business networks, such as guilds and support groups, and the interpersonal ties, including followers, suppliers and friends, upon which craft entrepreneurs draw to gain access to resources (see Chapters 7). Building on Anderson et al. (2007: 264), who argue that 'social capital is more akin to a key, rather than the resource itself', this study interprets social capital conversion as the process of unlocking resources that reside in the nexus between network partners.

One reason for the variation of definitions of social capital is its latent nature: it can be accrued and released (Nahapiet and Ghoshal, 1998), but its actual value is uncertain until entrepreneurs deploy it (Smith et al., 2017). Thus, it is hard to measure social capital objectively (Adler and Kwon, 2002) and to 'determine if a

stock of resources (e.g. knowledge) can be fully attributed to network relationships' (Gedajlovic et al., 2013: 459). As Lester et al. (2008) argue, it can be difficult to delineate social capital from other forms of capital (e.g. human capital). Furthermore, as social capital is context-specific, the transferability of research findings cannot be assumed (Foley and O'Connor, 2013). For example, it has been shown that the norms governing relational interactions and the processes required to access available resources within networks differ between contexts (Batjargal, 2010; Lee and Jones, 2008; Zhang, 2015).

Scholars usually differentiate between two forms of social capital: bonding and bridging (De Carolis and Saparito, 2006; Portes, 1998). The bonding perspective, which is primarily based in Coleman's (1988) work, proposes that the value of social capital to businesses emanates from strong connections and frequent interactions between actors who share similar values, resulting in reciprocity and trust (Lee and Jones, 2008; Light and Dana, 2013). This perspective suggests that social capital is collectively owned and that it creates solidarity among network members, which may otherwise be unobtainable (Gedajlovic et al., 2013). However, the resulting 'network closure' (i.e. the presence of cohesive ties, see Gargiulo and Benassi, 2000), which is usually enforced through social norms, has been shown to limit entrepreneurial agency (Foley and O'Connor, 2013). In contrast, the bridging perspective, which builds on Burt's (1992) work, highlights the benefits of open networks that feature diverse connections and provide access to non-redundant resources (Davidsson and Honig, 2003; Lans et al., 2015). Research within this vein has found that entrepreneurs employ brokers to bridge structural holes in their networks and to link them to non-overlapping clusters of people (Mosey and Wright, 2007; Steier and Greenwood, 2000).

There has been considerable debate regarding the importance of bonding versus bridging social capital (Anderson et al., 2007; Lee et al., 2011). Scholars increasingly recognise that their efficacy depends on both the contexts in which firms are embedded (Stam et al., 2014) and their stages of development (Burt and Merluzzi, 2016). As 'bridging social capital is closely related to the concept of weak ties' (Hernández-Carrión et al., 2017: 65) and bonding social capital is associated with strong ties, this study refers to the strength of ties connecting network members

rather than their bridging/bonding qualities. As discussed in the review of embeddedness theory (see below), scholars do not only differentiate between bonding and bridging social capital, which mainly describe structural characteristics of networks, but also between structural and relational embeddedness in networks (Bhagavatula et al., 2010; Payne et al., 2011) and their effects on entrepreneurs (Batjargal, 2003; Gedajlovic et al., 2013: 459). Furthermore, an emerging body of research explores the cognitive dimension of social capital (Lee and Jones, 2008; Nahapiet and Ghoshal, 1998), which encompasses the 'similarities in actors' cognitive schemes and systems of meaning' (Maurer and Ebers, 2006: 264). However, as Light and Dana (2013: 606) convincingly argue, this multidimensional view of social capital increases 'conceptual clutter' and conflates aspects of social and cultural capital. Thus, this study does not distinguish between these different dimensions of social capital in the analysis of data.

Concerning social capital conversions, a considerable number of studies have explored ways in which entrepreneurs can access and develop other forms of capital by drawing on network ties (Ozdemir et al., 2016; Semrau and Werner, 2014; Sullivan and Ford, 2014). For example, it has been shown that social connections can provide entrepreneurs with access to economic capital in the forms of financial gifts (Brüderl and Preisendörfer, 1998; Caputo and Dolinsky, 1998) and loans provided under favourable conditions (Le and Nguyen, 2009; Uzzi, 1999). Furthermore, prior research highlights the ability of entrepreneurs to access embodied cultural capital (in the form of knowledge) through network members (Davidsson and Honig, 2003; Hughes et al., 2014), which is particularly important during the early stages of venture development (Huggins et al., 2015; Yli-Renko et al., 2001). Knowledge sharing can be achieved through participation in informal events, including 'receptions, club meetings, or even through conversation at the school gates' (Casson and Della Giusta, 2007: 231), and through more formal arrangements, such as mentoring relationships (Grossman et al., 2012; Ozgen and Baron, 2007) and interorganisational collaborations (McAdam et al., 2014; van Burg et al., 2014). Becerra et al. (2008) emphasise that, because such exchanges of knowledge can involve a degree of risk, they rely upon trust between partners. Entrepreneurs can also benefit from connections with government support agencies, which can provide them with

economic and cultural capital (Casson and Della Giusta, 2007; Lee et al., 2011). As cultural capital is socialised within family networks (Anderson and Miller, 2003; Jayawarna et al., 2014), membership in upper-class families can bestow entrepreneurs with cultural understanding (Harvey et al., 2011a; Wright and Zammuto, 2013) and grant them access to elite schools (Maclean et al., 2010; 2014), which, in turn, facilities the development of knowledge, skills and abilities.

Furthermore, by acting as brokers, network partners connect entrepreneurs with other parties and, in so doing, increase the available stock of social capital (Ozdemir et al., 2016; Stam, 2010). According to Wong and Boh (2010: 129), 'an individual's affiliation with third parties affects that person's reputation because observers tend to evaluate connected individuals similarly'. Thus, social capital can be converted into symbolic capital not only through endorsements of network partners (Navis and Glynn, 2010; Stuart et al., 1999) but also through the simple act of associating with reputable others (Pollock et al., 2010; Svejenova et al., 2007). Such affiliations usually create positive spill-over effects (Partanen and Goel, 2017; Reuber and Fischer, 2005), but they can equally send negative signals should the reputation of the network partner be tarnished (Fischer and Reuber, 2007) and consequently impede a firm's status enhancement (Barlow et al., In Press). Nevertheless, extant research indicates that the advantages entrepreneurs derive from connections with network partners far outweigh the risks associated with developing and converting their social capital (Semrau and Werner, 2014; Sullivan and Ford, 2014).

## 4.2.5. Symbolic Capital and its Conversions

According to Bourdieu (1984: 251), symbolic capital represents 'name, renown, prestige, honour, glory [and] authority ... everything which constitutes symbolic power as a recognized power'. Entrepreneurship research that employs the concept usually refers to Bourdieu's work as its source, either directly by citing Bourdieu (Karataş-Özkan, 2011; Stringfellow et al., 2014) or indirectly (Lewis, 2013) by citing the work of De Clercq and colleagues (cf. De Clercq and Honig, 2011; De Clercq and Voronov, 2009a), who discuss Bourdieu's theories in detail. Furthermore, a number of studies employ similar concepts, such as reputation (Dimov et al., 2007; Reuber and Fischer, 2005), status (Packalen, 2007; Ruef et al., 2003) and prestige

(Jung et al., 2017; Pollock et al., 2010), often in explanations of how firms gain legitimacy (Bitektine, 2011; Fisher et al., 2016). Accordingly, this section discusses research progress on these related concepts in its review of relevant literature.

Symbolic capital (also termed symbolic resources, see Ertug et al., 2016) can significantly benefit entrepreneurs (Fuller and Tian, 2006; Lee and Shaw, 2016; Scott, 2012), as it facilitates crucial entrepreneurial activities, such as attracting customers (Clarke, 2011; Reuber and Fischer, 2009), gaining access to finance (Dimov et al., 2007; Shane and Cable, 2002) and growing business (Abeysekera, In Press; O'Cass and Sok, 2014). Scholars generally agree that, by accumulating symbolic capital, entrepreneurs can gain power (Harvey et al., 2011a; Shaw et al., 2013), which allows them to increase product and service prices (Rindova et al., 2005; Svejenova et al., 2007) and influence other agents in the field (De Clercq and Voronov, 2009a; Stringfellow et al., 2014). In fact, De Clercq and Honig (2011: 362) argue that 'the concept goes beyond the mere notions of quality, visibility or prestige ... [It] represents the ultimate basis of power through which actors can impose their vision of the way the field should be organized'. Furthermore, once acquired, symbolic capital can compel established businesses to acknowledge the quality, or even superiority, of products and services offered by entrepreneurial firms (De Clercq and Voronov, 2009b).

While prior research recognises the need for entrepreneurs to engage in symbolic management to maintain and increase their symbolic capital (Garud et al., 2014; Überbacher et al., 2015), little is known about what 'entrepreneurs actually do to distinguish themselves from their competing peers' (Zott and Huy, 2007: 71). A number of studies highlight the importance of positive reputation for gaining competitive advantage (Barney, 1991; Partanen and Goel, 2017) and for achieving entrepreneurial success (Abeysekera, 2017; Reuber and Fischer, 2011; Svejenova et al., 2007). However, Lange et al. (2011) lament that 'multiple definitions, conceptualizations, and operationalizations ... have emerged across studies'. Scholars employ the concept of reputation variously to connote being known (Rindova et al., 2005), being known for something (Fischer and Reuber, 2007) and generalised favourability (Love and Kraatz, 2009). Yet, most studies agree that reputation relies on subjective evaluations by external stakeholders (Bitektine, 2011;

Fisher et al., 2016), which are based on a firm's past performance and comparisons to other businesses in the same category (Dimov et al., 2007; Shane and Cable, 2002). Thus, drawing on Gioia et al. (2000: 67), this study defines reputation as 'relatively stable, long-term, collective judgments by outsiders of an organization's actions and achievements'.

It has been shown that entrepreneurs can develop their reputation by displaying signifiers, such as 'founders' track records, high-status partners, or wins in certification contests' (Fischer and Reuber, 2007: 54), as well as favourable media coverage (Deephouse, 2000; Schultz et al., 2014). Indeed, status is considered a primary means for signalling positive reputation (Ertug et al., 2016) and studies often combine both concepts in empirical analyses (Dimov et al., 2007; Milanov and Shepherd, 2013). However, as Sauder et al. (2012) note, while reputation is based on prior behaviours and accomplishments, status derives from the position that actors or firms occupy within a hierarchy (e.g. a social order or market place). Still, research has found that they reinforce one another as long as the information conveyed is consistent (Ertug et al., 2016). It has been shown that the status of firms can be increased by improving the perceived quality of products or services in relation to competitors (Podolny, 1993) and by associating the business with influential partners (Lin, 1999; Pollock et al., 2010). Importantly, possession of high status can lead consumers to purchase outputs from producers with whom they have not previously done business (Packalen, 2007), which highlights its importance for entrepreneurs (Stuart et al., 1999).

Scholars generally argue that entrepreneurs seek to acquire symbolic capital in the form of reputation, status and other symbols of prestige (e.g. awards, see Anand and Watson, 2004; Reuber and Fischer, 2007) in order to gain legitimacy (Fisher et al., 2016; Navis and Glynn, 2010; Stringfellow et al., 2014). Defined as a 'generalized perception or assumption that the actions of an entity are desirable, proper or appropriate within some socially constructed system of norms, values, beliefs and definitions' (Suchman, 1995: 574), legitimacy is crucial for acquiring resources, overcoming the liability of newness and achieving entrepreneurial success (Aldrich and Fiol, 1994; Lounsbury and Glynn, 2001; Zimmerman and Zeitz, 2002). Accordingly, a large body of research explores legitimacy and its various effects (for reviews of this literature, see Suddaby et al., 2017; Überbacher, 2014). As Bitektine (2011) notes, research has identified 18 distinct types of legitimacy which impact business survival and success, including procedural (Suchman, 1995), moral (Greenwood et al., 2002) and regulative legitimacy (Kostova and Zaheer, 1999). Thus, given the considerable amount of existing research on the subject, this study explores how symbolic capital can be transformed into other forms of capital (and vice versa), rather than investigating how it can be used to generate legitimacy (for in-depth discussions, see De Clercq and Voronov, 2009a, 2009b, 2009c).

Prior research identifies several conversions of symbolic capital conducted by entrepreneurs (Scott, 2012; Shaw et al., 2013). Regarding transformations into economic capital, positive reputation and high status have been shown to increase sales (Abeysekera, 2017) and facilitate access to other financial resources (e.g. through IPOs, see Martens et al., 2007; Pollock et al., 2010). Possession of prestige can also reduce business expenses, as 'those with high status receive overt deference, gifts, hospitality, and services from others who have lower status' (Packalen, 2007: 877). Reputation can increase social capital by raising external stakeholders' awareness of a firm (Fuller and Tian, 2006) and by supporting the development of industry-specific networks (Shaw et al., 2017). It has also been shown that high status generates an 'accumulative advantage', providing its holder(s) with a disproportionate share of network ties compared to those with low status and amplifying 'future changes in the structural characteristics of past networks by reinforcing the brokerage position of prominent actors over time' (Zaheer and Soda, 2009: 4). Regarding conversions into cultural capital, extant studies demonstrate that positive reputation can provide access to exclusive opportunities for education (Lawrence, 2004; McLeod et al., 2009).

It has been shown that accumulation of symbolic capital is particularly beneficial for entrepreneurs in the cultural industries (O'Connor, 2015; Townley et al., 2009), as it helps create 'buzz' (i.e. 'rumours and recommendations circulating through dense cultural intermediary networks', see Scott, 2012: 244), which increases exposure to customers and reduces the amount of economic capital spent on advertising. Furthermore, agents endowed with symbolic capital can influence understanding of what forms of artistic expression are desirable and thus enhance

the perceived value of their own products (Jones et al., 2016). They are also more likely to be invited to award ceremonies (Anand and Watson, 2004), which can provide them with further symbolic capital. Contrary to widely held assumptions, research has also demonstrated that winning awards and gaining reputation can impede business growth, as it can raise the expectations of customers and thus induce more critical assessments of business performance (Kovács and Sharkey, 2014). Furthermore, it has been shown that reputation can 'stick' (Love and Kraatz, 2009: 314), leading to 'evaluations (positive or negative) becom[ing] entrenched such that they have an ongoing impact on the firm's performance' (Fischer and Reuber, 2007: 67). Finally, there is a danger that consumers may associate similar organisations and products with one another (Barlow et al., In Press), which can result in entrepreneurs being negatively affected by the mistakes of other firms in the same category (Zhao et al., 2013).

# 4.3. Households and Families

Entrepreneurs are not 'lone heroes', but 'social animals' (Drakopoulou Dodd and Anderson, 2007) who are affected by a range of actors, including customers (Reuber and Fischer, 2005; Shepherd and Zacharakis, 2003; Yli-Renko et al., 2001), business advisors (Jack et al., 2010; Ram et al., 2015; Smallbone et al., 2003) and policy makers (Casson and Della Giusta, 2007; Lee and Shaw, 2016; Neergaard and Ulhøi, 2006). Among their closest connections are family members (Belenzon et al., 2016; Jack, 2005; Renzulli et al., 2000), who have been shown to impact various entrepreneurial processes (Aldrich and Cliff, 2003; Dyer, 2003; Jennings and McDougald, 2007), including opportunity recognition (Khavul et al., 2009; Ram et al., 2001), resource management (Anderson et al., 2005; Bird and Wennberg, 2016) and venture development (Fletcher, 2010; Nordqvist and Melin, 2010). Thus, family business research has emerged as a separate field of study (for detailed reviews, see Evert et al., 2015; Gedajlovic et al., 2012; Sharma et al., 2012), which has greatly advanced understanding of the role of family members in entrepreneurship (Karra et al., 2006; Miller et al., 2011; Morris et al., 2010b). However, research in this domain investigates family-owned firms and it has been shown that its findings are not readily transferable to businesses not operated by families (Bird and Wennberg, 2014; Miller et al., 2008; Zahra et al., 2004).

As this study explores the experiences of sole proprietors, it therefore primarily draws on family-related theory from outside the family business literature. Heeding the calls for research into the immediate family units (i.e. households) of entrepreneurs (Alsos et al., 2014; Baines et al., 2002; Wheelock and Oughton, 1996), this study explores *how household members impact the practices of entrepreneurs*. This section reviews extant literature on the household context, before discussing the resources of the household and examining the role of emotions.

#### 4.3.1. Household Context

There is a growing recognition within the entrepreneurship literature that business and private spheres are often blurred (Aldrich and Cliff, 2003; Azmat and Fujimoto, 2016; Cruz et al., 2012) and that households play an influential role in the development of entrepreneurial businesses (Baines and Wheelock, 1998; Gras and Nason, 2015; Oughton et al., 2003). Accordingly, an emerging body of research explores the ways in which interactions within the household affect venture creation and growth (Alsos et al., 2014; Carter, 2011; Sonnenberg, 2008). As this study seeks to contribute to these contemporary debates, it reviews research progress in this domain. Specifically, this section explores conceptualisations of entrepreneurial households, the interwoven relationship between household and business, and household-based entrepreneurship research.

The household perspective of entrepreneurship involves viewing and exploring entrepreneurs 'within the context of their immediate family unit' (Carter et al., 2017: 93). However, such investigations confront challenges, as the size and shape of households can vary significantly, from nuclear families (Baines and Wheelock, 1998; Livingston, 2014) to extended household units, which can include wider kin, such as grandparents and parents-in-law (Ram et al., 2001; Sanders and Nee, 1996), and unrelated individuals, such as tenants (Ellickson, 2006). In addition, the composition of households can change over time (Sund and Smyrnios, 2005), through transitions such as marriages, divorces and births (Aldrich and Cliff, 2003). Consequently, it is difficult to generalise empirical findings from one household

context to another (Gras and Nason, 2015). Furthermore, there is no universal definition of a household that applies across contexts (as cultural differences can affect perceptions of its meaning, see Peng, 2004). Similarly, definitions of the concept of family vary (Rodriguez et al., 2009). This study considers 'all individuals related by blood, marriage, or adoption' (Distelberg and Sorenson, 2009: 69) to be family.

Within the small firm literature, Wheelock and Oughton (1996) offer two commonly applied definitions of households. First, a household can be described according to its function, as 'the basic unit of society in which the activities of production, reproduction, consumption and the socialisation of children take place' (Roberts, 1991: 61). Second, it can be characterised as a 'group of people, their relationships and activities, who acknowledge a common authority in domestic matters ... [and] who have a common fund of material and human resources and rules for practices and exchange within it' (Messer, 1990). This study adopts the latter conceptualisation, as it accounts for the resources that are exchanged within households as well as the rules that guide the behaviours of its members, which is consistent with Bourdieu's (1996) perspective. Focused on the 'owners and occupants of a dwelling space' (Ellickson, 2006: 230), this study recognises that household contexts are permeable, can include both kin and non-kin and are a product of the shared histories and cultures of their members (Wheelock and Oughton, 1996).

This research explores entrepreneurial households (Oughton et al., 2003), which are distinguished by the fact that at least one of its members is a business owner or self-employed individual, who generates resources required by the household (e.g. income or necessaries) through these business activities (Carter et al., 2017). While this definition incorporates both family-owned (Miller et al., 1999) and home-based businesses (Vorley and Rodgers, 2014), this study is particularly interested in households where a single member engages in entrepreneurial activities outside of the family home. By focusing on this specific form of entrepreneurial household, this study seeks to minimise the likelihood that participants are impacted by factors which prevail in family-owned and home-based businesses, such as ulterior motives for altruistic behaviours (Karra et al., 2006; Schulze et al., 2001) and time

management issues (Thompson et al., 2009; Winter et al., 1993). This study also contends that selecting sole proprietors as research participants facilitates exploring the ways in which households affect entrepreneurial practices, as prior studies have shown that cohabitants have especially pronounced effects on solo entrepreneurs (Jennings et al., 2010; Oughton and Wheelock, 2003; Renzulli et al., 2000).<sup>5</sup>

While scholars have acknowledged that household members both enable and impede entrepreneurial processes (Aldrich and Cliff, 2003; Jennings and McDougald, 2007), 'the entrepreneurial household remains under-researched' (Carter et al., 2017: 92). A primary reason for this lack of attention is the prevailing understanding of economic and family domains as separate realms, rather than interconnected spheres (Mwaura and Carter, 2015). While businesses are usually portrayed as result-driven and objectively managed, families are often depicted as emotion-driven, unpredictable and even irrational (Dyer, 2003). Accordingly, scholars have recently suggested that households should instead be considered as contexts 'where normative systems (affect, altruism, tradition) and utilitarian systems (economic rationality) are combined' (Brannon et al., 2013: 111). Outside entrepreneurship research, in such fields as sociology, development studies and rural studies, the economic importance of households has long been established (Mulholland, 1997; Ram, 2001; Wheelock and Oughton, 1996). However, as most research in these domains does not differentiate between employed, unemployed or entrepreneurial households, there is a need for further investigations into the role of households in the decision-making of entrepreneurs (Alsos et al., 2014; Carter et al., 2017).

By adopting the household as the unit of analysis, scholars are able to address the call for entrepreneurship research to adopt a family embeddedness perspective (Aldrich and Cliff, 2003; Arregle et al., 2015; Cruz et al., 2012). Aldrich and Cliff (2003) base the need for this perspective on their analysis that socio-historical developments in the sizes, structures and compositions of families affect the roles of and relationships between family members and how families impact businesses. As families change over the course of the lifetimes of their members, business opportunities may arise as a result of shifting dynamics in the household structure (Alsos et al., 2014; Discua Cruz et al., 2013; Sanders and Nee, 1996). Accordingly,

<sup>&</sup>lt;sup>5</sup> Following Ellickson (2006) and Sonnenberg (2008), this study uses the terms 'household member' and 'cohabitant' interchangeably.

Aldrich and Cliff (2003: 592) suggest 'focusing on households, regardless of size, and not simply multi-person units in which two or more people are related', and present a framework to facilitate such research. This framework proposes that family system characteristics (i.e. household transitions, resources and norms/values) influence venture creation processes/outcomes and vice versa. Despite their persuasive argument that 'families and businesses truly are ... inextricably intertwined' (p: 590), only a limited number of studies have adopted a family embeddedness perspective (e.g. Azmat and Fujimoto, 2016; Powell and Eddleston, 2013; Wiklund et al., 2013). Furthermore, the majority of these studies focus on individuals or family businesses rather than on households (Carter at al., In Press).

Within the field of entrepreneurship, households are mainly considered sources of data for exploring and measuring the activities of individual entrepreneurs (cf. Boden and Nucci, 1997; Petrova, 2012; Reynolds et al., 2004), resulting in a disregard of household settings and their impacts on entrepreneurial activities. Entrepreneurship studies that do focus on household settings tend to employ them as contexts for analysing (1) ethnic minority businesses in western societies (Ram et al., 2001; Rodriguez et al., 2009), (2) portfolio entrepreneurship in rural areas (Alsos et al., 2014; Carter, 1998) and (3) entrepreneurial activities in developing countries (Azmat and Fujimoto, 2016; Gras and Nason, 2015). In comparison, there is a dearth of research on households which are located in industrialised regions (Parker, 2008) and which contain entrepreneurs commonly explored in the literature, such as founders of 'high-growth, technology-enabled, venture capital-backed businesses' (Welter et al., 2017: 312).

First, research on ethnic minority enterprise highlights that familism (defined as 'family ideologies and practices that privilege collectivism', see Valdez, 2016: 1620) is a common feature in ethnic households and encourages its members to prioritise the (economic) interests of the family over the concerns of any individual (Karra et al., 2006; Sanders and Nee, 1996). Among the first to study this subject was Ram (1994), who illustrated this behaviour with the example of an ethnic-minority business which was split into five firms, not in response to market needs, but rather to occupy several male members of the household. A further example is provided by Mulholland (1997), who investigates an ethnic-minority family that

expanded its business in order to accommodate the siblings of the firm founder. As Ram et al. (2001) note, such practices can protect minority groups from discrimination within the labour market and provide clear paths for career advancement to its members.

Second, studies of portfolio and farm-based entrepreneurship have similarly demonstrated the significant impact of households on entrepreneurial actions (Anthopoulou, 2010; Carter, 1998). As discussed by Iacobucci and Rosa (2010), portfolio entrepreneurs develop sets of (usually small) companies, which remain connected through joint ownership and/or management and draw on the resources of the more established businesses within the portfolio to support the development of new ventures (see, Sieger et al., 2011). Such clusters of firms are often found in farm contexts (Carter and Ram, 2003; Carter et al., 2004) and can involve complex ownership arrangements, especially when multiple people from the same household are engaged in portfolio entrepreneurship (Discua Cruz et al., 2013; Rosa, 1998). Even when organised as independent entities, it has been shown that firms operated by members of the same household are greatly interconnected (Alsos et al., 2014), providing less experienced members with opportunities for relational learning (Plate et al., 2010).

Third, the most recent body of research to embrace a household perspective explores entrepreneurship in developing economies (Azmat and Fujimoto, 2016; Khan et al., 2007). These studies highlight the crucial role that households play in starting and growing businesses in impoverished settings (Khavul et al., 2009; Ramachandran et al., 2012). For example, Gras and Nason (2015) reveal that diversity of skills, perspectives and knowledge within the household can foster business creativity. However, it should be reiterated that understandings of households within developing economies may differ (Peng, 2004). As Verver and Koning (In Press) note, such studies often move beyond conventional conceptions of the household and incorporate members of the wider community, who are considered kin on the basis of other forms of relatedness, such as shared dialect or descent, and who 'mimic the ties and transactions within the family' (Iyer, 2004: 247). Consequently, scholars are hesitant to generalise findings from developing countries to developed contexts (Gras and Nason, 2015).

Beyond household-based studies of ethnic minority entrepreneurship, portfolio entrepreneurship and entrepreneurship in developing countries, 'the adoption of a household perspective remains relatively unusual within the entrepreneurship research domain' (Carter et al., 2017: 123). Thus, further empirical research is required to explore the impact of households on entrepreneurs.

# 4.3.2. Resources of the Household

Resource transactions within households and families have attracted sustained scholarly interest (Caputo and Dolinsky, 1998; Coleman, 1988; Khayesi et al., 2014), as they can help entrepreneurs overcome resource scarcity and thus facilitate entrepreneurial activities (Kotha and George, 2012; Sanders and Nee, 1996; Sieger et al., 2011). Households can provide their members with access to a range of resources, such as human and social capital, which support venture development and growth (Danes et al., 2009; Kim et al., 2006; Renzulli et al., 2000). To date, research has focused on the effects of household income (i.e. financial capital) on entrepreneurs (De Clercq et al., 2013; Klyver and Schenkel, 2013; Newbert and Tornikoski, 2013). Consequently, there have been repeated calls for entrepreneurship research to investigate other types of resources that become available through embeddedness in households (Rodriguez et al., 2009; Steier, 2009). This section reviews research progress in the domain of household capital, its exchange norms and its effects.

While prior research has explored the relationships between individual forms of capital provided by household and family, such as financial and human capital (Bird and Wennberg, 2016; Caputo and Dolinsky, 1998) or financial and social capital (Edelman et al., 2016; Niehm et al., 2008), a broader range of resources is rarely incorporated into empirical analyses (Matzek et al., 2010). Thus, Rodriguez et al. (2009) and Steier (2009) developed the concept of household capital to encourage explorations of the various resources that become accessible to entrepreneurs through embeddedness in households. As household capital represents an extension of the concept of family capital (Danes et al., 2009; Sorenson and Bierman, 2009), this concept is introduced first.

For the purposes of this study, family capital is defined as 'the human, social, and financial resources that are available to individuals or groups as a result of family affiliation' (Dyer et al., 2014: 266). Alternatively, it has been conceptualised as a special form of social capital, which is more enduring, intense and readily available than other forms (Hoffman et al., 2006) and it 'has been more or less equated' with the concept of 'familiness' (Dyer et al., 2014: 267).<sup>6</sup> A few studies have also focused on family human capital (Sirmon and Hitt, 2003) or family financial capital (Harvey and Evans, 1995), leading to a largely disjointed body of research. Thus, it has been suggested that all three forms of capital should be combined in future analyses (Sorenson and Bierman, 2009). Prior research shows that family members continuously draw on and add to the available stock of resources within their family circle (Rodriguez et al., 2009), indicating that levels of family capital are constantly in flux. Danes et al. (2009) also highlight that the value of the resources possessed by the family can be greater than the sum of its individual amounts of capital, as combinations of resources can amplify their overall worth and volume. At the same time, debts of family members, such as educational loans or lawsuit settlements, can reduce the level of family capital, as the liability for their repayment can rest with or fall to other members of the collective (Bird and Wennberg, 2016; Edelman et al., 2016).

The concept of family capital is similar to Oughton and Wheelock's (2003: 7) 'household endowments', which encompass the 'physical and financial assets, human and social capital' accessible to the household. However, their concept and approach are largely focused on the constraints that derive from limited stocks of capital and the impacts of associated insecurities and risks. This study therefore employs the concept of household capital, which emphasises that access to jointly created and held resources can support firm development (Steier, 2009). By highlighting that resources are not only shared by family members, but also other individuals living within the same household, household capital extends the concept of family capital (Rodriguez et al., 2009). Furthermore, by focusing on household

<sup>&</sup>lt;sup>6</sup> Habbershon et al. (1999) introduced the term 'familiness' to family business research to describe the unique bundles of resources that are created by involvement of family members in the business and which provide family firms with strategic advantages over other firms.

members only, this approach avoids the inherent difficulties in operationalising the concept of 'family' (Aldrich and Cliff, 2003; Carter et al., 2017).

Household capital represents the financial, human and social capital possessed by the household and accessible to entrepreneurs (Rodriguez et al., 2009; Steier, 2009). Several studies explore the financial resources (i.e. household income, see Petrova, 2012; Reynolds et al., 2004) and associated benefits that become available to entrepreneurs through their households (Caputo and Dolinsky, 1998; Carter, 2011). Importantly, *household financial capital* incorporates not only monetary assets, but also 'other tangible assets such as the family home as the business office, family vehicles, phones, and computers that may be used to start a new business' (Dyer et al., 2014: 269). As such, the concept is markedly similar to Bourdieu's (1986) economic capital. It has been shown that financial means are usually provided by household members without an expectation of repayment, but in exchange for gratitude and love (Arregle et al., 2007; Edelman et al., 2016). Household financial capital therefore creates an invaluable safety-net for entrepreneurs as it acts as a buffer against external shocks (Dyer et al., 2014; Sirmon and Hitt, 2003). If it is not provided for free, accessing this type of capital usually holds lower transaction costs and fewer strings attached than alternative forms of funding (Steier, 2003). However, feelings of indebtedness may still arise and lead entrepreneurs to indirectly pay household members back, such as by taking care of those who are ill or unemployed (Steier, 2009). Danes et al. (2009) argue that small firm owners are particularly likely to draw on the savings of the household to develop their firms, as it is harder for them to acquire funding from external lenders (Bird and Wennberg, 2016; Sanders and Nee, 1996). Thus, it has been suggested that this type of support from household and family is the 'lifeblood of new ventures' and crucial for entrepreneurs (Edelman et al., 2016: 431).

*Household human capital* is another important asset that is available to entrepreneurs (Coleman, 1988; Kotha and George, 2012). It has been interpreted either as the labour that household members are able to provide (Rodriguez et al., 2009) or as the combined abilities, skills and values of cohabitants (Caputo and Dolinsky, 1998). This study adopts the latter conceptualisation, as labour in the form of time used to work in the entrepreneurial firm is considered a manifestation of

economic capital. Human capital theory suggests that individuals with considerable skills and specialist knowledge are more likely to achieve entrepreneurial success (Marvel et al., 2016; Shane, 2003), which implies that entrepreneurs who have access to households with larger amounts of human capital (i.e. partners who possess valuable skills) possess a competitive advantage (Bird and Wennberg, 2016; Coleman, 1988). As Danes et al. (2009) note, entrepreneurs can develop household human capital by passing on firm-specific knowledge to their partners and children, which, in turn, enables these household members to provide better support to the business. According to Dyer et al. (2014: 267), such processes often involve 'informal conversations over the dinner table', watching the entrepreneurs work or helping them, which can lead subsequent generations to adopt the trades and professions of the entrepreneurs (Laspita et al., 2012). At the same time, knowledge of cohabitants can 'spill over' to entrepreneurs and expand their horizons (Kim et al., 2013). Given that advice and information provided by family members is considered to be more trustworthy than from other sources (Bird and Wennberg, 2016), such support tends to be highly valued by entrepreneurs.

*Household social capital* can also provide entrepreneurs with considerable advantages (Coleman, 1988; Steier, 2009). It refers to the household's 'networks of relationships in which personal and organizational contacts are closely embedded' (Edelman et al., 2016: 431). Thus, it resides in the relationships between cohabitants and their external networks (Danes et al., 2009) and it can grant entrepreneurs access to discounted or complimentary resources, such as financial or material contributions (Bullough et al., In Press; Edelman et al., 2016). Dyer et al. (2014) argue that, by drawing on the long-term relationships of cohabitants, entrepreneurs can acquire valuable, rare assets, such as technological innovation or insider information, which can provide them with competitive advantage. In other words, entrepreneurs are able to utilise (or 'borrow') the strong ties of household members, which would otherwise be unavailable to them or require extended periods of time to develop (Arregle et al., 2007; Khavul et al., 2009). As Edelman et al. (2016) note, access to existing social networks, including supplier and customer contacts of household members, is particularly important for young entrepreneurs, who lack the social connections that can help them achieve entrepreneurial success. However, there is an inherent danger that extant trust can be damaged during this process if entrepreneurs do not adhere to the norms and values that were established between household members and their external ties (Danes et al., 2009), indicating that successful utilisation of household social capital requires entrepreneurs to access the human capital of cohabitants.

Household capital can therefore provide entrepreneurs with a variety of advantages, including access to resources which are inimitable, readily available and free from (or requiring low) transaction costs (Dyer et al., 2014). Nevertheless, it requires entrepreneurs to abide by norms and expectations set by household members, which need to be taken into consideration. A number of factors can influence access to household capital and reduce the ability of entrepreneurs to acquire its benefits (Dyer et al., 2014; Pearson et al., 2008).

First, resource exchanges rely on trust, which is usually considered to be high within families (Arregle et al., 2007), but can differ between individuals and be affected by negative past experiences, such as breaches of confidence. If portrayals in the media are to be believed, the archetypal entrepreneur is a 'lone hero', who is 'isolated and rather remote from even his closest relatives ... a person under a lot of stress who often upsets [others] by his seemingly "irrational", impulsive activities' (Kets de Vries, 1977: 35-36). Such depictions have, of course, been challenged (Drakopoulou Dodd and Anderson, 2007; Nicholson and Anderson, 2005), but if entrepreneurs are to act in this manner, they are unlikely to engender trust and generate support from their household members. Furthermore, it might be expected that levels of trust and the willingness to exchange resources are lower for unrelated cohabitants than for family members living in the same household (Kim et al., 2013; Rodriguez et al., 2009).

Second, extant norms and obligations can reduce the ability of entrepreneurs to draw on the resources of the household. As Stewart (2003: 385) notes, relationships between family members are directed by an order that is 'at odds with the amoral logic of markets'. Prior research has shown that this order consists of norms which encourage openness, teamwork, egalitarianism, altruism and reciprocity (Arregle et al., 2015; Karra et al., 2006) and thus promote collective actions over self-seeking behaviours (Pearson et al., 2008). However, strong norms which emphasise reciprocity may dictate that entrepreneurs provide help or access to required resources, preventing them from engaging in entrepreneurial activities (Adler and Kwon, 2002). Prevailing household values may also discourage the use of household financial assets for business purposes (Carter, 2011), necessitating the use of external funding provided at higher costs to entrepreneurs (Carter et al., 2007; Khayesi et al., 2014).

Third, a lack of time spent within the household can reduce the willingness of cohabitants to impart their resources to entrepreneurs (Pearson et al., 2008). Limited interactions with cohabitants can also negatively impact opportunity identification, as it can prevent awareness (and therefore utilisation) of available resources (Carter et al., 2017). In addition, conflicts between household members can temporarily, or even permanently, affect their relationships, which can reduce the willingness of cohabitants to engage in altruistic behaviours and result in inferior resource exchange rates (Miller et al., 2003; Shelton, 2006).

Changes in the composition of the household can also affect access to resources (Aldrich and Cliff, 2003; Jaskiewicz et al., 2017). On the one hand, alterations such as marriages, adoptions or accommodations of lodgers, can extend the pool of resources available to entrepreneurs (Khavul et al., 2009; Matzek et al., 2010). On the other hand, losses within the household, induced by divorces, deaths or the relocation of children or tenants, can reduce the supply of available resources (Baines et al., 2003; Dyer et al., 2014). As Mwaura and Carter (2015) note, such transitions impact the overall needs of the household and can put emotional strain on entrepreneurs (Shepherd, 2009). Furthermore, household members can choose to withdraw resources in order to use them outside of the business of the entrepreneur. Thus, household capital is not fixed in scale, size and availability (Alsos et al., 2014), but rather fluctuates over time.

It has been also argued that over-reliance on the support of household members can lead entrepreneurs to discount alternative sources of capital and increase the risks of business failure (Dyer, 2006). As Alsos and Carter (2006) note, the resources household members provide may be inferior to those available from external suppliers. By relying on household capital, entrepreneurs may therefore sacrifice some of their competitiveness in terms of product or service quality. Significant reliance on the resources of the household may also result in a lock-in situation, which can affect performance (Sanders and Nee, 1996). Furthermore, by withdrawing resources from the household, entrepreneurs can put the wellbeing of their families at risk, which in turn can negatively affect their business operations (Carter et al., 2017). In other words, household capital not only provides benefits that can help entrepreneurs develop their firms; it also poses certain risks and incurs costs that entrepreneurs must weigh.

# 4.3.3. The Role of Emotions

In addition to providing financial, human and social capital, household and family members are important sources of emotional support for entrepreneurs (Brüderl and Preisendörfer, 1998; Renzulli et al., 2000). While prior research has shown that starting and developing businesses are emotional processes (Cardon et al., 2005; Patzelt and Shepherd, 2011; Welpe et al., 2012), the role of emotions has, until recently, been largely overlooked in the field of entrepreneurship (Baron, 2008; Foo, 2011). In response to repeated calls for more dedicated research on the subject (Cardon et al., 2012; Morris et al., 2012; Shepherd, 2015), there has been significant interest in the powerful emotions experienced by entrepreneurs, such as entrepreneurial passion (Biraglia and Kadile, 2017; Cardon et al., 2009; Stenholm and Renko, 2016) and grief from business failure (Byrne and Shepherd, 2015; Cope, 2011; Shepherd, 2003). In comparison, little attention has been afforded to the emotions aroused by others in entrepreneurs (Biniari, 2012; Foo et al., 2009). As Jennings et al. (2015: 114) argue,

many fundamental questions remain 'sorely in need of study' ... some of the most important of these enduring questions stem from the field's historic glamorization of the lone entrepreneur – a preoccupation that is out of step with the increased recognition that much entrepreneurial activity occurs within teams or at least relies upon inputs from other individuals. This preoccupation is reflected in the entrepreneurial emotion literature, as evidenced by the relative preponderance of studies addressing *intra* rather than *inter* personal questions.

Thus, research on the emotional impacts of household members on entrepreneurs can make valuable contributions to knowledge. While the importance of emotions within families has long been recognised by family business research (Berrone et al., 2012; Gómez-Mejía et al., 2007; Holt and Popp, 2013; Kellermanns et al., 2012), empirical investigations outside of family-run firms are rare (Eddleston and Powell, 2012; Powell and Eddleston, 2013). Although many studies touch upon this process by acknowledging the emotional support that household members provide to entrepreneurs (cf. Alsos et al., 2014; Danes et al., 2009; Steier, 2009; Stewart, 2003), there has been scant dedicated research on the subject (Arregle et al., 2015; Edelman et al., 2016). Consequently, to analyse and interpret its empirical findings regarding the field of entrepreneurship, which are examined in this section. First, the social support perspective is introduced, before research on the work-family balance is reviewed. Subsequently, the concept of psychological capital is presented.

There has been sustained scholarly interest in exploring the effects of mental health in the workplace and its role in preventing stress (Cobb, 1976; Cohen and Wills, 1985; Uchino, 2004). Prior research has shown that embeddedness in social contexts not only causes stress, but also helps alleviate it through the provision of support from social relationships, which can promote well-being and thus reduce health risks (Lazarus and Folkman, 1984). Social support has been defined as 'the perception or experience that one is loved, cared for by others, esteemed, valued, and part of a mutually supportive social network' (Edelman et al., 2016: 430). According to Uchino (2004) social support encompasses both the structures of the social lives of individuals, such as family and work relationships, and the functions served by these structures.

Social support functions are either instrumental (i.e. tangible), such as through the supply of resources and assistance in problem solving (Cohen and Wills, 1985; McIntosh, 1991), <sup>7</sup> or emotional, including through the provision of encouragement, attention and understanding (Adams et al., 1996). Social support can be received in the work domain from coworkers and supervisors (Abbas et al., 2014) and in the family domain from partners, children and other kin (King et al., 1995). The latter context is of primary interest to entrepreneurship research (Arregle et al., 2015; Brüderl and Preisendörfer, 1998), as it is considered the 'bread and butter'

<sup>&</sup>lt;sup>7</sup> This study employs the concept of household capital instead of instrumental social support, as it facilitates a more nuanced differentiation of the various types of resources that cohabitants provide to entrepreneurs.
source of social support for entrepreneurs (Powell and Eddleston, In Press: 3). Emotional social support in particular has been shown to be common among family members (Renzulli et al., 2000). It can influence how individuals think about events, solve problems and adapt to difficult situations (Uchino, 2004) and thus represents an essential asset for facilitating entrepreneurial success (Edelman et al., 2016; Powell and Eddleston, 2013).

Cohen and Wills (1985) propose that emotional social support helps create a buffer that alleviates or prevents the negative outcomes of stress and has an overall beneficial effect on well-being, irrespective of the stress levels that individuals experience (Taylor, 2011). This argument influenced scholarly debates on the work-family balance, as discussed below. While emotional social support from family members can benefit any working individual, it is particularly important for entrepreneurs (Brüderl and Preisendörfer, 1998), as many entrepreneurial firms are owned and run by enterprising individuals, who lack the opportunity to exchange empathy and encouragement with coworkers (Powell and Eddleston, In Press). Indeed, the Department for Business, Innovation and Skills (BIS, 2016: 2) estimates that '76% of [UK] businesses [do] not employ anyone aside from the owner'.

According to Edelman et al. (2016) emotional social support is particularly strong in families with high cohesiveness, meaning that family members have a deep connection and emotional bond with one another (Laspita et al., 2012). High cohesiveness is usually found in families that share strong norms and understandings and it creates a feeling of 'togetherness' and solidarity in the pursuit of common goals (Sharma, 2008; Wiklund et al., 2013). In other words, cohesive families are likely to be reliable sources of emotional social support for entrepreneurs. However, risk-adverse families that value stability and salaried employment may not support new venture creation (Arregle et al., 2007), which suggests that high cohesiveness can also impede entrepreneurial efforts should entrepreneurs seek to avoid relational conflicts with relatives who subscribe to such a perspective (Edelman et al., 2016).

Social support theory has generated considerable scholarly interest and debate, most notably concerning the effects on work-family balance (Adams et al., 1996; Greenhaus and Powell, 2006; Livingston, 2014). According to Jennings and McDougald (2007: 747), research on work-family balance represents 'a rich and

relevant body of literature that remains relatively untapped by entrepreneurship scholars' and has the potential to shed new light on the intertwined relationship between entrepreneurship and family. Research in this domain can generally be classed into one of two competing views. The dominant conflict perspective suggests that the needs of work and family are largely incompatible, causing various problems and tensions (Powell and Eddleston, 2013). Frequently observed issues include strain-based conflicts (e.g. spillovers of stress from one domain into the other), behaviour-based disputes (e.g. clashes between personal and professional conducts) and time-based issues (Boswell and Olson-Buchanan, 2007; Greenhaus and Beutell, 1985; Shelton, 2006). In contrast, the less espoused enrichment perspective proposes that fulfilling multiple roles can be advantageous for individuals, creating spillovers of positive behaviours, emotions and attitudes between work and family domains (Carlson et al., 2014; Eddleston and Powell, 2012; Gudmunson et al., 2009).

Prior studies have focused on exploring subjective outcomes, such as wellbeing, and revealed that work-family conflict generally increases stress and reduces satisfaction with family, work or both (Higgins and Duxbury, 1992; Parasuraman et al., 1992). While the majority of studies have examined employment contexts, entrepreneurship research has made similar observations concerning the perceived effects of work-family conflict (Parasuraman et al., 1996). Objectively measurable outcomes include increased absenteeism, decreased work performance and reduced commitment (Graves et al., 2007). A number of strategies have been identified which can be employed to minimise the potential for work-family conflict (Jennings and McDougald, 2007). Couples may adopt (1) a traditional approach, where one partner assumes the responsibilities of the household in a stay-at-home role, (2) a one-career/one-job approach, where one partner's career is prioritised over the other's job, or (3) a dual career approach, which usually necessitates certain concessions, such as postponing or forgoing children, hiring housekeepers and nannies or scaling back either in the work or the home domain. Indeed, it has been argued that those entrepreneurs who employ effective strategies to reduce workfamily conflict are better able to achieve business success (Shelton, 2006).

An emerging body of research has begun to focus on the ways in which families enrich the entrepreneurial experience by nurturing work-family balance (Eddleston and Powell, 2012; Powell and Eddleston, In Press). Originating in role theory (Kahn et al., 1964), enrichment theories explore how 'experiences in one role improve the quality of life in the other role' (Greenhaus and Powell, 2006: 72). Enrichment can transfer from work to family or from family to work, the latter of which is considered to have significantly stronger effects on individuals (Greenhaus and Powell, 2006). This 'family-to-business enrichment' perspective has generated interest within entrepreneurship research (Eddleston and Powell, 2012; Powell and Eddleston, In Press), as it is compatible with the family embeddedness perspective (Aldrich and Cliff, 2003), which highlights the importance of family for successful entrepreneurship and proposes that the well-being of entrepreneurs can be enhanced by participation in the family. Similar to the concept of social support (see above), family-to-business enrichment incorporates both instrumental and emotional components (Greenhaus and Powell, 2006). Individuals experience instrumental enrichment when they transfer behaviours and skills gained in the family to the business, whereas emotional enrichment occurs when positive affect (e.g. happiness) is transferred from family to work (Powell and Eddleston, 2013). It has been suggested that emotional enrichment is particularly beneficial for entrepreneurs, as positive affect can stimulate creativity, improve problem-solving abilities and support opportunity identification (Foo et al., 2009; Hayton and Cholakova, 2012).

Prior research suggests that experiences of work-family balance differ for male and female entrepreneurs (Eddleston and Powell, 2012; Powell and Eddleston, 2013). Gender roles which are instilled through socialisation processes over time (cf. Ahl, 2006; Bird and Brush, 2002; Elam, 2008) have been shown to prompt women to start firms in order to gain the flexibility required to fulfil their perceived family responsibilities and achieve better work-family balance (Brush, 1992; DeMartino and Barbato, 2003). In addition, traditional gender roles may encourage female entrepreneurs more than male entrepreneurs to develop positive connections between their work and family roles (Eddleston and Powell, 2012). Female entrepreneurs tend to bear higher family demands, which are rarely reduced following start-up (Cliff, 1998; Jurik, 1998), whereas male entrepreneurs usually find relief from domestic responsibilities upon establishing their ventures (Parasuraman et al., 1996). Accordingly, it has been suggested that female

entrepreneurs adopt different coping strategies than men, often in the form of prioritising home life over work (Jennings and McDougald, 2007). Empirical research in this vein has been limited and findings have not been conclusive (Jennings et al., 2010), but it is anticipated that such strategies would serve to constrain growth of female entrepreneurs' businesses (Shelton, 2006).

Emotional support from family members has been shown to help individuals understand risks, cope with uncertainty (Cohen and Wills, 1985; Cordes and Dougherty, 1993) and overcome exhaustion by improving motivation, confidence and commitment (Baruch-Feldman et al., 2002; Lee and Ashforth, 1996). While these processes have frequently been observed for employees of larger organisations, there is limited understanding of the effects of emotional social support on entrepreneurs (Arregle et al., 2015; Brüderl and Preisendörfer, 1998). Specifically, little is known about the psychological strengths and capabilities that are developed with the help of household and family members. Thus, to analyse and interpret the experiences of participating entrepreneurs, this study draws on the concept of psychological capital (Luthans and Youssef, 2004; Luthans et al., 2007), which has only recently been introduced to the entrepreneurship literature (cf. Baron et al., 2013; Zou et al., 2016).

Psychological capital was developed as part of a new movement called positive psychology, which redirected the focus of psychological research from fixing the problems of people (i.e. concentrating on the negative) towards making their lives more productive and rewarding by enhancing their extant qualities and strengths (Luthans, 2002; Seligman and Csikszentmihalyi, 2000). Psychological capital was originally conceived to facilitate positive organisational behaviour through proactive management and workplace interventions (Luthans and Youssef, 2004). Luthans et al. (2007: 3) define psychological capital as

an individual's positive psychological state of development that is characterized by: (1) having confidence (self-efficacy) to take on and put in the necessary effort to succeed at challenging tasks; (2) making a positive attribution (optimism) about succeeding now and in the future; (3) persevering toward goals and, when necessary, redirecting paths to goals (hope) in order to succeed; and (4) when beset by problems and adversity, sustaining and bouncing back and even beyond (resiliency) to attain success.

As such, psychological capital represents emotional, cognitive and behavioural resources upon which individuals can draw to respond to challenging circumstances (Avey et al., 2010; Newman et al., 2014b). For example, entrepreneurs who have high levels of *self-efficacy* believe that they can 'get the job done' and accomplish what they set out to do (Baron et al., 2013: 747). A number of studies have shown that self-efficacy increases both entrepreneurial intentions and new venture performance (Bullough et al., 2014; Cardon and Kirk, 2015; Forbes, 2005). Similarly, entrepreneurs high in *optimism* think that they will experience favourable conditions and outcomes in most situations (Hmieleski and Baron, 2009). They take credit for their successes and consider setbacks as opportunities rather than failures, allowing them to mitigate potential stresses (Anglin et al., In Press; Dushnitsky, 2010; Ucbasaran et al., 2010). Entrepreneurs who have large amounts of hope maintain both short and long term business goals, motivation to achieve these objectives and the ability to envision multiple ways to overcome potential challenges (Snyder et al., 1996), all of which supports business start-up and development (Baron et al., 2013; Hmieleski et al., 2015; Zou et al., 2016). Finally, highly resilient entrepreneurs are able to encounter challenging situations, develop means of overcoming them and draw on those experiences to cope with future difficulties (Bullough et al., 2014). Within entrepreneurship research, such capabilities have mainly been explored in the contexts of business failure and dealing with adversity (Byrne and Shepherd, 2015; Cope, 2011; Williams and Shepherd, 2016).

Given that self-efficacy, optimism, hope and resilience all help prevent stress (Abbas et al., 2014; Avey et al., 2010), it may be assumed that strategies employed to maximise these psychological capacities will facilitate achieving work-family balance. Interestingly, this link has not previously been explored in the literature. While self-efficacy, optimism, hope and resilience have individually been explored within the entrepreneurship literature (see above), there is a lack of research incorporating more than one of these psychological assets (Bullough et al., 2014; Newman et al., 2014a). Furthermore, those studies that do employ the concept of psychological capital tend to focus on the experiences of individual entrepreneurs (Baron et al., 2013; Hmieleski et al., 2015; Zou et al., 2016). Consequently, little is known about the effects of collectively developed capacities, such as 'collective

optimism [which] impacts individual decision making because decisions are frequently made in a social context where individuals are influenced by the views, expectations, and beliefs of others' (Anglin et al., In Press: 2). Accordingly, 'calls have been made to investigate the potential for a collective version of the construct' of psychological capital (Dawkins et al., 2015: 926). Research into the role of household members in developing positive psychological capacities would therefore make value contributions to the literature.

### 4.4. Embeddedness in Communities

Entrepreneurs are embedded in and affected by a 'multiplicity of contexts' (Welter, 2011: 168). Prior research has not only explored the influence of embeddedness in families and households, but also in networks (Hite, 2005; Ozdemir et al., 2016; Simsek et al., 2003), places (Jack and Anderson, 2002; Kalantaridis and Bika, 2006; Korsgaard et al., 2015) and institutions (Battilana, 2006; Greenwood and Suddaby, 2006; Kisfalvi and Maguire, 2011). Among others, it has been shown that embeddedness in these contexts can provide entrepreneurs with access to resources (Batjargal, 2003; Grichnik et al., 2014; Newbert and Tornikoski, 2013), opportunities for collaboration (Coulson, 2012; Kuhn and Galloway, 2015; Sorenson et al., 2008) and valuable information and advice (Gras and Nason, 2015; Johannisson et al., 2002; Shaw et al., 2017). In comparison, there is limited understanding of the effects of embeddedness in communities on entrepreneurs (Martí et al., 2013; McKeever et al., 2015), particularly in those which are not bound by geographic proximity (Piva et al., 2012; Reuber and Fischer, 2011). Thus, this study seeks to advance understanding by exploring how and why embeddedness in communities affects the practices of entrepreneurs. This section therefore reviews extant literature on embeddedness theory and on communities and entrepreneurship. Subsequently, it examines research on social value creation, as it was found to be significant for the findings of this study (see Chapter 9).

### 4.4.1. Embeddedness Theory

The term embeddedness was first introduced to the business literature by Polanyi (1944: 46), who proposed that 'man's economy, as a rule, is submerged in his social relationships.' Drawing on this perspective, Granovetter (1985) argues that economic theory has adopted an 'undersocialised' approach and that researchers should account for the social contexts in which economic exchanges occur. Rather than being seen as impersonal and strictly rational, transactions in the marketplace should be understood as embedded in ongoing systems of social relations (Uzzi, 1997). Entrepreneurship scholars have echoed this assertion, suggesting that entrepreneurs should be studied within their contexts to better understand their behaviours (Aldrich and Fiol, 1994; Welter, 2011; Zahra et al., 2014). While individual definitions of the term embeddedness vary (for a detailed analysis and comparison, see Vestrum, 2014), there is a general consensus that 'embeddedness is broadly defined as individuals' exposure to social relationships that may facilitate or impede economic action' (Bird and Wennberg, 2016: 690).

Zukin and DiMaggio (1990) were amongst the first to provide a typology to differentiate between four types of embeddedness. First, cognitive embeddedness refers to 'the regularities of mental processes that limit ... economic reasoning' (Hess, 2004: 171). Second, cultural embeddedness relates to the shared meanings and beliefs about appropriate behaviours, which direct economic actions (Greenman, 2013). Third, structural embeddedness involves the contextualisation of economic exchanges in interpersonal relationships (i.e. social networks) (Granovetter, 1985). Finally, political embeddedness refers to the impact of social institutions (e.g. tax policies or legal codes) which affect individual behaviours (Kistruck and Beamish, 2010). As Hess (2004) argues, this typology is useful for analytical purposes, but it has contributed to a fuzziness around the concept of embeddedness. Indeed, classifications are being increasingly complex (Vestrum, 2014) as additional forms of embeddedness are being identified, such as emotional (Biniari, 2012), placial (Korsgaard et al., 2015) and venture-identity embeddedness (Fisher et al., 2016).

Within the entrepreneurship literature, research has focused on exploring how entrepreneurs become embedded in social structures in order to access opportunities and resources (Hite, 2005; Jack and Anderson, 2002; Uzzi and Gillespie, 2002). Early studies developed understanding of the importance of *structural embeddedness* (i.e. the position of an entrepreneur in a broader network) by investigating the characteristics of ego-centred networks, including their diversity, density and size (Donckels and Lambrecht, 1995; Dubini and Aldrich, 1991; Simsek et al., 2003). However, results have been mixed, with some studies highlighting the value of closed networks (consisting of dense groups of actors) that engender trust, reciprocity and continuity (Coleman, 1988; Steier and Greenwood, 2000) and others demonstrating the benefits of sparse networks, which are rich in structural holes and thus facilitate the exchange of non-redundant information (Burt, 1992; Zaheer and Soda, 2009). Consequently, there has been a growing recognition that the effects of structural properties of networks are dependent upon context and venture development stage (Batjargal, 2010; Hite and Hesterly, 2001).

Entrepreneurship research has also investigated the role of *relational* embeddedness. According to Hite (2005: 114), 'ties that enable exchange and influence the firm's economic choices on the basis of the relationship are relationally embedded'. Within the entrepreneurship literature, there has been considerable debate concerning the importance of weak versus strong ties (Elfring and Hulsink, 2007; Granovetter, 1973; Jack, 2005), which are differentiated by the amount of intimacy, emotional involvement, time and reciprocal services that characterise the tie. As explained in Chapter 1, no consensus has been reached on this subject, as different ties have been shown to be useful for different purposes at different times (Jack, 2010; Rasmussen et al., 2015). Trust (conceptualised as the 'willingness to be vulnerable', see De Carolis and Saparito, 2006: 44) plays an important role in these discussions, as it facilities the exchange of valuable information (Welter and Smallbone, 2006) and eliminates the need for costly, legal contracts (Newbert and Tornikoski, 2013). Embeddedness generates a specific form of trust, called relational trust, which engenders the confident belief that trusted partners (i.e. strong ties) will act beneficially towards one another because they care about each other's welfare (Jonsson and Lindbergh, 2013). It has been shown that entrepreneurs can develop 'arm's length ties' (i.e. weak ties) into relationally embedded ties by expressing goodwill, earning trust and developing personal relationships over time (Hite, 2005:

132). This process has been found to increase the access to resources for entrepreneurs (Hoang and Antoncic, 2003), particularly to knowledge and financial capital (Bhagavatula et al., 2010; Uzzi, 1999).

Indeed, there has been considerable interest in the impact of embeddedness on the abilities of entrepreneurs to access and utilise different resources (Bird and Wennberg, 2016; Grichnik et al., 2014). As Steier and Greenwood (2000) point out, the fact that a contact is in possession of a valuable resource does not guarantee the dispensation of that resource to the entrepreneur. Ozdemir et al. (2016) provide a detailed analysis of the implications of structural and relational embeddedness for resource acquisition. They propose that, while relational embeddedness helps create the trust and mutual understanding necessary to facilitate resource exchanges, it requires significant amounts of time and energy to maintain the requisite strong connections. Accordingly, it has been argued that structural embeddedness (in sparse networks) is a cost-effective alternative, as it provides entrepreneurs with links to agents who can act as brokers and character references without requiring significant investments into relationship development (Ahuja et al., 2012; Stam, 2010).

As Aldrich and Zimmer (1986) note, embeddedness not only facilitates, but also impedes entrepreneurial actions. Prior research reveals that embeddedness can turn into a liability when social responsibilities override economic imperatives, which Uzzi (1997: 57) considers a result of 'overembeddedness' (i.e. a state where a group of people is connected through a dense network of intensive relationships, which leads them to avoid building new connections with outsiders). For instance, friendships and social obligations can lead entrepreneurs to distribute and share resources that they should instead use to develop their own ventures (Khayesi et al., 2014; Stewart, 2003). In addition, being 'trapped-in-one's-own-net' (Lechner and Leyronas, 2009: 658) by overly relying on existing strong ties can restrict the abilities of entrepreneurs to exploit new opportunities and grow their businesses (Gargiulo and Benassi, 2000). Furthermore, the ability of entrepreneurs to influence groups (by themselves) was found to be limited (Lechner and Leyronas, 2009), highlighting the interdependency of embedded agents (Anderson et al., 2007; Chell and Baines, 2000). Despite these advances in understanding, there have been calls for research to further explore the ways in which embeddedness can inhibit entrepreneurial actions (Battilana, 2006; Kilduff and Brass, 2010) and how entrepreneurs respond to the constraints that result from embeddedness (Maurer and Ebers, 2006).

# 4.4.2. Communities and Entrepreneurship

As explained above, extant research has focused on exploring the structural and relational embeddedness of entrepreneurs (Jack, 2010; Slotte-Kock and Coviello, 2010). Thus, considerable knowledge has been developed regarding the ways in which certain network characteristics (i.e. structures and tie strengths) either enable or impede entrepreneurial processes (Aarstad et al., 2010; Jack et al., 2010; Ozcan and Eisenhardt, 2009). However, while Granovetter's (1985: 487) embeddedness in 'concrete, ongoing systems of social relations' is usually interpreted as embeddedness in networks (Kilduff and Brass, 2010; Simsek et al., 2003; Uzzi, 1996), it can also refer to other contexts, such as families (see above), regions (Kalantaridis and Bika, 2006) and cultures (Greenman, 2013). Of particular interest to this section is embeddedness in communities (McKeever et al., 2015; Vershinina et al., 2011), which has received limited attention within the entrepreneurship literature (Jennings et al., 2013). As Hindle (2010: 600) notes, 'different communities, at different times, will affect different entrepreneurial actors and processes in different ways', indicating that research findings are not readily generalisable across communities. There have therefore been calls for research to explore a wider variety of community contexts in order to extend knowledge of their influences on entrepreneurs (Johnstone and Lionais, 2004; Parkinson et al., 2017).

Martí et al. (2013: 13) criticise existing studies for tending 'to reduce communities to geographical arenas', such as local communities (Johannisson, 1986; Mair et al., 2012) and regional clusters (Aldrich and Kim, 2007; Tan et al., 2013). This spatial focus is due to the prevailing conceptualisation of communities as geographically bound groups of people (Jennings et al., 2013). Consequently, there is limited knowledge of the impact of embeddedness in virtual (Morse et al., 2007; Piva et al., 2012) and common interest communities (Biraglia and Kadile, 2017; Reuber and Fischer, 2011) on entrepreneurs, despite their increasing prevalence in the modern age (Smith et al., 2017). In order to account for these contexts, including

craft communities, this study employs the definition of Marquis, Lounsbury and Greenwood (2011: xvi):

Communities are collections of actors whose membership in the collective provides social and cultural resources that shape their action. Membership can result from a number of factors including propinquity, interest in a common goal, or common identity.

This approach incorporates communities which share interests other than economic gain, such as a passion for crafts (Biraglia and Kadile, 2017), and whose members do not necessarily reside in close proximity to each other. Furthermore, though this research does not limit itself to discussing only social and cultural capital, this conceptualisation takes into account such resources becoming available through membership in a community.

Research has demonstrated that embeddedness in communities allows entrepreneurs to access local resources (Besser et al., 2006). Based on a conceptual discussion of regional communities, Hindle (2010: 600) differentiates between the physical resources (e.g. land and infrastructure), human capital (i.e. skills embodied in the local labour force) and social capital (e.g. network contacts) that communities provide to entrepreneurs. In addition, Ring et al. (2010) propose that 'smaller communities ... may offer a multitude of resources such as volunteer labor, access to ... donated equipment, technical information, and advice that may make very significant contributions to ... new enterprise[s]'. However, most of these scholarly discussions are theoretical in nature and there are only a limited number of empirical studies which focus on the provision and sharing of resources within communities.

Such studies have shown that embeddedness in local communities grants entrepreneurs access to financial support (Khavul et al., 2013) and provides them with contacts, knowledge and advice (Jack and Anderson, 2002; Zhang et al., 2016). Furthermore, they highlight the willingness of community members 'to share information about new techniques, suppliers, customers [and] technology' (Miller et al., 2007: 653). Indeed, this type of knowledge exchange has also been observed in virtual communities (Kuhn and Galloway, 2015), where posted information (e.g. on blogs or forums) becomes available to a broad range of users (O'Mahony, 2003). In

comparison, there is scant empirical evidence of other resources that community members exchange with each other (Vershinina et al., 2011).

Prior research has shown that entrepreneurial activities can help develop local communities (Johnstone and Lionais, 2004; Maclean et al., 2013) and that communities can, in turn, 'rally around and support entrepreneurial ventures to the benefit of the local areas' (Korsgaard et al., 2015: 574). The relationship between communities and established entrepreneurs has been described as one of 'reciprocity, mutuality and common purpose' (McKeever et al., 2015: 61), which relies on the fairness and trust of community members (Miller et al., 2007; O'Mahony, 2003). As Greenman (2013) argues, the commitment to shared values and reciprocities leads to the creation of community boundaries, which can result in the exclusion of outsiders (Martí et al., 2013) despite the efforts of policy makers to promote openness (Lee et al., 2011; McKeever et al., 2014). By developing standard patterns of interaction, communities act as protective shelters which enable community members to distinguish themselves from others (Martí et al., 2013).

Entrepreneurs who seek to join such communities must learn accepted norms in order to 'fit in' (De Clercq and Voronov, 2009a, 2009b). According to Lave and Wenger (1991), individuals can acquire understanding of the nature and identity of a community by participating in its shared practices.<sup>8</sup> Thus, they argue that it is through a process of situated, collective learning and practice-based apprenticeships that newcomers can become 'practice masters' and 'legitimators of practice' within communities (McLeod et al., 2011: 116) – activities which entrepreneurs in craft industries have been shown to pursue (Slavich and Castellucci, 2016). Furthermore, institutional theorists (Bruton et al., 2010) have suggested that 'community' represents an institutional logic (i.e. a set of 'material practices and symbolic constructions' which constitute the organising principles of a society, see Friedland and Alford, 1991: 248), which 'competes with or compliments the governance systems of other orders of the interinstitutional system' (Thornton et al., 2012: 71). However, as this perspective suggests that different communities act in similar ways

<sup>&</sup>lt;sup>8</sup> Lave and Wenger (1991) formulated a dedicated theory called 'communities of practice' (for detailed reviews, see Omidvar and Kislov, 2014; Pyrko et al., 2017). Mutch (2003: 396) highlights the contradictions and 'tension between this perspective and Bourdieu's' (e.g. in relation to the different uses of such concepts as practice and habitus), which is why this study does not draw on this theory in order to avoid potential misinterpretations.

(Barth et al., 2015; Marquis et al., 2011), it contrasts with the context-focused approach adopted by this study and was therefore not selected as a conceptual lens through which to explore the experiences of participants.

Importantly, becoming embedded in a community does not only facilitate value creation (Di Domenico et al., 2009; Korsgaard and Anderson, 2011) and opportunity discovery and exploitation (Khavul et al., 2013; Vershinina et al., 2011); it can also inhibit entrepreneurial efforts (Johnstone and Lionais, 2004; Parkinson et al., 2017). As Hindle (2010) notes, strict governance mechanisms (i.e. processes by which communities are directed, controlled and held accountable) and certain institutional arrangements can limit the entrepreneurial potential within communities (see, O'Mahony and Ferraro, 2007). Furthermore, Jack and Anderson (2002) emphasise that social and moral obligations can constrain entrepreneurial processes, such as when business relationships deteriorate because entrepreneurs fail to meet the expectations of community members. Accordingly, it has been shown that violations of community norms can be punished through the exclusion of nonconforming parties from necessary resources and support (Maurer and Ebers, 2006). Thus, communities have been described as 'apolitical spheres which develop unconditional acceptance of prevalent structures of power' (Martí et al., 2013: 11) that discourage open forms of protest and can hinder entrepreneurial efforts. In addition, it has been shown that the need to develop trust and durable connections with community members increases the risk of overembeddedness (due to the isolation from outsiders) and reduces the incoming flow of novel information required to facilitate innovation (Burt, 1992; Uzzi, 1997).

Interestingly, it has been argued that virtual communities are spared from some issues which affect local communities (Fowler et al., 2004). Virtual embeddedness (defined as 'the establishment of interorganizational connections through the use of electronic technology', see Morse et al., 2007: 142) connects entrepreneurs with people from all over the world (Kuhn and Galloway, 2015; Smith et al., 2017) and enables the sharing and archiving of expert knowledge by various contributors (Fowler et al., 2004) who often come from different backgrounds and are therefore able to add unique perspectives (von Krogh et al., 2003). While local communities may achieve information richness, they usually lack substantial reach –

an issue that virtual communities rarely face due to their global connections (Morse et al., 2007). In addition, the ability to provide and receive online reviews and rankings (Clemons et al., 2006), which are available to community members at any time and do not require personal connections between them (Füller et al., 2008), can help entrepreneurs engender trust in their products and signal positive reputation (Reuber and Fischer, 2011). Embeddedness in virtual communities also allows entrepreneurs to become 'instant internationals' and stay connected with customers from around the globe, which has been shown to help craft entrepreneurs who might otherwise struggle to find collectors (Fillis, 2004). Despite these recognised advantages and the increasing engagement of entrepreneurs in online communities (Fischer and Reuber, 2014; Piva et al., 2012), research has not paid adequate attention to embeddedness in virtual contexts (Cumming and Johan, 2010; Morse et al., 2007). Specifically, there is a need for empirical studies into the online context because it 'is so different that offline research findings may not apply' (Smith et al., 2017: 22). Thus, dedicated analysis of virtual communities is both timely and warranted (O'Mahony and Ferraro, 2007).

## 4.4.3. Social Value Creation

One catalyst for the emerging interest in embeddedness in communities and its effects on entrepreneurial behaviours (Hindle, 2010; Jennings et al., 2013) is the realisation that 'enterprising people [develop] small-scale and simple collaborative solutions to address the social needs that exist in their local communities' (Zahra et al., 2009: 525). While, historically, 'there has been a strong tendency to see the value being created in entrepreneurship processes as solely economic' (Korsgaard and Anderson, 2011: 135), scholars have become more aware of the social value that is created by entrepreneurs who are embedded in and affected by their communities (McKeever et al., 2015; Vestrum, 2014). Similarly, there is an increasing recognition that the concept of growth, which is an important part of entrepreneurship (Schumpeter, 1942) and typically understood to refer to economic expansion (Audretsch et al., 2006; Shane et al., 2003), should incorporate social and community growth (Haugh, 2007; Niehm et al., 2008), both of which play significant roles in the advancement of society (Zahra and Wright, 2016). Consequently, there

has been growing interest in exploring the social value and social wealth creation of entrepreneurs (Chell et al., 2010; Dacin et al., 2011; Mair and Martí, 2006).

The most notable advances in understanding of social value creation have been achieved within the field of social entrepreneurship (Bacq and Janssen, 2011; Short et al., 2009; Zahra et al., 2009). While there is no single agreed upon definition of what social entrepreneurship entails (Nicholls, 2010; Shaw and de Bruin, 2013), scholars generally understand the driving motivation of social entrepreneurs to be the desire to create social value (Dacin et al., 2010; Nicolopoulou, 2014). As Peredo and McLean (2006: 59) propose, 'the social entrepreneur aims in some way to increase "social value," i.e. to contribute to the welfare or well being in a given human community'. Within entrepreneurship research, there is a growing recognition that the lines between social and commercial activities are often blurred (Austin et al., 2006; Pache and Santos, 2013) and that businesses adopt 'hybrid' organisational forms (Battilana and Lee, 2014; McMullen and Warnick, 2016), which combine 'value systems ... and agendas that traditionally were seen as incompatible' (Markman et al., 2016: 676). Accordingly, scholars are becoming increasingly aware of the various co-existing motivations which can drive entrepreneurs (other than economic gain, see Jayawarna et al., 2013; Rindova et al., 2009). Furthermore, it is more widely appreciated that entrepreneurs may engage in social value creation, but not consider it their first and foremost goal (Dacin et al., 2010; Wry and York, 2017). As shown in the analysis of findings (see Chapter 9), this study will therefore argue that craft entrepreneurs engage in a form of social entrepreneurship.

Compassion and personal conviction are usually understood to be main causes of an entrepreneur's desire to create social value (Grimes et al., 2013; Miller et al., 2012). However, recent studies have shown that context-related factors can also encourage social entrepreneurial activities (Tapsell and Woods, 2010), particularly in the aftermath of disaster (Lewis, 2013; Williams and Shepherd, 2016). As Smith and Stevens (2010: 583) argue, 'relatively little attention has focused on how and why an embeddedness perspective may be useful in the domain of social entrepreneurship'. Consequently, there have been repeated calls for research to investigate the effects of embeddedness in 'the heterogeneous contexts in which social enterprise and social innovation can occur' (Shaw and de Bruin, 2013: 742). Prior research has explored the impact of embeddedness in social networks (Kistruck and Beamish, 2010; Ko and Liu, 2015; Shaw and Carter, 2007) and institutional contexts on social entrepreneurs (Barth et al., 2015; Pache and Santos, 2013; Tracey et al., 2011). In comparison, little is known about the effects of embeddedness in communities on social entrepreneurial practices (Maclean et al., 2013; Peredo and Chrisman, 2006). Specifically, there is limited understanding of how and why social and moral obligations within communities may compel entrepreneurs to adopt socially oriented business practices. As De Clercq and Honig (2011: 354) argue, 'social entrepreneurship [is] a phenomenon that is intrinsically intertwined with the very fabric of contemporary society ... [and] linked with the need to both comply with and challenge existing expectations.' Thus, it has been suggested that further research into social norms and their influences on social entrepreneurial activity is needed (Di Domenico et al., 2009; Mair et al., 2012).

As outlined above (see also Chapter 3), there has been sustained academic interest in exploring how entrepreneurs gain legitimacy (Fisher et al., 2016; Lounsbury and Glynn, 2001; Stringfellow et al., 2014) by achieving a balance of 'fitting in' (i.e. conformity) with dominant norms, as well as 'standing out' (i.e. distinctiveness) as acceptably deviant innovators (De Clercq and Voronov, 2009a, 2009b, 2009c). Based on a review of the extant literature, Suddaby et al. (2017) differentiate between three main strategies used to signal *conformity*. First, existing pressures and uncertainties can lead entrepreneurial firms to adapt their characteristics and practices to meet prevailing standards, regulations and norms (Glynn and Abzug, 2002) and thus partake in isomorphism (defined as 'a constraining process in which organizations in a population resemble one another when they face the same environmental conditions', see Tan et al., 2013: 84). Second, firms may choose to engage in 'decoupling' by superficially conforming to normatively mandated business practices (e.g. by aligning their hiring and organisational policies, see Choi and Gray, 2008), while actually conducting activities which are closer aligned with the core functions and goals of their businesses (a closely related concept is 'selective coupling', see Pache and Santos, 2013). Third, entrepreneurs may gain acceptance of their innovations from audiences and stakeholders by integrating novel (or superior) ideas with existing designs and approaches (Slavich and Castellucci, 2016).

While displaying conformity can provide entrepreneurs with certain benefits, such as resource acquisition (Djupdal and Westhead, 2015; Fisher et al., 2016) and approval from customers and critics (Beverland, 2005; Voronov et al., 2013), it has been shown that there are also advantages to acquiring and demonstrating a degree of distinctiveness (Drakopoulou Dodd, 2014; Überbacher et al., 2015). For example, by creating products or services that are valuable, rare, inimitable and nonsubstitutable (Alvarez and Busenitz, 2001; Barney, 1991) and by staking out a unique position in the marketplace (Porter, 1996), entrepreneurs can gain valuable competitive advantage (Brush et al., 2001; Kellermanns et al., 2016). Scholars have developed three concepts that focus on the achievement of distinctiveness. First, 'legitimate distinctiveness' ensues from a firm associating itself with a certain category of organisations (Navis and Glynn, 2011), while simultaneously highlighting its differences from extant rivals (van Werven et al., 2015). Second, 'contextual distinctiveness' is attained by distinguishing a firm from organisations in the same basic category through membership in a specific subcategory (Gehman and Grimes, In Press). Third, 'optimal distinctiveness' (Alvarez et al., 2005; Zhao et al., 2017) is achieved by positioning a firm 'as different[ly] as legitimately possible' (Deephouse, 1999: 147) from its competitors.

As van Werven et al. (2015) note, in practice, it can be difficult for entrepreneurs to achieve a balance between conformity and distinctiveness. Furthermore, while there is an abundance of conceptual research on the subject (Deephouse, 1999; Lounsbury and Glynn, 2001; Navis and Glynn, 2011; Zhao et al., 2017), empirical studies are comparatively rare (Stringfellow et al., 2014; Voronov et al., 2013). Thus, there is a need for research to empirically explore the ways in which entrepreneurs decide how to fit in or stand out in different contexts (De Clercq and Voronov, 2009a, 2011). Given that the strategies and approaches developed through theoretical analyses are not context-specific, it might be expected that they will not apply to all communities.

As Downing (2005) points out, 'entrepreneurship, like the rest of social life, is a collaborative social achievement'. Accordingly, it has been suggested that collaboration between organisations has the potential to increase their positive impacts on society (Austin et al., 2006; Di Domenico et al., 2009). As Pearce and

Doh (2005: 32) note, 'each partner [venture] benefits when the other brings resources, capabilities or other assets that it cannot easily attain on its own.' Consequently, there have been repeated calls for research to investigate how collaboration among entrepreneurs facilitates the creation of social value (Daskalaki et al., 2015; Zahra et al., 2009). Within the social entrepreneurship literature, community-based enterprise has been identified as a distinct form of collaboration which seeks to create social wealth (Short et al., 2009; Vestrum, 2014). Communitybased enterprises are conceptualised as communities of people who act corporately as both entrepreneurs and enterprises in order to provide sustainable benefits to individuals and groups (Peredo and Chrisman, 2006). They are typically registered as not-for-profit businesses and emerge in rural or disadvantaged areas to revitalise local communities (Haugh, 2007; Vestrum et al., In Press). As a result of their nonprofit status, the economic value that community enterprises create rarely covers the costs of the resources they use, requiring them to rely on reciprocity and noneconomic exchanges to continue their operations over time (Shaw and Carter, 2007). As such, the sharing of resources within communities has been shown to play an important role in supporting local development and social entrepreneurial efforts (Dana and Light, 2011; Morris et al., 2011).

Indeed, collaboration and resource sharing has been found to facilitate social value creation, not only for community enterprises, but also for individual entrepreneurs (Barth et al., 2015; Ko and Liu, 2015). According to Smith and Stevens (2010), a common exchange between entrepreneurs 'consists of the provision of various forms of capital for the creation of social value'. However, the particular resources that entrepreneurs exchange to facilitate these processes are not well understood (Barth et al., 2015). Di Domenico et al. (2009: 888) propose that social entrepreneurs may draw on the 'local knowledge, social capital and social legitimacy' of their partners. Furthermore, collaboration among social entrepreneurs has been found to grant access to financial support (Maclean et al., 2013) and tacit knowledge (Chalmers and Balan-Vnuk, 2013). As entrepreneurship research has mainly focused on exploring the commercial benefits that derive from business collaboration (cf. Larson, 1991; Maurer and Ebers, 2006; Sorenson et al., 2008), further research is required to gain a better understanding of the collaboration and resource sharing practices that entrepreneurs employ to create social value.

# 4.5. Conclusion

This chapter examined entrepreneurship research on resource and capital management, households and families, and embeddedness in communities. As shown in Figure 4.1, this chapter brought together a diverse range of literature upon which this study draws in order to link its findings to contemporary scholarly debates. This review of the literature revealed a number of enduring gaps in understanding, which are summarised below.

On the subject of resource and capital management, this chapter demonstrated that, while it has long been established that access to resources is essential for entrepreneurial success, comparatively little is known about the ways in which entrepreneurs combine and convert their available forms of capital. The importance of economic capital, in the form of financial assets, is taken-for-granted and a multitude of studies explore how entrepreneurs can gain access to financing. In comparison, few studies investigate the ways in which entrepreneurs employ various types of tangible and intangible economic assets to augment their other resources. Similarly, there has been sustained interest in the effects of human and social capital on entrepreneurs, but further empirical research into their convertibility is required. Prior research also highlights the benefits of reputation and status for entrepreneurial activities, but understanding of what entrepreneurs actually do to acquire these types of symbolic capital is limited. Furthermore, the existing body of capital-related research is disjointed, using various, sometimes conflicting, definitions of key terms, and the majority of studies examine only one or two forms of capital. This study therefore seeks to extend knowledge by exploring how entrepreneurs convert their economic, cultural, social and symbolic capital.

Regarding entrepreneurship research on households and families, this chapter revealed that business and private spheres are often blurred for entrepreneurs. While entrepreneurship research has included households in empirical analyses, they are mainly utilised as sources of data for measuring the activities of entrepreneurs, rather than as subjects of research in their own right. It has been argued that access to the resources of household and family members can help entrepreneurs overcome resource scarcity and thus facilitate venture start up and development. However,



# Figure 4.1. Overview of Relevant Entrepreneurship Theories

prior studies have primarily focused on the effects of household financial capital on entrepreneurial activities, rather than incorporating a wider range of resources into their investigations. Accordingly, the concept of household capital was developed to encourage explorations of the financial, human and social capital of cohabitants and their effects on entrepreneurs. It has been acknowledged that household and family members can also serve as valuable sources of emotional support, but this has rarely been explored by entrepreneurship research. Instead, there is a tendency to employ a conflict perspective and consider business and family as 'rivals'. Thus, this study seeks to adopt a more broad-minded approach and account for both positive and negative effects by investigating *how household members impact the practices of entrepreneurs*.

On the subject of embeddedness in communities, this chapter revealed that entrepreneurship research has focused on exploring the structural and relational embeddedness of entrepreneurs (i.e. network-related characteristics). In comparison, limited attention has been afforded to communities, especially virtual and common interest communities, and their impacts on entrepreneurs. Prior studies have shown that embeddedness in communities can provide entrepreneurs with access to resources, such as financial support and knowledge, but there is lack of empirical research on the subject. Furthermore, there is a growing recognition of the social value that can be created by entrepreneurs as a result of community interaction. However, there is limited understanding of the ways in which social and moral obligations within communities may compel entrepreneurs to adopt socially oriented business practices. Prior research suggests that entrepreneurs modify their behaviours to fit in with prevailing norms and meet the expectations of community members, but empirical research of these practices is rare. Accordingly, this study seeks to address these gaps and advance knowledge by exploring how and why embeddedness in communities affects the practices of entrepreneurs.

Overall, this chapter provided a detailed analysis of research progress on the capital management of entrepreneurs, as well as the effects of embeddedness in households and communities on their practices. The following chapter presents an in-depth review of the philosophical and methodological underpinnings of this study and discusses the adopted data collection and analysis techniques.

### **CHAPTER 5: RESEARCH METHODS**

#### 5.1. Introduction

Empirical research should adopt a philosophical position and a methodological approach that facilitates its objectives (Gartner and Birley, 2002; Jennings et al., 2005). The overarching goal of this study is to *develop a theory of practice for craft entrepreneurs* which explains how embeddedness in multiple contexts affects entrepreneurs' resource management practices. As this study deals with 'soft issues, which are not amenable to quantification' (Jack and Anderson, 2002: 473), it will not attempt to measure the experiences of participants or address research questions of 'how many'. Rather, it seeks to gain an understanding of 'how and why' people live their lives the way they do (McKeever et al., 2015; Pratt, 2009) by investigating the subjective views of participants.

Consequently, this study adopts a phenomenological approach (Conklin, 2007; Smith et al., 2009). As Berglund (2007) explains, phenomenological research explores the meanings that participants attribute to their lived experiences in order to gain in-depth knowledge of their personal and social worlds. Inherently qualitative in nature, phenomenological studies generate theory inductively, from the 'bottom-up', through analyses of individual experiences (Cope, 2005). Given that phenomenological approaches are comparatively rare in entrepreneurship research (Grant and Perren, 2002; Hlady-Rispal and Jouison-Laffitte, 2014), the purpose of this chapter is twofold: (1) to introduce and discuss phenomenological research and its underlying assumptions and (2) to present the specific research approach adopted in this study. This chapter is structured as follows. It first explores philosophical assumptions of phenomenological research, before the research aim and objectives are recounted. Next, the research design is explained, followed by a discussion of the data collection and analysis. Finally, methodological reflections are provided and conclusions are drawn. This chapter demonstrates the rigorous process used to gather and analyse the data presented in the findings chapters (Chapters 7-9).

### **5.2.** Philosophical Assumptions

Whether implicitly or explicitly, scholars base their research on 'philosophical assumptions regarding ontology, epistemology, and human nature, which have methodological consequences' (Jennings et al., 2005: 145). It is important to acknowledge and discuss these underlying assumptions, as they generate different philosophical research paradigms which represent fundamentally different approaches to interpreting and analysing social phenomena. Accordingly, this section first examines major research paradigms and explains the differences between their perspectives, before reviewing phenomenological approaches and considering associated paradigmatic concerns.

# 5.2.1. Philosophical Research Paradigms

Kuhn (1962) introduced the term 'paradigm' to scientific inquiry as a means of describing the philosophical assumptions and practices that dominate a research field. Within the social sciences in general and entrepreneurship research in particular, Burrell and Morgan's (1979) paradigmatic taxonomy (see Figure 5.1) is commonly applied to position research along two continuums: subjective-objective and radical change-regulation (Howorth et al., 2005; Pittaway, 2005).

Figure 5.1. Burrell and Morgan's (1979) Paradigmatic Framework



Adapted from Grant and Perren (2002: 187)

As shown in Table 5.1, Burrell and Morgan (1979) identify several philosophical assumptions that underlie the subjective-objective dimension. First, on an ontological level, they distinguish between scholars who consider reality a socially constructed product of the mind (nominalism) and those who view it as an objective entity (realism). Second, with respect to epistemology, Burrell and Morgan (1979) differentiate between researchers who recognise that all observations are value-laden (anti-positivism) and those who aim to accumulate knowledge through neutral observations (positivism). Third, in relation to human nature, actions can be interpreted as directed by either the free will of an individual (voluntarism) or the environment (determinism). Fourth, concerning methodologies, scholars either deploy methods that reveal the internal logics guiding human behaviours (idiographic) or derive their methods from the natural sciences (nomothetic) (Johnson and Duberley, 2000). Burrell and Morgan (1979) propose that radical humanists and interpretivists adopt a subjective perspective, while radical structuralists and functionalists take an objective stance.

**Table 5.1.** Subjective-Objective Dimension of the Paradigmatic Framework

	Subjective	Objective
Ontology:	Nominalism	Realism
Epistemology:	Anti-positivism	Positivism
Human Nature:	Voluntarism	Determinism
Methodology:	Ideographic	Nomothetic

Importantly, scholars also hold differing views regarding the nature of society, represented by the radical change and regulation dimensions of this paradigmatic framework (see Figure 5.1). Burrell and Morgan (1979) suggest that researchers interested in radical change explore 'structural conflict, modes of domination, contradictions, emancipation, deprivation and potentiality' (Grant and Perren, 2002: 188). In comparison, those who investigate how society is regulated are concerned with the 'status quo, social order, consensus, social cohesion, solidarity, individual or system needs satisfaction, and actuality' (Jennings et al., 2005: 145). Importantly, the latter perspective also accounts for conflict, but it

considers it to be a temporary aberration. It has been shown that most research in the social sciences adopts a regulation-focused approach and is positioned either in the interpretivist or functionalist paradigm (Johnson and Duberley, 2000).

As might be expected, Burrell and Morgan's (1979) framework has been criticised for a number of reasons (for an in-depth discussion, see Deetz, 1996). Within the entrepreneurship literature, Pittaway (2005) examines prevailing debates, including questions of whether paradigms should unify fields of study and whether they over-simplify complex philosophical positions. Another frequently mentioned criticism of Burrell and Morgan's (1979) work is the theme of incommensurability: they argue that paradigms are mutually exclusive and their boundaries immutable. Scholars who disagree with the proposition that paradigms develop independently have devised models that allow for paradigm communication (Aram and Salipante, 2003; Watkins-Mathys and Lowe, 2005) and multiple-paradigm research (Howorth et al., 2005). Despite these criticisms and alternative approaches, paradigms are useful tools for exploring the dominant philosophical orientation within a field of research (Grant and Perren, 2002). As this study is firmly set within the interpretivist paradigm, it elects to draw upon Burrell and Morgan's (1979) perspective.

Kuhn (1962) argues that research fields go through prolonged periods of 'normal science', during which an established paradigm prevails, interrupted by episodes of 'revolutionary science', during which an alternative paradigm challenges and eventually supersedes previous conventions. Unlike the natural sciences, which have largely adopted the functionalist paradigm, Kuhn (1962) considered the social (or human) sciences a pre-paradigmatic field. As Kenworthy and McMullan (2013) note, there is some debate whether this belief still holds true, but many scholars agree that some fields of social science, including entrepreneurship research, have yet to establish distinctive boundaries and legitimate methods and are therefore 'pre-paradigmatic' (Busenitz et al., 2003; Maclean et al., 2015; Nicholls, 2010). However, opinions differ as to whether it is beneficial to agree on a single paradigm in an 'applied discipline', such as entrepreneurship research, which focuses on exploring the issues of practitioners (Pittaway, 2005).

Indeed, there have been repeated calls for entrepreneurship research to embrace greater paradigmatic and methodological diversity (Karataş-Özkan et al.,

123

2014; Shepherd, 2015; Welter et al., 2017), 'to enable debate, friction, creativity and ultimately new theories and understandings' (Grant and Perren, 2002: 202). Given that the majority of entrepreneurship studies are functionalist in orientation (Grant and Perren, 2002; Hlady-Rispal and Jouison-Laffitte, 2014) and 'overly reliant on quantitative methods and positivist thinking' (Suddaby et al., 2015: 2), this study contributes to the paradigmatic diversity of the field by adopting an interpretivist stance.

#### 5.2.2. Phenomenological Approaches

This study adopts a phenomenological approach, which is firmly set within the interpretivist paradigm (Burrell and Morgan, 1979), because (1) it rejects the ontological separation of reality and consciousness and instead asserts that reality is a product of the mind (nominalism); (2) it appreciates that all observation is value-laden (anti-positivism); (3) it believes that human actions are not pre-determined, but a product of human will (voluntarism); and (4) it seeks to uncover the internal logics that drive individual practices by accessing cultures (ideographic methodology) (Aram and Salipante, 2003).

Phenomenological research has a long tradition, most notably perpetuated by German philosophers Edmund Husserl (1859-1938) and his student Martin Heidegger (1889-1976) (Sanders, 1982). Phenomenology literally means 'the study or description of phenomena as experienced by people' (Lewis, 2015: 667). Accordingly, phenomenological studies aim to portray the essences of experiences (phenomena) that cannot be revealed through ordinary observation and to uncover their underlying foundations and meanings (Cope, 2005). In a phenomenological studies sense, things do not exist in themselves, but in the meanings that people attach to them (Berglund, 2007). This does not, of course, mean that phenomenologists reject the notion of a material world, but rather that the world is understood and interpreted through the meanings that individuals develop through their experiences (Conklin, 2012). In other words, while objects, such a craft items, may exist as concrete entities, it is through interactions with and interpretations of these objects that reality is created. Phenomenological research is therefore particularly useful for exploring

'what meaningful experiences and strategies are associated with different situations' (Berglund, 2007: 76), such as exchanging resources or developing a craft business.

As shown in Table 5.2, a number of entrepreneurship studies have adopted phenomenological and interpretivist approaches. Apart from a few exceptions (e.g. Kisfalvi, 2002), most of these studies were published following Cope's (2005) seminal article, which first advocated the use of phenomenological perspectives in entrepreneurship research and established methods for its implementation. This table highlights the range of subjects that can be investigated using phenomenological approaches, from networking and embeddedness to entrepreneurial leadership and learning. Accordingly, the findings of these studies are broad and provide insights into such varied topics as the ways in which knowledge is shared between business partners and how and why entrepreneurs adopt certain strategic behaviours.

Interestingly, theoretical articles, which largely examine knowledge creation, still incorporate empirical elements in their explanations of conceptual ideas (a notable exception is Sanders, 1982). As might be expected, all empirical studies that adopt a phenomenological perspective are qualitative in nature and perform in-depth analyses of the subjective experiences of participants. Notably, some entrepreneurship studies are not based in the interpretivist paradigm, but still adopt phenomenological interview techniques (e.g. McKeever et al., 2015; Spivack et al., 2014), which highlights that 'qualitative methodologies and analytical procedures are not ... mutually exclusive' (Cope, 2005: 172). Finally, phenomenological studies have been conducted across the world, although the UK and USA appear to be particularly popular contexts.

## 5.2.3. Paradigmatic Concerns

There are several underlying paradigmatic issues associated with phenomenological research which require further exploration. Cope (2005) suggests that the presuppositionless philosophy and the concept of the lived-world are in need of explanation, because they are antithetical to prevailing functionalist assumptions. Thus, this section examines these aspects before addressing Bourdieu's criticisms of phenomenology.

Subject	Article	Key Finding	Main Method Country	I Country
Embeddedness	McKeever et al. (2015)*	Social bonds enable entrepreneurship to create a positive identity of place.	Qualitative	Ireland
	Shaw et al. (2017)	Embedding a firm in its industrial context is an emergent, evolving process.	Qualitative	UK
Internationalisation	Lamb et al. (2011)	Phenomenography enables investigations of collective understandings.	Qualitative	Australia
Knowledge Creation	Conklin (2007) <sup>†</sup>	Doubt and reflective inquiry facilitate deeper understanding of phenomena.	. Theoretical	USA
	Leitch et al. (2010)	Quality must be internalised within the underlying research philosophy.	Theoretical	UK
	Sandberg (2005) <sup>†</sup>	Epochés help demonstrate the validity and reliability of interpretive studies.	. Theoretical	Sweden
	Sandberg & Dall'Alba (2009) <sup>†</sup>	Adopting a life-world perspective helps show how practice is constituted.	Theoretical	Multiple
	Sanders (1982) <sup>†</sup>	Phenomenology facilitates exposing frequently overlooked phenomena.	Theoretical	N/A
Leadership	Leitch et al. (2013)	Leaders' human capital is enhanced through development of social capital.	Qualitative	UK
	Lewis (2015)	Entrepreneurial leader identity evolves in tandem with firm development.	Qualitative	ZN
Learning	Cope (2005)	Phenomenological commitment to research translates into deep knowledge.	. Theoretical	Multiple
	Cope (2011)	Recovery from failure is a function of distinctive learning processes.	Qualitative	Multiple
Marketing	Lam & Harker (2015)	Entrepreneurs interact with customers to shape their marketing activities.	Qualitative	China
Networking	Al-Dajani et al. (2015)	Collaborative networks enable women to challenge constraining situations.	Qualitative	Jordan
	McAdam et al. (2014)	Horizontal SME networks enable open innovation and knowledge sharing.	Qualitative	UK
	Shaw (2006)	Small firm networking relationships are multiplex and informal in nature.	Qualitative	UK
Professions	Conklin (2012) <sup>†</sup>	Experiencing a job can help individuals discover their 'calling' for it.	Qualitative	USA
Repeated Start-up	Fiet et al. (2013)	For a search to be systematic, it must relate to specific prior knowledge.	Qualitative	USA
	Spivack et al. (2014)*	Start-up experiences can create behavioural addiction to entrepreneurship.	Qualitative	NSA
Social Value Creation	1 Korsgaard & Anderson (2011)	Social Value Creation Korsgaard & Anderson (2011) Entrepreneurship is as much a social as an economic phenomenon.	Qualitative	Denmark
	Parkinson & Howorth (2008)	Social entrepreneurs draw legitimacy from a local or social morality.	Qualitative	UK
Strategy	Kisfalvi (2002)	Entrepreneurs pursue strategic directions that reflect their set of life issues.	Qualitative	NSA
	Zou et al. (2016)	Entrepreneurs choose conflict management strategies according to context.	Qualitative	China
<ul> <li>* Article employs a ]</li> <li>* Article does not student of the selected articles were</li> </ul>	Article employs a phenomenological interview techniq Article does not study entrepreneurship, but provides a lected articles were published in journals ranked at ABS	<ul> <li>Article employs a phenomenological interview technique, rather than a phenomenological philosophy.</li> <li>Article does not study entrepreneurship, but provides a seminal argument regarding phenomenology in business and management research. Selected articles were mublished in journals ranked at ABS Grade 3 or higher (Wood &amp; Peel 2015)</li> </ul>	ent research.	

Table 5.2. Phenomenological and Interpretivist Entrepreneurship Research

Husserl's work was driven by a desire for a *presuppositionless philosophy*, where all judgements and assumptions about the nature of activities are suspended until they can be founded on the experiences of individuals (Conklin, 2012). He argues that suspending all prior knowledge about phenomena is necessary for scholars to perceive things as they are and to provide authentic descriptions free from prejudice (Cope, 2005). Husserl claims that accounts of experiences are all too often biased by what the researcher assumes they would be like, rather than being understood 'from within' (Moran, 2000). To achieve a state free from presuppositions, Husserl proposes that scholars should adopt a 'natural attitude' and engage in a process of bracketing (Conklin, 2007). Sanders (1982: 355) defines bracketing as 'the temporary suspension of all existing personal biases, beliefs ... or assumptions in order to get straight to the pure and unencumbered vision of what a thing "essentially is". As Cope (2005) points out, detaching oneself from personal beliefs about the subject under investigation facilitates going beyond the 'natural standpoint' of the attached observer, which is why Husserl's approach is often described as 'transcendental'. His technique of phenomenological reduction and detachment is also called epoché (Throop and Murphy, 2002).

While Heidegger (1962) endorsed much of the ideas of his teacher, he opposed the concept of bracketing as a means of gaining true knowledge. Instead, Heidegger believed that human beings always already exist in the world and that 'the world has always already had natural meaning for us' (Berglund, 2007: 90). Meaning is located in situated, ongoing activities, which are affected by prior learning and historical understandings of the world (Seymour, 2006). Thus, Heidegger (1962) argues that, to gain a deep understanding of phenomena, scholars must acquire knowledge of the wider contexts in which individuals are embedded. The ability to acquire privileged access and the capacity to understand the stories of others therefore have significant influence on the depth and quality of scholars' analyses of human experiences (Berglund, 2007). In a sense, the researcher becomes an interpreter who needs to be immersed in a context to fully understand it. In Finlay's (2009: 11) words, 'it is precisely the realisation of the intersubjective interconnectedness between researcher and researched that characterizes phenomenology'. In other words, scholars' expert knowledge can guide their enquiries and make them more

revealing (Lopez and Willis, 2004). Nevertheless, it is important that researchers remain open to the meanings participants attribute to their experiences and strive to see the world through their eyes.

While this study initially adopted Husserl's approach, including the practice of bracketing, the researcher quickly realised that his industry knowledge made it almost impossible to entirely remove himself from the topic under investigation. Participants frequently asked the researcher about his experiences as a craft maker and drew on his understanding of craft processes to explain their perspectives. In addition, as Cope (2005: 181) points out, 'one's perceptions of the phenomenon in question are [always] influenced, both explicitly and implicitly, by exposure to extant theory prior to engagement with the participants'. Not only is this inevitable, but without such prior knowledge, it would be difficult to make meaningful contributions to understanding. Thus, Heidegger's approach was embraced after the first round of interviews was completed.

Another essential aspect of phenomenology is 'Die Lebenswelt', which is translated as '*the lived-world*' (Moran, 2000). First introduced by Husserl (1954), the lived-world refers to 'the world of the ordinary, immediate experiences and is the background for all human endeavours' (Cope, 2005: 167). Heidegger (1962) places particular emphasis on this concept and the impact of 'Dasein', or 'being-in-the-world', on individual practices. He argues that human beings need to be studied in their lived-world to fully explicate their behaviours, as their subjective experiences are inextricably intertwined with cultural, social and political contexts.

Seymour (2006) highlights several aspects of 'being-in-the-world' that can help inform phenomenological research. First, Heidegger proposes that humans learn from the past, which evokes instinctive feelings that attune them to their present surroundings (e.g. fear can reveal potential threats). Second, he suggests that being entangled in the present situation leads humans to fall into routines and accept existing conventions, which affects their decision-making. Third, he contends that humans are not static, but always in the process of becoming and of anticipating future possibilities that can be seized. Thus, Heidegger's (1962) approach calls attention not only to the impact of embeddedness in context, but also to temporal dimensions (individual's past experiences, present involvements and future opportunities). Heidegger also emphasises that interpretations are revelatory and allow scholars to go beyond their initial, unreflective understanding of phenomena (Seymour, 2006). As Lopez (2004) points out, meanings are not always apparent to research participants, but can be deduced by interpreting their stories. Heidegger's (1962) approach is therefore often described as 'hermeneutic' or 'interpretive' phenomenology. Rather than focusing on what participants consciously know, phenomenological studies are supposed to explore their lived experiences in context. This appreciation of context and its effects on individuals is the main reason why this study adopts an interpretative phenomenological approach.

Finally, *Bourdieu's criticisms of phenomenology* require discussion. Essentially, Bourdieu (1977) argues that phenomenologists focus too much on the immediate experiences of individuals and their interpretations of the social world. As several studies provide in-depth analyses of Bourdieu's stance on phenomenology (cf. Atkinson, 2010; Throop and Murphy, 2002), only his most important and relevant critiques are outlined below.

First, Bourdieu suggests that phenomenological research 'remains decidedly deficient in its ability to penetrate the obscuring veils of doxa' (Throop and Murphy, 2002). However, it could be argued that, despite being labelled 'descriptive' phenomenology (Finlay, 2009), Husserl's approach in particular facilitates exposing taken-for-granted assumptions (doxa) through the suspension of all prior knowledge of a phenomenon. Second, Bourdieu contends that phenomenologists place too much emphasis on the conscious decisions and actions of individuals (Throop and Murphy, 2002). As such, it is important to consider the potential subconscious motivations that affect the decision-making of participants. Third, Bourdieu (1977) proposes that phenomenological approaches focus too much on individual agency, rather than on structural dimensions. Indeed, while this study explores the role of context in entrepreneurial practices, it does not seek to analyse structural arrangements within social fields. Thus, agency-related considerations are given priority in this study. Furthermore, Bourdieu criticises phenomenologists for failing to account for the impact of durable dispositions (habitus) on subjectivity. However, this seems to overlook Husserl's exposition that lived experience 'results in forgotten "traces" or "substrates" that may later direct our attention to similar objects in a horizon of "typical familiarity and precognizance" (Throop and Murphy, 2002: 193). In other words, Husserl did express an appreciation for the influence of personal dispositions on individual practices. Similarly, Heidegger (1962) emphasises the impact of prior knowledge and learning on the creation of meaning.

It has been suggested that phenomenological approaches could, in fact, help remedy some of the criticisms of Bourdieu's own work (Atkinson, 2010). For example, Throop and Murphy (2002) argue that Bourdieu's focus on habitus and subconscious decision-making leads to a flawed model of human mentation. In comparison, a phenomenological perspective promotes exploring the consciously felt ideals, objectives and emotions of diverse agents. Bourdieu has also been criticised for not adequately accounting for the heterogeneity of habitus within fields. A phenomenological approach, which seeks to reveal the subtleties of human lives, can help explain such diversity (Atkinson, 2010). Finally, Bourdieu's underlying assumption that human behaviour is generally driven by a deep-seated, competitive inclination and a desire for power has similarly been critiqued (Özbilgin and Tatli, 2005). Phenomenological research, with its emphasis on examining individual goals and experiences, is able to provide a more nuanced and complex picture of motivations and behaviours. Thus, while there are differences between phenomenological and Bourdieusian perspectives, they are able to complement one another. A phenomenological study may never be able to achieve Bourdieu's (1977, 1984) goal of bridging the gap between subjective and objective positions, but its findings can nevertheless make significant contributions to knowledge.

#### **5.3. Research Aim and Objectives**

Having discussed the philosophical assumptions underlying phenomenological research, this section recounts the research aim and objectives which influenced the design of this study. The overarching goal of this research is to *develop a theory of practice for craft entrepreneurs* which explains how embeddedness in multiple contexts affects entrepreneurs' resource management practices. Thus, this study is exploratory in nature and seeks to generate new knowledge about phenomena (Cope, 2005; Leitch et al., 2010), rather than to (dis-)prove hypotheses derived from a review of the extant literature. Accordingly, it examines the lived experiences of

entrepreneurs to gain in-depth understanding of their contextualised practices (Berglund, 2015). In order to extend knowledge of the behaviours of 'everyday entrepreneurs' (Welter et al., 2017) and to support 'ordinary people who want to become entrepreneurs' (Sarasvathy et al., 2015: 227), this study chose to investigate craft entrepreneurs, who are under-researched, despite having been shown to find creative ways of mobilising resources and exploiting opportunities (Bhagavatula et al., 2010; Kuhn and Galloway, 2015). The analysis of participants' experiences determined this study's focus on how embeddedness in household and community contexts affects the resource management practices of craft entrepreneurs. A review of the relevant literature revealed several gaps in understanding, which this study was able to address with its empirical evidence. Three research questions were therefore developed to focus the analysis of data and the presentation of findings:

- 1) How do entrepreneurs convert their economic, cultural, social and symbolic capital?
- 2) How do household members impact the practices of entrepreneurs?
- 3) How and why does embeddedness in communities affect the practices of entrepreneurs?

In answering these research questions, this study is able to develop a theory of practice for craft entrepreneurs which explains how embeddedness in multiple contexts affects entrepreneurs' resource management practices. Chapter 6 presents the conceptual model of this theory and explicates the connection between the subsequent findings chapters (Chapter 7-9), each of which is dedicated to the discussion of one of the research questions. Chapter 7 focuses on the micro-level, discussing the various capital conversion processes that craft entrepreneurs conduct, many of which have been overlooked in the entrepreneurship literature. Chapter 8 explores the meso-level, revealing ways in which household members support craft entrepreneurs, particularly through the provision of psychological capital, and also inhibit their practices. Chapter 9 is concerned with the macro-level. It demonstrates how craft entrepreneurs adapt their behaviours to fit in with community norms and how such norms can lead entrepreneurs to engage in socially-oriented business practices. Importantly, this study's aim and objectives are based on the phenomenological commitment to understanding what it feels like to run and develop a craft enterprise.

It should be noted that entrepreneurship scholars have adopted various alternative ways of defining micro-, meso- and macro-levels (Greenman, 2013; Kim et al., 2016; Kloosterman, 2010). For example, while Shepherd (2009) agrees that the micro-level refers to individuals and the macro-level to groups of people, he conceives of the meso-level as the emotions that connect individuals and groups. Another variation is presented by Özbilgin and Tatli (2005: 860) who draw on Bourdieu's work and regard 'the concepts of capital and dispositions at the individual [micro] level, habitus at the meso level, and the field at the macro level of analysis'. Acknowledging these other conceptualisations, this study chooses to follow: (1) Kloosterman (2010: 27), who perceives 'the micro-level [as] the individual entrepreneur and his or her resources', (2) Gras and Nason (2015: 559), who 'introduce the meso-level institution of the family household' and (3) Kwon et al. (2013: 980), who conceive of the macro-level as 'units such as communities or regions'. This paradigm was selected as it best suited the research context studied.

# 5.4. Research Design

This study conducts an interpretative phenomenological analysis (Smith et al., 2009). First introduced to entrepreneurship research by Cope (2011), this approach is gaining traction within the field (cf. Lewis, 2015; Shaw et al., 2017; Zou et al., 2016), as it allows researchers to explore various dimensions of phenomena and processes as they unfold. This section reviews methodological concerns associated with phenomenological research, before discussing the interpretative phenomenological analysis technique and explaining the participant selection process.

# 5.4.1. Methodological Concerns

This section explores a number of methodological issues surrounding the use of interpretive phenomenological approaches. According to Cope (2005), relevant concerns are related to (1) adopting a qualitative approach, (2) interacting with participants and (3) working within the context of discovery.

Given that phenomenological studies seek to explore the lived experiences of individuals, they must adopt methods that enable the collection of rich qualitative

data (Berglund, 2015). As Thompson et al. (1989) argue, individual experiences of the world can differ significantly from 'objective descriptions', as all human experience is linked to distinct contextual settings. Thus, it is impossible to provide generalisable accounts which are 'true' across diverse contexts and times. Instead of stripping participants from their context (Lincoln and Guba, 1985), phenomenological research embraces methods which allow for the exploration of participants' subjective, situated views. According to Cope (2005), this is best achieved by employing various techniques to decipher, translate and interpret the meaning, rather than the frequency, of phenomena which occur in the social world. In other words, qualitative methods are required to gain sufficiently deep insights into phenomena and to account for the individual nature of human experience (Gartner and Birley, 2002).

While qualitative methods are no longer considered a 'special case' in entrepreneurship research (Suddaby et al., 2015), they are still comparatively rare in highly ranked journals (on average, only 10% of articles published in *Entrepreneurship Theory and Practice* and the *Journal of Business Venturing* employ qualitative methods, according to Hlady-Rispal and Jouison-Laffitte, 2014). Nevertheless, there is general consensus that qualitative methods are preferred when exploring social phenomena that require detailed analysis (Bansal and Corley, 2012; Pratt, 2009). They are particularly well suited for investigating complex issues, processes and practices (McKeever et al., 2015) and addressing research questions of 'how' and 'why' (Jack, 2010; Pratt, 2009). Yet, exploring the lived experiences of participants through qualitative methods is demanding and brings with it a number of issues.

First, it can be challenging to interpret the accounts of individuals, as they are always partial and no single method is able to 'grasp the subtle variations in ongoing human experience' (Denzin and Lincoln, 1994: 12). Second, real-life phenomena are rarely neat and organised, but rather cluttered and confusing, which complicates the analysis and presentation of findings (Gartner and Birley, 2002). Third, even longitudinal studies are only able to collect data in certain intervals and each communication with participants represents only a snapshot in time (Lincoln and Guba, 1985). Consequently, interpretations of the stories of participants may vary depending on the circumstances surrounding a research visit and participant perspectives may change over time (Cope, 2005). Given that certain events and experiences will stand out in the minds of participants, phenomenological research is always contextually and temporarily situated, as it represents their 'being-in-theworld' (Thompson et al., 1989).

Furthermore, scholars are situated in the lived-world in the same way as their participants. Accordingly, researchers can never be impartial, neutral or detached observers; their perspectives are always subjective and limited by their own knowledge (Heidegger, 1962). Thus, to gain a deep understanding of the meanings that participants associate with different phenomena, scholars must build relationships with them (Cope, 2005). A close connection to both the subject of study and participants can greatly improve the quality of phenomenological research, as knowing is created through the interactive relationship between scholar and participant (Larkin et al., 2006). As this study collected data over the course of 18 months, the researcher was able to form durable connections with participants. To facilitate the rapport-building process, the researcher disclosed information about himself during informal conversations, which allowed participants to learn more about his background and motivations. In addition, the researcher purchased craft items of participants (e.g. his wedding rings) to gain a deeper appreciation for their work. These efforts led participants to become more invested in the research and thus open up more, allowing deeper insights into their lives. Still, it is important to note that phenomenological descriptions always represent the researcher's interpretations of how participants interpret their own experiences (Smith, 2004). Readers need to be aware of this, as they will, in turn, interpret the findings of such research.

Phenomenological research is firmly set in 'the "context of discovery" rather than the "context of justification" (Cope, 2005: 171). In other words, phenomenological approaches are inductive; they describe phenomena based on the analysis of participants' perspectives, rather than engaging in hypothesis testing (Parkinson and Howorth, 2008). As such, their techniques contrast with the hypothetico-deductive framework which prevails in entrepreneurship research (Leitch et al., 2010). Indeed, phenomenological research facilitates the discovery of new and unforeseen insights into social phenomena (Gartner, 2008). It is therefore important that scholars adopt a holistic approach and 'relate descriptions of specific experiences to each other and to the overall context of the lived-world' (Thompson et
al., 1989: 137). In so doing, phenomenological approaches enable the development of interpretive theories from the 'bottom-up' (i.e. grounded in the lived-world, see Cope, 2005). Such theories are not expected to be generalisable to a variety of contexts, but should instead create 'local knowledge' (Steyaert, 1997) and provide a trustworthy account of phenomena (Lincoln and Guba, 1985). It is through the iterative process of applying such theories to new contexts that their descriptive power is improved, which entails that theory-building should be understood as an evolutionary process in which multiple scholars are engaged (Kempster and Cope, 2010). In the case of this study, the developed theories are applicable to craft makers, but it is expected that entrepreneurs in related fields, such as other cultural industries, may experience similar issues and employ comparable strategies. By providing 'thick descriptions' (Geertz, 1973), which scholars can draw upon and compare to others contexts, this study seeks to facilitate knowledge creation and accumulative theory development.

## 5.4.2. Interpretative Phenomenological Analysis

This study adopts an interpretative phenomenological analysis approach, which was conceived by Smith (1996) to conduct psychological research. After 20 years of application, this qualitative approach is still comparatively new, but has been successfully employed within a range of contexts (for a detailed review, see Smith, 2011) and been subject to in-depth analysis (e.g. Giorgi, 2010; Larkin et al., 2006). As Zou et al. (2016: 450-451) explain, 'this research design posits that the meanings an individual ascribes to events are of central concern but are only accessible through an interpretative process'. A particular strength of interpretative phenomenological analysis is the systematic set of procedures that it provides for interpreting the perspectives of participants. These procedures are explained throughout the remainder of this chapter. It is therefore considered a rigorous method that facilitates analysing the lived experiences of entrepreneurs (Cope, 2011; Lewis, 2015) and obtaining an 'insider's perspective' of their lives (Smith et al., 2009). Unlike other, more descriptive forms of phenomenological analysis, this technique seeks to go beyond 'first-order' analysis and, instead, reach a higher level of conceptualisation (Larkin et al., 2006). Interpretative phenomenological analysis

follows Heidegger's (1962) hermeneutic tradition, as it appreciates both the active role that researchers play in the analysis and interpretation process and the impact of social worlds on individual practices (Smith and Osborn, 2008).

Its goal of providing an insider's perspective into reality differentiates interpretative phenomenological analysis from other forms of qualitative research (Millward, 2006). Grounded theory, for example, is also both interpretative and inductive in nature, but focuses more on the investigation of social and cultural processes (Suddaby, 2006). Furthermore, several methods can be employed within grounded theory research, whereas interpretative phenomenological analysis entails a particular philosophical and methodological approach (Smith and Osborn, 2008). Discourse analysis is another inductive and interpretive technique, which is often compared to phenomenological methods (see Berglund, 2015). However, discourse analysis 'is concerned with the social functions of language in precisely defined linguistic contexts or narrative analysis, which look[s] to identify particular forms of narrative used by people as justifications for their behaviour' (Millward, 2006: 319). Thus, while there are a number of approaches which facilitate interpreting the perspectives of entrepreneurs, none are as well suited for exploring their lived experiences and 'living experience' as interpretative phenomenological analysis (Steyaert, 2007).

Regarding criticisms of this approach, Giorgi (2010) provides an overview of commonly raised concerns. First, interpretative phenomenological analysis has been criticised for being unscientific, as it is neither prescriptive nor replicable. However, as Smith (2010) points out, while the approach does not dictate precisely how to perform its methods (e.g. interviews), it does provide scholars with guidance on best practice concerning data collection techniques to help them generate high quality findings. Furthermore, Leitch et al. (2010) argue that replication and generalisability of results are not a concern to idiographic studies of individuals. Instead, such research allows for the deduction of common mechanisms and practices within specific contexts based on the analysis of several individual experiences (Kempster and Cope, 2010). It therefore enables inductive theorising and 'a degree of theoretical generalisation' (McAdam et al., 2014: 849). Another criticism of interpretative phenomenological analysis is 'that it is a "simply descriptive" methodology' (Larkin

et al., 2006: 102). Thus, this study compared emerging findings to extant research to gain a higher level of conceptualisation (Cope, 2011). Finally, Giorgi (2010) also criticises interpretative phenomenological analysis for a weak grounding in the philosophy of hermeneutics. However, this criticism seems to overlook the vast body of research which explores the relationship between the method and its underlying philosophical assumptions (e.g. Larkin et al., 2006; Smith, 2004; Smith et al., 2009).

## 5.4.3. Participant Selection

Interpretative phenomenological analysis suggests employing 'purposive' sampling (Patton, 2002) and defends the use of small sample sizes to generate in-depth understanding (Zou et al., 2016: 451), as long as a 'variety of contextual experiences' are considered (Kempster and Cope, 2010: 12). Thus, participants should be selected based on their knowledge of a phenomenon (in this case, craft entrepreneurship) to enable the researcher to gain deep understanding of related issues. Patton (2002: 169) suggests selecting interesting and 'information-rich cases' that maximise the potential for learning from the data analysis. Distinctive to interpretative phenomenological analysis is 'a commitment to producing a fine-grained interpretative account that is grounded in, and does justice to, each participant's unique lived experience' (Cope, 2011: 608-609). Accordingly, Smith (2004) suggests using a maximum of 10 participants and emphasises that scholars should be pragmatic during the selection process. This study heeded this advice and explores the perspectives of 10 craft entrepreneurs.

While it has already been established why the stories of participants are deemed interesting and worth studying (see Chapter 2), this section seeks to provide further insights into the participant selection process. Given that the study was initially conceived to follow in Husserl's tradition (i.e. including the bracketing of any prior knowledge and presuppositions), the search for participants commenced before an in-depth review of the literature was conducted. In terms of location, issues of convenience affected the decision to approach participants in South West Scotland and North East England. Apart from the fact that these areas share several similarities (see Chapter 2), both were easily accessible to the researcher. In Scotland, it was possible to meet various craft makers within one location (Artisan

Village). The researcher first advertised the proposed study there by e-mailing local makers in January 2013, before travelling to various studios in the area to judge the owners' interest in the project and to gain initial insights into their perspectives. In England, an acquaintance circulated the call for participants via e-mail to makers across Craft County. The researcher then followed up with those who responded via telephone and video-chat. These initial, unrecorded discussions lasted between 20-60 minutes each. As there was considerable interest in the research project, these conversations helped determine which participants to select for this study.

As explained in Chapter 1, this study drew upon Fillis (2004: 71) to identify craft entrepreneurs (i.e. makers who are willing to take risks, recognise the significance of developing a customer base and exhibit ambitions for growth). Thus, in choosing a sample, the researcher sought to identify craft owner-mangers who embodied these traits, practised craft as their primary profession and operated their businesses from a dedicated studio or workshop. Table 5.3 provides an overview of the 10 participants who fulfilled the inclusion criteria. Specifically, this table illustrates the breadth of perspectives that participants bring to this study. For instance, participants were aged between 29 and 64 years at the start of the data collection and they engage in a broad range of crafts, from contemporary artistic teddy bear making to more traditional basket weaving and pottery making. Similarly, the levels of reputation of participants, which are based on a subjective evaluation by the researcher, are spread between emerging, national and international.<sup>9</sup> The variation in business age (the oldest business was started in the late-90s and the youngest in 2013, only weeks before the start of the study) further enriches the study's consideration of a 'variety of contextual experiences' (Kempster and Cope, 2010: 12).

Other characteristics of the sample reveal commonalities that are equally significant. Notably, the sample consists of eight women and two men. As is typical for phenomenological research, participants were chosen irrespective of their sex (cf. Cope, 2011; McKeever et al., 2014). However, this distribution is (inadvertently) reflective of the craft industry in the UK, where makers are '69% female [and] 31% male' (Burns et al., 2012: 18). The majority of participants share their households

<sup>&</sup>lt;sup>9</sup> This evaluation involved considering various reputation-related aspects, such as invitations to exhibit craft pieces at national or international shows and exposure in craft magazines.

Name	Age	Craft	Start-Up	<b>Studio Opening</b>	Reputation	Children	Househo	ld Member	Children Household Member Cohabitants Since
Amy*	29	Accessory making	2011	2012	National	0	Adam	Adam (Partner)	2006-2013
Betty*	32	Book making	2013	2013	Emerging	0	Brenda	Brenda (Partner)	2013
Carol	36	Yam dying	2006	2007	International	0	Clara	Clara (Partner)	2001
Diana	39	Silversmithing	2001	2004	International	0	Daniel	Daniel (Partner)	2010
Emily	44	Hand stitching	2003	2007	National	1	Evan	Evan (Spouse)	1990
Frank*	46	Pottery making	2012	2013	Emerging	0	Felix	Felix (Friend)	2014
George	50	Basket weaving	1999	2004	International	2	Grace	Grace (Spouse)	1986
Holly	52	Teddy bear making	1998	2013	International	2	Henry	Henry (Partner)	1988
Isla*	63	Souvenir making	2010	2010	National	1	Iain (	(Spouse)	1975
Julie*	64	Printmaking	2006	2010	National	2	Jack	Jack (Spouse)	1984

5
H
sipar
ä
[c]
iti
Pai
-
5
5
8
S
0
R
4
Ö
verview of Research
N.
<u>е</u> .
>
H
ě
5
$\mathbf{O}$
5
0
-9

with partners or spouses and half of them have one or two children, most of whom still live at home (only Isla's and Julie's daughters have left home). As participants are all sole-traders, these household members have significant impacts on their practices (see Chapter 8). Overall, as Table 5.3 indicates, selected participants provide a broad range of perspectives, while also holding similarities which facilitated the deduction of common practices.

#### 5.5. Data Collection and Analysis

Following the selection of research participants, data were collected over the course of 18 months starting in March 2013. As such, this study adopts a longitudinal design, which is regarded as under-represented within entrepreneurship research despite its capacity to generate valuable insights into entrepreneurial processes (Jack, 2010; McMullen and Dimov, 2013). Thus, it is not only able to portray the subjective views of participants, but also to distinguish how external factors influenced them over time. This section first explains the data collection process, paying particular attention to the phenomenological interviewing technique, before reviewing ethical concerns. Subsequently, the data analysis and interpretation are discussed.

### 5.5.1. Data Collection Approach

This study sought to gain in-depth understanding of the practices of craft entrepreneurs and therefore employed methods which facilitated 'getting close' to participants (Greenman, 2012) and gathering detailed descriptions of their lived experiences. As interviews are considered the most powerful tools for exploring the lived-worlds of participants (Cope, 2011; Thompson et al., 1989), data collection was conducted primarily through four rounds of interviews, which were carried out in intervals of around six months starting in March 2013. This longitudinal approach enabled the researcher to penetrate the realities of participants and to build rapport with them (Shaw, 2006). Table 5.4 provides a detailed overview of the 40 in-depth interviews, which produced a total of 63.5 hours of audio recordings. These interviews lasted an average of 95 minutes each and their transcripts provided large amounts of information-rich data (over 1,000 single-spaced pages).

		Interview 1	w 1			Interview 2	w 2			Interview 3	w 3			Interview 4	v 4	
Participant	Date	Duration	Words	Pages	Date	Duration	Words	Pages	Date	Duration	Words	Pages	Date	Duration	Words	Pages
Amy	12/04/13	90min	16,470	31	26/09/13*	50min	7,698	16	11/03/14*	56min	8,068	17	11/09/14	80min	13,829	32
Betty	10/04/13	121min	17,307	29	11/10/13*	70min	8,770	16	28/02/14*	70min	8,272	15	10/09/14*	113min	15,265	25
Carol	13/03/13	125min	18,768	35	18/09/13	71min	10,917	19	28/03/14	80min	12,053	22	09/09/14	100min	16,154	29
Diana	20/03/13	157min	24,686	42	10/10/13	84min	12,910	22	27/03/14	91min	13,349	24	12/09/14	106min	15,899	30
Emily	22/03/13	147min	24,125	36	19/09/13	98min	15,572	23	21/02/14	80min	12,312	22	20/08/14	118min	20,429	33
Frank	18/04/13	159min	23,917	35	03/10/13*	59min	8,549	16	10/03/14*	81min	12,472	21	04/09/14	108min	18,469	30
George	21/03/13	138min	21,772	49	30/09/13	60min	8,395	18	25/03/14	51min	7,507	20	21/08/14	75min	11,836	27
Holly	19/03/13	113min	17,173	36	15/10/13	63min	11,454	26	20/02/14	56min	8,010	18	22/08/14	70min	11,448	25
Isla	02/04/13	153min	20,058	32	25/09/13	72min	8,545	15	04/02/14*	94min	10,176	27	26/09/14	105min	12,462	23
Julie	03/04/13	142min	20,601	31	27/09/13	98min	15,437	24	16/02/14*	70min	9,379	15	03/09/14	136min	21,697	35
Totals	1	1,345min 204,877	204,877	356	-	725min	108,247	195	-	729min	101,598	201	ı	1,221min	157,488	289
Averages	1	135min	20,488	36	ı	73min	10,825	20	-	73min	10,160	20		101min	15,749	29
* The inter	rview was (	The interview was conducted via video-chat or telephone.	via video-	-chat or	telephone.											

Table 5.4. Overview of Interviews

Overall totals: 63.5 hrs (3,810 min) of recordings On 572,210 words across all interviews 1,041 pages of interview transcript

On average: 1.6 hrs (95 min) per interview 14,300 words per interview 26 pages per interview (single-spaced) In addition to these 40 in-depth interviews and the 10 initial discussions during the participant selection process (which were not tape recorded, but documented through field notes), 10 further interviews were conducted with the household members listed in Table 5.3. These interviews were not originally planned. Instead, the idea emerged from conversations with participants during the second round of interviews. These 10 interviews lasted an average of 20 minutes each and primarily served to produce further insights into the ways in which household members affect the practices of participants. They were all carried out via video-chat or telephone in October and November 2013, between the second and third rounds of participant interviews. Chapter 8 presents excerpts from these interviews to better explain the influence of household members on entrepreneurs.

A total of 60 interviews thus provided the primary source of data for this study and offered deep insights into the perspectives of participants. While interpretative phenomenological analysis does not require nor, in fact, encourage the use of additional data collection methods (see Larkin et al., 2006; Smith and Osborn, 2008), it does acknowledge that 'under certain circumstances other approaches such as ... participant observation (where the activity being observed includes discussion of experience) may work' (Smith et al., 2009: 73). Observational data were therefore collected in the studios (on the days of the interviews) and at craft shows, when participants invited the researcher to attend these events. During these observations, the researcher took note of participants' working practices and interactions with community members (McKeever et al., 2014). These field notes were handwritten and primarily served to enhance understanding of the behaviours of entrepreneurs. In addition, when participants referred to other sources of information during interviews, such as blogs and online forum discussions, these were later accessed as well. These online sources extended knowledge of the contexts in which participants are embedded (Lewis, 2015). While this thesis provides a multitude of quotations from the interviews 'to allow the participants' voices to be heard' (Cope, 2005: 174) and it refers to observations in the discussion of findings (Chapters 7-9), it does not quote material from online sources. As most information on the internet is freely available for anyone to access, it would not have been appropriate to include quotations from websites, as doing so may have compromised the anonymity of participants, which had to be maintained (see the below section on ethical concerns).

In-person interviews were conducted in the workshops or homes of participants, using everyday language to make them feel comfortable (Patton, 2002). As shown in Table 5.4, there were only a few occasions where video-chat and telephone conversations took place instead, particularly when participants became too busy to meet in their studios. As this study initially embraced Husserl's (1954) phenomenological approach, it adopted Cope's (2005) associated interviewing technique for the first round of interviews. Thus, prior knowledge of entrepreneurial practices was bracketed to facilitate a discovery process, rather than imposing assumptions from the literature. The goal of these interviews was to gain first-person descriptions of entrepreneurial experiences (McKeever et al., 2014). Accordingly, participants were allowed to steer the conversations towards topics that mattered most to them (Thompson et al., 1989). Interviews were therefore unstructured and began with the broad, open-ended questions, 'Tell me about your background' and 'Tell me about your business'. Follow-up questions were developed ad-hoc, which enabled participants to expand upon their stories (Cope, 2005). The level of openness and emotion displayed by participants was surprising. They seemed to welcome the opportunity to discuss their experiences and share privileged information, knowing that their personal details would be kept confidential (see below).

Having the interviews directed by what the entrepreneurs wanted to share also presented difficulties. As some control over the interviews was given up (Thompson et al., 1989), participants frequently recounted topics unrelated to their business activities. For example, when asked about her background, Julie launched into a 4,820-word monologue (lasting 30 minutes), before the researcher was able to ask any follow-up questions that emerged from her descriptions. However, while some topics, such as discussions of former workplaces, shed comparatively little light onto the business practices of craft entrepreneurs, other seemingly unrelated subjects had significant impacts on the development of this study. Stories about spouses and children, for example, led to the exploration of the effects of household members on entrepreneurs. Thus, these unstructured interviews were critical in shaping the overall focus of this study.

During the analysis of the first round of interviews, it became clear that a Heideggerian approach to phenomenological research was more appropriate for this study. Specifically, Heidegger's (1962) appreciation of the impact of embeddedness reflected the significance of context to the stories of participants. Furthermore, his recognition of the value of the researcher's prior knowledge appeared more suitable to this study. Thus, the researcher discontinued the practice of bracketing and embraced his understanding of the craft world during subsequent interviews, which adopted a semi-structured format to explore topics that emerged during the first round of interviews. Each interview commenced with the opening question, 'What has affected your work over the last 6 months?' Several participant-specific questions were then asked to pursue issues raised during the prior interview, such as outcomes of award ceremonies or successes at craft exhibitions. After these initial 15 to 30 minutes of conversation, topic guides were used to direct the discussions.

As shown in Appendix 1, the second round of interviews primarily explored questions related to the support that participants receive from their family members, their connections with other craft makers and the effects of working in rural areas. The third round of interviews covered a broad range of subjects, including the reasons for participants to pursue their craft, their interactions with customers, their relationships with family members and their past and present practices. The final round of interviews explored such issues as participants' connections with local traders, embeddedness in the creative community and opinions about education. While most questions emerged as a result of the data analysis process, others were included because a topic was raised by participants during a prior interview. For example, Amy and Frank referred to the importance of photographs of their products during the first round of interviews, which is why all participants were asked about the subject during the second interview. Not all topics discussed with participants are included in this thesis. For instance, the final round of interviews included questions about the creative practices of participants, which turned out to be largely irrelevant for this study. However, given that phenomenological research engages in a process of discovery (Cope, 2005, 2011), it was important to pursue a range of topics which might have been important to participants.

#### 5.5.3. Ethical Concerns

When undertaking research that seeks to get close to participants and explore their lived experiences, it is important to consider associated ethical concerns (Easterby-

Smith et al., 2012). Prior to commencing this study, ethics approval was attained from the University of Strathclyde. This process required the researcher to develop a strategy to ensure the appropriate conduct of this study. The participant selection procedure, outlined above, was helpful in ensuring voluntary participation and informed consent. It introduced the study to craft makers via e-mail, allowing them to form an opinion about participating prior to meeting the researcher. During the initial, unrecorded conversations with interested craft makers, the researcher explained the broad aim of the study, which was to explore the practices of craft business owners. The researcher assured the craft makers that their anonymity would be maintained. Accordingly, all names presented in this study have been anonymised (including those of towns, firms and family members).

All interested craft makers were then presented with a participant information sheet, which reiterated that participation in the study was voluntary and that withdrawal from it was an option at any time and did not require explanation. Furthermore, it highlighted that private information would not be disclosed and that no identifying information would be made publicly available. Makers who were selected to participate in this study were then asked to provide a signed copy of the form to the researcher. In so doing, participants also gave permission for the interviews to be recorded and transcribed. In addition, participants were reminded at the start of each interview that their details would be kept confidential. Participating household members were introduced to the study by the core participants. They were given the same explanations and their consent to participate was also sought prior to conducting interviews. The researcher took great care to anonymise all interview transcripts and to ensure that the identities of participants would be kept private, including from fellow participants. Research outputs, such as journal articles and conference papers, were shared with participants. They all appeared satisfied with the work of the researcher and it is anticipated that the same will hold true for this thesis.

## 5.5.3. Data Analysis and Interpretation

As recommended for inductive research approaches (Lincoln and Guba, 1985), data collection and analysis occurred simultaneously. Following each interview, audio recordings were transcribed verbatim to enable systematic analysis of raw data

(Cope, 2011). The data analysis adhered to the principles of interpretative phenomenological analysis, which set forth a series of clear, accessible guidelines (Smith et al., 2009). While the approach allows for flexibility and individuality in its procedures, interpretative phenomenological analysis follows a basic systematic process, which starts with detailed analyses of individual interviews and proceeds with comparisons between the different perspectives of participants (Larkin et al., 2006). As explained below, this process requires the researcher to move from descriptive to interpretative activities (Smith and Osborn, 2008). Kempster and Cope (2010) provide a detailed discussion of the analysis approach, which involves performing the six steps described in Table 5.5.

Table 5.5. Data Ana	lysis Process
---------------------	---------------

Step	Description
1. Familiarising (Material reading)	Interview transcripts, observation notes and other documents are read and re-read to gain an appreciation for the participant's whole story and to recall the interview(s) both cognitively and affectively (Zou et al., 2016).
2. Sense-making (Note taking)	Transcripts are annotated employing a 'free-textual analysis' technique (Smith and Osborn, 2008: 67), which involves writing up notes to summarise, form associations and develop preliminary interpretations.
3. Categorising (Theme developing)	Initial notes are used to develop emerging themes that appear salient to a particular interview. This process is completed for each individual transcript, leading to a range of idiosyncratic coding schemes (Cope, 2011).
4. Associating (Theme clustering)	A meta-level analysis is conducted across interviews. Coding schemes are compared between participants to search for links between their accounts, identify patterns and develop superordinate concepts (Leitch et al., 2013).
5. Interpreting	Interpretive summaries are written up, consisting of narrative accounts of
(Summary writing)	each participant's experiences (Marlow and McAdam, 2015). They are shared with participants to gain their feedback on the interpretations of the researcher.
6. Explaining (Theory building)	Emerging findings are compared to concepts in the literature. This abstraction process involves moving iteratively between existing theory and the data to gain a higher level of conceptualisation (Edmondson and McManus, 2007).

Adapted from Kempster and Cope (2010: 15).

Following each round of interviews, transcripts, observation notes and other documents were (re-)read to achieve familiarisation with the stories of participants (Lewis, 2015). Potentially significant passages in the transcripts were then highlighted and annotated with notes to facilitate the discovery and coding of emerging themes (Leitch et al., 2013). This process was crucial, as interviews often

went into different directions than anticipated, resulting in disjointed accounts, which are representative of the messy and complex nature of human experience (Cope, 2005). The preceding steps were completed with the data from each participant, prior to comparing emerging themes between individual entrepreneurs and searching for patterns from which superordinate concepts could be developed (Smith et al., 2009). Each round of interviews uncovered instances of similar phenomena affecting multiple participants, leading to the identification of several aggregate themes. As explained above, topic guides were consequently developed to explore emerging themes and to direct subsequent rounds of interviews. The most influential concepts that were revealed during the analysis of data informed the foci of the findings chapters (Chapters 7-9); they relate to (1) the capital management and conversion practices of craft entrepreneurs, (2) the impact of household members on their behaviours and (3) the influence of embeddedness in communities on their practices.

After each round of interviews, interpretive summaries were written and shared with participants to gain their feedback (see Appendix 2 for an example). Participants mostly agreed with the researcher's interpretation and frequently expressed a desire to follow up on aspects of the previous discussion. Accordingly, any misunderstandings were corrected and participants were reminded of follow-up topics during subsequent interviews. The final stage of analysis involved 'putting the new knowledge about the phenomena and relations back into the context of how others have articulated the evolving knowledge' (Marlow and McAdam, 2015: 800). In other words, emerging findings were compared to existing theories to gain a higher level of abstraction (Cope, 2011). This iterative process of going back and forth between the data and the literature ensured that resulting findings made a valuable contribution to current academic debates, 'whilst remaining sensitive to the unique situated experiences of the participants' (Kempster and Cope, 2010: 15).

The data coding process changed over time as a result of the developing experience of the researcher and suggestions from co-authors and reviewers. Initially, data coding was conducted manually on printed out versions of the interview transcripts using highlighters and sticky notes. Based on the analysis of the first round of interviews, the researcher decided to submit a paper related to the capital management and conversion practices of craft entrepreneurs to the Institute for Small Business and Entrepreneurship (ISBE) Conference in 2013. During the development of this paper, it quickly became apparent that the data would benefit from the application of a theoretical framework. As Lopez and Willis (2004: 730) note, 'an interpretive approach does not negate the use of a theoretical orientation or conceptual framework as a component of inquiry'. Rather than using theory to develop or test hypotheses, theoretical frameworks can help focus phenomenological research (Cope, 2011). Based on the analysis of the experiences of participants, Bourdieu's (1986) capital theory was selected as an appropriate conceptual lens through which the behaviours of entrepreneurs could be interpreted. Thus, the data were re-coded, this time utilising the computer-based data processing tool NVivo.

Table 5.6 illustrates how the researcher moved from the raw data to the refined themes and shows how the different types of capital were operationalised. As suggested by Corley and Gioia (2004), this table presents 'first-order concepts' that are directly linked to raw data. 'Second-order themes' connect several of these concepts and are themselves amalgamated into 'aggregate theoretical dimensions', which, in the example shown, represent the different forms of capital. It should be noted that the use of this method, which was first developed to demonstrate the rigour of grounded theory research (Gioia et al., 2013), was chosen in response to the suggestion of an anonymous reviewer of the ISBE paper. As Suddaby (2006) argues, grounded theory and interpretative phenomenological approaches share similarities, as they both seek to develop theory inductively, grounded in the lived experiences of participants. Thus, it appeared sensible to illustrate the data coding and analysis process using Gioia et al.'s (2013) approach, which is gaining popularity in entrepreneurship research (cf. Hampel and Tracey, In Press; Phillips et al., 2013).

The themes included in Table 5.6 are part of the overarching concept of capital conversion (see Chapter 7). While the approach of Gioia et al. (2013) advocates for the presentation of three levels of analytical abstraction, NVivo allows for myriad sub-themes. Consequently, the 'coding tree' that this thesis employs is more complex than Table 5.6 suggests. To provide an indication of the various sub-themes included within a single concept, Figure 5.2 depicts all themes subsumed under 'symbolic capital'. Concepts and themes that are featured in both Table 5.6 and Figure 5.2 are highlighted to show the link between these two illustrations.

# Table 5.6. Data Coding Example

Examples from raw data	Second-order themes	
Aggregate theoretical dimension: Economic C	apital	1
'There are big highs and lows some months you could make thousands of pounds and other months not so much.' (Emily)	Income	Financial
'I've always been careful with my money, because if I had the money, then I knew I could do my own thing'. (Amy)	Savings	Assets
'Anything written, you have copyright over. But once you do a pattern it's much more difficult even if you actually register it'. (Julie)	Designs	Intangible Assets
'You can't make stuff unless you've got enough equipment'. (Frank)	Equipment	
'From when I was a student silver has quadrupled in price'. (Diana)	Materials	
'At least we're not paying that rent anymore. We bought it [studio] outright, so in terms of cash-flow, it's spectacular'. (George)	Property	Tangible Assets
'I can't go into creating product ranges and ordering stuff if I can't sell it, 'cos the last thing I want is a load of stock that I can't shift'. (Isla)	NTOOK	
Aggregate theoretical dimension: Cultural Co	apital	
'I don't pretend to be all-knowing, but when you've got a few years under your belt I think it's [very] valuable'. (Julie)	Experience	
'If you collect bears, your group of bears is called a hug, it's a hug of bears – see, all this knowledge is important'. (Holly)	Knowledge	Embodied Cultural Capital
'It helps that I can draw cos some people come in and they've got a specific thing in their head and we design it together'. (Emily)	Skills	
'It's very personal, as well, a piece of pottery it's sort of personal taste and experience and all sorts of things come into it'. (Frank)	Craft Items	Objectified
'People want to do things that are unusual they are not keen on doing commercial kits that everybody else can get'. (Julie)	Craft Kits	Cultural Capital
'I studied English literature even if you can't find immediate use for it you never know when it'll come in handy'. (Carol)	Institutionalised	
'I went to, like, a workshop and had a go at binding books'. (Betty)	Cultural Capital	
Aggregate theoretical dimension: Social Cap	pital	
'There may well be a fellow who's worked all their life in that field and could give insights'. (Isla)	Guilds	Business
'If there is anything you want in the newsletter, you get in touch with them [business incubator]'. (Betty)	Support Groups	Networks

'I have some of the nicest collectors I call them my lovely collectors, Henry calls them my stalkers - somewhere in between I think'. (Holly)	Collectors	
'Some people don't even look at websites, they [use] Facebook, you know, "Does anybody know somebody who can do baskets?""(George)	Online Followers	Followers
'I get wholesale requests all the time people that have bought stuff from me for personal use may also own a yarn shop'. (Carol)	Wholesalers	
'It's good to have a little network of a few people and you can ask people things, 'cos it's quite technical'. (Frank)	Makers	Deerre
If I'm having a problem with something technically, I know I could phone up my old tutor to ask him [for] advice'. (Diana)	Teachers	Peers
I'm gonna get some interns as well who will help me'. (Amy)	Apprentices	
I go back to the same sort of people lots of people want to know what's next. I have a database of people for workshops'. (Julie)	Students	Novices
If you can, talk to them [suppliers] they can unwittingly give you help and ideas for all sorts of things, even new stuff'. (Isla)	Material Suppliers	S1i
My web hosting [is] provided by the same people and it's a small independent company as well, which is really, really helpful'. (Carol)	Website Developers	Suppliers
'My boyfriend was a massive, massive help!' (Amy)	Family	
'I have a friend, who is a hotel inspector, who is trying to get them [products] into different hotels and places, which is quite fun'. (Emily)	Friends	Confidantes
Aggregate theoretical dimension: Symbolic Ca	pital	
So the accolade, I've got the certificates up there on the wall'. (Diana)	Certificates	
That's my award cabinet over there [I'm] very pleased with the top shelf of the cabinet!' (Holly)	Trophies	Awards
'You can see a piece in [Scottish art museum] one in [national gallery], you can see one in [London art museum]'. (Diana) Galleries		
'I'm doing another, a trunk show in this kind of small world people know who you are and they are excited to meet you'. (Carol) Shows		
'I've done articles for magazines which has publicised the book Magazine it's also obviously got my name out there'. (Julie) Articles		
She is quite a prominent knitwear designer, she blogged about how fabulous the studio was and I got an order off the back of that'. (Emily)	Online Features	Publicity
'She said she could always recognise the baskets when they came in, because they were so much neater and more beautiful'. (George)	Esteem	Reputation
because they were so much heater and more beautiful . (George)		

'It would only take one negative review ... you need to make sure you are customer orientated the whole time'. (Betty)

Adapted from Pret et al. (2016: 1019-1021)

Reputation

# Figure 5.2. Coding Tree for Symbolic Capital



The figure indicates how difficult it is to portray multiple layers of coding in a non-digital format, which is why the researcher decided to provide only representative examples of the wider coding process. Accordingly, additional tables are included throughout the presentation of findings (Chapters 7-9) to demonstrate the link between the raw data and refined themes.

Naturally, with each additional round of interviews, new data were collected and further codes were created. As the questions of later interviews were directed by topic guides, the researcher was able to explore emerging concepts, particularly those regarding the effects of household and community members on the practices of craft entrepreneurs. Furthermore, the researcher continued to attend academic conferences to gain feedback and disseminate emerging findings. Table 5.7 provides an overview of the conferences attended and articles published, which affected the analysis process. For example, in 2013 an anonymous reviewer for the ISBE paper made the constructive suggestion to employ Bourdieu's full theory of practice rather than only his capital theory. This advice positively affected the analysis of data, as the introduction of the concepts of habitus and field helped explain the effects of different contexts on entrepreneurial practices.

Theme	Article	Conference/Journal	Location
Capital	Pret & Shaw (2013)	Institute for Small Business and Entrepreneurship (ISBE) Conference	Cardiff
Conversion	Pret et al. (2016)	International Small Business Journal	Online
	Pret & Shaw (2014a)	Economic and Social Research Council (ESRC) Seminar	Glasgow
Household Context	Pret et al. (2014)	Babson College Entrepreneurship Research Conference (BCERC)	London, Canada
_	Pret & Shaw (2014b)	Institute for Small Business and Entrepreneurship (ISBE) Conference	Manchester
<b>a</b>	Pret (2016)	Social Innovation and Entrepreneurship Research Conference (SIERC)	Auckland, NZ
Community Context	Pret & Carter (2016)	Academy of Management (AoM) Conference	Anaheim, USA
	Pret & Carter (In Press)	Entrepreneurship & Regional Development	Online

**Table 5.7.** Conference Papers and Journal Articles

### 5.6. Methodological Reflections

The preceding explanations about the study's underlying philosophical assumptions, research design and data collection/analysis provide a wealth of information about the research approach of this study. However, further methodological reflection is required to provide a balanced picture and to address common criticisms of phenomenological methods. Accordingly, this section reviews limitations of phenomenological research, explains the validity and reliability criteria adopted in this study and considers the importance of reflexivity.

Hycner (1985) provides an in-depth discussion of various *criticisms* levelled against phenomenological research. First, phenomenological research is criticised (primarily by positivists) for a lack of *randomness* in the participant selection process. Indeed, participants are typically selected because they had particular experiences and are able to articulate them (Cope, 2005). However, this approach is consistent with the phenomenological goal of investigating phenomena in a detailed and revealing manner, rather than developing generalisable findings (Sandberg, 2005). Second, a related concern is the *limited number of participants* that phenomenological studies investigate, which further inhibits generalisability of results (Lewis, 2015).

Furthermore, phenomenological interview techniques are criticised for providing a *retrospective viewpoint* (Seymour, 2006). However, from a phenomenological perspective, this is not of particular concern, as descriptions of experiences are always considered different from the experiences themselves (Cope, 2011). An associated concern is the practice of *confabulation* in which participants engage when trying to fill gaps in their memory (i.e. recall bias) or when altering descriptions in an effort to please the researcher (Hycner, 1985). While this is usually a subconscious practice, it does affect the collection of data. Nevertheless, as phenomenological research is interested in the ways in which participants think they have experienced situations, the potential effects are of limited concern (Cope, 2011). In addition, it is believed that longitudinal research approaches minimise 'distortion, omission and (self)-deception' (Lam and Harker, 2015: 328).

Phenomenological research is also often criticised for the *subjective influence* of the researcher. As Smith et al. (2009: 3) highlight, phenomenological research

relies upon 'double hermeneutics': scholars interpret events as relayed to them by participants who have themselves interpreted these experiences. However, as Leitch et al. (2010) emphasise, research in the social sciences can never be free from interpretation. Thus, it could be argued that, by embracing the resulting subjectivity, phenomenological approaches adhere to a fundamental principle of the field. Another criticism of phenomenological studies is their lack of *replicability* (Sandberg, 2005). Scholars are likely to interpret the same interview transcripts in different ways. Once again, given that interpretivist approaches embrace subjectivity, differences in interpretation are generally appreciated by phenomenologists (Smith and Osborn, 2008). Similarly, concerns about the *absence of hypotheses, control groups and prediction* are based on a functionalist understanding of the principles of research (Hycner, 1985). It is therefore important to evaluate phenomenological research based on its underlying assumptions, which is why this study provided such a detailed discussion in this respect.

As this study does not conform to the 'dominant positivist-quantitative orthodoxy' (Smith et al., 2013: 366), its value cannot be judged according to traditional positivist criteria, such as objectivity and generalisability (Leitch et al., 2010). Instead, interpretivists must adopt alternative approaches to demonstrate the *validity and reliability* of their work (Finlay, 2006). While several different approaches have been developed to achieve this goal (cf. Leitch et al., 2010; Sandberg, 2005; Yanow, 2006), they all serve to verify the empirical findings of interpretivist research (for a detailed comparison, see Finlay, 2006). Heeding Berglund's (2015) advice, this study employs Sandberg's (2005) well-established technique, which involves the use of four criteria: communicative, pragmatic and transgressive validity (Kvale, 1989; Lather, 1993) and reliability as interpretive awareness (Sandberg, 2000).

First, *communicative validity* was established by pursuing consistent interpretations of the perspectives of participants (Kvale, 1989). This entailed allowing participants to explain their experiences in detail by asking open-ended questions and subjecting the resulting interpretations 'to critical appraisal as a result of the [supervisors'] reviewing process and presentation at conferences as well as openly discussing findings with participants' (Lamb et al., 2011: 679). Second, to

achieve pragmatic validity (Kvale, 1989), emerging knowledge was tested by asking participants probing questions and by taking into account their reactions to the researcher's interpretations of their experiences. This process facilitated correcting any discrepancies and enabled participants to clarify their thinking (Sandberg, 2000). Third, transgressive validity was attained by exploring the differences between individual views, rather than solely investigating the similarities between the perspectives of participants (Lather, 1993). This approach helped account for the diverse and complex nature of lived experiences. Fourth, to ensure reliability as interpretive awareness (Sandberg, 2000), the researcher acknowledged the subjectivity that derives from prior knowledge of a subject and from gaining a profound understanding of the lived-worlds of participants. The stories of all participants were treated as equally important and the researcher continuously checked to ensure that emerging themes could not be better explained through alternative conceptualisations (Sandberg, 2005). This comprehensive approach was employed to verify the findings of this study and to demonstrate the quality of its research design.

As Finlay (2006: 321) points out, 'the role of the researcher needs to be acknowledged and accounted for in the documentation of ... research'. *Reflexivity* is particularly important in phenomenological studies, where researchers interpret the experiences of participants (Heidegger, 1962). Thus, the researcher shares relevant personal reflections below. In Finlay's (2006: 319) words, these critically self-aware evaluations are intended to 'show you, the reader, different dimensions of my thinking'. It is important to recognise that, whether consciously or subconsciously, researchers always bring a 'hidden agenda' to the research process (Cope, 2005). By making readers aware of the researcher's involvement in a study, underlying goals and beliefs can be revealed. In addition, Alvesson and Sandberg (2013) argue that a reflexive attitude encourages one to consider alternative explanations for phenomena, which can lead to greater analytical insight. However, as Conklin (2007, 2012) notes, reflexive practices are not without their challenges and it is essential that the focus of phenomenological research remains on the perspectives of participants rather than those of the researcher.

Bourdieu (1977) also highlights the role of reflexivity in the research process (for in-depth discussions, see Karataş-Özkan and Chell, 2010; Swartz, 2008). According to Golsorkhi et al. (2009: 786), Bourdieu suggests that 'researchers must be able to become aware [not only] of the illusio and the doxa of the fields they investigate but also of that of their own scientific fields'. Drakopoulou Dodd et al. (2014) provide a detailed analysis of the doxa and illusio in the field of qualitative entrepreneurship research. They find that the need to fit-in with the prevailing positivist perspective affects the abilities of qualitative scholars to tell the stories of their participants, so much so that some abandon the pursuit of journal publication in favour of writing book chapters and books. Fortunately, this study is not limited by the need to meet the expectations of journal reviewers, but rather is addressed to those who appreciate the importance of adopting alternative paradigms and approaches in entrepreneurship research.

Personal reflections: First, it is important to consider my own engagement in the craft field. When I moved to the UK in 2005, I was trained in woodworking and later taught this craft to students at a college in England. Although I never started my own firm, I did seize several opportunities to apply and develop my skills in a range of capacities, including being a theatre technician in Scotland, a set designer in the USA and a carpenter in Australia. These experiences provided me with deep insights into cultural production and fuelled my interest in exploring the practices of craft entrepreneurs. While my perceptions of craft practices will have inadvertently influenced my understanding of the accounts of participants, my embeddedness in this context helped me establish rapport with participants and allowed them to build on my knowledge in order to explain their perspectives in more detail.

Second, my upbringing affected my choice of exploring entrepreneurship in rural surroundings. I grew up in a village east of Berlin, which is why I am aware of the peculiarities of living in an accessible rural area. Among other things, I received help not only from my neighbours and friends, but many other people, including local business owners, postal workers and even the mayor, who knew me personally and took an interest in my life. However, this interest in the affairs of other people did, of course, also encourage gossip and inhibit privacy. Thus, I am familiar with the benefits and drawbacks of residing in a rural area. After moving to the UK, I have also spent time living and working in smaller towns in England and Scotland, so my experience goes beyond the German context. Accordingly, I was able to draw on this understanding in my discussions with participants, but it will also have affected my interpretations of their experiences.

Third, it is also important to reflect upon my learning experience throughout the research process. Over the course of this study, I honed many crucial skills (e.g. interviewing techniques) and advanced my knowledge of research practices based on experiential learning – particularly from challenges and failures. For instance, I learned that adding participants, such as household members, to an ongoing study is challenging. While participating entrepreneurs were surprisingly open and happy to share their experiences, I found that their household members were much less eager to divulge personal information. I suspect that this was mainly due to a lack of rapport (the importance of which I learned to appreciate more as the research progressed). Despite my reassurances that all information would be kept confidential, it struck me that some household members still feared that they might say 'the wrong thing' about the entrepreneurs. Thus, I have come to the conclusion that any future research which incorporates the accounts of household or family members should be designed as such from the beginning.

The reflections presented in this section provide insights into the researcher's personal engagement in the craft industry, his background and the experiential learning that emerged from conducting this study. By introducing the researcher's lived experience and highlighting how it impacted the development of this study over time, these reflections demonstrate a deeper level of temporal embeddedness within this study. Not only did the researcher analyse how participants' lived experiences shape their actions over time, but also considered how his own history impacted that analysis. It should be noted that while the selected examples of reflections presented above are not exhaustive, they were chosen because they represent the reflective approach that was adopted in this study.

# 5.7. Conclusion

This chapter provided an in-depth review of the philosophical and methodological underpinnings of this study. In line with its overall objectives and the researcher's own beliefs, this study adopts an interpretivist stance and employs an interpretative phenomenological analysis approach. Thus, it interprets the subjective views of participating entrepreneurs to create situated, local knowledge, rather than generalisable findings. Theory is developed inductively, based on the livedexperiences of participants. Consequently, this study is able to generate new theoretical constructs, particularly in relation to entrepreneurs' management of various forms of capital and the impact of household and community members on their practices. Furthermore, it enhances existing theories, especially Bourdieu's theory of practice, through analysis of the real-life behaviours of entrepreneurs. In adopting a phenomenological approach and collecting data over the course of 18 months, this study addresses repeated calls for more paradigmatic and methodological diversity as well as for more longitudinal designs in entrepreneurship research. In explaining the study's purposeful participant selection, in-depth interviewing technique and systematic data analysis, this chapter demonstrated the quality of the adopted research design and the validity and reliability of the resulting findings discussed below.

#### **6.1. Introduction**

In proposing a theory of practice for craft entrepreneurs, this chapter seeks to fulfil the overarching goal of this thesis. Whereas the findings chapters (Chapters 7-9) present the empirical data collected and discuss them in relation to the extant literature, this chapter provides a conceptual synthesis of the results of this study. As such, it does not draw directly on examples from the raw data to justify its arguments, but rather refers to the following chapters, which provide detailed analyses of the behaviours of participants.

The conceptual model presented in this chapter explains how embeddedness in multiple contexts impacts entrepreneurs' resource management practices. It demonstrates how micro-, meso- and macro-level contexts, subsumed within the overarching temporal context, affect the behaviours of craft entrepreneurs. In so doing, it illustrates the connections between the subsequent findings chapters. While generalisability of the results cannot be assumed (Leitch et al., 2010), future research can compare the theory of practice presented in this study to the experiences of entrepreneurs in other industries, adapt (or confirm) the conceptual model and thus engage in accumulative theory development (Cope, 2005). This chapter is structured as follows. The conceptual model is introduced in the next section, which discusses the aggregate stock of capital, inhibitors and facilitators of capital conversion, competing habitus and the temporal context. Next, the relationship to Bourdieu's theory of practice is examined. Finally, conclusions are drawn.

### 6.2. Conceptual Model

Figure 6.1 displays a theory of practice for craft entrepreneurs, which was developed through an analysis of the experiences of participants. This conceptual model integrates multiple contexts and facilitates investigations that transcend micro-, meso- and macro-levels, addressing calls for research to develop multilevel theories of entrepreneurial practice (De Clercq et al., 2013; Kim et al., 2016; Shepherd, 2009). Specifically, this model shows that the resource management practices of craft



# Figure 6.1. A Theory of Practice for Craft Entrepreneurs

entrepreneurs are affected by embeddedness in household and community contexts, and that the interplay between these contexts is in turn governed by the temporal context. Given that accumulation and conversion of capital are crucial for the successful operation of craft enterprises (e.g. crafting products for sale involves the conversion of cultural into economic capital) and that the day-to-day activities of participants revolve around these practices, they permeate this model.

# 6.2.1. Aggregate Stock of Capital

The stock of capital available to craft entrepreneurs greatly affects their ability to develop sustainable businesses. Accordingly, participants were found to engage in a range of conversion processes that allow them to maximise their amount of capital. Chapter 7 discusses in detail how craft entrepreneurs convert their economic, cultural, social and symbolic capital. It provides a multitude of examples of various conversions, which are shown to lead to the development of individual as well as multiple forms of capital. As indicated in Figure 6.1, this study finds that each form of capital is convertible into all other forms, as well as different variations of the same form of capital. Since prior research has mainly focused on only one or two forms of capital (Bhagavatula et al., 2010; Davidsson and Honig, 2003) and conceptual discussions of their convertibility have rarely been verified by empirical research (Scott, 2012), the findings of this study advance contemporary debates. Importantly, Figure 6.1 illustrates only the conversions between Bourdieu's (1986) generic forms of capital. Chapter 7 provides a more nuanced account of different variations of these capitals and their conversions. As explained in Chapter 3, the term 'capital conversion' is understood to entail the creation of new capital and does not necessitate the loss of the original form of capital (apart from economic capital which is usually lost during its conversion). As shown in Figure 6.1, newly created capital is therefore added to the entrepreneur's available stock of capital.

It should be noted that the processes discussed in Chapter 7 present only a partial picture. Without accounting for the household and community contexts in which craft entrepreneurs are embedded, their aggregate stock of capital cannot be determined. As shown in Chapter 8, household members provide entrepreneurs with a multitude of readily available resources, which are free from (or require low)

transaction costs. Whereas prior research on household capital is limited to analyses of financial, human and social capital (Rodriguez et al., 2009; Steier, 2009), this study demonstrates that cohabitants supply craft entrepreneurs with a much broader range of resources. Thus, these household resources become part of the stock of capital that entrepreneurs draw upon to engage in capital conversion processes. Indeed, Chapter 8 provides examples of crucial activities (e.g. website design and maintenance), which participants can only conduct with the help of cohabitants and their various capitals.

In addition, Chapter 9 reveals how embeddedness in craft communities leads to the open sharing of resources among community members. This study shows that entrepreneurs not only gain access to financial support, network contacts and tacit knowledge, as demonstrated by the extant literature (Chalmers and Balan-Vnuk, 2013; Maclean et al., 2013), but a number of additional resources, including cocreated cultural and symbolic capital. Craft entrepreneurs were found to collaborate with and draw upon the support of community members regularly, allowing them to increase their aggregate stock of capital. However, this support is not unidirectional: craft entrepreneurs also provide household and community members with resources, which can reduce their amount of capital. Thus, in order to understand the mutable stock of capital available to craft entrepreneurs, the evolving resource contributions and needs of household and community members over time need to be considered.

## 6.2.2. Inhibitors and Facilitators of Capital Conversion

As shown in Figure 6.1, there are both inhibitors and facilitators to capital conversions. Based on the analysis of participants' experiences, Chapter 7 identifies a range of conversion-specific inhibitors, which have largely been overlooked in the literature. Among others, the findings show that economic capital conversions can be inhibited by associated risks, cultural capital conversions by lapses in memory, social capital conversions by the high expectations of contacts and symbolic capital conversion by a lack of self-confidence. As shown in Figure 6.1, the list of these factors is extended once household and community contexts are taken into account.

Concerning additional inhibitors of capital conversion, Chapter 8 highlights that the problems, needs and opinions of household and family members can negatively impact craft entrepreneurs. Cohabitants' problems, such as job insecurity or illness, can lead entrepreneurs to temporarily relegate their business operations to a secondary role and therefore neglect vital activities such as capital conversions. Similarly, household members' needs, including those related to scheduling and relocation, are found to restrict the abilities of entrepreneurs to engage in the development of new capital. Furthermore, expressions of disapproval, doubt and disregard from wider family members are shown to inhibit craft entrepreneurs, as they decrease their self-confidence, which is crucial to successful creative practices. As most of these inhibitors result from temporary circumstances, they highlight how the temporal embeddedness not only of entrepreneurs, but also of their household and family members affects entrepreneurial actions.

Embeddedness in communities can further inhibit the abilities of craft entrepreneurs to engage in capital accumulation and conversion. Chapter 9 shows how gossip acts as a control mechanism which ensures that community members adhere to extant norms. This compels participants to refrain from a number of activities, including copying ideas, which prevents them from conducting certain cultural capital conversions. In addition, Chapter 9 reveals that embeddedness in craft communities can lead to a dis-embeddedness from local communities, which limits access to local resources. At the same time, Figure 6.1 indicates that these byproducts of embeddedness can also act as facilitators of capital conversion. This study shows that the control mechanism of gossip can free craft entrepreneurs to engage in collaborative and relational creative practices without fearing plagiarism. Similarly, by focusing exclusively on their craft communities, rather than local communities, participants can avoid spending resources on business unrelated activities.

Regarding other facilitators of capital conversion, Chapter 8 shows that household members do not only provide instrumental, but also emotional support to craft entrepreneurs. By boosting their self-efficacy, hope, optimism and resilience, cohabitants increase the psychological capital of entrepreneurs (Baron et al., 2013; Zou et al., 2016). As such, they provide them with the means of combating inhibitors to conversion processes, such as a lack of self-confidence. Furthermore, Chapter 9 reveals that participants engage in a range of strategic changes of practice, which allow them to temporarily adapt their conduct to meet the expectations of community members, depending on the surrounding temporal context. These strategic actions facilitate a number of crucial activities, such as finding and maintaining collectors and gaining invitations to shows and exhibitions, which can increase their social and symbolic capital. In other words, embeddedness in household, community and temporal contexts both inhibits and facilitates the practices of craft entrepreneurs.

## 6.2.3. Competing Habitus

As craft entrepreneurs are embedded in multiple contexts, they are affected by each context's divergent 'rules of behaviour' (habitus). On a micro-level, Chapter 7 shows that their actions are driven by the need to accumulate new resources and to maximise available capital. Thus, the practices of participants are aligned with those of most business owners who operate in capitalist societies (Calás et al., 2009; Schumpeter, 1942). Craft entrepreneurs convert their capital not only to generate income, but also to create new resources that can be reinvested into the business and support the development of the firm. As personal dispositions, these underlying objectives affect the practices and agency of entrepreneurs.

However, the fields in which entrepreneurs are embedded feature their own habitus, which can oppose one another and create conflict (Friedland, 2009). In the case of the household context (Chapter 8), this study shows that habitus prescribes an open and free exchange of resources and that cohabitants are expected to provide unconditional support to one another. Similarly, embeddedness in craft communities leads entrepreneurs to engage in collaboration and social value creation because the prevailing habitus requires community members to demonstrate high levels of camaraderie and generosity (Chapter 9). As shown in Figure 6.1, this study finds that, depending on participants' temporal contexts, they will prioritise meso- or macro-level habitus over micro-level habitus, as they seek to maintain active membership in their households and craft communities. For example, when they are in contact with members of either field, such as on the weekends with their families or when interacting with other makers at craft shows, the maximisation of capital is often relegated to a secondary role. This does not mean that participants do not continue to engage in capital conversion processes, but rather that the importance of

fitting in with these contexts is valued more than the need to accumulate new capital. As explained in Chapter 9, however, compliance with extant norms is also a prerequisite for enjoying the benefits that membership in communities provides, including access to a range of capitals. As such, though entrepreneurs may, at times, prioritise the habitus of one field, this does not necessarily entail a complete neglect of the habitus of another.

# 6.2.4. Temporal Context

In line with this study's main findings, and thus the discussions presented in Chapters 7-9, Figure 6.1 focuses on the social and spatial contexts in which participants are embedded. However, the model also illustrates how these contexts exist within an overarching temporal context. Thus, this temporal dimension of craft entrepreneurs' practices needs to be taken into consideration. As time and history are highly abstract concepts, it is difficult to directly observe their impacts upon daily activities (Lippmann and Aldrich, 2016b). However, throughout this study, it became clear that participants' actions within myriad contexts are shaped by their temporal contexts. First, craft entrepreneurs' lived experiences (i.e. their personal histories, traditions, shared memories and collective identities) are what enable them to enact the habitus of the different fields in which they are embedded. Second, the constantly fluctuating temporal context, in which micro-, meso- and macro-level contexts are subsumed, affects how participants negotiate these competing habitus and manage their resources.

The generative function of temporal context becomes clear when viewed as lived experience. Taking first its role on a micro-level, craft entrepreneurs' nonmaterial capitals are largely the products of past experiences (Lee and Shaw, 2006). They are the result of investments into education (cultural capital), connections (social capital) and reputation (symbolic capital) – the products of a lifetime spent building upon knowledge, competencies, relationships and values. Within the mesolevel and macro-level contexts of the household and craft community, it is the sharing of experiences and memories with others, and the resulting collectivism, which enables craft entrepreneurs to successfully operate within and benefit from these contexts. For example, as discussed in Chapter 8, the value of psychological capital provided to participants by their household members is dependent upon a sense of mutual understanding created through collective memories of shared experiences. Similarly, Chapter 9 finds that habituation into the craft community is contingent upon past participation in collective activities and sharing of collective history. Thus, entrepreneurs' embeddedness within these contexts is a consequence of their temporal embeddedness.

Furthermore, as indicated in the above discussion of competing habitus, the manner in which participants prioritise one field over another when habitus clash is determined by their temporal context. While both meso- and macro-level habitus were found to naturally take precedence over that of the micro-level (see Figure 6.1), this process occurs within specific temporal contexts, as mentioned above. Similarly, the conscious prioritisation of habitus was found to be dependent upon surrounding circumstances. During peak times (e.g. in the lead up to major exhibitions or holidays), when craft entrepreneurs focus on the creation of products, they interact less with their peers and households and, as a result, engage in fewer mutual support activities. At times when members of the household are in particular need, however, participants were found to focus their energies and capital on coping with these problems. Periodically, participants' and other community members' businesses slow, which is when they tend to share more resources with one another and undertake collaborations. As such, entrepreneurs' temporal contexts not only impact their prioritisation of competing habitus, but also their resource management practices. Thus, accounts of entrepreneur's aggregate stock of capital, as well as inhibitors and facilitators of capital conversion, are always only snapshots in time.

### 6.3. Relationship to Bourdieu's Theory of Practice

This study draws on Bourdieu's (1977, 1982, 1986) work to develop a theory of practice for craft entrepreneurs. Thus, it is important to examine the differences between the conceptual model presented in this study and other interpretations of Bourdieu's work.

First, Bourdieu's theoretical framework allows for investigations of a multitude of fields in addition to family and creative communities, such as religion and education. As explained in Chapter 1, this study focuses on the impacts of

household and community contexts on entrepreneurs' resource management practices as a result of the analysis of participants' perspectives. Incidentally, none of the participants are active members in religious groups. Furthermore, their engagement in their respective craft communities leads them to experience a degree of dis-embeddedness from their local communities. Thus, only two influential social and spatial contexts are featured in this study's theory of practice, as shown in Figure 6.1. However, other entrepreneurs (e.g. religious individuals) will be influenced by a greater number of contexts, as suggested by Bourdieu (1977).

Second, Bourdieu (1984, 1986) argues that human behaviour is generally driven by a struggle for power within different fields. While the findings of this study suggest that, on an individual level, craft entrepreneurs seek to accumulate and maximise their available capital, Chapter 8 and 9 clearly show that the habitus of household and community can override such conventional market-driven behaviours. As such, the theory presented in this study allows for a more nuanced picture of entrepreneurial practices to emerge.

Third, while convertibility of capital is an important aspect of Bourdieu's (1986) theory, only a limited number of conversions are discussed in his work (see Chapter 3). In comparison, this study accounts for a multitude of conversions that craft entrepreneurs conduct on a regular basis. Furthermore, this study offers a more detailed list of inhibitors and facilitators of capital conversion than presented by Bourdieu, who focuses on the time required to engage in transformation processes. In addition, while Bourdieu (1984) considers capital to be the 'stakes of the game' that individuals both strive for and possess, he pays little attention to the impact of capital sharing among agents (a notable exception is in his discussion of the family field, see Bourdieu, 1996). Thus, the conceptual model presented in this study extends Bourdieu's theorising.

As discussed in Chapter 3, there are a number of studies that employ Bourdieu's theory of practice in the context of entrepreneurship. However, the majority of these studies draw on individual aspects of his framework, such as capital theory, to explain the behaviours of entrepreneurs. Consequently, there are few comprehensive Bourdieusian theories within the field of entrepreneurship to which this study's theory of practice can be compared. Within transnational entrepreneurship scholarship, Drori et al.'s (2006) work has sparked several empirical studies which account for multiple contexts. For example, Patel and Conklin (2009) examine the importance of bifocality (i.e. the ability to operate in two different environments) and Terjesen and Elam (2009) explore macroenvironment related factors, such as cultural repertoires and power relations, which affect the practices of transnational entrepreneurs. However, neither of these studies considers the influence of household or family members on entrepreneurs and both largely overlook the importance of cultural and symbolic capital in their empirical investigations. In comparison, De Clercq & Voronov (2009a, 2009b, 2009c), who engage in in-depth conceptual discussions, focus on the importance of cultural and symbolic capital for generating entrepreneurial legitimacy, but at the expense of economic and social capital. Furthermore, Karatas-Özkan (2011) examines entrepreneurial learning, which she interprets as a meso-level construct, and capital transformations, but omits field-related considerations. Thus, while there are a number of entrepreneurship studies that draw on Bourdieu's work to develop theories of practice, the conceptual model presented in this study differs in its depth and focus on micro-, meso- and macro-level environments.

## 6.4. Conclusion

This chapter presented a theory of practice for craft entrepreneurs. It illustrated how entrepreneurs accumulate and convert their various forms of capital and how embeddedness in multiple contexts affects these practices. This chapter demonstrated that the stock of capital available to entrepreneurs is an aggregate of the various resources generated by the individual and shared by members of the household and community. It also showed that there are a number of inhibitors and facilitators to capital conversion, which derive from micro-, meso- and macro-level factors. This chapter highlighted that the competing habitus, which results from embeddedness in multiple fields, leads craft entrepreneurs to prioritise fitting in with the norms of household and community over the individualistic desire to maximise available capital. Furthermore, this chapter revealed that these contexts, and entrepreneurs' actions within them, exist within a wider temporal dimension, which needs to be taken into consideration. Craft entrepreneurs' temporal contexts are what enable them to enact different fields' habitus and affect how they negotiate these competing habitus and manage their resources. By presenting a comprehensive theory of practice, this study adapts Bourdieu's work to better describe the context of craft entrepreneurship. It differs from prior work that has employed Bourdieu's theoretical framework and facilitates future investigations into other contexts and industries.

#### 7.1. Introduction

As explained in the previous chapter, which introduced the theory of practice for craft entrepreneurs, each of the findings chapters explores a different environment. This chapter focuses on the individual actions of participants (micro-level), before the following chapters examine the influences of household (meso-level) and community contexts (macro-level) on entrepreneurs. Given the paucity of empirical research into the convertibility of capitals (Scott, 2012), this chapter seeks to advance knowledge by addressing the research question: *How do entrepreneurs convert their economic, cultural, social and symbolic capital*?

This chapter is structured to demonstrate the distinctive conversions found between each form of capital. First, it discusses conversions of economic capital, revealing their comparatively minor role in the wider capital transformation processes conducted by craft entrepreneurs. Second, it examines conversions of cultural capital, highlighting their high exchange rates and resulting value for participants. Third, the chapter investigates social capital conversions, demonstrating the significance of network ties to the accumulation of resources and the brokerage of new connections. Finally, it explores symbolic capital conversions, illustrating the transmutability of awards, exhibitions and publicity, which have largely been overlooked in the entrepreneurship literature. Importantly, this chapter examines not only direct conversions from one form of capital into another, but also accounts for more complex transformations which result in the accumulation of multiple forms of capital. Furthermore, it exposes various inhibitors to capital conversion processes. In so doing, this study advances the findings of prior research on capital conversion (Nicolopoulou, 2014; Scott, 2012).

## 7.2. Economic Capital Conversions

As illustrated in Table 7.1, craft entrepreneurs utilise their economic capital to expand their available stock of resources. For example, participants were found to
employ profits from sales to acquire tangible business assets (economic capital). Furthermore, they acknowledge that the payment of fees can be a prerequisite for learning new skills (cultural capital), obtaining memberships in business networks (social capital) and securing publicity (symbolic capital). These conversions of economic capital appear to be purposeful and considered; participants do not spend their business income without weighing the costs and benefits of doing so.

## Table 7.1. Conversions of Economic Capital

Outcome	Representative Quotation
Economic Capital	'Everything I bought, you know, the chairs, tables and things it all came from the money I made selling the prints'. (Julie)
	'If a bear sells and I need stuff, I'll buy it out of the profit from that bear'. (Holly)
Cultural Capital	'I have paid to go on workshops ones that teach me a new technique I've always updated my professional skills'. (Julie)
	'I paid tuition fees for that course [and] I did learn stuff from it'. (Frank)
Social Capital	'You have to pay an annual fee for that [membership], but I would say it's worth it, even just for having met [new supplier], you know'. (Diana)
	'It's [membership] not very much. It's only £22 a year or something like that'. (Carol)
Symbolic Capital	'For actual product shots you do need to pay to get some professional photography done you have to go the extra mile [for publications]'. (Isla)
	'You pay to get into that one [magazine] and it's very important for me to be seen to be in that'. (Holly)

Indeed, participants communicate that their investments of economic capital can generate more than a single form of capital, making such conversions even more lucrative. For instance, Isla shares,

[Guild] membership is £160 a year ... they do some seminars ... and you get the use of the society house in London ... you can use their library, their business room ... they have their own website and you can ... put your project up and ask for input or help.

Thus, the returns on investment into membership include learning new skills and gaining creative advice from peers (cultural capital), as well as complimentary access to tangible and intangible business assets (economic capital), rendering it more beneficial than costly to Isla. Similarly, Diana explains,

You pay for that [service] ... You get photographs done at a reduced rate with ... the best craft photographer in Scotland ... I've been taking advantage of that ... images are very important, if you're submitting for shows ... They [also] put your images forward to design magazines and write a little article on you.

In other words, paying a single service charge provides Diana with valuable contacts (social capital) and affordable, high-quality photographs, which are assets in their own right (economic capital) and which facilitate show attendance and publicity in magazines (symbolic capital).

While conversions of economic capital may appear effortless and straightforward, the data also show that such activities can be impeded, as demonstrated in Table 7.2. At times, the complexity of the process, the risk involved and the sheer cost convinces participants against transforming economic capital into other forms of capital. Once again, these decisions are the result of participants appraising the benefits of pursuing such conversions and, in some instances, determining that the benefits do not offset the costs.

Inhibitor	Representative Quotation
Complexity	'I thought about applying to be a member of the society of master craftsmen but it's expensive and you have to jump though so many hoops'. (Julie)
	'You have to fill out an application form and apply with images and pay a fee and you're still not guaranteed to get in [the course]'. (Amy)
Risk	'Because the material is so expensive, just to make pieces on spec that are not commissioned it's trying to weigh up whether it's worth doing'. (Diana)
	'To buy the Harris Tweed [in bulk] it's an expensive thing to do'. (Emily)
Cost	'The issue there was that they [gallery] wanted to charge a kind of membership fee every month, and I said, "I can't do that I'm not selling enough". (Frank)
	'Exhibiting in the new gallery space it's too expensive'. (Carol)

#### 7.3. Cultural Capital Conversions

While participants approach economic capital conversions as necessary activities, which are a 'means to an end' rather than a particularly enjoyable practice, they seem to embrace cultural capital conversions much more emphatically. Specifically,

craft entrepreneurs were found to readily transform their embodied, objectified and institutionalised cultural capital into other forms of capital.

Table 7.3 provides examples of how craft entrepreneurs benefit from converting their *embodied cultural capital*. For instance, knowledge and skills enable participants to apply for grants and garner orders for commissions (economic capital). They also allow craft entrepreneurs to run workshops and give talks, through which they can obtain feedback on existing products, develop ideas for new techniques (cultural capital) and spread recognition for their ventures (symbolic capital). Furthermore, participants draw on their experience in their respective craft worlds to gather industry contacts and collectors (social capital). Due to the tacit nature of embodied cultural capital, its conversions were found to occur almost organically.

<b>Table 7.3.</b> Conversions of Embodied Cultural Capital
--

Outcome	Representative Quotation
Economic Capital	'I applied for a start-up grant you had to include a business plan, which I knew how to do so that was very fortunate'. (Diana)
	'She [wholesaler] thought my baskets were so much neater and more beautiful so anything funny [orders] that came in they were given to me'. (George)
Cultural Capital	'I'm running some workshops, so I'm hoping to get some feedback on the new [products]'. (Julie)
	'The workshops are quite fun to run 'cos when somebody else tries something that I don't normally do that sometimes gives me ideas of how to do stuff'. (Carol)
Social Capital	'I got a job at a fashion house which wasn't my thing, but at least I made really good contacts'. (Amy)
	'The longer you've been making, the more people get to know you'. (Holly)
Symbolic Capital	'I'm going to give a talk at [museum] it's all about, you know, lifting my profile you never know, they might invite me for an exhibition in the future'. (Julie)
	'I did more workshopswhich started to get my name out there'. (Emily)

Skills in particular seem to be well-suited for conversion into multiple forms of capital. For example, the skill of pitching allows craft entrepreneurs to generate myriad benefits for their businesses, as demonstrated by Julie: 'I've sent in a proposal to [gallery] ... to do a weekend there ... it's good exposure and I feel they are going to be a good fit ... profitable in a marketing sense ... and in a financial sense'. Thus, through her aptitude for proposal writing, Julie managed to secure an

opportunity to exhibit her products at a reputable gallery (symbolic capital), which she is confident will lead to sales (economic capital). Similarly, Frank 'pitched an idea to [museum], which is a big museum, and it has hundreds of thousands of visitors every year and ...that got them interested in selling my mugs, which is really good'. In this way, Frank used his skills not only to gain a valuable wholesaler (social capital) but also to associate his work with a popular, well-established museum (symbolic capital).

Despite these benefits and the apparent ease with which participants engage in converting their embodied cultural capital, they can also be hindered from performing this activity. As shown in Table 7.4, craft entrepreneurs are affected by IT ineptitude, lack of time and lapses in memory, which can prevent them from being able to capitalise upon their skills, knowledge and experience. Interestingly, participants appear to view all of these inhibitors as inherent and unavoidable, rather than conditions they could work to correct, further demonstrating the organic nature of such capital conversions.

 Table 7.4. Inhibitors to Embodied Cultural Capital Conversions

Inhibitor	Representative Quotation
IT Ineptitude	'I had a few videos to put on, but I'm not really good with this technology I just dread when things don't work for me'. (George)
	'I'm not really that happy with my website it doesn't show the intricacy of my stitching but it never seems a priority, because I'm not very computer literate'. (Emily)
Lack of Time	'It was a lot more work obviously than two hours my first class, it [took] most of the day to plan it and then I had to fire [their] stuff in the kiln'. (Frank)
	'It's not that easy because you've gotta prepare everything if you teach'. (George)
Memory Lapses	'I ran a workshop and then I came in here and then I was busy catching up with the orders and by the time, you know, I was ready to work on the [new range], I'd forgotten most of the [students'] ideas'. (Carol)

Craft entrepreneurs have ready access to *objectified cultural capital*; it is the very basis of their ventures in the form of the craft products they create. The ubiquitous nature of this form of capital enables participants to profit from its conversion in several ways, as illustrated in Table 7.5. Not only are cultural goods transformed into sales (economic capital), which is a core business function, but participants also convert them into more ideas for products (cultural capital). In

addition, the cultural worth of their craft products facilitates developing relationships with collectors and wholesalers (social capital) as well as accumulating awards and generating media attention (symbolic capital).

 Table 7.5. Conversions of Objectified Cultural Capital

Outcome	Representative Quotation
Economic Capital	'The kits really are the way to go as far as money making is concerned because people want original [work] and they don't want to have to think for themselves'. (Emily)
	'The whole point about it is that you're kind of selling a lifestyle a piece of pottery it's sort of personal taste and experience and all sorts of things come into it'. (Frank)
Cultural Capital	'Well, it started really with the book and the realisation that people wanted that, which generated this one [book] and this one has generated these products'. (Julie)
	'I had made something for Valentine's Day with lots of different colours of felt and then that kind of developed into other things'. (Emily)
Social Capital	'They [collectors] said, "I've looked at loads of websites, and yours [baskets] are the best", you know, they can see somehow that integrity [that's] in the work'. (George)
	'I took along some examples of my work and one of them said "You've got talent I know some people who might be interested in selling your work". (Isla)
Symbolic Capital	'You send in photographs of your creations and the sort of big wigs in our world vote on them and then the top five get through to the public choice awards'. (Holly)
	'Magazines sometimes call and say, "We like this basket, can we put it in and give you a mention?" (George)

Indeed, objectified cultural capital is highly versatile in the hands of craft entrepreneurs, as demonstrated by Carol:

Every year I do a new ... colourway collection ... and the club is a preview of the new collection ... it's limited places. They give me £200 each ... on the expectation that I'm gonna give them something good ... and it seems to be quite popular ... the first year I had 49 people ... and then the third year I had about 80 people apply.

In other words, Carol uses an objectified cultural resource (i.e. a unique colour scheme of yarns) to generate considerable profits (economic capital), increase her reputation (symbolic capital) and attract a growing number of collectors (social capital). Thus, a single cultural good can provide craft entrepreneurs with access to various other forms of capital.

Given that objectified cultural capital is a central component of craft ventures, its conversion is an essential activity which is vital to entrepreneurial success. However, the ability to transform this type of capital can be impeded by the amount of time it takes to create cultural goods, the cost of such valuable, luxury items to customers and the years of practice required to produce quality crafts (see Table 7.6). As such, while converting cultural goods into other forms of capital represents the foundation of craft enterprises, it is a process that involves considerable effort and relies upon the purchasing power of consumers.

Table 7.6. Inhibitors to	Objectified	Cultural (	Capital	Conversions
--------------------------	-------------	------------	---------	-------------

Inhibitor	Representative Quotation
Lack of Time	'The main factor is time it's all hand made and I'm the only person doing it, so there is only so much I can physically produce as one person'. (Carol)
	'There is a lot of potential there for [new] kits, but I haven't got the time there is a whole lot of work to do there, to get that off and up and running'. (Julie)
Product Cost	'It is a luxury you do have to have a certain amount of disposable income to be able to afford to buy hand-made stuff at hand-made prices'. (Betty)
	'Indie hand-dyed stuff is not cheap, it's not a cheap hobby, so to be able to kind of invest time and money in it, you need to have a reasonable income level'. (Carol)
Required Experience	'To produce high quality products it's just years of making I would say years and years of just making and experience really, more than anything else'. (Diana)
	'I don't think I was a natural at it. I just I had to persevere with it I don't think my first stuff was brilliant or that I had an innate skill for it I had to practise'. (Frank)

As shown in Table 7.7, there are several ways in which craft entrepreneurs convert *institutionalised cultural capital* into other resources. Qualifications allow entrepreneurs to charge more for their products and undertaking degrees provides them with access to physical business assets (economic capital). University and college attendance also confers important business knowledge and skills (cultural capital) and facilitates building contacts with wholesalers and suppliers (social capital). Furthermore, the lifelong association with respected institutions allows participants to raise their standing in the community and reinforce their reputation (symbolic capital). Importantly, the ability of craft entrepreneurs to transform their degrees into other resources appears to be rooted in the cultural value that others place upon these qualifications and what they represent (e.g. standards, principles and identity).

# Table 7.7. Conversions of Institutionalised Cultural Capital

Outcome	Representative Quotation
Economic Capital	'The idea is that once I've got a qualification you know, I'll be able to [get] a bit more money'. (Betty)
	'Being at [art school] got me the workshop straight way'. (Diana)
Cultural Capital	'I think any degree makes you quite good at writing short articles it doesn't matter how good my pottery is, if I can't write a good little tweet or article about it'. (Frank)
	'Studying at [university] involved business studies and accounts. I never had anybody else do my accounts, cos I studied it at college, so that helps'. (Emily)
Social Capital	'There's a girl who I lived with when I was at uni. I still keep in touch with her She's a freelance stylist and she's got a lot of contacts in London, which I use'. (Amy)
	'He came and mended the kiln [and] I know him originally through college'. (Frank)
Symbolic Capital	'I kind of bring it up sometimes you kind of mention, you know "I went to [prestigious university]". (Emily)
	'I've been to art school, so if people ask how I got into it, then I'll talk about it'. (Amy)

Despite these benefits, the capacity of participants to convert institutionalised cultural capital can also be restricted (see Table 7.8). For instance, a dearth of degree programmes within a specific craft, the failure to undertake relevant training and variations in course quality can undermine the value of qualifications. Thus, for a degree to sufficiently equip craft entrepreneurs with the cultural capital required to support their acquisition of other forms of capital, it must be relevant to their business activities.

Table 7.8. Inhibitors to Institutionalised Cultural Capital Conversions

Inhibitor	Representative Quotation
Lack of Courses	'There aren't that many [pottery] courses, there are none in Scotland anymore I was going to do one [in Wales] and then I got a letter to say it had been cancelled'. (Frank)
	'There is no degree in dying yarn'. (Carol)
Lack of Training	'It's a bit unfortunate that I don't actually have any qualification to back up what I'm doing, you know it would be good just to have one under my belt'. (Betty)
	'I haven't done any courses, my paperwork is a nightmare I only just scrape by on my tax return'. (Holly)
Course Quality	'I don't think they [teachers] got me at all. They were used to dealing with people, who wanted to set up, I don't know, businesses to sell paper plates'. (Julie)
	'I did learn stuff from it [course], but there wasn't enough practical stuff'. (Frank)

## 7.4. Social Capital Conversions

In addition to transforming economic and cultural capital, participants were found to engage in a range of social capital conversions. As the impacts of family and peers on craft entrepreneurs' practices are the subjects of dedicated discussion (see Chapters 8 and 9), they are omitted from this section. Instead, this section focuses on the ways in which business networks, followers, suppliers and friends enable craft entrepreneurs to access and develop various resources.

Table 7.9 demonstrates how craft entrepreneurs draw on *business networks* to expand their stock of capital. Membership in guilds and networks provides participants with reduced rates for tangible resources and access to funding (economic capital), as well as opportunities to showcase their products (symbolic capital). Network members were also found to impart valuable knowledge and skills to craft entrepreneurs (cultural capital) and act as brokers for relationships with collectors and suppliers (social capital).

Outcome	Representative Quotation
Economic Capital	'They [business network] gave me a deal on the rent it was only £300 a month, when it was supposed to be £390 it's a fantastic, it's a really good deal'. (Carol)
	'[Business network] have funding available through grants and things'. (Betty)
Cultural Capital	'They [business network] have been really helpful with marketing ideas I do get some business support and advice, which has been really helpful'. (Amy)
	'I'm keen to go to as many sessions with them [business network] as I possibly can free workshops on selling, setting up a business plan'. (Betty)
Social Capital	'It was excellent for networking I met [IT supplier] at [business network] so that's how I got that help with my website'. (Isla)
	'[Guild members] were very kind, they gave me customers when I set up'. (George)
Symbolic Capital	'I probably get more publicity because although I don't like my photograph taken I realise it's important for the [business network] so she uses me'. (Diana)
	'The woman that organised it [guild event], she said, "We're organising this exhibition do you want to be involved?" (Isla)

Indeed, the data show that a range of resources can simultaneously become available to craft entrepreneurs through such memberships. As Diana explains,

They [craft guild] paired us with Scottish celebrities ... to make an item ... It's been in Japan, New York, high-profile places ... a lot of free on-going publicity ... one gentleman who saw the collection commissioned a coffee pot ... people have seen the images of [celebrity] with the piece and have gone, 'Oh, you were the girl who made the piece for [celebrity]'.

In this way, becoming a guild member opened up the opportunity for Diana to undertake a high-profile project she would not have managed to attract on her own, which resulted in long-lasting publicity (symbolic capital), payment for her work and subsequent commissions (economic capital).

While membership in business networks can therefore be a valuable means of developing available resources, conversions of this type of social capital were also found to be impeded by both entrepreneurs and the organisations themselves (see Table 7.10). For instance, business networks can suffer from disorganisation and members who lack expertise. Equally, by not setting aside sufficient time for involvement, entrepreneurs can miss out on resulting benefits. Such factors limit the abilities of participants to transform membership into additional resources, demonstrating the importance of selecting high-quality, reputable business networks and investing time into participation.

Table 7.10.	Inhibitors to	Social Capi	tal Conversion	s: Business I	Networks

. . . **.** . . . . . . . .

Inhibitor	Representative Quotation
Disorganisation	'It's difficult, 'cos they [business network] tend to say, "We've got an exhibition in two months' time". Well, I can't work towards an exhibition that quickly'. (Diana)
	'They're [business network] a bit lackadaisical, I would say. Not following through on things that they said they would do'. (Amy)
Lack of Time	'I should use, make use of them [business network] more, I suppose, but I'm just too busy'. (George)
	'One of the business support groups that I'm in I went along, but I haven't done much with them it was too much [I] don't have the time'. (Amy)
Lack of Expertise	'You don't necessarily have to be a brilliant embroiderer to become a member of the [guild] so I don't get much from them'. (Julie)
	'Their website people couldn't help me there is a hell of a difference between doing a business website and doing somebody's creative business website'. (Isla)

*Followers* are another source of social capital that craft entrepreneurs utilise to generate additional resources (see Table 7.11). For example, collectors not only supply a steady stream of orders and commissions (economic capital), but also

encourage the creation of new product ideas (cultural capital). As active members of their respective craft communities, they can widen the networks of entrepreneurs by acting as relationship brokers (social capital). Furthermore, wholesalers and online followers were found to promote the businesses of participants through providing exposure in other regions (symbolic capital).

Table 7.11. Conversions of Social Capital: Followers

Outcome	Representative Quotation
Economic Capital	'I've got a good customer base on the internet, I've got some really nice collectors, who come back to [buy from] me time and time again'. (Holly)
	'I was taking orders for bespoke things, because that's what people wanted me to do, so the way that the business built up was that I was taking orders from [regulars]'. (Emily)
Cultural Capital	'I don't want people who come in quite regularly, I don't want them to get bored of the same styles, so I'm always trying to develop a different range'. (Diana)
	'I get a lot of repeat customers asking for different things and I think, "If somebody wants it, then somebody else will want it", so I make so many different products'. (Amy)
Social Capital	'I get wholesale requests all the time people that buy stuff from me for personal use may also own a yarn shop or their friend owns a yarn shop'. (Carol)
	'A lot of collectors come from word of mouth I'll get a few, say, from Ireland, within a couple weeks, and think, "Oh, well, that person must've told that person". (Holly)
Symbolic Capital	'Next month I'm gonna send stuff to [wholesaler], which is in a town about 15 miles south I probably won't sell lots but it's good exposure'. (George)
	'I have never advertised anywhere people who buy my yarns go on to the forums and say, "You should buy from this person, because her yarn is really nice". (Carol)

The data show that followers can provide access to multiple resources, as illustrated by Emily,

[Collector] came in ... and said, 'Is there any chance that you could make a [new product] for my daughter?' ... So I came up with the design ... and when she came to pick it up, she told me her daughter worked for [national magazine] ... and they put it in the magazine and ... I've made over 100 now, because people are still ordering them ... they are a best seller.

Emily's experience demonstrates how a single repeat customer can generate new product ideas (cultural capital), connections to publishers (social capital) and publicity for craft products (symbolic capital), which results in numerous sales (economic capital). Consequently, the value of a dedicated collector is not limited to direct sales, but encompasses the generation of various forms of capital, indicating the significance of maintaining such relationships.

As important as followers are for obtaining resources, fulfilling their special requests, confronting their sometimes impossible demands and neglecting creative work in favour of satisfying their needs can reduce their utility in developing other forms of capital (see Table 7.12). Thus, it appears that, when cultivating followers, craft entrepreneurs recognise that not all will equally facilitate capital conversions.

Table 7.12. Inhibitors to	Social	Capital Convers	sions: Followers
---------------------------	--------	-----------------	------------------

Inhibitor	Representative Quotation
Special Requests	'Commissions can get a wee bit stiflinglyyou're trying to get what's in somebody's head into your head and it's not gonna look like what's in their head'. (Holly)
	'I have a couple of regulars who will say, "I've thought of this thing you can make" and I'll come up with a design and they'll say "Well, I want this instead". (Emily)
Impossible Demands	• 'Everything I make is one-off I had to tell her that she will get a different variation of it, just by the techniques that I do she was very stubborn about it'. (Diana)
	'She just wanted that [specific] piece [and] it wouldn't have been the same'. (Julie)
Creative Frustration	'It's just what people want and I'm making it for them [If] I didn't have so many commissions I would have more time to do more creative things. (George)
	'It wouldn't bother me if I never had to stich a name on anything ever again'. (Emily)

As shown in Table 7.13, participants draw upon their *suppliers* not just to purchase materials, but also to gain access to a range of resources. Building relationships with suppliers encourages them to provide craft entrepreneurs with free goods and reduced rates on services (economic capital) and to share valuable business knowledge (cultural capital). In addition, suppliers maintain large networks of makers and collectors (social capital), which they can make available to craft entrepreneurs. Friendly relationships can also motivate suppliers to help promote craft ventures by providing them with bespoke materials and augmenting their online visibility (symbolic capital).

Furthermore, the data show that a single supplier relationship can generate multiple forms of capital. For example, Julie shares, 'I told [supplier], "I made something out of the fabric" ... and he said, "Send me an image" ... They put it on their website ... and I had about 1,000 hits and ... got lots of orders and a commission in America'. In this way, Julie's interactions with her supplier prompted him to publicise her products (symbolic capital), which provided her with sales and commissions (economic capital). Accordingly, participants seek to create goodwill in their suppliers so that they can gain advantages from these relationships.

#### Table 7.13. Conversions of Social Capital: Suppliers

Outcome	<b>Representative Quotation</b>
---------	---------------------------------

Economic Capital	'I order from the [fabric supplier] quite regularly and she sends me down [free] samples and bits and pieces which is great, which is really nice'. (Emily)
	'I got a quote from this company and it was £400 or £500 to mend the kiln and then [clay supplier] came and did it and it was £250'. (Frank)
Cultural Capital	'The company that I buy dye from when I first started they were super helpful, sent me out leaflets [on] how to do stuff, you know, what the process was'. (Carol)
	'[Suppliers] know lots about things and they are really enthusiastic and if you build up a relationship with them you kind of get their knowledge'. (Isla)
Social Capital	'I get to know people through [supplier] especially in London when I go down there to get fabric, you see the same faces [and] they're good contacts to have'. (Amy)
	'She [supplier] has a huge network and that's where I got all these orders from'. (Julie)
Symbolic Capital	'[Supplier] looked at the website, optimised it [and] overnight I was on the first page for "willow baskets" and there's millions of websites there [laughs]'. (George)
	'I've got 2, 3 yarns that are from a smaller mill in England and those are spun specifically for me, they don't produce them for anyone else'. (Carol)

Nevertheless, the data analysis showed that there are also instances were these capital conversion processes are unsuccessful. As Diana explains, 'The chap who did the website disappeared on me after doing it ... When I wanted images put online, he wasn't available ... I couldn't get a hold of him'. Thus, the unreliability of suppliers can be an impediment to craft entrepreneurs' abilities to convert social capital, emphasising the significance of building goodwill in order to fully benefit from all the resources suppliers can provide.

While participants spend comparatively little time with *friends* outside of their craft communities, those with whom they do interact appear to be valuable sources of capital for their business operations (see Table 7.14). The data show that friends offer craft entrepreneurs free services (economic capital) and business advice (cultural capital). Some friends can even broker contacts with wholesalers and suppliers (social capital) and provide participants with publicity through advertising their products in local publications and other media (symbolic capital). Accordingly, participants were found to maintain certain friendships for the express purpose of gaining access to the resources they possess.

#### Table 7.14. Conversions of Social Capital: Friends

Outcome	Representative Quotation
Economic Capital	'A friend of ours is just in university doing media studies he's gonna do a cool video for us and that might get us started on YouTube'. (George)
	'My closest friend is like "Just ask me if you ever need a pair of hands" so when I do need an extra pair of hands I've got some really good people'. (Betty)
Cultural Capital	'[Friend] gave me really good advice you can use postage discounts as a way of encouraging people to spend more so I'm trying that out'. (Frank)
	'He [friend] is very supportive he's got a lot of experience with running a business and he's helped me become more commercial'. (Julie)
Social Capital	'She told him [wholesaler], "My friend makes [product] from Harris Tweed," and he said, "Oh, I'd really be interested in speaking to her". (Emily)
	'I have an old friend and she's been looking at suppliers for things and she put me in touch with [supplier], who's turned out to be great'. (Isla)
Symbolic Capital	'I put an advert into [magazine] the editor is a friend and gave me a little bit of editorial, so there's that and a nice photo of one of my mugs in [magazine]'. (Frank)
	'He [friend] product places for television and films he [said] to me, "Send me two or three of your [products] and I'll see where I can get them."" (Emily)

When utilising their friends in social capital conversions, craft entrepreneurs can generate multiple forms of capital at a time. As Emily explains, 'My friend got me a contract ... she is a quality and standards advisor ... I had all these different samples and she would take them to big meetings with her ... [and] they ordered from me ... I'm an approved supplier now'. As such, Emily was able to capitalise upon her friendship to secure sales (economic capital) and gain access to a valuable wholesaler (social capital), without having to pitch the products herself. Importantly, these types of social capital conversion appear easily accomplished, requiring little input from craft entrepreneurs.

Despite the ease with which friendships can be transformed into other forms of capital, the data show that this process is not entirely unconditional. As Betty notes, 'Occasionally somebody will come in and they will be a friend and they will want something quite quickly and I will do it at home at night ... I don't think they even realise'. Thus, when friends are utilised in social capital conversions, they can hold expectations of reciprocity or the entrepreneurs can maintain feelings of indebtedness. This requires investments of time and materials into returning favours, which can lead craft entrepreneurs to refrain from capitalising upon these easily convertible sources of capital.

## 7.5. Symbolic Capital Conversions

Unlike economic, cultural and social capital, which are considered useful in their own right, craft entrepreneurs value symbolic capital almost exclusively for its convertibility. Specifically, participants highlighted the importance of awards, exhibitions, publicity and reputation for generating other forms of capital, illuminating why their conversions are significant for venture growth.

Table 7.15 illustrates how *awards* expand craft entrepreneurs' stock of capital. Entering and winning competitions gains them both one-off commissions and sustained sales (economic capital). Attending awards ceremonies also provides participants with knowledge and further habituates them with the norms in the craft field (cultural capital). In addition, awards are a useful means of attracting new collectors (social capital), whether through providing the opportunity to meet them at competitions or through convincing them of the quality of the craft products. Furthermore, awards help promote the businesses of entrepreneurs by boosting their reputations and providing media exposure (symbolic capital), which is especially beneficial during venture start-up.

Table 7.15.	Conversions	of Symbolic	Capital: Awards

Outcome	Representative Quotation
Economic Capital	'I can certainly see a difference, even in the last couple of years I have been making more profit. I think the awards have definitely helped'. (Julie)
	'There was a sort of competition there for new potters and I didn't win, but somebody ordered two of my little bowls through that, so that was definitely worth doing'. (Frank)
Cultural Capital	'Winning a couple of the awards made me realise, actually, I can do this when I won the big awards it made me realise I can make a living from it'. (Diana)
	'I quite fancy doing it [international awards ceremony] just for the experience'. (Holly)
Social Capital	'There is like a gala evening, after the awards and a dinner and it's all about getting out [there] and meeting collectors and hopefully picking up new collectors'. (Diana)
	'They [awards] are important especially new collectors might say, "Oh, I really like the bear, but is it made well?" so being able to say, "I'm an award winner" helps. (Holly)
Symbolic Capital	'It's also really good publicity for your business The fact that people know I'm nominated is brilliant, absolutely brilliant'. (Betty)
	'I did win a prize once the competition was not really my thing but when I first came to [town] I was looking for a way of sort of promoting myself'. (George)

Awards were also found to enable the accumulation of several forms of capital simultaneously, as exemplified by Isla's experience:

I entered a design prize ... [and won] a travel bursary, so ... I got into the studios in Milan and ... was given letters of introduction to [famous designer] ... that was very useful experience ... I learned a lot.

Winning a single award therefore gained Isla a monetary prize (economic capital), the addition of a high-profile maker to her network (social capital) and the opportunity to learn new skills (cultural capital). Similarly, upon entering a national silversmithing competition, Diana 'won first prize ... and the piece was presented to [museum] ... it's on permanent display, so the accolade ... people took note and ... from one commission came the next commission'. Thus, the benefits of Diana's award included permanent exhibition in a prestigious national museum, which expanded awareness of her talents (symbolic capital) and resulted in sustained commissions (economic capital).

While entrepreneurs can convert awards into various resources, this process was found to be inhibited by both internal and external limitations (see Table 7.16). In addition to there being few competitions for certain crafts, participants were found to impede their own abilities to gain and convert awards by their lack of interest or confidence in doing so. As such, it appears that the convertibility of such symbolic capital is dependent upon the appreciation by craft entrepreneurs of its benefits.

Inhibitor	Representative Quotation
Lack of Competitions	'I have never heard of any competitions for yarn it's not really seen as an art and a lot of the kind of, the creative awards are for kind of more artistic stuff'. (Carol)
	'I haven't found anything [competition] that's a good fit'. (Isla)
Lack of Interest	'I have never put myself forward for anything [awards] I'm not that bothered about things like that'. (Amy)
	'Maybe I should be more bothered about that kind of thing [awards], but I'm really not'. (George)
Lack of Confidence	'I'd never win it, it's, you know, my stuff is fairly solid and basic and just it's nice pottery, but it's not gonna win any prizes'. (Frank)
	'They sent a polite rejection, you know that was it, I just gave up after that'. (Emily)

Table 7.16. Inhibitors to Symbolic Capital Conversions: Awards

The symbolic capital that participants accrue from presenting their products at *exhibitions* is also highly convertible (see Table 7.17). Displaying their crafts at fairs enables entrepreneurs to sell stock (economic capital) and attending shows with fellow makers leads them to learn new skills, develop new product ideas (cultural capital) and grow their networks (social capital). Not only does showcasing products at exclusive shows provide craft entrepreneurs with prestige, but the very act of exhibiting work raises their esteem (symbolic capital).

In addition, exhibitions were found to facilitate the attainment of multiple resources, as illustrated by Amy:

At [show] ... [maker] also had a stall ... we got talking and she said, "Oh, you should get in touch with [head of business network]" ... so I got her email address and she said, "I could support you, business-wise" ... and she was brilliant, just full of ideas ... and they run all these courses.

In other words, a simple conversation with a fellow stall-holder generated Amy's involvement in a business network (social capital) and provided her with useful business advice and knowledge (cultural capital). It thus appears that physical attendance is a crucial facilitator of such symbolic capital conversions, which explains why participants prefer attending shows in person, rather than simply sending products out for exhibition.

Outcome	Representative Quotation
Economic Capital	'By the end of it [show] I had two bears left people were, like, grabbing bears, going, "I'll have this one," and throwing money at me absolutely amazing'. (Holly)
	'I sold about £1,000 worth of stuff at [festival] in three days'. (Frank)
Cultural Capital	'This big festival of basket making there are some makers coming from other countries that will be teaching [I'll] learn different materials, different things'. (George)
	'When I go to a fair or anything I always like to pick people's brains, find out what's new, what's trending'. (Amy)
Social Capital	'[Show] is the biggest in Britain [makers] come from all over the world, so that is a lot of different contacts and new collectors and possibly some new shops'. (Holly)
	'I did a craft fair and then [customer] rung me she said, "I bought a bowl off you and I've got a shop and we like your things", so that was good'. (Frank)
Symbolic Capital	'I make them aware that it is not just you fill out the form and send it off. They [show] have to send the invitation to you to apply the customer really likes that'. (Diana)
	'Exhibiting is a way of showing people that you're serious about what you do'. (Julie)

In spite of the marked benefits of transforming symbolic capital derived from exhibitions, a number of inhibitors to such conversions were discovered. As demonstrated by Table 7.18, the cost involved, the amount of effort required and the variation in exhibit quality can all impede the ability of entrepreneurs to undertake such capital conversions. Consequently, participants were found to evaluate the costs and benefits of exhibition attendance and seek out low-risk, reputable outlets for displaying their work.

**Table 7.18.** Inhibitors to Symbolic Capital Conversions: Exhibitions

Inhibitor	Representative Quotation
Cost	'I had my fingers burned, as far as a few shows are concerned. I wasted quite a lot of money doing the [festival] in 2010 lost about £700'. (Julie)
	'It's such a big outlay it is expensive to go to a show and obviously your stay and everything down there [in London], so you do think, "Is it worth it?"" (Diana)
Effort	'Talking to people and going to places loading the car especially if you have to load it again on the way back, that's the worst thing, if you don't sell much'. (George)
	'It is a lot of work If I've got an exhibition coming up then I'll probably be there till 8, 9 at night, you know, I'll spend long days there'. (Julie)
Devaluation	'At a lot of [fairs], the majority of people are selling jewellery for £5 and they bring the whole thing down to a kind of, it's just crap really'. (Frank)
	'If I pitch myself with my £40 [product] next to somebody else who has made something else and it's a fiver it brings down the value of my work'. (Emily)

Table 7.19 demonstrates that *publicity* can also be easily transformed into other forms of capital. Specifically, the data show that media coverage leads to increased sales (economic capital) and sets expectations which require entrepreneurs to enhance their skills and develop new product ideas (cultural capital). Furthermore, it draws the attention of wholesalers and collectors, adding these contacts to the networks of participants (social capital). Media attention can also generate opportunities for further publicity and product promotion (symbolic capital).

In addition to enabling the accumulation of individual resources, media coverage allows craft entrepreneurs to generate several forms of capital simultaneously. For example, Carol reports that '[Magazine] is London-based, it's really well-known, really classy ... so I'm happy to be featured in it ... [major collectors] found me through that'. Thus, media coverage provides Carol not only with publicity, but it also raises her status through affiliation with a high-class institution (symbolic capital) and attracts collectors from a wealthy demographic (economic and social capital). Accordingly, the data analysis showed that participants' strategic positioning of articles and online features is essential to the success of this capital conversion process.

## Table 7.19. Conversions of Symbolic Capital: Publicity

Outcome	Representative Quotation
Economic Capital	'I was in quite a lot of magazines and lots and lots of blogs, and I think that will have helped, definitely, with the sales'. (Amy)
	'By putting the article about the [product] in the magazine, I've sold a lot'. (Betty)
Cultural Capital	'They put it in the magazine [and] there was this huge flurry [so] I had to figure out a way to make [product] much quicker'. (Emily)
	'Because it's a stitch magazine, I knew that they were looking for unusual stitches, so I made them specifically different and people, you know, were drawn to that'. (Julie)
Social Capital	'I got into [magazine] and through that [wholesaler] ordered my [product] and they have gone really well immediately, so she has ordered some more'. (Isla)
	'When I was featured in [magazine] I did get some people who said "I saw this and I really wanted to come". (Carol)
Symbolic Capital	'The American magazine had done a feature on me and they [British magazine] saw that so that's how that got lined up'. (Holly)
	'I've done articles for magazines which has publicised the book it's also obviously got my name out there'. (Julie)

The ability of entrepreneurs to conduct such transformations was found to be hindered by several factors (see Table 7.20). Media attention that is of high enough quality to facilitate capital conversions can be costly to attain. Furthermore, available media outlets may be substandard and promised opportunities for publicity can fall through. Importantly, media attention also carries a degree of risk, as published stories may not find approval with target audiences, potentially driving customers away. Consequently, participants are careful in selecting outlets in which to be featured.

<b>Table 7.20</b>	Inhibitors to	Symbolic	Capital C	onversions:	Publicity
-------------------	---------------	----------	-----------	-------------	-----------

Inhibitor	Representative Quotation
Cost	'I haven't actively tried to get it into any publication, primarily because it's hugely expensive'. (Isla)
	'To me to be in a magazine seems to be quite a big investment I don't think I would want to do it'. (Betty)
Limited Benefit	'I kind of have to assess whether or not it would be beneficial to me 'cos there is some, I suppose, some publications that aren't as well thought of'. (Carol)
	'We do have kind of trade magazines I can't think of their names at the moment yeah not necessarily magazines that you buy'. (Diana)
Lost Opportunities	'There was a television programme somebody from the BBC said he was gonna come that would have been quite an exposure, but it didn't happen'. (George)
	'A lot of people say a lot of things about [promotion opportunities] to me and 90%, 99% of the time, nothing comes of it'. (Emily)

Finally, *reputation* was found to be highly convertible into other forms of capital (see Table 7.21). It is vital for convincing potential customers of the quality of products and for generating sales (economic capital). Reputation also provides craft entrepreneurs with access to training opportunities and expert advice (cultural capital) and helps attract wholesalers and collectors (social capital). Furthermore, the data show that reputation can raise the esteem of entrepreneurs within their craft communities and facilitate their procurement of media coverage (symbolic capital).

## Table 7.21. Conversions of Symbolic Capital: Reputation

Outcome	Representative Quotation
Economic Capital	'They can buy with confidence because they know I'm not just gonna disappear tomorrow there is some reputation behind the business'. (Holly)
	'I have noticed a lot of people buying stuff that haven't bought stuff off me before which means that my reputation is spreading'. (Carol)
Cultural Capital	'All these [famous] projects kind of brought me into their attention I was very fortunate to spend a couple of days with him [expert] giving me a bit of tutoring'. (Diana)
	'She had heard about me [so] I could ask her for advice on how to do it'. (Amy)
Social Capital	'16 years I have been doing this and you know I'm quite well known shops come to me, people come to me'. (Holly)
	'The people who are buying from me, because they know that my product is well received keep coming back, and so I have more repeat custom now'. (Emily)
Symbolic Capital	'Although that reputation is quite hard to live up to, it's a good place to be, because people appreciate you'. (Julie)
	'Because I've been going at it for so long, if I want anything in the magazines I phone them up and say, "Can I have this in please," and they usually do it'. (Holly)

Reputation is one of the most significant sources of symbolic capital for craft entrepreneurs and participants consider it essential to business growth. Accordingly, they were observed expending considerable efforts to earn and maintain a good reputation, both online in their presentation of work and communication with craft community members and in person at shows and exhibitions. The extent to which craft entrepreneurs value reputation indicates their appreciation for the myriad resources it makes available to them.

Despite the importance of reputation to craft entrepreneurs, the data show that its conversion into other forms of capital can be impeded. As Table 7.22 illustrates, entrepreneurs themselves can inhibit the conversion process. At times, participants can lack confidence in their reputation, preventing them from drawing upon it to acquire additional resources. Furthermore, in spite of craft entrepreneurs' considerable efforts, customers can maintain impossible expectations, project unrelated aggravation upon entrepreneurs or simply be ignorant of the consequences of their online reviews, all of which reduce the convertibility of reputation. It therefore appears that, as crucial as craft entrepreneurs believe reputation to be, it is a comparatively fickle source of symbolic capital to rely upon for needed resources.

Table 7.22. Inhibitors to Symbolic	Capital Conversions: Reputation
------------------------------------	---------------------------------

Inhibitor	Representative Quotation	
Lack of Confidence	'If I had a big reputation and was well established, then I think that would be, you know that would be something else'. (Betty)	
	'Reputation is something that worries me, 'cos I think people will think, "Oh that's rubbish" I am always worrying about, "Oh, is this good enough?"' (Frank)	
Negative Feedback	'Reputation is everything, everything [and] it would only take one negative review to mess it up'. (Betty)	
	'I'd rather give the money back than have even neutral feedback, it's just a disaster she probably didn't realise how damaging it was'. (Frank)	
Difficult Customers	'It just takes one person who's having a bad day [at a workshop] You have to work really hard to stop them from having a bad experience'. (Carol)	
	'The woman said, "The colour wasn't as bright as in the photo" I mean, the colour won't necessarily be exactly the same as on the computer monitor'. (Amy)	

#### 7.6. Analysis and Interpretation

The findings presented in this chapter show that craft entrepreneurs engage in a multitude of capital conversion processes in order to expand their stock of capital. Indeed, it appears that the maximisation of capital represents a main goal that craft entrepreneurs pursue to develop sustainable businesses. Given that such practices are common in capitalist societies (Calás et al., 2009; Schumpeter, 1942) and usually inculcated in children through their upbringing in such contexts, they represent personal dispositions (i.e. micro-level habitus) that affect the behaviours of entrepreneurs over time. While prior research has acknowledged the convertibility of capital (Shaw et al., 2009; Vershinina et al., 2011), empirical studies of this entrepreneurial practice are rare (Scott, 2012). This study thus advances knowledge in a number of ways. First, it provides a more comprehensive account of capital transformations than prior research by distinguishing between different types of cultural, social and symbolic capital. Consequently, this study not only identifies a wider range of capital conversions, but also accounts for transformations which result in the accumulation of different variations of the same form of capital (e.g. conversions of objectified into embodied cultural capital). Second, this chapter presents a range of conversions which lead to the development of multiple forms of capital simultaneously. Third, this study reveals various inhibitors to capital conversion, which have been overlooked in the extant literature. Following the structure of the above presentation of findings, this section compares the economic, cultural, social and symbolic capital conversions conducted by craft entrepreneurs with those discussed in the literature.

#### 7.6.1. The Convertibility of Economic Capital

Conversions of economic capital into other forms of capital are largely taken-forgranted (Drakopoulou Dodd et al., 2016) and have thus received scant empirical attention within the entrepreneurship literature. The findings of this study therefore advance understanding by demonstrating how craft entrepreneurs employ their economic capital to increase their stock of resources. Rather than securing loans and accumulating debts (Riding and Haines, 2001), participants were found to rely on the profits from their business operations to acquire needed materials (Hoegl et al., 2008) and other tangible business assets (Alsos and Carter, 2006), which suggests that they engage in a form of bricolage (Baker and Nelson, 2005; Stenholm and Renko, 2016). Craft entrepreneurs also pay to be featured in prestigious media outlets in order to raise and maintain their standing in the community (Reuber and Fischer, 2009). In addition, they regularly invest into further education and business network memberships to develop their cultural and social capital, corroborating the findings of extant research (Harvey et al., 2011a; Shaw et al., 2013). Whereas prior studies identify time and effort as the inhibitors to such transformation processes (Greve and Salaff, 2003; Jayawarna et al., 2014), this study highlights that associated costs and potential risks can also prevent entrepreneurs from engaging in these conversions. Although participants do not necessarily lack economic capital, they are still purposeful and deliberate in choosing how to best invest their economic resources (Grichnik et al., 2014; Winborg and Landström, 2001).

A surprising and significant finding is that craft entrepreneurs appear to give no primacy to economic capital when converting the various forms of capital at their disposal. Rather, its transformations were found to be only one element of a much larger process of capital conversion. Participants discussed a multitude of ways in which they employ and convert their available resources and it appears that economic capital does not play a dominant role in this process. As such, the findings of this study contradict Bourdieu's (1986: 252) proposition that economic capital is the underlying 'currency of exchange', which is 'at the root of all the other types of capital'. The enduring focus on the importance of economic capital is, in fact, likely to obscure transformation processes which do not involve economic exchanges at all. Based on the experiences of craft entrepreneurs, this study identifies a large number of such conversions and highlights their potential for supporting venture development and growth.

The interest in economic capital and its significance is, of course, not unique to Bourdieu; it also prevails in much of the entrepreneurship literature (Rosenbusch et al., 2013; Shane and Cable, 2002). Indeed, it has been suggested that economic capital is the most important asset for entrepreneurs (Allison et al., 2015; De Bettignies, 2008) and research has expended considerable efforts to examine the

various ways in which entrepreneurs can acquire different types of financing (Huang and Knight, 2017; Uzzi, 1999). Given the growing number of studies which demonstrate the influence of alternative forms of capital on entrepreneurship (Bhagavatula et al., 2010; Davidsson and Honig, 2003; Grichnik et al., 2014), this fixation with economic capital does not appear warranted. It could be argued that, by focusing on the financial stakes of the 'economic game', entrepreneurship scholars have bought into the illusio of market rhetoric. It is time to question these taken-forgranted assumptions (doxa) regarding the primacy of economic capital (Amit et al., 2001; Rindova et al., 2009) and account for the impact of different contexts on the valuation patterns of various capitals (Golsorkhi et al., 2009; Özbilgin and Tatli, 2005).

Particularly within the cultural industries, economic capital may play a secondary role (Scott, 2012). While achieving sales is a core business function, this study shows that transformations to and from economic capital represent only a portion of the conversions conducted by craft entrepreneurs. Thus, the rules of the game (habitus) in this context appear to favour commitment to alternative stakes (illusio), such as cultural, social and symbolic capital. It is therefore essential to consider habitus and illusio when exploring the capital transformations and exchanges conducted in different fields (De Clercq and Voronov, 2011; Vincent, 2016). By maintaining an economic-centred approach, scholars risk importing a neoclassical, market-driven perspective into contexts which may have different priorities and patterns of practice.

## 7.6.2. The Convertibility of Cultural Capital

The findings of this study confirm the importance of cultural capital for entrepreneurs who seek to start and develop firms in the cultural industries (Drakopoulou Dodd, 2014; Jones et al., 2016; Randle et al., 2015). Specifically, the data show that cultural capital in its various guises is crucial to the operations of craft entrepreneurs and highly convertible into a number of resources. Drawing on the concept of human capital (Becker, 1964), entrepreneurship scholars have shown that knowledge (Markman and Baron, 2003), skills (Chell, 2013) and abilities

(Jayawarna et al., 2014) significantly influence entrepreneurial success. This study contributes to these scholarly debates by demonstrating that craft entrepreneurs transform such embodied cultural capital into economic (Brinckmann et al., 2011; Haber and Reichel, 2007), cultural (Scott, 2012), social (Bhagavatula et al., 2010; Mosey and Wright, 2007) and symbolic capital (Beverland, 2005; Clarke, 2011). As such, this research also demonstrates the explicit role that entrepreneurs' temporal contexts – their past lived experiences, knowledge gained and skills obtained – play in their everyday lives and entrepreneurial practices.

In addition, this study extends knowledge by highlighting that, in the context of craft entrepreneurship, objectified cultural capital plays a similarly important role. To date, even studies that do employ Bourdieu's multifaceted perspective of cultural capital tend to emphasise the value of embodied cultural capital (Karataş-Özkan, 2011; Patel and Conklin, 2009). For example, while acknowledging the existence of objectified and institutionalised cultural capital, Lehner (2014) focuses almost exclusively on the impact of knowledge and understanding on venture sustainability and success. In contrast, the findings of this study show that objectified cultural capital, in the form of cultural goods, represents the very basis of venture activity for craft entrepreneurs. It is not only convertible into sales, but also facilitates the acquisition of knowledge, social connections and awards. Furthermore, cultural goods are a vital aspect of craft entrepreneurs' personal histories and experiences: they form the tradition of their craft from which new, innovative goods can be developed (De Massis et al., 2016). Thus, cultural goods are significantly more transmutable than extant research suggests (Barlow et al., In Press).

Furthermore, craft entrepreneurs were found to employ their institutionalised cultural capital to develop their stock of resources. In line with the findings of prior research (Jayawarna et al., 2014; Shaw et al., 2009), the data show that craft entrepreneurs consider their academic qualifications and further education to be valuable. However, participants emphasise that formal schooling in itself is not sufficient; it needs to be related to their craft and business responsibilities in order to provide substantial benefits. Thus, rather than acquiring broad functional experience (Kacperczyk and Younkin, In Press) and becoming 'jacks-of-all-trades' (Lazear, 2004: 208) to perform a range of tasks independently (Åstebro et al., 2011),

participants were found to acquire task-specific knowledge and qualifications (Estrin et al., 2016). Importantly, while transforming objectified cultural capital usually involves the release of the resource and its loss to the entrepreneur (similar to employing financial assets to purchase memberships or education), conversions of embodied and institutionalised cultural capital do not deplete the stock of existing resources. As such, investments into either type of cultural capital can provide exponential and sustained benefits to craft entrepreneurs, further reinforcing their value in this context.

The findings show that cultural capital can be transformed into multiple resources concurrently. Among others, participants were found to turn their pitching and proposal writing skills not only into economic capital (Chen et al., 2009; Foo et al., 2005), but also into new connections with wholesalers and access to prestigious museums. Similarly, objectified cultural capital was shown to be convertible into income, collectors and reputation at the same time. These practices suggest that transforming cultural capital is a more organic and multifaceted process than might be assumed based on prior scholarship (DiMaggio, 1979; Nicolopoulou, 2014).

Despite the apparent ease with which participants engage in cultural capital conversions, the findings show that there are also a number of factors that can impede this process. For example, lapses in memory (Anheier et al., 1995) and IT ineptitude (Martinez Dy et al., 2017) can inhibit conversions of embodied cultural capital. Furthermore, high product costs and a lack of time or experience were found to limit the abilities of craft entrepreneurs to produce and sell cultural goods. Similarly, a dearth of degree programmes, failure to undertake relevant training and variations in course quality impede conversions of institutionalised cultural capital. In comparison, this study does not find evidence of academic qualifications being seen as irrelevant or disparaged by field members, as suggested in the literature (Vershinina et al., 2011). Neither does a lack of cultural understanding or sophistication affect participants' abilities to transform their available cultural resources (De Clercq and Voronov, 2009a; Perry-Rivers, 2016). Craft entrepreneurs consider inhibitors to cultural capital conversions as largely inherent and unavoidable, which could explain why prior research has paid little attention to them.

## 7.6.3. The Convertibility of Social Capital

This study provides substantial evidence of the importance of network ties to the accumulation of various resources (Ozdemir et al., 2016; Semrau and Werner, 2014; Sullivan and Ford, 2014). Although a number of studies discuss the positive effects of social capital on entrepreneurs (Cope et al., 2007; Gedajlovic et al., 2013; Stam et al., 2014), it has been noted that the processes required to access resources within networks differ between contexts (Batjargal, 2010; Lee and Jones, 2008; Zhang, 2015). Thus, the findings of this study contribute to scholarly discussions by demonstrating the various social capital conversions conducted by entrepreneurs in the cultural industries (Antcliff et al., 2007; Jones et al., 2016).

First, the data show that membership in guilds and business networks grants craft entrepreneurs access to financial resources (Brüderl and Preisendörfer, 1998), knowledge (Davidsson and Honig, 2003), customer contacts (Casson and Della Giusta, 2007) and opportunities to showcase their products. Second, followers were found to provide participants not only with income (Shepherd and Zacharakis, 2003) and knowledge (Yli-Renko et al., 2001), but also new contacts and publicity. In addition, their needs can compel entrepreneurs to develop new patterns and product ideas (Lam and Harker, 2015; Marvel, 2013). Third, friends represent rich sources of support (Bullough et al., In Press) and advice (Hmieleski et al., 2015) and were found to broker business connections (Aldrich and Kim, 2007) and enhance the reputations of entrepreneurs (Jack, 2005). While many of these capital conversions have been individually observed by prior research, few studies have considered the wide applicability and convertibility of social capital (Scott, 2012). Furthermore, little attention has been afforded to suppliers and their influence on entrepreneurs (Le and Nguyen, 2009). Thus, the data expand existing knowledge by highlighting that relationships with suppliers can provide craft entrepreneurs with access to free goods, contacts to customers and peers, market knowledge and highly prized, bespoke materials.

The findings also offer insights into conversion processes that result in the accumulation of multiple forms of capital. For example, business network memberships were shown to provide participants with opportunities to create high-profile products, which are linked to celebrity endorsements (Stuart et al., 1999),

large remunerations and subsequent commissions. Similarly, interactions with collectors were found to not only produce sales (Dalpiaz et al., 2016), but also facilitate the development of cultural, social and symbolic capital (Crossland and Smith, 2002; Ertug et al., 2016). Given that these types of network connections represent weak ties (Granovetter, 1973), the findings of this study indicate that high intimacy and emotional investments are not essential for craft entrepreneurs to derive substantial benefits from their social capital (Aldrich and Zimmer, 1986; Vardaman et al., 2012). However, the findings also show that even these weak ties are founded through shared experiences and histories, and are thus rooted in entrepreneurs' temporal contexts. While extant research suggests that entrepreneurs tend to develop arms-length ties into relationally embedded ties (Hite, 2005; Larson and Starr, 1993; Steier and Greenwood, 2000), the data show that craft entrepreneurs usually seek to maintain their weak ties over time, as they help them gain access to a broad range of resources (Uzzi, 1997) and potential opportunities. Indeed, the findings demonstrate that, regardless of the strength of ties connecting network members, most social capital conversions are easily accomplished and require less input from craft entrepreneurs than might be assumed based on prior research (Becerra et al., 2008; McAdam et al., 2014).

Despite the relative ease with which participants transform social capital into other forms, the data show that there are also factors which can impede these processes. To date, scholars have argued that maintaining ties with significant conversion potential is demanding (Nahapiet and Ghoshal, 1998; Newbert et al., 2013) and that developing knowledge through relational learning can be timeconsuming (Hughes et al., 2014; Yli-Renko et al., 2001). The findings contribute to these discussions by highlighting that the quality and reliability of business networks affects entrepreneurs' abilities to accrue benefits from network memberships. Furthermore, high expectations of both collectors and friends can compel craft entrepreneurs to prioritise their needs (Coleman, 1988; Kim et al., 2013), which can lead them to neglect their creative practice and therefore negatively affect their abilities to develop cultural and economic capital. Consequently, entrepreneurs sometimes refrain from capitalising upon these easily convertible sources of capital because they recognise that not all connections equally facilitate capital transformations. Nevertheless, participants generally display a positive attitude towards social capital conversions, which is in line with the findings of prior research (Semrau and Werner, 2014; Sullivan and Ford, 2014).

## 7.6.4. The Convertibility of Symbolic Capital

The data show that symbolic capital is a particularly mutable resource, which is readily transferable into economic, cultural, social and symbolic capital. Entrepreneurship research has demonstrated that possession of symbolic capital can facilitate various significant activities (Fuller and Tian, 2006; Scott, 2012), such as attracting customers (Clarke, 2011; Reuber and Fischer, 2009) and gaining access to financing (Dimov et al., 2007; Shane and Cable, 2002). However, the extant literature on related concepts, such as reputation (Reuber and Fischer, 2005), status (Packalen, 2007) and prestige (Pollock et al., 2010), is largely disjointed (Bitektine, 2011) and has paid scant attention to the management of symbolic resources (Zott and Huy, 2007). This study therefore contributes to contemporary debates by reconciling these fragmented discussions and advancing knowledge of the symbolic capital conversions conducted by entrepreneurs.

First, the data highlight the value of awards, which help generate sales, attract new collectors, boost reputation (Partanen and Goel, 2017) and facilitate habituation with field-specific norms and conventions (habitus) (Anand and Watson, 2004). In comparison, entrepreneurship research has largely overlooked the importance of awards and their transmutability into other resources (Zhao et al., 2013), apart from their reputation-building capacity (Reuber and Fischer, 2007; Svejenova et al., 2007). Second, the findings show that exhibitions are highly convertible into income, contacts (Hjorth, 2005), esteem and opportunities for learning. While the positive effects of presentations at exhibitions have long been recognised by management research (Alexander, 1996; Le Breton-Miller and Miller, 2015), they have received little consideration within the entrepreneurship literature (Hjorth and Holt, 2016).<sup>10</sup> Third, the data show that media coverage is an effective means of developing not

<sup>&</sup>lt;sup>10</sup> This lack of attention could be explained by the 'excessive and exclusive interest in high growth ... businesses in entrepreneurship' (Welter et al., 2017: 312), which are less likely to present their work at exhibitions.

only economic and social capital (Lehner, 2014; Lewis, 2013) but also cultural and symbolic capital. As extant research has focused on entrepreneurs' abilities to employ publicity to gain legitimacy (Fisher et al., 2016; Navis and Glynn, 2010; Überbacher, 2014) and only recently begun to examine the effects of social media attention on entrepreneurs (Fischer and Reuber, 2014; Nambisan, In Press), the findings of this study offer new insights into the convertibility of this type of symbolic capital.

In contrast, prior research provides abundant evidence of the value of reputation to entrepreneurs (Dimov et al., 2007; Milanov and Shepherd, 2013; Reuber and Fischer, 2005) and its transmutability into economic (Abeysekera, In Press), cultural (McLeod et al., 2009) and social capital (Fuller and Tian, 2006). The scarcity of research into alternative types of symbolic capital, such as awards and media attention, could be explained by scholars' prevailing interest in nascent and early stage entrepreneurs (cf. Davidsson and Honig, 2003; Dimov, 2010; Reynolds et al., 2004), who are less likely to have accumulated such recognition. As participants ranged in their level of experience from new entrants to established entrepreneurs, this study was able to present a more comprehensive picture and account for the accumulation and transformation of symbolic resources in multiple temporal contexts. Furthermore, as the value of symbolic capital differs highly between contexts (De Clercq and Voronov, 2009c; Shaw et al., 2017) and has been shown to matter greatly in the cultural industries (Glynn and Lounsbury, 2005; Svejenova, 2005), the focus on craft businesses facilitated the exploration of these under-studied resources. In doing so, it revealed a cause for symbolic capital's context-specific nature: the worth of awards, exhibitions, publicity and reputation are dependent upon collective notions of value, which are created through shared experiences and traditions. Symbolic capital is thus a temporally, as well as socially and spatially, specific product.

Within this context, the data demonstrate that symbolic capital can be converted into multiple forms of capital simultaneously. For example, individual awards were found to not only carry cash prizes (McLeod et al., 2011), but also provide opportunities for learning (Karataş-Özkan, 2011) and for exhibiting work in prestigious venues. Similarly, publicity was shown to generate economic, social and symbolic capital at the same time (Ertug et al., 2016; Svejenova et al., 2007). Although these findings highlight the convertibility of symbolic capital, the data demonstrate that these processes can also be inhibited by several factors. This study agrees with prior research (Kovács and Sharkey, 2014), which suggests that winning awards can raise the expectations of customers and increase their critical assessments of business performance. The findings also show that a lack of relevant competitions, as well as a lack of confidence, can discourage award submissions and prevent access to resulting benefits. Furthermore, not only can the effort required to gain admittance into exhibitions be prohibitive, but the variations in exhibit quality can also limit their value and utility. Thus, the data support the view that entrepreneurs are affected by the mistakes of other makers in the same category (Zhao et al., 2013). The findings also show that customers' unrealistic expectations and personal aggravations can lead to negative online reviews, which can impede the abilities of entrepreneurs to draw on their reputation to gain access to other resources (Reuber and Fischer, 2011). The gravity of the repercussions of others' actions upon entrepreneurs further attests to the role of context in influencing entrepreneurial activities.

## 7.7. Conclusion

This chapter set out to answer the research question: *How do entrepreneurs convert their economic, cultural, social and symbolic capital?* It provided abundant evidence of the various capital conversion processes performed by craft entrepreneurs and highlighted their positive effects on entrepreneurial activities. In so doing, this study identified a range of transformations which have not previously been observed in the academic literature. This chapter also revealed that craft entrepreneurs give no primacy to economic capital when converting the various forms of capital at their disposal. Instead, transformations of economic capital were found to be only one part of a much larger process of capital conversion. Indeed, the findings demonstrated the high convertibility of cultural, social and symbolic capital and their significance to entrepreneurs in the cultural industries. Importantly, the data showed that previously under-explored manifestations of these capitals, such as objectified and institutionalised cultural capital, are as, if not more, convertible than the resources

commonly investigated by the extant literature (Bhagavatula et al., 2010; Davidsson and Honig, 2003). However, this study also highlighted that valuation patterns and commitments to the stakes at play (illusio) are field-specific, which suggests that empirical results cannot readily be transferred between contexts. Furthermore, the data showed that conversions are not only direct, from one form of capital into another, but also result in the accumulation of multiple forms of capital. Finally, this chapter revealed a range of inhibitors to capital conversion processes, which have not previously been considered by entrepreneurship research.

### **CHAPTER 8: HOUSEHOLD CONTEXT - MESO-LEVEL**

#### 8.1. Introduction

Following discussion of the capital conversion practices of craft entrepreneurs (micro-level), this chapter explores the effects of household contexts on participants (meso-level) before the next chapter investigates their embeddedness in community contexts (macro-level). This chapter seeks to shed new light on the interwoven relationship between work and family (Eddleston and Powell, 2012; Jennings and McDougald, 2007) by addressing the research question: *How do household members impact the practices of entrepreneurs*?

This chapter is structured as follows. First, it describes the household contexts in which participants are embedded. Second, it discusses ways in which household members support craft entrepreneurs through the provision of various forms of capital. Specifically, this study provides rich examples of how participants develop their economic, cultural, social and symbolic capital with the help of cohabitants.<sup>11</sup> Third, this chapter highlights the importance of the emotional support that household members provide. Drawing on the concept of psychological capital (see Chapter 4), this study explains how cohabitants can help craft entrepreneurs cope with stress and respond to challenging circumstances by boosting their self-efficacy, hope, optimism and resilience. Fourth, this chapter reveals ways in which household members impede entrepreneurial practices. As such, this study demonstrates that the lines between private and business spheres are often blurred (Aldrich and Cliff, 2003; Dyer, 2003) and that household members both facilitate and inhibit entrepreneurial processes (Alsos et al., 2014).

It should be noted that this chapter not only presents data gathered from participants, but also from their cohabitants to better explain their impacts on entrepreneurial practices (see Chapter 5). Excerpts from interviews with household members are presented in cursive to highlight that their perspectives are being presented, rather than those of the entrepreneurs. However, given that this study

<sup>&</sup>lt;sup>11</sup> Although cohabitants also participated in this study (see Chapter 5), the term 'participant' is only used in reference to craft entrepreneurs to avoid any misunderstandings.

focuses on the experiences of craft entrepreneurs, it primarily draws upon and analyses their accounts.

## 8.2. Household Characteristics

Before exploring the ways in which cohabitants impact the practices of craft entrepreneurs, this section describes the characteristics of participants' households. Specifically, it presents key attributes and discusses the involvement of household members in the craft field.

As explained in Chapter 5, participants were not selected based on the composition of their households. During the data analysis process (following the first round of interviews) it emerged that cohabitants significantly affect the practices of craft entrepreneurs and therefore warrant further investigation. The households portrayed in this study are comparatively similar. Most participants share their households with spouses or partners (apart from Amy and Frank, who broke up with their respective partners either during or immediately prior to the start of the study) and half of them have one or two children, most of whom still live at home. Thus, participants describe their households in similar ways, as 'a very, very small family' (Isla) or as 'only Henry [partner] and the boys ... that's kind of it really' (Holly).

Since most participants made the conscious decision to move into a rural area to enjoy the natural surroundings (see Chapter 2), they do not reside in close proximity to their wider families. As Frank notes, 'only my mother has actually visited up here of my family', and as George explains, 'in terms of other family, they have very little influence, I mean they all live down in England and [the Baltics]'. Still, participants do contact wider family members, mainly via telephone. Frank reports that, 'occasionally, I speak to them [parents], 'cos it is a bit isolated otherwise'. Furthermore, they visit one another, as Diana describes, 'during the summer, my parents, they are retired, they will come down and visit'. In comparison, cohabitants have a significantly bigger and more immediate impact on participants, which is why this study focuses on their influences on the practices of craft entrepreneurs. Participants reside in close proximity to their studios (less than a 10-minute drive) in small houses or flats. The researcher was invited to visit several of these residences, mainly towards the end of the study once the relationships with participants had been developed. These properties might best be described as modest buildings with distinctive interiors. In Diana's words,

if you go into a maker's home, it's always different to somebody who works in an office ... everything is just a bit more quirky or they don't follow fashion ... they figured out their own individual mind and their houses just reflect that.

Participants 'live quite a frugal lifestyle' (Amy) and often bring work home with them, as described by Emily: 'every night I'm stitching, you know, I bring my business home with me'. Consequently, as Carol explains,

it is quite messy ... if she [partner] is coming home in the winter and going, 'Oh, I'm freezing and I really would love to soak in the bath', I'm thinking 'No you can't', because there is yarn drying there [laughs].

The households of participants all rely on more than one income (while Julie's spouse does not work, he receives a small pension), which entails that cohabitants primarily connect in the evenings after work hours and during the weekends. According to Betty, 'I definitely try to have Sunday off'. The children of participants are all in their late teens or early twenties and, as such, 'they are fairly independent' (George). As Holly describes, 'because they are older, they're 18 and 16 ... I felt I could take this place [studio] and get me out the house ... so now they don't really affect me'. Emily agrees that 'the reason I took the studio in the first place was because I could be away from the house ... [my daughter] is hardly at home anymore'.

Table 8.1 demonstrates that cohabitants can be differentiated by their degree of involvement in the craft field and their own creativity. First, household members who consider themselves to be creative were also found to actively engage with the work of participants. For example, in Holly's case, the researcher was able to observe Henry (partner) prepare props for teddy bear displays and pre-cut fur in the lead-up to a major craft show. Second, there are cohabitants who do not consider themselves creative, but who show an interest in the craft field. As Grace (spouse) recounts, 'I've made some baskets, but ... I don't think they were any good'. They still get involved and help the craft entrepreneurs, but they do not usually engage in creative activities. As Evan (spouse) admits, 'as somebody who is not really creative and who doesn't stitch ... I don't particularly have an eye for colour and things like that'. The majority of partners and spouses fall into this category. Third, some cohabitants were found to be non-creative and not involved in the craft field. As Clara (partner) notes, 'She probably wishes sometimes that help would be a bit more forthcoming, but in being totally honest ... I find it hard to get over excited about what she does ... but I do support her, that's only natural'. In other words, these household members still support participants, but do not show much interest in crafts. Interestingly, none of the participating craft entrepreneurs have cohabitants who consider themselves creative but are not interested in the craft field.

Table 8.1. Cohabitant	s' Involvement in the	Craft Field and Perceived	Creativity
-----------------------	-----------------------	---------------------------	------------

Involvement	Representative Quotation
Involved and Creative	'My other half [Henry], well he's a scenic artist, so he can paint on 32 foot by 16 foot backdrops. He also makes props and things like that for me'. (Holly)
	'I would see myself as a creative person in the art side of thing. As I said, I used to make bags and used to do some designs on it'. (Adam, partner)
Involved, but Non-Creative	'She always says that she is not creative, but she definitely does, like, appreciate my hand made things'. (Betty)
	'I have respect for creative people and for what they do'. (Evan, spouse)
Uninvolved and Non-Creative	'Daniel [partner] isn't really involved and doesn't really go to any of the shows, he doesn't really see anybody else's work. Sometimes I come home with things and he looks at it and goes, "Why have we got this?" but then he gets used to it'. (Diana)
	'It's not a world that she [partner] has ever really been interested in. She is just not into crafts at all'. (Carol)

Participants are also highly aware of their children's creativity and involve them in craft production. Craft entrepreneurs actively encourage their children to develop their creativity and describe them in ways similar to Julie: 'Both my children are very good artists as well'. While none of their children have taken up their crafts as a profession, participants are generally open to this idea. George explains that 'my son is a little bit interested ... He's not got much staying power at the moment. I don't know whether that's just his age ... [but] he is quite artistic'. Accordingly, participants are proud of the creative work of their children. As Isla notes, 'as a mother with a daughter who is ... working in the fashion industry ... I'm very pleased to report she has been exceedingly successful'. As discussed below, children support craft entrepreneurs in a variety of ways, including by attending craft shows and contributing creative ideas.

Importantly, the perspectives of household members were also found to change over time. For example, Daniel (partner), who was observed at the start of the study actively avoiding entering Diana's studio during a special display of her work, suddenly developed an interest in the silversmithing community. As Diana explains, 'it's only in the last couple of months that he has gone [to shows] and actually is really kind of understanding why it's important for me to go to these events and he really enjoys them'. Similarly, Brenda (partner) admitted that her perspective changed, due to her relationship with Betty:

I'm not a very creative person and I'm quite sort of logical ... so when she talks about things, I could quite ... easily kind of think, 'Oh well, that just sounds like crazy creative things, like, I don't know anything about that' ... which might be the way I might have reacted in the past, but I think ... I have changed and I appreciate her work a lot more.

Concerning other household members, Frank explains that 'Felix [cohabitant, non-family] is always interested in what I'm doing. He came to the workshop the other day ... to see my new range of mugs'. Similarly, Amy, who moved back in with her family after breaking up with her partner, describes her parents as 'just very supportive in any way'. Thus, while cohabitants may be 'always wearing suits and things' (Amy), they are still found to encourage the craft work of entrepreneurs. Furthermore, regardless of whether cohabitants are involved in crafts or not, there appears to be a feeling of joint responsibility for the success of the ventures of craft entrepreneurs. As Jack [spouse] explains, *'it's very much, you know, her leading it, but I think, yeah, we feel very much part of it'*. Accordingly, participants are grateful for the support their household members provide. As Holly notes, 'Henry [partner] is a big help, I must admit, he really is a big help – grumpy, but a big help [laughs]'.
# 8.3. Provision of Capital

Craft entrepreneurs in this study benefit from the support of their household and family members in a variety of ways. Adopting Bourdieu's (1986) approach, this section differentiates between the economic, cultural, social and symbolic capital that participants develop with the help of cohabitants.

The data show that cohabitants facilitate the development of *economic capital* of craft entrepreneurs. Table 8.2 presents an overview of common types of economic capital support that household members provide. First, a number of participants were gifted financial assets during the venture start-up phase, which they valued as this helped them maintain their independence. As Julie explains, 'I would never take a loan, I would never spend money that I didn't have'. Similarly, Carol declares that 'I would sooner be putting that money out there myself rather than getting a loan ... and running up debts'. Importantly, participants were found to reject financial business support post start-up, mainly because they want to avoid breaking into the savings of the household: 'I'm not dipping into the family coffers' (George), and to remain in complete control over their businesses: 'I don't want somebody else coming in and telling me what to do' (Diana).

Туре	Representative Quotation
Money	'Clara [partner] supported me financially, especially at the start, 'cos I wouldn't have been able to start the business if she hadn't had a decent paying job and been willing to live on one income to start with'. (Carol)
	'A few years ago he [partner] helped me with rent for the workshop'. (Diana)
Equipment	'Evan [spouse] helped me to pay for the refurbishment of the studio'. (Emily)
	'I changed my car, [ex-partner] helped me with that, with nearly all of it [cost] so that's really helpful, 'cos I do need a reliable car for the business, because, you know, I do go around and deliver things'. (Frank)
Material	'Like yesterday, he [Adam, partner] said, "Do you want some legs?" [laughs], 'cos he is at college at the minute and they are getting rid of a load of mannequins, so he's bringing me some mannequin legs'. (Amy)
	'I got lots of Harris tweed and different things from him [spouse]'. (Emily)
Labour	'My husband now posts them [products] out, because, you know, it's something he can do, 'cos things were getting rather hectic'. (Julie)
	'I wouldn't want pay 'cos I kind of see us as a team I've now spent two weekends helping her out at [craft shows]'. (Brenda, partner)

Second, participants receive tangible assets from their cohabitants. They are given equipment, as discussed by Frank: 'he's [ex-partner] given me ... the pug mill, which is worth a lot of money ... it was really nice of him ... really supportive'. They also obtain free materials, as illustrated by Betty: 'She's [partner] got big, massive dresses that she inherited from her grandma ... which I can use'. Furthermore, craft entrepreneurs benefit from the free labour that their household members are willing to contribute to their businesses, such as *'helping her out over the weekends'* (Iain, spouse). By providing free labour, household members accomplish tasks that craft entrepreneurs would otherwise need to spend time performing, enabling them to instead create more products that they can sell.

Even members of the wider family were found to provide such benefits. As Emily explains, 'my mum volunteers in the studio for me when I'm busy [before Christmas]. She comes, she stays in my house ... and goes to the studio ... every day to handle the customers'. Craft entrepreneurs also valued the childcare support that their wider families provided during the school holidays when their children were younger. As Isla reports, 'I didn't have family nearby that could do school pick-ups or whatever, but my husband's parents loved to have my daughter ... in the school holidays'. Participants shared that they appreciated such support as it allowed them to focus on their businesses.

Furthermore, participants value the knowledge that cohabitants contribute an income to the household. As Holly explains in reference to her work, 'it's not like a guaranteed wage at the end of the week ... you never know how much it's gonna be ... whereas Henry [partner], although it's contractual, we know exactly how much he is getting in'. However, participants also highlight that 'he [spouse] financially doesn't support my business ... [but] he kind of regulates our [household] cash flow'. In other words, the regular wages of partners and spouses provide a sense of stability, which allows entrepreneurs the freedom to concentrate on practicing their crafts.

Craft entrepreneurs also draw on the *cultural capital* of household members to develop their businesses. As shown in Table 8.3, participants particularly benefit from the skills and advice of cohabitants.

### Table 8.3. Cultural Capital Support

Туре	Representative Quotation
Skills (corollary)	'I usually do her tax return for her, 'cos she is even worse at accounting than me'. (Brenda, partner)
	'I have helped her physically sort of painting the place when she first got it and fitting the units that she uses, kitchen units and things like that'. (Clara, partner)
Skills (essential)	'He [son] put the website together and he did all the photography for my books, he is very good at editing, he is much better at spelling and grammar than I am'. (Julie)
	'On a more practical level, it's making stuff making little bits of dressing and props to help present her products in a, hopefully, flattering light'. (Henry, partner)
Advice (business)	'The majority of my customers are so nice that when you get one who is a problem customer you kind of don't know how to deal with that almost so yeah, there have been a few times where I have said, "I can't believe this e-mail, you [partner] have to read it and tell me how to answer it". (Carol)
	'I sort of ask him [partner] for advice sometimes, 'cos he is not a creative person, but he is very practical, which I'm not always'. (Diana)
Advice (creative)	'He [cohabitant] gives me criticism of what I'm doing and things and says what he likes and stuff'. (Frank)
	'I've always sort of tried helping out in a sense, if I've looked at something and said 'Have you ever thought of, I don't know, changing that buckle?' or I've sent her through inspirational pictures that I thought she might like'. (Adam, partner)

Participants often utilise the technical skills of their household members. While most of these skills are used for corollary business operations, such as 'doing up the studio ... a lot of building it up and decorating' (Iain, spouse) or dealing with 'the accounts at the end of the year for the tax office' (Jack, spouse), some skills are essential to the daily operations and success of craft entrepreneurs. For example, several participants require their partners to manage their websites. Amy admits, 'I couldn't do that without him [partner], I mean I'd end up spending a fortune buying, well paying someone to make a website, so I'm really lucky that he can do that'. However, Amy's evolving situation equally reflects the disadvantages of dependence on household or family members for such integral skills. After breaking up with Adam (partner), she explained that 'I can't really rely on him anymore'. Consequently, she had to invest into enlisting a graphic designer to help develop her website.

Emily describes another disadvantage that arises from reliance on the skills of cohabitants for day-to-day business activities: 'with the website, I've got to wait until Evan [spouse] comes home'. Surprisingly, participants do not attempt to learn these essential skills from their household members, but rather depend upon their continued help over time. As Holly explains, 'I don't have time and I don't have any interest'. The only types of skills that craft entrepreneurs seem to value enough to acquire themselves relate to their creative practice. As Adam (partner) recounts, 'I have shown her how to properly use a couple of tools for the leather'. Certain skills are also imparted to entrepreneurs by members of their wider families, particularly during their upbringing. As Emily describes, 'my grandmother was a dressmaker and so she passed on certain skills to me, and my mum knitted and so I picked that up'.

The second main form of cultural capital support that cohabitants provide is advice. Participants regularly benefit from business advice, as illustrated by Carol:

Financial stuff I will ask her advice about because she [partner] is more sensible than I am ... I think often you find with creative people they are not always the most business savvy, or the most organised people [laughs], so yeah sometimes, if I'm struggling to organise my workload ... I'll ask for her advice.

Participants also obtain creative input and opinions from cohabitants. As Diana explains, 'it's sometimes helpful for someone who is not involved in this [craft] world in any way to kind of bring a different perspective to it'. Furthermore, Grace (spouse) discloses, 'I think he probably appreciates my help more than he would most people, simply because I probably understand ... his creativity better than most people do'. In many cases, household members also inspire the development of new work or have even triggered the establishment of the craft business in the first place. As Isla describes, 'it was just my husband saying, "That's really nice, you should be doing something with that!" was the absolute trigger that took it forward'. Nevertheless, participants also note that they frequently ignore the advice given to them. As Betty reveals, 'I'll ask her [partner] for advice about stuff, but I don't always listen'. Henry (partner) makes a similar observation that he will often 'suggest how she might want to do things and then she does something totally different'. This behaviour underscores how important autonomy is to craft entrepreneurs (see Chapter 2).

*Social capital* support, in the form of new ties brokered by household members, seems to be of lesser importance to participants than other forms of capital support. For Carol and Diana, this is due to the fact that their partners are 'not involved in the art world at all' (Carol). Still, as shown in Table 8.4, participants do gain business contacts

though their cohabitants. As Amy describes, 'He [supplier] designed the logo at a cheaper cost than normal, but his work was amazing ... and if I hadn't known Adam [partner] I wouldn't have got that'. Similarly, Isla explains that 'of course we share contacts ... we help each other out', and Holly notes that 'I will maybe get contacts through ... people he [partner] knows through different [business] clubs'.

### Table 8.4. Social Capital Support

Туре	Representative Quotation
Business Contacts	'I've managed to put her in contact with a few fairs, sort of children's fairs and craft fairs in the area'. (Adam, partner)
	'There have certainly been a few contacts that I've passed on to her'. (Iain, partner)
Customers	'Daniel [partner] is quite a sociable person and probably through him, he has introduced me to new clients and things'. (Diana)
	'She has had sales from me just mouthing off and being generally proud of what she does'. (Evan, partner)

Household members also help craft entrepreneurs gain new customers. As Emily reports, 'I get custom through him [partner]. He works at the council and it's a big kind of office building ... a lot of people found out about my business'. Felix (cohabitant) makes a similar remark: '*I know people who may be interested in buying things, or going for lessons, and I will pass his contact details to them*'. Indeed, the provision of new clients is also fostered by wider family members. As Betty explains, 'I think my mum and dad ... they tell their friends about [business name] and then they go online ... and buy things'. According to Frank, 'when people ask "What does your son do?" then they [parents] will say, "Well, he does ... [pottery]", "Oh right, I didn't know. Could he do this?" "Yes, go and see him"".

As shown in Table 8.5, craft entrepreneurs also develop their *symbolic capital* with help from cohabitants. Household members support participants in this endeavour by attending shows and exhibitions where collectors expect to meet the partners or children of craft entrepreneurs. As Carol explains,

It's not just about the making [at shows], it's the kind of family as well ... Even [famous maker] goes along with his wife and you meet ... his son, you meet ... the whole family and it is one big happy family really, everyone gets along.

The importance of family attendance at such exhibitions demonstrates how significant perceptions are to successful craft ventures and the role that 'family values' play in the craft field. In other words, family seems to be connected to reputation in this context.

 Table 8.5. Symbolic Capital Support

Туре	Representative Quotation	
Exhibitions	'At the convention, it was very much we were sort of on show as a couple'. (Holly)	
	'We've done fairs together occasionally where I play the role of supportive husband, which I like to think I am actually'. (Evan, spouse)	
Reputation	'Some people aren't really aware of teddy bears, what it is as a product, they just think, "Oh cuddly toy", and you've really got to say, "Well it's hand made, it's been designed by the person that made it and there has been a lot of work going into the actual bear and it's not a toy for kids, it's for collectors"'. (Henry, partner)	
Promotion	'My son put pictures [of baskets] up on Twitter and he's got a ridiculous amount of followers'. (George)	
	'I've also been promoting her work through the likes of social networking, like Facebook and Twitter'. (Adam, partner)	

Family members can also help improve the reputation of entrepreneurs through positive word-of-mouth, as described by Brenda (partner): '*All my friends* ... *are people who can put her name out there* ... *so yeah, I definitely talk to my friends about what she is doing*'. Julie reports that her daughter has actively helped promote her work at fairs: 'she is very good with people, she is very charming ... an artist came around and said ... "Your daughter has been talking about your work all morning ... I'm really interested". Similarly, Diana explains,

Something as simple as you're out of an evening, you are never really able, especially when you are self-employed, you are never really away from your work, there is always somebody who wants to talk about what you do and if it's people that we've just been introduced to and they don't know about my work ... I'm just kind of like, 'This is what I do', and that's it, whereas Daniel [partner] will be the one who will say, 'Well, you know, she has done this and she has done that', and ... he will be the one promoting.

Thus, by highlighting their achievements and praising the quality of their work, household members can help develop symbolic capital for entrepreneurs.

### **8.4. Emotional Support**

Though largely absent from Bourdieu's (1977, 1984, 1986) theory of practice, emotional support is amongst the most highly valued forms of support that participants receive from household members. This section illustrates ways in which cohabitants provide emotional support and explains when and why entrepreneurs value this support most.

The data analysis revealed that cohabitants provide emotional support to craft entrepreneurs by developing their self-efficacy, hope, optimism and resilience. Table 8.6 offers deep insights into the perspectives of participants and explains the specific ways in which entrepreneurs benefit from these forms of emotional support from their household members.<sup>12</sup>

First, by expressing their admiration for entrepreneurs and their products and by providing reassurance, cohabitants boost the confidence of craft entrepreneurs in their abilities (i.e. self-efficacy). For example, Evan (partner) expresses his pride for Emily by *'applauding her from the side lines'*, and Jack (partner) explains that he dispels Julie's doubts by assuring her, *'You've always made the right decisions for your business and for yourself and for our family'*.

Second, household members offer encouragement and invigoration to entrepreneurs. In so doing, they motivate craft entrepreneurs to achieve their goals and thus generate hope. As Amy expresses, 'He [partner] is just encouraging me ... which is really, I think that's the main thing'. Furthermore, Frank explains that 'It's quite hard in the workshop to get motivated ... when it's cold and damp ... so in some ways, you know, Felix [cohabitant] ... getting ready and going to work every day kind of motivates me ... [to] get started'.

Third, cohabitants share their belief in the quality of the work of craft entrepreneurs and provide them with validation. As a result, they foster expectations of positive outcomes for the businesses of craft entrepreneurs (i.e. optimism). For instance, Brenda (partner) proclaims her conviction that Betty will succeed by declaring, 'the stuff she designs is amazing ... she just has all these ideas that come out of her head and ... when she executes them they are really good'. Holly also emphasises the importance of counsel offered by household members, 'even if it's

<sup>&</sup>lt;sup>12</sup> The design of Table 8.6 was inspired by McKeever et al. (2015: 55).

Descriptive Categories < to to			Anal	Analytical Categories
Examples from raw data	This tells us about Core to this entrepreneurs: is:	Core to this is:	This in turn produces:	The consequence is:
'I like to think that he [partner] is quite proud of my achievements'. (Holly) 'Grace [spouse] sometimes reads me praise about my baskets e-mails from happy customers'. (George)	The value of			
'It wasn't until somebody said 'she [daughter] is very proud of the fact that you [own] a studio' that I realised how she felt. (Emily)	expressions of pride Admiration	Admiration	Confidence	
<i>I m very proud of what she does, and all of the volume (clard, partner)</i> 'You will have odd moments and that's when you do need [cohabitants] to help you and say "You're on the right track" just to bridge that gap of confidence'. (Isla)			in one's abilities	Self-efficacy
'I know she struggles sometimes with the feeling of pressure so again, it's just – I have to give her reassurance'. (Jack, spouse)	The importance of	Reassurance		
'I think she needs a bit of a confidence boost; she needs to be told how artistic she is'. (Henry, partner)	aispelling doubt			
'I comment on how nice things are, how well executed they are'. (Evan, spouse)				
'They are brilliant really they are always very encouraging'. (Amy)				
'Clara [partner] is very supportive She can, you know, kind of encourage me to keep doing the thing that I'm doing'. (Carol)	Ę			
'Sometimes you can get a wrap over the knuckles and "For goodness sake, mum, what's your problem? Just get on with it!" which is quite often what you need'. (Julie)	Inc powers of persuasion	Encouragement		
'I just supported her and gave her a little push to say "What's the harm in trying it, you can always back out if it doesn't work out". (Brenda, partner)			Motivation to	
'Sometimes I just need him to kind of go, "Yeah, that is a good plan". (Diana)			achieve goals	норе
"When you see your child have a large degree of success, that gives you a huge amount of buzz it's almost as if someone [says] "You can do it, too". (Isla)	3° - 10-33° - 111			
"My partner sometimes passes on comments, which can be useful in the sense that they are either triggers or kicks up the arse". (Betty)	arousing enthusiasm	Invigoration		
'It's always a help if he [cohabitant] compliments my work, 'cos that kind of helps to keep you motivated and keep you going with it'. (Frank)				

Table 7.6. Emotional and Moral Support

Examples from raw data ent			•	CATTOSAND TRATE CITY
	This tells us about Core to this entrepreneurs: is:	Core to this is:	This in turn produces:	The consequence is:
'She [partner] has tremendous belief in me, even if I don't have belief in myself sometimes'. (Betty)				
'They support me in whatever I choose to do They trust that I'll do well at it'. (Amy) The impact of	The impact of			
'The finished products that he comes up with, they are always amazing stuff that con blows me awav every time'. (Grace. spouse)	conviction in one's success	Belief		
she is doing, I'm glad that she is happy eople do like it'. (Iain, partner)			Expectation	
'It's nice that Felix [cohabitant] is always interested in what I'm doing and what			of positive	Optimism
I've managed to get done He makes me feel good about my work'. (Frank)				
'Sometimes I just need people to validate the decision that I've made or to, you The know kind of so "Veah that is a good alan", (Emilv)	The benefits of	Validation		
rd I think'. (Adam, partner)	interest and counsel			
'He sounds things off me sometimes I think he doesn't always have somebody				
else to talk to about it to get positive feedback '. (Felix, cohabitant)				
'I know it's there when I need it [emotional support], so, you know, if it wasn't, I would probably flounder quite a bit'. (Holly)				
can always go [to them]'. (Amy)	The value of			
anyway	unconditional	Security		
	support			
'The fact that she [daughter] offered it [help] and it was there if I did need it was quite a source of comfort and support'. (Isla)			Ability to bounce back	÷
'He [spouse] is very calm. When I ring up, when I'm in a panic and think 'I've			and adapt to	Keshlence
'I do work on quite high adrenaline levels [so] when I hit a brick wall that's				
	The power of	Commother		
'She [partner] is a lot calmer than I am I tend toward the anxious and I tend to kind of panic unnecessarily over things'. (Carol)	consolation	oympaury		
'You need to be able to go home if something hasn't gone the way you expected or wanted it to and have somebody tell you "It will be ok"'. (Diana)				

just asking Henry's [partner's] opinion ... it's just, there is somebody there as a sounding board ... to tell me that what I'm doing is alright'.

Fourth, by providing a sense of security and expressing their sympathy, cohabitants increase the ability of craft entrepreneurs to bounce back from setbacks and adapt to changes (i.e. resilience). As Emily explains, 'they [cohabitants] are very kind and very supportive whenever I'm doing anything', highlighting the value of this unconditional support. Participants also rely on consolation from their household members, as illustrated by Isla: 'I think everybody needs their wailing wall occasionally ... to unload and vent and then move on'.

Participants require the support of cohabitants to respond to challenging circumstances and to gain the motivation and persistence required to succeed in their crafts. Implicit in the shared stories of entrepreneurs is the importance of trust to building self-efficacy, hope, optimism and resilience. Without a strong degree of trust, it is questionable whether participants would value and be convinced by their household members' expressions of admiration, reassurance and invigoration .This indicates that transactions of emotional support depend upon entrepreneurs placing trust within others and that a lack of trust can diminish the effects of such support.

Participants explain that they value the emotional support from household members most when they are anxious, such as when they are struggling with stress or a lack of confidence. For Holly, emotional support is imperative 'when I am stressed before a show', and for Amy it is 'when I'm all stressed out and running into a deadline'. Carol needs this type of support 'when it's been a bad day, you know, when something has not gone quite right or a client is not quite happy'. As craft entrepreneurs need to continuously develop their styles to retain the interest of customers, they need to believe in their abilities to create new products that are aesthetically pleasing. Therefore, Isla expresses, 'I think it's when your confidence is knocked, that probably is the most important time when you need emotional support from family'. Similarly, Julie explains that

the pressure of continuing to create something new when you're feeling well and on top of things, you know, it's not an effort, but when the pressure starts to get to you, I suppose that's when you need emotional support the most There are several reasons why participants value this type of support so highly. For Emily it is 'because I'm a very emotional person ... and because I know that my work is so much linked to my personality'. Crafting products alone in their studios all day can cause certain difficulties for craft entrepreneurs, which emotional support can help overcome. As Betty notes, 'I had sort of moments with depression ... and without that support, you know, I don't think I would have ... gone through them or I wouldn't have got through them as well'. Frank agrees that 'sometimes it can be such a lonesome job, working away by yourself, that you just want that reassurance'. Participants thus see emotional support as integral to their abilities to craft products for their business. Diana most clearly reveals why it is the emotional support of household members that she particularly values:

because he [partner] has been there from the very beginning ... he has seen how I have grown and how the work has grown ... and [therefore] has a greater understanding and knowledge of what's involved.

In other words, the understanding that household members have of the entrepreneurs, their crafts and their businesses is what makes their emotional support so valuable. As Julie describes, 'he's [spouse] always there for me, ... he knows how I can get carried away and get quite chaotic sometimes ... he will just sit me down ... [and] help me see things in perspective'.

The importance of understanding is also evident in the distinction participants make between the emotional support provided by cohabitants and members of their wider families. Craft entrepreneurs open up more to cohabitants mainly because they are much closer to them. As Amy notes, 'the other family don't really have a clue ... what's going on'. George explains that this is due to 'I suppose distance ... I don't really have huge amounts of contact with them ... I'll sort of speak on the phone or e-mail occasionally, but not on a day-to-day basis'. Negative experiences in particular appear to be discussed more with household members, as illustrated by Emily:

there is a lot I don't tell my parents, ... I tell them the good stuff, but I don't tell them any of the bad stuff, so I don't go to my parents for so much, for emotional support any more, I used to go for more ... but that was when I was younger.

As Carol explains, 'I sometimes think that they don't really get what I do, my parents in particular ... I think a lot of the time they haven't really understood, you know, couldn't understand what ... I go through'. Thus, it appears that craft entrepreneurs mainly draw on the emotional support of their household members, because they are the ones who truly understand them.

# 8.5. Impediments

As beneficial as such moral support is for participants, cohabitants were found to not only facilitate, but also impede entrepreneurial processes. The data analysis highlighted a number of ways in which the needs, problems and opinions of household and family members negatively affect participants, as discussed in this section.

Amongst the most influential factors that impede entrepreneurial practices are the *needs* of household members. As shown in Table 8.7, when their children were younger, craft entrepreneurs were greatly affected by the need to either organise childcare or supervise their children themselves. As Julie explains,

The fact that you were at work and they [children] had to be picked up and go to a child minder and ... it was an absolute minefield trying to get good childcare ... That was a very tricky thing to handle and caused me personally quite a lot of anguish.

The data also show that there is an expectation that craft entrepreneurs schedule their work according to the needs of household members, as their jobs are usually more flexible. As Carol notes,

In terms of taking time off ... because my job is the one that's adaptable, you know, with Clara [partner], unless something really badly goes wrong, she has to be at work Monday till Friday 9 till 5, whereas mine is a little more flexible ... [so] if and when stuff does come up ... I'm the one that kind of has to deal with it.

In the most extreme cases, this perceived flexibility can even result in an expectation that craft entrepreneurs relocate whenever the jobs of cohabitants demand it. As Betty describes, 'Brenda [partner] just wants me to pack everything in

and go live in Aberdeen'. Participants were found to accommodate such requests, despite the potential negative impacts on their businesses. As Isla complains, 'It's the fact that you can't do what you want to do when you want to do it and that can be ... frustrating'.

Need	Representative Quotation
Supervision	'It did impact work life, because you can't just treat them [children] like a piece of work and say, 'I'll put you in that tray and deal with you later'. (George)
	'If you've gotta work full time and you've got small children you are hugely conflicted, there is just no two ways about it'. (Isla)
Scheduling	'He [partner] works on the local ferries and because he is week-on and then he is off for a week, so the week that he is on, I can work late if need be and then the week that he is off, I have to be nine till five'. (Diana)
	'I've had to adjust my time with the children's holiday's and things'. (George)
Relocation	'If he [partner] had to be relocated with his work, I would move with him'. (Amy)
	'Because she is self-employed, we've been able to set up wherever my job took us, so she has been able to follow me and take her work with her'. (Evan)
Housekeeping	'House work is very low on my priority list I'll do the bare minimum so we've all got clean clothes and we don't get diseases and things [laughs]'. (Holly)
	'Clara [partner] cooks, I don't cook at all, I'm really bad at it I tend to do a lot of the hoovering and laundry [but] it doesn't take that long. Our house is really tiny, so it takes like 20 minutes to hoover – you can do it all from one plug'. (Carol)

While the need to carry out chores around the house does take the attention of participants, they were found to have a minimal impact on craft entrepreneurs, mainly because participants share such responsibilities with their household members. As Emily reports, 'Evan [partner] does a lot of stuff ... so we kind of share the responsibility'. In addition, most participants do not consider such tasks to be particularly essential, as illustrated by George: 'We don't pay too much attention to tidying everything, cleaning, I don't have to iron my shirts and things like that, so we're kind of relaxed about it'.

The *problems* of household and family members can further impede entrepreneurial practices (see Table 8.8). As the households of participants all rely on more than one income, potential threats to the job security of cohabitants were found to greatly affect craft entrepreneurs. As Carol explains, At Clara's [partner's] work ... an entire department came in two weeks ago to be told, 'None of you have a job anymore'. No warning ... "You can all go home". So the people that are keeping their jobs are the people that come in early and leave late and ... if they don't, they're gonna be the ones that are classed as under-performing and ... are gonna lose their jobs, so ... I kind of just have to hope that people keep buying yarn, 'cos otherwise we're in trouble.

According to Diana, such uncertainty is unsettling and distracting: 'it cuts into time, it cuts into concentration'.

Table 8.8	Problems o	f Household	and Family
-----------	------------	-------------	------------

Problem	Representative Quotation
Income Insecurity	'We did have quite a worrying summer, because in her department she [partner] is a team leader basically and what they did was restructure which means that a lot of the team leaders were no longer required [so] yeah, it was a bit of a worry'. (Betty)
	'He [son] has a small benefit, disability allowance which is under review at the moment. I don't know what we'll do if that disappears, because obviously he lives under our roof if we weren't around, I don't know how he would survive'. (Julie)
Illness	'I haven't really marketed them [products], again, lack of time. My daughter has not been very well, so there is a whole lot of work to do'. (Emily)
	'When the kids were younger, I was always there if they were ill if there was an emergency I would always be there'. (Holly)
Loss	'I had a certain amount of [product] developing that I was doing when the shit hit the fan with Iain's [spouse's] mum dying and all the rest of it'. (Isla)
	'There's always a hole then, which sometimes gets, you know, kind of fills up again, but other times it doesn't ever go away'. (Frank)

Craft entrepreneurs are also impacted by the illnesses of cohabitants. As George notes, 'If somebody is seriously ill or something, then obviously you're gonna be distracted'. Permanent disabilities of household members can be even more disruptive, as discussed by Julie:

My son's illness affects me and has done for, you know, 20 odd years, the fact that he, although he is very, very bright, he can't ... hold down a job or, you know, he doesn't achieve ... the potential that I know he has, because of his illness, so that has a negative impact most of the time.

Furthermore, the loss of family members significantly impacts craft entrepreneurs, as observed for Amy, who lost her mother around six months into this study. She discontinued all work and broke off contact with her craft community for several months. In Amy's words, 'I had to take time out when my mum [died] ... for two months I've really, I've been cut off from everything, we [household] went abroad for a little while'. Thus, craft entrepreneurs need to adjust their work to support household and family members and to deal with grief and uncertainty.

As explained above, advice and constructive criticism can help craft entrepreneurs develop their products. However, dissenting *opinions* can also impede their practices (see Table 8.9).

Opinion	Representative Quotation
Disapproval	'My parents' sort of default setting seems to be, they are a bit anti-change, they don't really think outside of the box whereas I'm all about that and sometimes they can be a bit full-on with their criticism of, "Why do you want to do that?"' (Betty)
	'My mum railed against it from the minute I said I really wanted to do technical drawing at school I had to fight to do art'. (Emily)
Doubt	'My mum kept saying "You're gonna have to get a part-time job, you're not getting enough from it". I mean, it was paying for my studio rent, but that was it, so she was saying, "Could you not just do, like, cleaning?" and I was like [gasps]'. (Amy)
	'I don't know if they [parents] think I can make any kind of living out of it'. (Frank)
Disregard	'She [mother] actually said"Why can't you do something normal like your sisters? Why can't you just do secretarial studies or something like that?"" (Diana)
	'I think she [mother] just thought I was wasting my time she wanted me to get a real job'. (Carol)

The data show that participants are particularly affected by family members' expressions of disapproval, doubt and disregard for their chosen profession. For example, Frank recounts that 'I did talk to my father ... when I was thinking about ... [getting] a workshop and he said that it would be too expensive for what I'd make ... [and] he thought it was ... too much exposure for me'. Consequently, Frank delayed getting his own workshop for over a year, passing up the chance to move into a property which would have been more suitable for his business than the studio he later obtained. Similarly, Betty describes that 'I think she [mother] just thinks it's, like, something I'm not really that serious about. It'll pass, it's a phase that'll pass'. These expressions of disapproval, doubt and disregard were found to reduce the self-efficacy, hope, optimism and resilience of craft entrepreneurs. As Julie explains,

The creative side, that's difficult because what you're doing is you're tapping into your emotions and it's your emotions that are driving ... your creativity, so if you're disturbed emotionally, you are not able to settle into the creative aspect of the job.

Interestingly, such critical opinions are predominantly expressed by members of the wider family, rather than cohabitants.<sup>13</sup>

With regards to household members, participants are more worried about unvoiced concerns than outright expressions of displeasure. As Holly explains, while her family members do not complain about her working hours, 'I do feel guilty when I'm working late and I haven't – we spent like, we live in the same house and I have seen them, like, 5 minutes a day'. Similarly, Julie notes that 'I'm very aware, you know, I can't just ... be doing my own thing all the time and not spending quality time with them'. As George describes, 'That's what relationships are, isn't it, you kind of have to think about other people'.

This distinction between household members and family further highlights the impact of emotional support upon participants. Whereas the emotional damage caused by unsupportive family members impedes the creative abilities of entrepreneurs and their motivation to drive their businesses forward, it is the strength of relationships with household members that encourages entrepreneurs to turn their attention away from their ventures for periods of time. Though both can be seen as impediments, the latter is arguably less damaging to business operations than the former.

# 8.6. Analysis and Interpretation

Following the presentation of household-related findings, this section 'recontextualises' the findings 'by putting the new knowledge about the phenomena and relations back into the context of how others have articulated the evolving knowledge' (Marlow and McAdam, 2015: 800). Mirroring the structure of the above

<sup>&</sup>lt;sup>13</sup> Participants were asked several times, both directly and indirectly, about potentially negative impacts of household members on their practices (particularly during informal, unrecorded conversations) to confirm this observation. Only Isla acknowledged that occasionally fighting with her spouse affects her work: 'Yesterday, I had a row with my husband ... and that really rattles the cage'.

findings, this section first discusses ways in which households can be explored, before suggesting how the concept of household capital should be extended. It then evaluates the role of psychological capital in this context, followed by considerations about work-family conflict.

## 8.6.1. Exploring Entrepreneurial Households

In introducing the households of participants, this study took an approach different to that which prevails in the literature (cf. Caputo and Dolinsky, 1998; Sonnenberg, 2008). Instead of presenting distinguishing socio-economic features, such as household income or social class, this study drew on the experiences of participants to provide insights into their subjective views (Berglund, 2007; Cope, 2005). In other words, rather than engaging in 'an economic (capitalistic) evaluation' (Johnstone and Lionais, 2004: 218), this study provided 'an evaluation ... in relation to lived life and experience' (Korsgaard et al., 2015: 579). The differences between these techniques are therefore similar to the distinction between spaces and places, which has been explored in regional entrepreneurship research (Steyaert and Katz, 2004; see also Chapter 2). While space-related factors may demonstrate the capacity for profit of a region or, in this case, a household, a focus on place-related aspects provides insights into the meanings that entrepreneurs connect with a location (or home).

Participants in this study share similar experiences in that they are embedded in small households which are largely removed from their wider families (both emotionally and geographically). Consequently, they lack extended social support mechanisms available to other entrepreneurs, who live closer to their kin (Khayesi et al., 2014; Verver and Koning, In Press). Participants often continue their work at home in the evenings and on weekends, which can lead to some degree of tension between work and family (Harrison and Wagner, 2016; Shelton, 2006). As the children of participants are now relatively independent, they rarely restrict entrepreneurial practices. Indeed, the increasing independence of their children was a main factor in the decisions of participating craft entrepreneurs to open their own studios, demonstrating the influence of temporal contexts on entrepreneurial activities, including entry. In Holly's case, the liberation from childcare responsibilities led to the development of a previously home-based business into a firm based outside of this physical location (Markantoni and van Hoven, 2012; Mason et al., 2010). As such, the findings highlight the impact of shifting dynamics in the household structure (Alsos et al., 2014; Mwaura and Carter, 2015), which are caused by transitions in the family (Aldrich and Cliff, 2003), on the practices of entrepreneurs.

This study also offers a new approach to differentiating the household members of entrepreneurs in cultural industries. To date, the limited number of studies which have focused on the households of entrepreneurs typically classify their members according to such features as their level of business experience (Gras and Nason, 2015), ethnicity (Ram et al., 2001; Rodriguez et al., 2009) and formal involvement in the businesses of the entrepreneurs (Baines and Wheelock, 1998). In comparison, prior research has paid little attention to alternative approaches, which focus on providing context-specific descriptions of household members (Alsos et al., 2014). The analysis of the perspectives of participants suggests that cohabitants can be distinguished based on their involvement in the cultural field of the entrepreneurs (i.e. crafts) and their perceived level of creativity. Figure 8.1 captures the four types of household members based on participants' reports.

		of the Ent	repreneur
	_	High	Low
ç	_	Contributor	Bystander
<b>У</b> Нідь		(Creative,	(Creative,
tivit		Involved)	Uninvolved)
Creativity		Promoter	Nonparticipant
O M	S C	(Non-Creative,	(Non-Creative,
		Involved)	Uninvolved)

Involvement in the Cultural Field

Figure 8.1. Household Members of Entrepreneurs in Cultural Industries

First, the 'contributor' is creative and helps entrepreneurs develop their products or services. In addition to some partners, the children of participants could be classified in this way, mainly due to their upbringing in a creative household and their resultant habituation with the work of their parent(s). Second, the 'promoter' is an individual who is not creative, but is involved in the cultural field, usually due to the influence of the craft entrepreneur. Most partners and spouses of participants fall into this category, as well as some parents and non-family cohabitants (i.e. flatmates). Third, 'the bystander' is creative but not interested in the cultural field or work of the entrepreneur. This study did not encounter such household members, but they may still exist. For example, non-family cohabitants who are not on friendly terms with the entrepreneurs could fall into this category, as could children who seek to emancipate themselves from their parents. Fourth, the 'nonparticipant' is neither creative nor interested in the cultural field. While some partners or spouses may fit this description, even nonparticipants were found to support craft entrepreneurs in some way, mainly through the provision of emotional support.

To date, research has focused on the gendered nature of support that cohabitants provide to entrepreneurs (Baines and Wheelock, 1998; Bittman et al., 2003). Such studies have shown that female partners tend to support male entrepreneurs in a variety of ways, including through managing childcare responsibilities and domestic chores (Livingston, 2014; Parasuraman et al., 1992). In comparison, male partners of female entrepreneurs are often only expected to provide a steady income to support the household (Bittman et al., 2003; Grunow et al., 2012). As this study is based on the experiences of both male and female entrepreneurs, some of which live in same-sex cohabiting households, the typology developed transcends normative gender divisions. Its transferability to other contexts cannot be assumed, given the small sample upon which this study drew to devise this typology. However, the findings do suggest that involvement of cohabitants in the fields of entrepreneurs may be based on attributes other than gender.

## 8.6.2. Extending the Concept of Household Capital

The findings show that craft entrepreneurs highly value the various resources that become accessible through their households. To date, entrepreneurship research has focused on the importance of financial capital (i.e. household income, see Boden and Nucci, 1997; Petrova, 2012; Reynolds et al., 2004) and its effects on entrepreneurs (Caputo and Dolinsky, 1998; Carter, 2011). In contrast, little is known about the

influences of other resources, such as the human (Bird and Wennberg, 2016; Coleman, 1988) and social capital (Arregle et al., 2007; Edelman et al., 2016) that household members provide to entrepreneurs. Accordingly, the concept of household capital, which consists of the financial, human and social capital of cohabitants (Rodriguez et al., 2009; Steier, 2009), was developed to encourage investigations of a broader range of resources. However, there is a lack of empirical studies applying this approach to different contexts (Sorenson and Bierman, 2009). Drawing on Bourdieu's (1986) perspective, this study demonstrates that the concept of household capital should be expanded to incorporate the economic, cultural, social and symbolic capital of the household.

First, the findings show that household members provide craft entrepreneurs with money, equipment, material and labour. Importantly, the data highlight that craft entrepreneurs are willing to accept direct financial contributions only during the venture start-up phase, not because they fear indebtedness (Steier, 2009), but to protect the savings of the household (Carter, 2011) and to maintain their independence. This finding challenges prevailing beliefs that entrepreneurs are willing to draw on financial support from their households over time (Bird and Wennberg, 2016) and consider doing so crucial to their success (Edelman et al., 2016). Thus, it highlights the temporality of entrepreneurial action and decision-making (Lippmann and Aldrich, 2016b). Although participants do not seek monetary support from cohabitants post start-up, they do value knowing that there is another, regular income which helps sustain the household. Participants also highlight the value of free labour that household members provide, which is in line with the findings of prior research (Alsos et al., 2014; Baines and Wheelock, 1998; Ram et al., 2001). Furthermore, they are happy to receive other tangible resources (e.g. materials) from the household that they can use to create products. Thus, although household financial capital accounts for the use of shared assets, such as cars and phones (Danes et al., 2009; Dyer et al., 2014), the findings of this study suggest that the concept of household economic capital may be better suited to explain the range of material contributions that cohabitants can make to the businesses of entrepreneurs.

Second, the data show that craft entrepreneurs value receiving advice from and drawing on the skills of their household members. Rather than teaching their household members craft-related skills in order to gain additional support from them and increase the overall level of household human capital (Danes et al., 2009; Dyer et al., 2014), the findings of this study demonstrate that entrepreneurs appreciate receiving an 'outsider's perspective' from trusted partners (Bird and Wennberg, 2016) who are not involved in making the products (e.g. through the provision of judgements about product quality). Instead of acquiring essential skills from cohabitants (e.g. website design), the data show that participants rely on the support of household members over time. Only those skills which relate to the creative practice of participants appear to be considered worthwhile learning. This suggests that craft entrepreneurs are more dependent on the human capital support of household members than is currently assumed (Caputo and Dolinsky, 1998; Kotha and George, 2012). Given that participants emphasise the importance of receiving creative advice and opinions from cohabitants, the quality of which depends on their ability to evaluate cultural goods (i.e. on their embodied cultural capital, see Bourdieu, 1984), the findings also suggest that it would be more appropriate to explore household cultural capital than household human capital, which does not account for the habituation of individuals with cultural norms.

Third, household members were found to pass on client and business contacts to craft entrepreneurs. However, this type of support is much less common in the craft industries than in other contexts (Bullough et al., In Press; Edelman et al., 2016). While household social capital can help entrepreneurs develop their businesses (Arregle et al., 2007; Khavul et al., 2009), it appears that the lack of embeddedness of cohabitants in the craft field prevents them from providing significant support in this regard.<sup>14</sup> In comparison, household members were found well-positioned to improve the abilities of participants to develop symbolic capital. Specifically, the data show that, since family members are expected to attend craft shows and exhibitions, they help the entrepreneurs maintain and raise their status in this field. Furthermore, cohabitants were found to augment the reputation of participants through positive word-of-mouth and by promoting their work. Accordingly, this study proposes that symbolic capital should also be incorporated into the concept of household capital.

<sup>&</sup>lt;sup>14</sup> As shown in Chapter 9, embeddedness in the craft field is required to gain access to various agents, such as collectors, who enable entrepreneurial success.

Whereas extant research points towards a range of factors that can negatively impact access to household resources, such as the impulsive behaviours of entrepreneurs (Kets de Vries, 1977) or the deliberate withdrawal of capital by cohabitants (Alsos et al., 2014), participants made little or no reference to such inhibitors. Prevailing norms and rules (i.e. habitus) within their households encourage altruism, openness and reciprocity (Arregle et al., 2015; Karra et al., 2006), but do not lead or require entrepreneurs to relinquish resources needed for their businesses (Adler and Kwon, 2002). In fact, the resources provided by household members are drawn upon freely and regularly, which suggests that craft entrepreneurs consider them a part of their aggregate stock of capital.

Furthermore, although participants spend significant amounts of time in their studios and bring work home with them, this does not seem to result in an unwillingness of cohabitants to support the entrepreneurs, as proposed in the literature (Pearson et al., 2008). Neither do such behaviours seem to provoke major conflicts between household members (Miller et al., 2003; Shelton, 2006). This indicates that collective identity and collaboration are based upon shared experiences reified by 'technologies of memory' (Lippmann and Aldrich, 2016a) – the strength of social units, including households, cannot be measured by time spent, but rather in emotions and history shared (Holt and Popp, 2013). Thus, the experiences of participants suggest a more positive outlook on the resource support practices within entrepreneurial households than is currently held. Indeed, the findings indicate that the 'rules of the game' (habitus) within entrepreneurial households prescribe an open and free exchange of available resources to support entrepreneurs in their business endeavours.

# 8.6.3. Reliance on Psychological Capital

The findings of this study highlight the importance of emotional support provided by household members to entrepreneurs (Brüderl and Preisendörfer, 1998; Renzulli and Aldrich, 2005). While there is growing recognition of the role that emotions play in entrepreneurial processes (Baron, 2008; Cardon et al., 2012; Foo, 2011), scant attention has been afforded to the interpersonal aspects of emotions, including how emotional states are affected by others (Jennings et al., 2015). Within the household

context, the value of emotional support from cohabitants has been acknowledged (Alsos et al., 2014; Danes et al., 2009; Steier, 2009), but still remains woefully under-explored (Arregle et al., 2015). Prior research suggests that emotional support involves listening and empathy (Adams et al., 1996) and that it can influence how entrepreneurs think about events, solve problems and adapt to difficult situations (Edelman et al., 2016; Powell and Eddleston, 2013). The findings of this study advance understanding by demonstrating that emotional support from household members increases the psychological strengths and capabilities of entrepreneurs. The 'dance of data and theory' (Anderson et al., 2010: 129) revealed that there is a concept in the extant literature which helps to explain the experiences of participants. Specifically, the data analysis showed that cohabitants provide entrepreneurs with psychological capital (Luthans and Youssef, 2004; Luthans et al., 2007) by developing their self-efficacy, hope, optimism and resilience (see Table 8.6).

First, by expressing admiration for their products and providing them with reassurance, cohabitants boost the confidence of craft entrepreneurs in their abilities. As Baron et al. (2013) argue, high levels of self-efficacy can help entrepreneurs accomplish what they set out to do and improve venture performance (Cardon and Kirk, 2015; Forbes, 2005). Second, by offering encouragement and invigoration, household members impart hope to entrepreneurs. According to Zou et al. (2016), feelings of positive expectation (i.e. hope) motivate entrepreneurs to achieve their business goals and therefore facilitate entrepreneurial success (Hmieleski et al., 2015). Third, by providing validation and sharing their belief in the entrepreneurs, cohabitants generate optimism. As shown by prior research (Dushnitsky, 2010; Hmieleski and Baron, 2009), anticipating favourable outcomes leads entrepreneurs to consider obstacles as opportunities, which promotes venture sustainability and growth (Anglin et al., In Press). Fourth, by providing a sense of security and expressing sympathy, cohabitants increase the abilities of craft entrepreneurs to adapt to changes and recover from setbacks. As Bullough et al. (In Press) note, high resilience enables entrepreneurs to handle challenging situations and develop means of overcoming them (Williams and Shepherd, 2016).

Thus, by providing entrepreneurs with the emotional, cognitive and behavioural resources (i.e. psychological capital) required to cope with stress and respond to challenging circumstances (Avey et al., 2010; Newman et al., 2014b), household members facilitate the business activities of entrepreneurs, such as conversions of capital. The findings of this study highlight the communal way in which these strengths and capacities are developed in craft entrepreneurs, which suggests that the concept of psychological capital is not an individual construct, but a collective one (Dawkins et al., 2015). Furthermore, given that emotional support within the household context is of such high importance to craft entrepreneurs, the data indicate that the concept of household capital (Rodriguez et al., 2009; Steier, 2009) should incorporate psychological capital be (re-)defined as *the economic, cultural, social, symbolic and psychological capital accessible to entrepreneurs through their households*. It is expected that adopting this definition will lead to a more holistic understanding of the resources that household members provide to entrepreneurs.

The data analysis showed that craft entrepreneurs value the emotional support of cohabitants most at times when they are anxious and that they consider such support integral to their abilities to carry on crafting products. In this way, it is the temporal context surrounding entrepreneurs that drives them to seek emotional support from their co-habitants. While extant research suggests that entrepreneurs can learn to deal with emotional challenges over time (Arregle et al., 2015: 334), the findings of this study demonstrate that craft entrepreneurs continue to rely on their household members for emotional support. It was also shown that understanding of entrepreneurs is key to useful support (Powell and Eddleston, 2013). Whereas cohabitants usually possess the necessary access and insights into the lives of entrepreneurs to provide meaningful support, members of the wider family, who do not share the same household, were found to lack the required understanding. As such, the data support the view that strong feelings of togetherness (i.e. high cohesiveness, see Laspita et al., 2012) increase the quality of emotional social support exchanged within families (Edelman et al., 2016). It is due to a history of shared experiences and resultant mutual understanding that entrepreneurs value, and therefore benefit from, the support of their household members (Holt and Popp, 2013). Consequently, this finding suggests that a focus on the household (Baines et

al., 2002; Wheelock and Oughton, 1996), rather than family, can provide richer insights into the emotional support received by entrepreneurs.

## 8.6.4. Work-Family Conflict

The analysis of data showed that household members not only facilitate, but also impede the practices of entrepreneurs. To date, only a limited number of entrepreneurship studies have explored the ways in which household members interfere with entrepreneurial processes (Jennings and McDougald, 2007; Shelton, 2006) and their conceptual ideas have rarely been empirically verified (Forson, 2013). As such, the findings of this study shed new light on the interwoven relationship between household and business (Baines and Wheelock, 1998; Ram et al., 2001). Specifically, the data show that the needs, problems and opinions of household and family members can have negative impacts on entrepreneurs and inhibit capital conversion processes.

Concerning the needs of household members, certain norms (habitus) of reciprocity appear to govern the practices of craft entrepreneurs (Stewart, 2003; Verver and Koning, In Press). Craft entrepreneurs receive a range of resources from cohabitants, which encourages them to make accommodations in their work and help out around the house (Bird and Wennberg, 2016). As Carter et al. (2017) argue, it is important to consider the power relations between household members to understand their decision-making processes. Participants in this study share the responsibility for generating household income with their partners, though they acknowledge that the income from their businesses is less stable than that of their partners' salaried employment. However, this does not mean that their partners take a dominant position in the household, as suggested by extant research (Fine, 2010). Instead of being seen as providing a secondary income (Bittman et al., 2003) or being locked into a subordinate role (Grunow et al., 2012), participants and their businesses are on (more or less) equal footing with their partners and their jobs.

Although participants did not consciously choose to start their own craft firms to accommodate the needs of the household (DeMartino and Barbato, 2003;

Thompson et al., 2009),<sup>15</sup> but rather because they spotted specific opportunities in the market and wanted to stay in complete control over their creative outputs (see Chapter 2), their chosen professions do allow for a degree of flexibility (Tregear, 2005). Consequently, participants were found to accommodate the scheduling and relocation needs of their household members, despite potentially negative affects on their businesses (Livingston, 2014). As such, the temporal contexts in which the family members of entrepreneurs are embedded also affect entrepreneurial decisions and actions. Furthermore, as suggested in the extant literature (Forson, 2013; Rouse and Kitching, 2006), the data show that organising childcare and dealing with unexpected issues with their children can put strain on craft entrepreneurs. Participants engage in 'role-sharing' with their partners to manage childcare responsibilities and domestic chores (Shelton, 2006: 290). Thus, while they maintain some household responsibilities, participants were generally found to adopt an approach of 'scaling back within the non-work domain' (Jennings and McDougald, 2007: 749) to minimise potential work-family conflict. These practices help them achieve work-family balance (Eddleston and Powell, 2012).

Regarding the problems of cohabitants, the data show that potential threats to the income security of their partners can impede the abilities of craft entrepreneurs to concentrate on their creative practice. Given that the financial rewards of entrepreneurship are generally variable and irregular (Carter, 2011), such situations compound income uncertainty and increase the level of risk within entrepreneurial households. Similarly, illnesses and losses of household and family members can have distracting effects on craft entrepreneurs and lead them to temporarily prioritise their families over their businesses. Although such occurrences are comparatively rare, they do affect entrepreneurs. Whereas extant research highlights the reduction of available resources that accompanies the loss of a household member (Baines et al., 2003; Dyer et al., 2014), the findings indicate that the resultant emotional impact is equally, if not more, harmful (Shepherd, 2009).

Relating to the negative influence of others' opinions, the data show that participants are particularly affected by expressions of disapproval, doubt and disregard from family members (Forson, 2013). Such critical opinions were found to

<sup>&</sup>lt;sup>15</sup> Holly is the only participant who operated her enterprise from her home before moving into a studio because she wanted to be closer to her children while they were younger and more dependent.

reduce the self-efficacy, hope, optimism and resilience (i.e. psychological capital, see Luthans and Youssef, 2004) of craft entrepreneurs and thus impact their abilities to develop new products. While participants worry about unvoiced concerns of their household members, outright expressions of disagreement and scepticism from wider family members have much stronger effects on their practices. Once again, it is the lack of understanding of these relatives which leads participants to fear their criticism and cause emotional disconnection from them. If their kin were more involved in their lives and resided in closer proximity, they might have a more positive impact on the practices of entrepreneurs, as suggested in the literature (Alsos et al., 2014; Khayesi et al., 2014; Verver and Koning, In Press). Overall, the experiences of participants are that household members primarily support their business activities, whereas members of the wider family are more likely to impede them. Thus, the findings show that it is beneficial to differentiate between household and wider family members when exploring the relationship between business and family and its impact on entrepreneurs (Carter et al., 2017; Wheelock and Oughton, 1996).

### 8.7. Conclusion

This chapter sought to answer the research question: *How do household members impact the practices of entrepreneurs?* In so doing, it first introduced an alternative approach to differentiating the household members of entrepreneurs in cultural industries, which challenges the current assumptions about the gendered nature of familial support. Second, it provided evidence that the habitus within households prescribes an open and free exchange of resources and that, contrary to prevailing beliefs, the cultural and symbolic capital provided by household members are more highly valued by craft entrepreneurs than access to their financial resources. Third, this chapter highlighted the importance of the psychological capital that household members help entrepreneurs develop, demonstrating that such support is most needed in times of anxiety and stress and that household members' ability to provide it is reliant upon their understanding of the entrepreneur. Furthermore, this chapter revealed that the needs, problems and opinions of household and family members can negatively impact entrepreneurs. While cohabitants mainly help entrepreneurs

build up their psychological capital, it was shown that wider family members can disrupt this process by expressing doubt and criticism. The findings of this chapter thus demonstrate that household and family members significantly influence the practices of entrepreneurs, both positively and negatively, adding to research which recognises that the lines between work and family are often blurred for entrepreneurs (Aldrich and Cliff, 2003; Arregle et al., 2015).

#### **CHAPTER 9: COMMUNITY CONTEXT - MACRO-LEVEL**

### 9.1. Introduction

Having discussed the individual capital conversions that participants conduct (microlevel) and the influence of household members on their practices (meso-level), this chapter focuses on the impact of community contexts on craft entrepreneurs (macrolevel). As discussed in Chapter 4, there is limited understanding of the effects of embeddedness in communities, especially virtual and common interest communities, on entrepreneurs (Jennings et al., 2013; Piva et al., 2012). Thus, this chapter seeks to advance knowledge by addressing the research question: *How and why does embeddedness in communities affect the practices of entrepreneurs*?

This chapter is structured as follows. First, it describes the craft communities in which participants are embedded and explores the effects of their embeddedness in co-existing fields. This section highlights that embeddedness in one field can lead to dis-embeddedness (i.e. disconnection from a field, see Leca and Naccache, 2006; Zietsma et al., 2017) from another. Second, this chapter examines the importance of fitting-in with community norms and 'rules' (habitus). It shows that craft entrepreneurs achieve compliance not only by learning accepted norms, but also by performing strategic actions which allow them to temporarily modify their practices to meet the expectations of community members. Third, this chapter explores the role that social value creation plays in craft communities. It demonstrates that community norms (e.g. camaraderie and generosity) can lead entrepreneurs to embrace the creation of social value by collaborating with peers and sharing resources with potential competitors. This chapter provides rich examples of the sharing of economic, cultural, social and symbolic capital. It also explains why participating craft entrepreneurs are considered to be social entrepreneurs. Based on the analysis of the experiences of participants, this chapter seeks to present a realistic picture of the actions of craft entrepreneurs, highlighting both their virtuous behaviours, such as collaboration and social value creation, and their darker practices, including pretending, concealing and performing.

#### 9.2. Embeddedness in Communities

This section first presents primarily descriptive findings which shed new light onto the characteristics of craft communities, before exploring their effects on participants' embeddedness in rural communities.

The data show that craft communities share a number of common features which differentiate them from other creative communities and suggest that the craft sector is a distinct entrepreneurial sub-culture. First, craft communities are primarily virtual. In Amy's words, 'it's all online. So, without the Internet ... I would have to move away, I would be totally cut off out here'. As Holly explains, 'I would say about 80 per cent [of community members] live in America ... then Australia, this country and sort of bits like, Lithuania ... and Russia'. Thus, while community members do meet at events, such as craft shows, they almost exclusively communicate through dedicated community websites. In Carol's case, the main portal is 'like Facebook for knitters, but it's a lot more than that ... it brings people together from all over the world'. In other words, craft fields combine elements of local, virtual and transnational communities. Thus, knowledge of embeddedness in these under-explored contexts can be advanced by research into craft communities.

Second, the blending of consumers and producers is a distinctive characteristic of craft communities. In Diana's words, 'they are all kind of mixed in somehow'. Participants not only create, but also collect craft pieces. Similarly, many customers are interested in learning how to design products and could therefore be described as hobbyists. Accordingly, participants run classes to show community members how to craft their own items, which is a demanding process. As an example, the researcher observed the preparation for a class that Holly runs on an annual basis for her collectors. This process is significantly more time consuming than might be expected: it involves not only developing a special design that hobbyists are capable of making and preparing the materials (i.e. pre-cutting the fur for the bears), but also requires organising the teaching space and frequent communication with attendees and others who want to be involved, but cannot make it in person. George, who weaves baskets, explains his reasons for tutoring as 'just for them to see that they can make something ... that's really good out of a bunch of little twigs – [to] come out with a basket'. While participants usually charge a small fee for tutoring community

members, financial rewards were not a primary motive for engaging in this activity. In fact, the income participants lose by not crafting products themselves is generally greater than any fees they charge for instructing others.

Third, craft communities are close-knit despite being spread across large geographic areas. Participants portray their craft fields in similar ways – as 'a very small, but very well-populated world' (Julie). As Frank explains,

the crafting community is very small and even if they [customers] are not gonna buy stuff, they might go away and say ... 'Oh I was in this interesting shop' ... You never know where your customers are gonna come from.

Holly describes her community as 'mainly mental people [laughs], seriously, there is some really odd people who collect bears and I would say 99.9 per cent are absolutely lovely, just a wee bit kinda off the radar'. This sentiment was reinforced for the researcher upon the observation that collectors would travel for hundreds of miles and queue up hours before the start of a craft show in order get a chance to meet the makers and be the first customers at their stands. In Carol's case, they brought along items they had knitted with the hand-dyed yarn as well as pictures to show where they had travelled to and worn their pieces to demonstrate their passion for the craft. Thus, as explained below, it is essential that craft entrepreneurs gain an understanding of the norms and expectations within these intimate and idiosyncratic communities.

Embeddedness in online communities carries several implications (see Table 9.1). Although it enables entrepreneurs to operate businesses from remote locations while still staying closely connected to a variety of customers, developing aesthetic and user-friendly websites requires a high level of Information and Communication Technology (ICT) skill and diverts attention from local customers. Interestingly, craft entrepreneurs were found to adopt a simple, yet effective solution to address the issues that result from divided loyalties between craft and rural communities: they consciously prioritise one field over the other. Chapter 2 provided detailed descriptions of the place-related meanings that participants associate with the areas in which they reside and indicated that participants enjoy living in their rural areas. Nevertheless, the analysis of data showed that participating craft entrepreneurs rarely interact with local community members, which is a significant finding of this study.

## Table 9.1. Implications of Embeddedness in Virtual Communities

Implication	Representative Quotation
Flexibility	'Because I'm internet based I could be in the middle of a field, as long as I have an internet connection and a post office in the field with me'. (Holly)
	'In this day and age, basically you can work anywhere There is so many makers who are out in the middle of nowhere, working away and making a living. You can send your work to galleries, so you don't have to be in the big smoke'. (Diana)
Connectivity	'It's all user-built [website] It's free and once you join you can upload details and photographs and if you had any problems with the pattern or that kind of thing there are forums, where you can chat to other people It's quite an amazing resource and it brings people together from all over the world'. (Carol)
	'In terms of online businesses, we do stuff online because we want it to be immediate, so I maintain a level of communication that I would like to receive as a customer. So if people do e-mail me, I will e-mail them back straightaway'. (Amy)
IT literacy	'My website is my online window. I didn't want a business website, I needed [one] that would attract people who were visually attracted by what they saw, not just a list of products, and it was really hard, I got quite dispirited at one point'. (Julie)
	'Some people don't even look at websites, they just do it through Facebook so I had to get the Facebook and the Twitter and Pinterest and I'm taking photographs now of everything'. (George)
Divided loyalty	'There is a dilemma, between your [online] shop looking nice and your [studio] window looking nice if there aren't many things in the window it's because I'm sending things out and doing things online all the time'. (Emily)
	'I'm online, but I want to still encourage people to come, so it's hard'. (Frank)

Due to their dedication to their crafts, participants spend most of their days making products and communicating with craft community members online. As Julie explains, 'I'm working here from sort of 10 in the morning till 9 ... sometimes 10 o'clock at night and getting home and having something to eat and then crashing out, going back in again'. Even at home, participants often find that they are 'kind of always thinking about the business' (George). According to Isla, 'the downside of an online business, particularly with international customers, is that you have to be always available'. As discussed in the previous chapter, participants mainly use their remaining free time to engage with household members. Rather than spending time socialising with others (e.g. in pubs), participants describe their time off in similar ways: 'I usually have a fairly quiet evening, just in front of the fire, just watch some telly or something' (Frank). In addition, most participants practise individual, rather than team sports, such as running or swimming to secure some 'alone time'. Amy even keeps gym equipment in her studio: 'I do a workout, 'cos I feel that clears my head ... I've got the rebounder, the little trampoline, [so I] just have a bounce'.

Consequently, this study shows that the independence which embeddedness in a virtual community affords, can lead to a dis-embeddedness from local communities. While the studios of participants are open to the public, their products are so specialised that locals are rarely interested in them. In Carol's words,

unless you are a serious knitter you're not just gonna come in off the street and buy something, but I do have people that have found me online and that then make a trip out to the studio or ... some of my international customers, if they're gonna be in Scotland ... they'll e-mail me and arrange a visit.

In addition to lack of interest, lack of knowledge about the craft community can add to the detachment between the entrepreneur and the local community. According to Diana, 'your average man in the street, not used to thinking and working in that way, would just go, "Oh, that looks nice" [laughs] and think that it was just hummed up in five, ten, fifteen minutes' and therefore be shocked at the high prices. As Holly explains,

it's a kinda high end, luxury product, it's like, people don't understand, they just think of teddy bears and don't understand everything that goes into them and making them and designing them and things like that, so I'm not too bothered if, you know, people off the street come in and don't understand it, because ... my main source of income is the internet and it always will be.

As Emily notes, 'I think there will be people who will completely forget that I'm there'. Similarly, Betty explains that 'people do say to me "Why are you here?"' Indeed, during observations in the studios, the researcher was rarely witness to interactions between rural community members and craft entrepreneurs. In reference to the frequency of her interactions with locals, Isla responded positively, 'absolutely none ... they don't sort of tend to come around the corner, which is why this studio is perfect for me'. Consequently, most participants designed their workplaces to look and function more like workshops than studios. As Julie admits, 'I haven't even put out a poster. I know I should have done, but I don't really, I don't really fit in'. In contrast, participants make significant efforts to fit into their craft communities, as explained in the following section.

# 9.3. The Importance of Fitting-in

The data analysis revealed the importance of compliance with community norms and expectations for gaining the support of community members. This section first reviews the habitus in craft communities, before analysing the strategic changes of behaviour that participants practise.<sup>16</sup>

Habitus in craft communities was found to have a significant impact on the practices of embedded entrepreneurs. As participants cater to 'such a very niche market' (Amy), it is crucial for them to join these communities and to continue their active membership over time. Consequently, the data show that craft entrepreneurs must adhere to a number of conventions in order to be accepted into their respective communities (see Table 9.2).

# Table 9.2 Craft Community Conventions

Convention	Representative Quotation
Positivity	'They don't like to hear [the] negative everything is supposed to be rosy [and] I get the impression that nobody ever has any problems and once you had a problem you're almost, like, tarnished'. (Diana)
	'You just have to be nice and sometimes that can be quite exhausting'. (Emily)
Camaraderie	'I do feel a connection with people who make things it's quite sort of close, you know, their needs and their concerns are the same [so] I help out'. (George)
	'I value their [peers'] friendship and I know what it's taken to get mine [business] off the ground, so I, you know, I try and help'. (Julie)
Moral Support	'When your confidence is knocked, that is when you need emotional support from colleagues maybe working in a similar field You will have odd moments and that's when you need somebody to help you and say "You're on the right track", you know, and you just bridge that gap of confidence' (Betty)
	'People that have got good ideas if they've got the courage and get up and do something with it I think all credit to them, because it's not easy! Whenever I come across [peers] I will always make a point of congratulating them'. (Isla)
Generosity	'I think people who make crafts, I mean obviously there are always nice and nasty people, but generally it's like a little community everybody knows nobody is making a lot of money, but people do try and help, people help out a lot'. (Frank)
	'A lot of people come to me for advice of varying kinds and I give it freely'. (Holly)

<sup>&</sup>lt;sup>16</sup> Adopting Bourdieu's (1977) terminology, this chapter refers to habitus instead of 'established norms' or 'rules of the game' that influence the behaviours of community members. As such, this chapter utilises the concept in its meaning as a constraining social structure (Friedland, 2009), rather than as personal dispositions.

Once accepted within the community, there appears to be a sense as well as an expectation of camaraderie among craft makers. As will be discussed in the following section, this fellowship results in a number of benefits for craft entrepreneurs, specifically related to collaborative work and the sharing of resources. At the same time, the need to demonstrate solidarity with community members also imposes obligations. As Carol explains,

part of the reason that people ... come to see stuff in person [at shows] is so they can meet you ... hang out with you ... it seems really conceited to say ... but in this kind of small world, the people who are the makers and the designers ... are sort of famous and people know who you are and they are excited to meet you, and so if something happened and ... even if I just got somebody else to take my stuff ... and sell it for me, it kind of diminishes the experience ... [so] I can't afford to get sick ... because I have to do this [attend].

Individuals who fall short of the expectations of community members become the subjects of gossip. As Betty describes, 'it is such a small community and, like living in a small town, people like to gossip ... It's just human nature. People like to talk about bad stuff a lot more than they like to talk about good stuff'. As a result, as Julie notes, 'I'm quite ... careful about reputation ... [so] I need to be wary, because I don't want to upset people, so I have to be quite cautious about ... what I do'. As such, it is the fear of being punished by becoming the subjects of gossip which drives participants to comply with habitus. In other words, gossip represents a control mechanism which enforces compliance with existing conventions and ensures that community members seek to fit in.

Entrepreneurs who learn their crafts at a dedicated vocational school or university are introduced to the prevailing habitus by their instructors and more experienced peers. In Diana's words, 'it is drummed into you at art school'. In comparison, lateral entrants were found to engage in both relational learning (e.g. when they met other makers at shows) and experimentation, which helps them figure out accepted practices. Amy explains that she learned by 'just kind of ... looking at what other people were doing'. As Frank notes, 'having contacts is really useful because you learn, you get ideas and you can start doing things, you know, without treading on other people's toes ... 'cos I wouldn't want to do that'. This learning and habituation process can be time-consuming, but participants consider it essential for legitimising their operations. Furthermore, not only must craft entrepreneurs learn about the existing habitus; they must also be seen complying with it over time. In George's words, 'you need some patience for a start', as 'people want to see you out there for a few years' (Diana).

Interestingly, the data show that, because craft entrepreneurs adapt their behaviours to meet the expectations of community members, they frequently experience a feeling of 'standing out' from the habitus of other fields. In particular, the expectations and conventions in craft versus rural communities can cause difficulties for participants. In order to illustrate the tensions that result from the differences in habitus between co-existing fields, participants referred to the example of acceptable dress:

If you're ... kind of an arty, crafty person, there is an expectation that you're going to look interesting ... if I had ... a really normal hair cut and just standard glasses and wore neutral clothes, I think people would be a bit disappointed in a way 'cos there is this expectation that you're going to be quirky or different. (Carol)

I found when I first moved to the village, there was lots of people saying, 'Gosh all those [colourful] tights ... what in earth is that?' ... people will look and go, 'What's that all about?' (Diana)

The other day when I was on the bus, the driver asked where I was from ... and I was like '[Here!]' and he said 'Really? I thought you were like German or Polish or something' ... so I think maybe I'm a bit different to the people around me. (Amy)

Once again, in such situations, craft entrepreneurs choose to prioritise fitting into one field over the other. Complying with the habitus of craft communities is paramount to participants. There are, of course, other ways in which craft entrepreneurs stand out from the members of other communities. For example, innate attributes, including a different accent or birthplace, can lead entrepreneurs to stick out from their surroundings. As Julie explains,

people, sometimes are, because of my accent, put me in a pigeon hole where I don't belong and think that because I speak the way I do, that I come from a wealthy background, which of course I don't and they have misconceptions or bad perceptions of who the southern-accent-person is, you know, they assume all sorts of things, which of course aren't true.
Consequently, participants do their best to adapt their practices to facilitate fitting in with existing conventions and to conceal unconventional traits.

The analysis of data highlighted a range of strategic actions that participants carry out to change their conduct and meet the exceptions of community members (see Table 9.3).

Table 9.3	Strategic	Changes	of Practice
-----------	-----------	---------	-------------

Strategic Action	Representative Quotation
Pretending	'If you are running an online business the persona that you are presenting online, even if it does get a little bit personal, is still your kind of business persona. You paint this picture so they think that you are this very interesting person'. (Carol)
	'I can pretend to be very confident - doesn't mean I am but people believe it and that's what counts'. (Isla)
Restraining	'I don't swear as much I'm trying to be more professional and stuff. I'm still quite chatty, but I think that sort of make[s] people feel at ease'. (Betty)
	'Obviously with clients I have to be totally professional I can be more upfront with family, 'cos you know, you can say what you mean'. (Amy)
Concealing	'Generally, I put a positive spin on everything that is happening especially if I'm talking to people that I don't know particularly well'. (Emily)
	'You don't want people to know that, when nobody sees you, you sit about wearing slippers shaped like dogs [so] you take the picture of the bit of the house that looks nice so you do have to edit to a certain extent'. (Carol)
Performing	'At shows and things like that, I'm always delightful and happy and a little ray of sunshine, and at home I'm horrible [laughs]'. (Holly)
	'I'm aware, when I'm talking to somebody who's always lived here, that I probably sound quite posh, so I change how I speak to people'. (Frank)

Importantly, participants were found to adapt their practices only temporarily, rather than permanently changing their behaviours. Specifically, they engage in activities, such as pretending, restraining, concealing and performing, in an effort to, as Holly puts it, 'play the game'. Most of these changes of practice were subtle and it took some time for the researcher to realise that they were occurring. It was particularly helpful to observe participants at craft shows and to compare these behaviours to their conduct in the studios. This process revealed that participants constantly judged how community members expected them to behave and adjusted their practices accordingly.

This study also found there to be limits to the abilities of craft entrepreneurs to join certain fields by adapting to their habitus. For example, regarding access to local family networks, Isla notes that,

In the North East, the families ... are very close ... it comes through the sort of ... pit villages, there is a community network of inter-social family networks, they are so closed ... It sounds very strange, but they don't need you, an outsider. You are a nice enough person and, if they had met you in the street they would be perfectly polite and friendly, but you would never get close to them.

Thus, although strategic changes of practices allow participants to fit into certain fields, such as craft communities, this study finds that, in some cases, fields are too exclusive to join or the expectations of members are too high to fulfil.

## 9.4. Social Value Creation

The need to comply with the habitus of communities therefore greatly affects the behaviours of craft entrepreneurs. Significantly, the data demonstrate that it leads participants to engage in collaboration and social value creation. This section explores the reasons for participants to adopt these socially-oriented business practices and reviews the capital sharing of craft entrepreneurs.

The data analysis showed that entrepreneurs' embeddedness in craft communities can compel them to disregard 'traditional' business practices, such as fiercely competing for clients and resources to increase profits (Kirzner, 1973; Knight, 1921), and instead embrace social value creation by collaborating and sharing resources with community members. As Isla describes, '[despite] two very bad experiences in terms of time wasting and no feedback whatsoever ... there is an expectation that you will help and demonstrate what you can do'. In spite of such experiences, participants continue to support their peers, because 'we are all in the same boat and we just have to get on' (Diana). As Frank notes, 'everyone knows how difficult it is to make a living from it, but we're all doing something that ... we love doing and [that] gives pleasure to people, so we all help each other out'.

Participants were found to collaborate with other makers in various ways. Among other activities, they co-create craft items, such as 'collaborating on making some buttons' (Emily), and develop shared projects. As Carol recounts, 'I did the [yarn] club with ... [designer]. That was a collaborative effort, because she was designing the patterns and ... then I was dying things and giving them back to her to see what she thought'. Furthermore, 'people are [always] sharing ideas and helping other people develop their ideas' (George). Although participants emphasise that they try not to offer unsolicited advice, they are keen to help. As Julie explains, 'if people [peers] ask me, I'm delighted to help them ... [but] I always think they know what they're doing, so it's only when people say, "But, I can't do that", I think, "Wow, I'll help you". Above all, craft entrepreneurs support each other through the provision of free resources. For example, Carol explains that,

If you're providing yarn support, you are giving it away for free ... It's generally a designer ... who contacts me and asks if I provide yarn support and then ... [I'll] decide whether or not it's something that I would like to do ... If I like their patterns, then I'll offer them yarn support.

There are a number of reasons why craft entrepreneurs share their resources and collaborate (see Table 9.4). Although these reasons are individual and vary between participants, they embody general principles to which craft makers must adhere and which they seek to inculcate in others. Participants highlighted the need to advance their respective industries, ensure that skills get transferred to the next generation of makers and sustain a genuine interest in hand-crafted products. While participants generally believe in the importance of these goals, they acknowledge that they also feel compelled to pursue them, as illustrated by Emily: 'I think there is an expectation that, because you make art ... keeping it alive has to ... [come] before your business'. In other words, participants act selflessly not only out of an internal desire to do good, but also because of the perceived external pressure of their field's habitus.

Importantly, while participants appreciate help from members of the community, they generally do not expect reciprocity for their actions. In Betty's words, 'I won't necessarily get back something for my efforts ... [but] that's totally fine by me'. Furthermore, the data show that, while collaboration and resource sharing help participants expand their networks (social capital) and maintain their standing in the community (symbolic capital), these are secondary reasons for craft

entrepreneurs to adopt such practices. Instead, participants perceive industry preservation and the support of peers to be their social responsibilities. They are made aware of these conventions through the habituation process of learning to fit into their respective craft communities (see previous section). As Holly explains, 'I think that it's the responsibility of all the people [craft makers]'.

Motivation	Representative Quotation
Conservation	'People doing things well and passing on those skills that's something that is really important for our generations to come, that there are still people who do that'. (Julie)
	'If we don't kind of help each other out when we can there is not gonna be any makers left. It's all gonna be stuff that you buy at Primark for two quid'. (Carol)
Solidarity	'I think it's because we're all in the same boat, we are all individual makers, that you want to kind of help one another out'. (Emily)
	'You tend to find you want to help one another out, because you would hope that somebody else would do the same for you'. (Diana)
Altruism	'I give free time, free design-work, to help [others] It's not driven to pay the mortgage it's nice I've got a real buzz going on inside'. (Isla)
	'I think it's important to help people and to give back in some way'. (Betty)
Empathy	'I've spoken to the fashion class, 'cos I know how hard it is when you're a student I would have loved somebody to have told me how they set up on their own, so I think the knowledge I can offer will be invaluable'. (Amy)
	'When you've made things that you sell, you know how much work goes into [it]. I always feel like, empathy for people [peers] at craft fairs [so] I always talk to them and take their cards and show an interest'. (Betty)

Even when the open exchange of ideas between makers generates considerable potential for plagiarism, this study found that craft entrepreneurs are not discouraged from helping others. Table 9.5 highlights a number of reasons why participants are willing to accept such risk. They believe that, as their styles are developed through their past experiences and training (cultural capital) and their individual personalities, nobody else would be able to create the same products. In Julie's words, 'nobody can copy exactly what I do, 'cos they don't have the same history that I have of how I got to that place'. More importantly, 'the unwritten rule is that you, you don't ... sell the same things as another maker. You try to come up with your own original ideas' (Emily). All participants made similar statements, such as Isla:

the most important rule is [that] you don't pinch anybody's ideas ... you're always going to find people who have better ideas than you do. It's the way of the world, that's how things progress and move on. You think you might be doing quite well and then, you turn around, you look at somebody else and think, 'Crumbs, how did they come up with that? That's fantastic!' But what you can't do is ... plagiarise their idea into your work. You can be influenced by them, and there's difference of ... taking that influence and developing it in your own way - that's fine, but to take the idea and parody it ... that's an absolute no-no.

If craft community members are found plagiarising, they are excluded from the community. Such events occurred on a few, rare occasions during the course of the study, but the response of the craft communities was the same: the offending parties were flagged up in the online forums and community members warned one another not to buy their products. As Julie explains, 'nobody likes somebody who tries to copy, you know, and everybody becomes aware of that person'.

Table 9.5 Reasons	for Indifference	Concerning Plagiarism
-------------------	------------------	-----------------------

Reason	Representative Quotation
Individuality	'I'm not worried at all, 'cos hopefully there will be other customers out there There is obviously the chance that they [students] will then start making and selling, but they are still not me and they will never do what I do'. (Holly)
	'We're all art school trained, but you can see there are so many different styles and hopefully there will be something [customers] for everybody'. (Diana)
Experience	'Somebody else might see that and think, "Oh, right, I'll copy that", but they wouldn't be able to copy it with the same sort of precision and skill'. (George)
	'I've noticed on Pinterest people have pinned my wallets and put things like "I'll try later", but nobody has done it I don't know that anyone would be able to do quite as good a job, 'cos it has taken me years to develop it'. (Amy)
Integrity	'The integrity of the design has to be right and you can't copy other people, it's my design reputation on the line if I copy something else'. (Emily)
	'As a small business owner, you need to, you are the face of the brand so it's important to do things in a way that reflect [your] personal beliefs'. (Frank)
Exclusion	'I've been doing it [style] for years and there was somebody who had stuff [copies] up on Etsy I had customers messaging me saying, "Do you know this person is doing this?" So, I don't think you can [plagiarise], really because it is such a small community and people like to gossip about the negative things'. (Carol)
	'If you work over time, you get to know people If you behave unethically I can tell you it will come around and bite you in years to come'. (Isla)

As mentioned above, craft entrepreneurs were found to freely share a number of resources, including economic, cultural, social and symbolic capital. With regard to economic capital, there appears to be a convention among makers to buy each other's products. As Frank explains,

When I make money ... I do try to spend [some] ... on buying something from another crafter because ... we kind of have to encourage each other ... [so] I do have, mentally, this sort of unwritten rule that when I have a bit of extra money I buy something from another maker.

In addition, craft entrepreneurs were found to engage in charitable giving. Participants give free labour, as illustrated by Betty: 'I've volunteered ... there are these things in the [town hall] ... they have, like, an exhibition in the library ... I was making books with people that came in'. Furthermore, participants donate their products to help their communities. As Emily describes, 'I have done craft fairs where it's for charity'. Thus, the norms of craft communities can also lead entrepreneurs to support their local communities.

The sharing of cultural capital was also found to be a common practice among craft entrepreneurs. Participants engage in knowledge exchange and the cocreation of products through a variety of means, such as online forums, guild meetings and craft shows. As Diana explains, 'we'll discuss things or moan about things ... always kind of ask each other's advice ... online or if we go to a show'. Indeed, within craft communities it is common practice that 'everybody is ... sort of feeding off each other' (Betty) and 'passing on skills' (Holly). Craft entrepreneurs continue to benefit from this custom over time, as illustrated by Amy:

When I first started, in fact even now, when I go to a ... fair or anything, I always like to pick people's brains, find out how they got started and where they sell ... like their lines of sale. Yeah, I think it's really important to communicate with other makers.

Participants were found to perpetuate this practice. As George reports, 'a couple of them [hobbyists] are becoming self-employed and they're asking me for business advice ... [and] I'm happy to help'. In Betty's words, 'in a way, we all work as a team, you know, to do crafts ... we pass on tips and that sort of thing'.

Concerning the exchange of social capital, the data show that craft entrepreneurs become brokers of relationships, as well as receivers of such support from community members. As Isla notes, 'you pass on contacts, networks, that kind of thing'. For example, participants introduce peers to individuals who can help them develop their businesses, as explained by Frank: 'I said, "Well, I don't do marketing, but I know a lady in [Craft County] Business Club who's been quite helpful and supportive to me, personally. I'll put you in touch with her". This privilege was also extended by peers to participants and, surprisingly, was found to involve not just the exchange of supplier contacts, but also the sharing of customer contacts. As George recounts, 'like the other day, she [customer] said, "He [peer] just gave me your number, because he doesn't really make square baskets ... it's not really his thing", and so she got in contact with me'. It appears that membership in craft communities is generally beneficial in this regard, as illustrated by Emily:

I do actively say to them, "Oh, you should go check out this place" ... and I kind of encourage people to go look at other stuff, as well ... I don't know if they [peers] are doing the same thing for me, but I hope so [laughs].

Regarding the sharing of symbolic capital, participants were found to support less established peers by affiliating with them. As Julie explains, 'she [newcomer] is quite insecure, I think, about her own skills, so ... any time she publishes a new design ... I put it up on my website'. Similarly, participants have benefitted from association with more well-known community members. In Carol's words,

Being associated with designers helps ... she [designer] has quite a massive following ... she is really talented and has quite a lot of people that really love her designs and she used one of my yarns in one of her patterns, when I was first starting out and that ... helped to raise my profile quite a bit.

Furthermore, prominent craft entrepreneurs draw on their reputation to improve the standing of the community as a whole. Holly describes her motivations for organising a dedicated bear show as, 'For our little, small world, to try and expand it ... to get more people interested in what the artist bear is ... to kind of raise the awareness'. As an observer of the event, the researcher was made aware of the amount of effort required to stage a craft show, which appears to greatly outweigh

the sales (economic capital) and standing in the community (symbolic capital) that the organiser earns. Instead, craft entrepreneurs were found to conduct such activities to help new makers, as well as the whole community.

# 9.5. Analysis and Interpretation

Following the presentation of community-related findings, this section compares the results of this study with current scholarly debates, demonstrating their contributions to knowledge. This section analyses field-, habitus- and capital-related issues and, as such, mirrors the structure of the preceding sections.

# 9.5.1. Craft Fields

The analysis of data highlighted a number of characteristics of craft communities which have received scant attention in the academic literature. To date, craft research has predominately focused on marketing (Fillis, 2004; Paige and Littrell, 2002), branding (Beverland, 2005; Verhaal et al., 2015) and tourism-related topics (Cohen, 2001; Revilla and Dodd, 2003) and explored the engagement of entrepreneurs in local fields (Blundel, 2002; Tregear, 2005) and in high society (Slavich and Castellucci, 2016; Svejenova et al., 2007). In contrast, there is a paucity of research which considers the online activities of craft entrepreneurs (Kuhn and Galloway, 2015). Thus, this study expands knowledge by revealing that craft communities interact primarily through virtual channels. Advances in ICT allow entrepreneurs to connect with a variety of customers and peers while residing in rural areas (Anderson et al., 2016; Cumming and Johan, 2010). As a result of such developments, craft communities transcend national boundaries and unite enthusiasts from around the world.

Another distinguishing feature of craft communities is the blending of consumers and producers. Particularly in the context of craft brewing, prior research has highlighted the engagement of hobbyists in the creation of products (Biraglia and Kadile, 2017). However, most entrepreneurship studies into craft brewing focus on 'coopetition' (Mathias et al., In Press; Ofstein et al., 2017), which 'describes situations where two companies compete in some areas while cooperating in others

for mutual benefit' (Kuhn and Galloway, 2015: 576). Outside of the craft field, entrepreneurship research has demonstrated the value of brand communities, which connect consumers and producers, for facilitating opportunity discovery, evaluation and exploitation (Füller et al., 2008; Reuber and Fischer, 2011). Still, this research has paid little attention to community contexts in which producers collaborate with consumers who engage in their own product creation and may turn into potential competitors. It might be argued that open source software communities (Bianchi et al., 2012; O'Mahony, 2003; O'Mahony and Ferraro, 2007) are closely related, as they blend producers and consumers, but these contexts differ from craft communities in that they bring together developers who collaborate, but who do not usually buy each other's products. Thus, this study explores a context which has received limited consideration within the entrepreneurship literature.

The data also highlight that virtual communities can be close-knit despite being spread across large geographic areas. Indeed, the sense of shared history and collective identity expressed by participants, even with field members whom they have never physically met, echo the collective memories of generational units referred to by Lippmann and Aldrich (2016a). To date, entrepreneurship research has focused on the advantages that virtual embeddedness provides, including global market reach and the expansion of networks (Cumming and Johan, 2010; Smith et al., 2017), which have been shown to help entrepreneurs overcome the liabilities of newness (Morse et al., 2007). In comparison, little is known about the wider effects of embeddedness in international, online communities which unite a relatively small number of people and which follow distinctive rules of behaviour. Thus, given the increasing importance of online markets for entrepreneurs (Drori et al., 2009a; Fischer and Reuber, 2014; Hallen and Eisenhardt, 2012), the findings of this study make valuable contributions to knowledge.

Embeddedness in virtual communities has both positive and negative implications. While the resulting flexibility and connectivity allows craft entrepreneurs to freely choose their base of operation while staying connected with community members, it requires a considerable level of ICT literacy and can cause divided loyalties between local and virtual communities. Indeed, this study shows that embeddedness in an online community can lead entrepreneurs to prioritise one field over another. In the case of craft entrepreneurs, this phenomenon was found to create a degree of dis-embeddedness from the rural communities in which participants reside. As Dacin et al. (1999: 343) note, there is a lack of research exploring dis-embeddedness and 'the relative "strength[s]" of different types of embeddedness'. Prior studies argue that dis-embeddedness can free entrepreneurs from institutional pressures and therefore facilitate innovative and disruptive behaviours (Battilana et al., 2009; Leca and Naccache, 2006). The findings of this study indicate that dis-embeddedness from one field can also be an unintended consequence of commitment to another. By prioritising their embeddedness in virtual communities, entrepreneurs may lose the advantages associated with local embeddedness (Hindle, 2010; Ring et al., 2010), such as financial support (Khavul et al., 2013), contacts and advice (Jack and Anderson, 2002; Zhang et al., 2016). Thus, their embeddedness in craft communities can inhibit capital accumulation and conversion. This is a downside of embeddedness, which has attracted little prior attention in the literature. At the same time, by focusing their efforts on their craft communities, entrepreneurs can avoid spending resources on business unrelated activities. As such, it could be argued that dis-embeddedness from local fields can also facilitate strategic resource management.

Dis-embeddedness from local contexts is compounded by the fact that the products of craft entrepreneurs are so specialised that customers are unlikely to be found in their immediate surroundings. Misperceptions of local community members about the value of craft items further reinforce the predisposition of craft entrepreneurs to focus their activities on their virtual communities. Thus, virtual embeddedness appears to be significantly stronger than local embeddedness for participants. Given the amount of time that craft entrepreneurs spend at work and with members of their virtual communities, it might even be argued that virtual embeddedness is stronger than household embeddedness. However, as the problems and needs of household members affect the way participants develop their businesses (see Chapter 8), it is likely that valuations of these two contexts are highly idiosyncratic.

In drawing attention to the co-existing nature of different contexts and the resulting effects on entrepreneurs, this study also advances understanding of the concept of fields. Bourdieu (1977, 1984) argues that fields are organised in a hierarchical order within an overarching field of power (Emirbayer and Johnson, 2008; Maclean et al., 2014). While this perspective implies that there is tacit agreement among field members concerning the value of different fields (e.g. that conventional business fields dominate artistic fields, see De Clercq and Voronov, 2011), the findings of this study suggest that the valuation of distinct fields will differ between individual agents. Craft entrepreneurs regard the craft community and its prevailing habitus of camaraderie and generosity to be more important than business fields and their traditional competition- and profit-oriented practices. In comparison, local community members appear largely unaware of the 'hidden' (i.e. virtual) world of crafts and, as such, are likely to give higher value to other fields. In other words, this study suggests that field theory should account for the individual nature of agents' valuations of fields.

# 9.5.2. Habitus of Craft Communities

This study contributes to debates in the academic literature concerning the role of habitus and its impact on entrepreneurial practices (Barrett and Vershinina, 2017; McKeever et al., 2014; Patel and Conklin, 2009). First, the findings confirm that compliance with extant conventions within communities is a prerequisite for enjoying the benefits that membership provides (Coleman, 1988; Khavul et al., 2013; O'Mahony, 2003). Furthermore, this study produces deeper insights into the practices that entrepreneurs employ to achieve fitting in. To date, scholarly discussions have been primarily theoretical (Lounsbury and Glynn, 2001; Navis and Glynn, 2011; Zhao et al., 2017) and have suggested that entrepreneurs should seek to achieve a balance of 'fitting in' with dominant norms, as well as 'standing out' as acceptably deviant innovators (De Clercq and Voronov, 2009a, 2009b, 2009c). In comparison, the findings of this study indicate that conformity with habitus is significantly more important for craft entrepreneurs than achieving distinctiveness. Within craft communities, entrepreneurs must exhibit significant camaraderie, generosity and positivity to gain the support of community members. In addition, while compliance with conventions was shown to provide considerable advantages for participants, including access to resources, it was also found to put certain pressures on them.

The analysis of data highlighted that, when craft entrepreneurs fail to conform to the habitus of communities, they adopt a range of strategic actions, including concealing, performing, restraining and pretending, which allow them to adjust their conduct. Prior research proposes that entrepreneurs achieve conformity by adapting their practices permanently (isomorphism, see Tan et al., 2013), by superficially complying with mandated practices (decoupling, see Überbacher et al., 2015) or by integrating new ideas into existing practices (Slavich and Castellucci, 2016). Although the concept of decoupling may appear similar to the behaviours of craft entrepreneurs, it differs in that participants do not employ such strategic actions to hide activities which are closer aligned to their core business functions and goals (Suddaby et al., 2017). Instead, they seek to conceal certain traits and practices which fall short of the expectations of community members. In other words, their changes of behaviour are mostly temporary and geared towards other individual community members, rather than to the community at large. By allowing entrepreneurs to gain and maintain new collectors and invitations to craft shows and exhibitions in this way, these strategic changes of behaviour represent facilitators of capital conversion. Prior conceptual studies have proposed that when entrepreneurs are unaligned with the habitus of a community, they either seek to effect change within the field (Battilana, 2006) or obtain all the resources required to fit in (De Clercq and Voronov, 2009c). In contrast, craft entrepreneurs were found to circumvent issues by hiding unconventional behaviours and traits, as opposed to pursuing change in themselves or their communities. As such, this study portrays entrepreneurial behaviour in a more complex and perhaps less complimentary way than commonly done in entrepreneurship research (Calás et al., 2009).

The data shed new light on the habituation process of learning to conform to prevailing conventions (Anderson et al., 2010; Ocasio et al., 2016). Specifically, it shows that this process is not only social – dependent on the willingness of community members to impart information about the peculiarities and expectations of a field, but also temporal – requiring extended participation in collective activities and sharing of collective history (Lippmann and Aldrich, 2016a). According to Lave and Wenger (1991), habituation can be achieved through practice-based apprenticeships, which allow newcomers to become 'practice masters' (McLeod et

al., 2009; Slavich and Castellucci, 2016). While the analysis of data confirms that such situated learning is a viable route towards understanding how to fit in, it also shows that experimentation is an alternative technique which craft entrepreneurs employ to determine accepted practices. Despite their various efforts to achieve or portray conformity, there appear to be limits to the abilities of entrepreneurs to fit in by adapting their behaviours (Mair et al., 2012; Solebello et al., 2016). Among others, innate attributes, such as lack of kinship within a location, can prevent access to certain community contexts. In addition, embeddedness in multiple, co-existing fields can create difficulties for entrepreneurs (De Clercq and Voronov, 2009b; Kislov et al., In Press). By seeking to conform to the habitus of one field, entrepreneurs may defy that of another. Craft entrepreneurs consider the costs and benefits of fitting in and standing out to decide which field to prioritise.

Prior studies have demonstrated that 'reciprocity, mutuality and common purpose' (McKeever et al., 2015: 61) and commitment to shared values (Greenman, 2013) can result in the creation of community boundaries, which can lead to the exclusion of outsiders (Martí et al., 2013). In comparison, the habitus of craft communities was found to encourage the inclusion of new members, while simultaneously discouraging debate about prevailing conventions (Parkinson et al., 2017). In other words, in an effort to increase their size, craft communities actively promote membership on the condition that new members abide by established rules. Extant research has examined the negative impacts of embeddedness, including the effects of social and moral obligations on entrepreneurial activities (Jack and Anderson, 2002; Uzzi, 1997), but there have been calls for research to further explore the potential downsides of embeddedness (Battilana, 2006; Kilduff and Brass, 2010). Whereas prior studies have highlighted the influence of structural characteristics, including the tight-knit nature of communities, which can impede information flow (Kwon et al., 2013; Tan et al., 2013), this study emphasises the significance of relational dimensions. Specifically, it shows that, once entrepreneurs are embedded in their communities, their behaviour is monitored and their conformity with its habitus is enforced through the control mechanism of gossip. By limiting the behaviours of entrepreneurs, gossip represents an inhibitor of capital conversion. As Rooks et al. (2016: 113) note, gossip 'deters uncooperative and selfish behavior because other members ... [will] learn about this undesired behavior'. In other words, this study demonstrates that craft entrepreneurs conform to the habitus of communities because of the benefits that doing so provides and the penalties it precludes.

## 9.5.3. Capital Sharing

Although embeddedness can have negative impacts on entrepreneurs, this study also demonstrates that it provides a number of benefits, both to entrepreneurs and society at large. Specifically, this study shows that embeddedness in craft communities leads entrepreneurs to share available resources and to collaborate in an effort to create social value (Di Domenico et al., 2009; Korsgaard and Anderson, 2011). Craft entrepreneurs were found to share their various capitals and provide frequent, generous support to members of the community in order to sustain a genuine interest in hand-crafted goods and to ensure the preservation of craft skills over time. This study shows that craft entrepreneurs consider such collaborative and supportive practices a social responsibility, which is shared among community members and passed down from one generation to another. As such, these findings challenge descriptions of entrepreneurs as lone heroes who are driven by profit and competition (Knight, 1921; Schumpeter, 1942) and highlight the roles of social and temporal contexts in shaping entrepreneurial behaviour. Rather than merely following personal gratification, entrepreneurs are driven by their shared experiences and resulting sense of collective identity (Lippmann and Aldrich, 2016a). Thus, the findings confirm the need to look beyond economic motivations (Bourdieu, 1986) to understand the behaviours of entrepreneurs (Jayawarna et al., 2013; Rindova et al., 2009).

It is, of course, important to acknowledge that social entrepreneurship research recognises the ability of entrepreneurs to advance society (Chell et al., 2010; Zahra and Wright, 2016). However, while most social entrepreneurship scholars consider compassion and personal conviction to be the main causes of an entrepreneur's desire to create social value (Grimes et al., 2013; Miller et al., 2012), the findings show that socially oriented business practices can also be embraced as a result of the habitus of communities. This study therefore contributes to emerging

256

debates concerning the influence of context on encouraging social entrepreneurial practices (Shaw and de Bruin, 2013; Tapsell and Woods, 2010). Furthermore, it demonstrates that entrepreneurs who would not describe themselves as social entrepreneurs and who operate in an industry that is not typically associated with social entrepreneurial practices (Dacin et al., 2010), can still devote considerable effort to creating social value.

The data highlight the range of resources that craft entrepreneurs share in order to create social value (Ko and Liu, 2015; Smith and Stevens, 2010). To date, entrepreneurship research has mainly focused on exploring the commercial benefits that derive from business collaboration (Larson, 1991; Maurer and Ebers, 2006; Sorenson et al., 2008). Within the social entrepreneurship literature, access to financial support (Maclean et al., 2013) and tacit knowledge (Chalmers and Balan-Vnuk, 2013) have been emphasised as the main outcomes of resource exchanges. In comparison, the findings show that entrepreneurs embedded in craft communities share a much wider range of resources. They were found to provide (1) economic capital, in the form of products and money, (2) cultural capital, through passing on skills and ideas, (3) social capital, by sharing customer and supplier contacts, and (4) symbolic capital, by using their standing in the community to support new members. Consequently, the data demonstrate the need to account for a broader range of capitals when seeking to comprehend social entrepreneurial practices (Di Domenico et al., 2009; Nicolopoulou, 2014). They also highlight that the aggregate stock of capital available to entrepreneurs is augmented by the resources that community members are willing to share.

Prior research has argued that intellectual property protection is of significant importance in most industries (Kale et al., 2000; O'Mahony, 2003). In contrast, the findings show that craft entrepreneurs are not particularly concerned about the plagiarism of their ideas. Participants believe that their past experiences and training (cultural capital) lead them to create products which are so individual that they are difficult to copy, accentuating how an entrepreneur's personal history 'shapes their perception, strategies and actions' (Hess, 2004: 180). More importantly, they also feel protected by the habitus of the craft field, which sanctions members who plagiarise through loss of standing and even ostracism from the community (Kotha

and George, 2012). As such, the data highlight that 'the rules of the game' within community contexts can have not only restricting, but also liberating impacts on entrepreneurs. This suggests that the control mechanism of gossip, which helps enforce the community habitus, represents a facilitator of capital conversion, as well. Thus, these findings advance scholarly debates, which have focused on the negative effects of norms within communities on entrepreneurial activities (Coleman, 1988; Kandori, 1992; Parkinson et al., 2017). Given how insistent participants are that plagiarism is out of the question in craft communities, it might even be argued that this represents part of the doxa of the field. As Spigel (2013: 808) notes, many rules 'appear to be so natural and unremarkable that they are unquestionable' (i.e. doxic). However, as some community members do challenge this rule (albeit mainly unsuccessfully) and as participants point it out so vigorously, it does not appear to be invisible. As such, for the purposes of this study, it is considered to be part of the habitus, rather than the doxa of the craft field.

### 9.5.4. Craft Entrepreneurs as Social Entrepreneurs

Drawing on both academic research and the findings presented above, there are several reasons why participating craft entrepreneurs are considered social entrepreneurs. First, their embeddedness in virtual communities allows them to operate sustainable businesses in rural regions, which have suffered from deprivation. Although their interactions with local community members may be limited, running businesses in such deprived areas supports their economic recovery. According to McAuley and Fillis (2002; 2004), craft firms typically employ one to five workers, which indicates that, as the businesses of participating craft entrepreneurs grow over time, they are likely to generate employment opportunities. In addition, this study found that craft entrepreneurs promote tourism to their rural areas, as their virtual communities consist of devoted collectors who go to great lengths to meet craft makers in person. Furthermore, as Schwarz and Yair (2010) discuss, by offering classes, which help raise the confidence levels of attendees, and by encouraging social inclusion through shared activities, craft entrepreneurs create social value in their local communities.

Prior research also shows that craft makers are driven by a number of motivations, including the creation of products that are not only commercially viable, but which stay true to their artistic self-expression (Al-Dajani et al., 2015; Fillis, 2004). Thus, they must find ways to reconcile diverging interests (Eikhof and Haunschild, 2007), such as 'goals of lifestyle, community involvement and commercial success' (Tregear, 2005: 11), which is a challenge most social entrepreneurs face (Wry and York, 2017). This study reveals that, by sharing resources and collaborating with community members, craft entrepreneurs enable the continued existence of their respective industries and, by extension, have a positive impact on their rural surroundings. Furthermore, it could be argued that maintaining diverse types of artistic expression for future generations is a social good in itself. Consequently, while the label of craft entrepreneurs was employed throughout this study, participants are also considered to be social entrepreneurs.

# 9.6. Conclusion

This chapter set out to answer the research question: How and why does embeddedness in communities affect the practices of entrepreneurs? First, this study showed that entrepreneurs are embedded in multiple, co-existing fields, rather than a single, homogenous context. Participants were found to make deliberate decisions regarding prioritisation of these fields, because embeddedness in one has repercussions on their ability to conform to the expectations of another. Second, it highlighted that the habitus of communities affects entrepreneurial practices, as the need for entrepreneurs to fit in can override conventional market-driven behaviours. Indeed, this study found that this need drives entrepreneurs to employ strategic actions to temporarily adapt their behaviours in order to comply with community expectations. Third, it revealed that the habitus of the craft world compels entrepreneurs to openly share resources, including economic, cultural, social and symbolic capital. Field members are taught that such support is a social responsibility. Thus, this study shows that social value creation and collaboration can be embraced due to a community's habitus. As such, the findings indicate that social entrepreneurial behaviour can only be understood within context and that such

practices are more widely practiced than is currently assumed. This chapter therefore demonstrates the significant impacts of embeddedness in communities upon entrepreneurial activities.

#### **CHAPTER 10: CONCLUSION**

## **10.1. Introduction**

The aim of this thesis was to *develop a theory of practice for craft entrepreneurs* which explains how embeddedness in multiple contexts affects entrepreneurs' resource management practices. Given the prevailing focus on high-tech and highgrowth entrepreneurs within the academic literature (Welter et al., 2017), this research set out to examine craft entrepreneurs, who engage in everyday entrepreneurship and who represent an under-studied entrepreneurial sub-culture. The analysis of participants' experiences revealed that embeddedness in households and communities has significant impacts on their practices. In addition, it showed that entrepreneurs' actions within these contexts are affected by the wider temporal context in which they are embedded. In comparing the accounts of participants to the extant literature, it became clear that the empirical data, gathered through four rounds of interviews over the course of 18 months, could address enduring gaps in understanding of the influence of household and community members on the capital conversion and sharing practices of entrepreneurs. As such, this thesis advanced knowledge of how micro-, meso- and macro-level environments impact the practices of entrepreneurs. The purpose of this chapter is to draw conclusions from this study. It first presents key findings, before considering limitations of the research. Subsequently, it examines implications for theory, practice and policy and summarises the contributions of this study.

# 10.2. Key Findings

This thesis developed a theory of practice for craft entrepreneurs. The conceptual model presented in this study demonstrates that the accumulation of capital is a core entrepreneurial activity. To maximise their available stock of capital, craft entrepreneurs do not only engage in capital conversion processes, but also draw on the resources provided by household and community members. These meso- and macro-level contexts were shown to affect craft entrepreneurs in a number of ways, such as by both facilitating and inhibiting capital conversion processes. Indeed, the 'rules of behaviour' that prevail in these contexts were found to contend with the

micro-level habitus of maximising capital, leading craft entrepreneurs to prioritise different fields at different times. Furthermore, entrepreneurs' embeddedness within these contexts was found to rely upon their lived experiences and shared histories with field members. Thus, the conceptual model also highlighted the impact of an overarching temporal context on entrepreneurial practices. In order to present a more detailed account of the results of this study, this section reviews the findings related to individual, household and community contexts in turn.

Prior research has highlighted the crucial role that access to resources plays in facilitating entrepreneurial processes (Baker and Nelson, 2005; Mosakowski, 1998) and scholars have shown growing interest in investigating the different resources accessible to entrepreneurs (Newbert and Tornikoski, 2013; Ozdemir et al., 2016). While financial, human and social capital have received considerable attention (Bhagavatula et al., 2010; Davidsson and Honig, 2003), the importance of cultural and symbolic capital has largely been overlooked (Shaw et al., 2017). Furthermore, as most empirical studies focus on individual forms of capital (Gedajlovic et al., 2013; Marvel et al., 2016), there is limited understanding of how entrepreneurs convert the variety of resources available to them (Scott, 2012). This study therefore sought to advance understanding by addressing the research question: *How do entrepreneurs convert their economic, cultural, social and symbolic capital*? The findings highlighted the transmutable, multifaceted nature of different forms of capital.

Conversions of economic capital were found to be necessary activities, which craft entrepreneurs consider a 'means to an end' rather than a particularly enjoyable practice. Indeed, transformations of economic capital were shown to be part of a much larger process of capital conversion and it appeared that participants give no primacy to economic capital when transforming their various resources. Thus, the findings challenge Bourdieu's (1986: 252) proposition that economic capital is the underlying 'currency of exchange'. Instead, this study argued that a focus on economic capital is likely to obscure conversion processes that do not involve economic exchanges at all.

For example, this study demonstrated a range of conversions from cultural capital directly into social and symbolic capital. In so doing, the data confirmed the

high transmutability of cultural capital and its value to entrepreneurs in the cultural industries. Despite the scant attention it has received in the extant literature, objectified cultural capital was shown to play an important role and to be simultaneously convertible into profits, collectors and esteem. Furthermore, when developing their embodied cultural capital through training, craft entrepreneurs were found to prefer acquiring task-specific qualifications, rather than becoming 'jacks-of-all-trades' (Lazear, 2004: 208). Due to their unique set of lived experiences, entrepreneurs are able to create products that are individual in nature and connected to their personal histories, providing them with a distinct craft tradition, which imparts greater value to their cultural goods.

This study also provided substantial evidence of the importance of network ties to the accumulation of various resources. Guild members, followers, friends and suppliers were shown to supply finances, knowledge, contacts and publicity through both linear conversions and dynamic ones which provided access to multiple resources at once. As most of these network connections represent weak ties, the findings suggested that high intimacy and emotional investments are not essential for craft entrepreneurs to derive substantial benefits from their social capital. However, it was also shown that even these weak ties are founded through belonging to communities with shared histories – real or imagined (Anderson, 1983) – and therefore impossible to separate from entrepreneurs' temporal contexts.

Furthermore, symbolic capital was found to be a highly mutable resource. The analysis of data revealed that awards, exhibitions, publicity and reputation are particularly useful to craft entrepreneurs, as they help them generate sales, gain opportunities for learning and attract collectors, among other resources. Similar to the other forms of capital, symbolic capital was shown to produce several resources simultaneously, highlighting its value and explaining why participants spend so much time developing it. The amount of time required to attain it reflects the temporal embeddedness of symbolic capital. Awards, exhibitions, publicity and reputation are all dependent upon collective notions of value, which are created through shared experiences. While devoting time and effort towards gaining awards and presenting at exhibitions are relatively secure investments, publicity carries a degree of risk and reputation is highly dependent upon positive online reviews. Thus,

craft entrepreneurs were found to be strategic when selecting media outlets and especially careful when dealing with customers.

Importantly, this study also revealed a number of inhibitors to capital conversion. For instance, it was shown that transformations of (1) economic capital can be inhibited by associated costs and risks, (2) embodied cultural capital by lapses in memory and IT ineptitude, (3) social capital by the high expectations of contacts and (4) symbolic capital by a lack of self-confidence. These inhibitors can change in severity over time and can lead craft entrepreneurs to adapt their practices and draw on the support of household and community members to counteract their negative effects. Furthermore, this study demonstrated that valuation patterns and commitments to the stakes at play (illusio) are field-specific, which suggests that empirical results regarding capital conversions cannot be readily transferred between contexts.

Indeed, there is growing recognition among scholars that entrepreneurs need to be studied within their contexts to gain deeper insights into their behaviours (Autio et al., 2014; Zahra et al., 2014). Despite the resulting increase in context-focused research (Audretsch et al., 2010; Wright et al., 2014), understanding of social contexts outside the business world remains limited (Welter, 2011). Specifically, little is known about the role of households in the development of entrepreneurial businesses (Alsos et al., 2014; Gras and Nason, 2015). Thus, this thesis sought to contribute to scholarly debates by addressing the research question: *How do household members impact the practices of entrepreneurs?* The findings showed that cohabitants significantly and continuously influence entrepreneurial practices, both positively, by providing entrepreneurs with a range of material and non-material resources, and negatively, by burdening them with problems and requiring them to accommodate their needs.

As a result of this exploration, this study introduced an alternative approach to describing the households of entrepreneurs. Instead of distinguishing the socioeconomic features of households, it was proposed that research should draw on the subjective views of participants to provide insights into the meanings they associate with their households. In so doing, this study offered a new method for differentiating the household members of entrepreneurs in cultural industries.

264

Specifically, it developed a framework that distinguishes cohabitants based on their level of creativity and their degree of involvement with the work of entrepreneurs. This approach challenges current assumptions about the gendered nature of familial support and encourages research to consider context-specific factors which impact household members' support.

The data analysis also revealed a range of resources that become available to entrepreneurs through their households. In combining the concept of household capital (Rodriguez et al., 2009; Steier, 2009) with Bourdieu's (1986) capital theory, this study showed that cohabitants provide craft entrepreneurs with economic, cultural, social and symbolic capital. Whereas access to the financial means and social networks of household members was found to be of limited value to craft entrepreneurs, participants depend upon the skills and advice of cohabitants over time. Furthermore, it was shown that they appreciate the help of household members in maintaining and raising their status by attending craft exhibitions with them. Resources provided by household members are therefore part of the aggregate stock of capital available to entrepreneurs and they are drawn upon regularly. As such, the data suggested that the habitus within entrepreneurial households prescribes an open and free exchange of available resources to support entrepreneurs.

The findings also highlighted the importance of emotional support provided by household members to entrepreneurs (Brüderl and Preisendörfer, 1998; Renzulli and Aldrich, 2005). Specifically, the data showed that cohabitants help develop the self-efficacy, hope, optimism and resilience (i.e. psychological capital) of craft entrepreneurs and, in so doing, provide them with the emotional, cognitive and behavioural resources required to cope with stress and respond to challenging circumstances. Furthermore, as such support allows entrepreneurs to compensate for inhibitors of capital conversion, such as a lack of self-confidence, it represents a facilitator of conversion processes. It was shown that participants require emotional support most when they are anxious and that the key to successful support is the level of understanding household members have of the entrepreneur. These two aspects highlight the temporal embeddedness of entrepreneurial behaviours: not only is emotional support sought out during particular circumstances in an entrepreneur's life, but the value of this support is contingent upon a sense of collective memory which ensues from shared experiences.

In addition, the data analysis revealed that the needs, problems and opinions of household and family members can negatively impact entrepreneurs. Participants were expected to accommodate the scheduling and relocation needs of their partners and to contribute to childcare and domestic duties within the household. They achieve a work-family balance only by role-sharing with their partners and scaling back as much as possible within the home domain (Jennings and McDougald, 2007; Shelton, 2006). Still, major problems of cohabitants were found to lead entrepreneurs to temporarily relegate their businesses to a secondary role. As such, embeddedness in the household context can also generate inhibitors of capital conversion processes. Furthermore, it appeared that while household members are largely supportive of participants, wider family members are more likely to express doubt and criticise entrepreneurs, which can cause emotional disconnection. Thus, the findings demonstrated the considerable impact of household and family members on entrepreneurs, adding to research which recognises that the lines between work and family are often blurred (Aldrich and Cliff, 2003).

As entrepreneurs are affected by a 'multiplicity of contexts' (Welter, 2011: 168), it is important to consider how embeddedness in different contexts influences their practices. To date, entrepreneurship research has focused on the effects of structural and relational embeddedness (Batjargal, 2010; Hite and Hesterly, 2001) in network contexts (Jack, 2010; Slotte-Kock and Coviello, 2010). In comparison, understanding of the impacts of embeddedness in communities is limited (Korsgaard et al., 2015; McKeever et al., 2015). Thus, this thesis sought to answer the research question: *How and why does embeddedness in communities affect the practices of entrepreneurs*? By revealing the profound effects of such embeddedness on entrepreneurs, this study advanced knowledge in a number of ways.

The findings indicated that entrepreneurs are embedded in multiple, coexisting fields, rather than a single, homogenous context. If embeddedness in one context is stronger than in another, this was shown to have repercussions upon the memberships of entrepreneurs in divergent fields. Specifically, this study demonstrated that embeddedness in virtual communities can lead to a dis-

266

embeddedness from local communities and, consequently, a lack of local support, representing an inhibitor to capital accumulation and conversion. While disembeddedness may be an unintended consequence of commitment to another field, participants were found to make deliberate choices when prioritising one field over the other.

Furthermore, the data highlighted the significant effects of habitus on entrepreneurial practices. To date, entrepreneurship research has focused on Bourdieu's forms of capital (Lehner, 2014; Nicolopoulou, 2014; Vershinina et al., 2011) and has rarely examined habitus in depth (Anderson et al., 2010; McKeever et al., 2014). The findings demonstrated that the habitus of craft communities leads entrepreneurs to experience a sense of camaraderie and fellowship, rather than a desire to gain competitive advantage. Indeed, as a result of this collective identity and their shared situated experiences (Lippmann and Aldrich, 2016a), craft entrepreneurs were found to collaborate and share resources with community members to create social value (Mair and Martí, 2006; Smith and Stevens, 2010). This suggests that field-specific habitus can override conventional market-driven behaviours.

This study also showed that, instead of seeking to simultaneously fit in and stand out (De Clercq and Voronov, 2009a, 2009b), entrepreneurs prioritise achieving conformity over gaining distinctiveness within craft communities. The findings revealed that, when craft entrepreneurs fail to meet the expectations of community members, they employ a range of strategic actions, such as pretending, restraining and performing, which are hidden from the community and facilitate the conversion of capital. This further attests to the temporal embeddedness of entrepreneurial action: entrepreneurs shift their behaviours depending on the situation at hand, requiring attention to be paid to the temporal context in order to understand entrepreneurial decision-making and activities.

Furthermore, this study highlighted that, by achieving embeddedness in a camaraderie-oriented community, entrepreneurs can pursue their crafts without the worry of product plagiarism, as the habitus of the field prohibits such disreputable behaviour. Thus, the control mechanism of gossip does not only limit entrepreneurial practices, it also facilitates capital conversions. In addition, the habitus of craft

communities was shown to encourage the open sharing of economic, cultural, social and symbolic capital. As such, the aggregate stock of capital available to craft entrepreneurs is not only increased by embeddedness in households, but also by embeddedness in community contexts. There appeared to be a general understanding among craft community members that the provision of resources and generous support is a social responsibility, which ensures the preservation of craft skills and interest in hand-crafted products over time. In other words, this study revealed that collaboration and social value creation can be embraced due to the habitus of communities. This supports the view that social entrepreneurial behaviour cannot be fully understood without accounting for the contexts in which entrepreneurs are embedded (Lewis, 2013; Shaw and de Bruin, 2013).

Running throughout the findings regarding how micro-, meso- and macrolevel contexts affect entrepreneurs' behaviours is the impact of the temporal context on their practices. Entrepreneurs' adherence to different habitus depends upon their temporal contexts, which require them to adapt their behaviours and shift priorities. Their temporal embeddedness is also the very basis of their connections to social and spatial contexts and their actions within them. It is due to their personal histories, traditions, shared experiences and collective memories that entrepreneurs are able to learn, abide by and enact the habitus of these contexts. As such, the temporal context is the overarching context in which entrepreneurship is embedded.

#### 10.3. Limitations

While this study extended knowledge of the influence of contexts on entrepreneurial practices, it also carried with it limitations which must be taken into consideration. First, the research approach and phenomenological perspective of this study limits the generalisability of its results to wider populations (Berglund, 2015). Due to its small sample size (Smith, 2004) and its interest in providing nuanced understandings of the subjective views of participants (Cope, 2011), interpretative phenomenological research does not produce findings that are readily transferable across contexts (Smith et al., 2009). Instead, such research seeks to generate in-depth knowledge of individual behaviours in order to facilitate the inductive development of theory (McKeever et al., 2014).

Second, qualitative research risks falling into the trap of 'anecdotalism', which Silverman (2013: 284) describes as the 'use of data extracts (like spectacular quotes from participants) out of context'. Accordingly, this study sought to present several representative quotations for each analytical theme to demonstrate that the data presented were not exceptional or taken out of context. Furthermore, the exploratory nature of this study gave rise to alternative explanatory positions, necessitating the prioritisation of some interpretations over others. In order to combat this potential weakness, this study attempted to strike a balance between presenting the voices of participants and critically analysing their stories. In addition, the researcher continuously checked that emergent themes could not be better explained through alternative conceptualisations (Sandberg, 2005).

Third, it might be argued that the researcher's personal experience in the craft industry and the close rapport he developed with participants over time introduced a level of researcher bias. However, as Leitch et al. (2010) emphasise, phenomenological research embraces the subjective influence of scholars on the research process and considers it an opportunity to improve the quality of empirical findings. Furthermore, in line with the suggestions of prior research (Alvesson and Sandberg, 2013; Conklin, 2007, 2012), the researcher sought to be reflexive and present a balanced view of participants, portraying their darker practices along with their virtuous ones.

A fourth criticism that could be levelled against this study is for what it did not explore. While it investigated the embeddedness of craft entrepreneurs, it did not examine the structure of networks in which they are embedded. It also did not explore gender differences in depth, despite the majority of participants being female. Indeed, on the whole, this research focused on similarities between participants, rather than their differences. The decision to examine the aspects presented, instead of other possibilities, was based on the analysis of data. The data collected did not suggest itself for an analysis of structural aspects of networks. Similarly, differences between gender groups were not discussed in detail because few discernible differences were discovered between participants. It was also an effect of the data analysis process, which focused on themes that applied across participants, that this study primarily presented findings that were relevant to various craft entrepreneurs.

Despite these limitations, the phenomenological approach adopted by this research allowed for the development of an in-depth, comprehensive understanding of entrepreneurial behaviours that could not be reached through positivist methods. Thus, this study provided evidence of the effectiveness of interpretivist phenomenological analysis techniques for exploring the impacts of different contexts on entrepreneurial practices. The interest of phenomenological research in delving beyond numbers and 'objective' descriptions enabled this study to truly appreciate the multifaceted interplay between agents and contexts.

## **10.4.** Theoretical Implications

A number of avenues for future research emerged from this study. Most importantly, given that interpretivist phenomenological analyses are not generalisable (Lewis, 2015), it is recommended that future studies test the conceptual model presented in this study within different settings. Specifically, quantitative research could test the relationships and patterns identified for a larger sample and qualitative research could adapt the theory of practice for craft entrepreneurs by applying it to entrepreneurs in other contexts.

On the subject of capital conversion, this study highlighted the range of transformations that craft entrepreneurs conduct on a day-to-day basis and identified a number of conversions which have not previously been considered. It would therefore be interesting to examine the degree to which such transformations occur outside of the cultural industries and whether economic capital may play a similarly minor role in the capital conversion processes conducted within other contexts. While the high value of objectified cultural capital may be limited to the cultural industries, it is likely that institutionalised cultural capital, as well as awards and publicity, will affect a broad range of entrepreneurs. More detailed research into inhibitors to capital conversion is also encouraged. This study revealed a range of factors that can prevent entrepreneurs from transforming available resources into other forms of

capital, but these inhibitors have received scant consideration by entrepreneurship scholars, rendering further research in this area both timely and warranted.

Concerning the impact of household members on entrepreneurial practices, this study demonstrated the range of resources that cohabitants provide to entrepreneurs. While prior research has focused on financial, human and social capital, this study showed that household members help entrepreneurs develop cultural and symbolic capital, as well. Accordingly, it is suggested that future studies account for these forms of capital and incorporate them into conceptualisations of household capital. Psychological capital is also a recommended addition, as it successfully frames the emotional support that cohabitants provide entrepreneurs. Given that this study explored the experiences of sole proprietors, it would be interesting to investigate whether institutional entrepreneurs and entrepreneurial teams also rely on such support from their household members. Furthermore, the data illustrated differences between household and family members with regard to their emotional impact on craft entrepreneurs. Thus, it is recommended that future studies not only distinguish between cohabiting family and external family, but also purposefully explore the differences between cohabitants and wider kin.

The community-related findings of this study have additional theoretical implications. It was shown that the habitus of communities can lead craft entrepreneurs to create social value by freely sharing their resources with peers. Accordingly, scholars should investigate whether field-specific rules in other market-driven contexts may prescribe such counter-intuitive behaviours. As craft entrepreneurs are not usually associated with social entrepreneurship, this study suggests that social value creation be examined in other domains not traditionally associated with such activities. This study also demonstrated that craft entrepreneurs perform strategic actions to temporarily adapt their behaviour and hide non-conforming traits. Given the covert nature of this entrepreneurial practice, further exploration into this subject is needed to increase understanding of such deceptive behaviours. In addition, the findings showed that entrepreneurs face challenges when seeking to fit in: there are limits to entrepreneurs' abilities to adapt their behaviours and, by fitting in with the norms of one field, they may simultaneously defy those of

another. Thus, further research into the associated struggles and opportunities that emerge from embeddedness in co-existing fields is encouraged.

Based on the findings of this study, it is also suggested that future research explore the importance of the temporal context in more detail. The data showed that entrepreneurs' embeddedness within different contexts – their connections to field members and commitment to the rules and stakes of the game in each field – is grounded in their lived experiences. Embedding was revealed as a temporal process that relies upon shared history, memory and collective identities. Thus, future research should investigate the shared experiences, histories, traditions and memories that facilitate embeddedness. Furthermore, this study demonstrated that the social and spatial contexts in which entrepreneurs are embedded are themselves subsumed within an overarching temporal context which influences how entrepreneurs negotiate these fields at different times. As such, it is recommended that studies which seek to comprehend the dynamics between the divergent contexts in which entrepreneurs are embedded pay attention to the temporal dimension of this relationship.

By drawing on Bourdieu's work to develop a comprehensive theory of practice for craft entrepreneurs, which can be tested and adapted through future research in other settings, this study engages in accumulative theory development (Cope, 2005). First, this study confirmed the utility of Bourdieu's conceptual framework for investigations into entrepreneurial practices. It provided myriad examples that illustrate how the concepts of field, habitus and capital are applicable and advantageous to explorations of the various dimensions of context. Second, while Bourdieu (1977, 1984) proposes that struggles for position and power drive agents across contexts, this research showed that field-specific habitus can override such self-serving practices and lead to an open exchange of resources. This study therefore calls for research to incorporate habitus when adopting a Bourdieusian perspective in order to develop comprehensive understanding of the inner logics of various fields. Third, as Bourdieu's theory of practice pays little attention to the importance of emotions, it is suggested that future research which seeks to account for emotions and their impact on entrepreneurial practices include psychological capital in analyses of different forms of capital. Fourth, although Bourdieu

introduces the concept of capital conversion, he provides only a limited number of examples of this crucial activity. This study offered a much more detailed account and also identified a range of inhibitors and facilitators to capital conversion, which are likely to affect entrepreneurs in different contexts, as well. Compared to other research that has applied Bourdieu's theoretical framework to entrepreneurship settings, this study developed a more comprehensive theory of practice, as it accounts for micro-, meso- and macro-level environments and their effects on the behaviours of entrepreneurs. It is recommended that future research follow the example of this study to develop multilevel theories of entrepreneurial practice.

### **10.5. Implications for Practice and Policy**

As noted by Kenworthy and McMullan (2013), there is a need for entrepreneurship research to provide more practical knowledge which is useful to practitioners and policy makers. This section therefore presents recommendations for entrepreneurs, especially craft entrepreneurs, to aid them in developing their businesses and for policy makers to help guide the creation of programmes to support entrepreneurship.

The findings concerning capital conversions offer numerous suggestions for entrepreneurs. By highlighting the convertibility of economic, cultural, social and symbolic capital, this study indicated that a lack of individual resources should not discourage entrepreneurs from starting a business. For example, the data showed that sustainable development can be achieved without first obtaining high levels of economic capital, as other forms of capital can be just as, if not more effective at supporting entrepreneurial growth. Thus, entrepreneurs should actively leverage their available capitals in order to gain access to additional resources. Based on the lived experiences of craft entrepreneurs, this study proposes several specific recommendations for how entrepreneurs can increase and manage the various forms of capital at their disposal.

First, regarding economic capital, it is suggested that entrepreneurs critically evaluate whether the benefits of converting this capital outweigh potential risks involved. Indeed, while economic capital conversions were found to reward entrepreneurs with high returns on investment, such occurrences are not guaranteed and require strategic decision-making to ensure that resulting gains offset expenses.

273

Second, this study recommends that entrepreneurs, particularly in the craft industries, consciously develop and draw upon their cultural capital, as this resource is especially versatile. While embodied cultural capital is often tacit and its transformations organic, entrepreneurs are encouraged to purposefully work to overcome impediments to its conversion, including perceived ineptitude and lack of time, and prioritise the acquisition of crucial skills. Similarly, it is suggested that craft entrepreneurs appreciate their goods as worth more than direct sales and instead deploy them to generate further products, relationships and business recognition. As demonstrated by this study, this can be accomplished through drawing inspiration from existing cultural goods, making quality goods visible to potential connections and entering products into competitions. Furthermore, entrepreneurs are encouraged to be discerning when selecting educational qualifications to pursue. This study found that the quality of the course and its relevance to core business activities affect the convertibility of the degree.

Third, entrepreneurs are called upon to rise above feelings of inferiority, especially at the start-up stage, and pursue awards, exhibit their products, seek publicity and draw upon their reputation to increase their stock of available capital. Entrepreneurs should appreciate that doing so generates more than monetary rewards: such instances of symbolic capital also stimulate entrepreneurs' abilities to attain knowledge of industry norms, skills and ideas, new industry connections, exposure and prestige. In addition, entrepreneurs are encouraged to physically attend awards shows and seek out low-risk, reputable outlets (media and retail) for featuring their products and business. Furthermore, while it is recommended that entrepreneurs maintain confidence in their reputation in order to capitalise upon it, they should also be careful when interacting with customers, as negative or even neutral online feedback can have significant repercussions on business development.

Household-related findings carry further practical implications. Entrepreneurs are encouraged to view their household members not only as potential sources of funding for their ventures, but as partners in business growth. Cohabitants were shown to help craft entrepreneurs develop their ventures by providing free labour, offering business advice, sharing business contacts and promoting their businesses. Even if cohabitants are not knowledgeable about core business activities, this study demonstrated that their support can still benefit entrepreneurs. Understanding of the entrepreneurs and their businesses appears to determine the value of the emotional support that household members provide. Expressions of admiration, reassurance and encouragement (among others) facilitate the development of self-efficacy, hope, optimism and resilience only if cohabitants are seen to truly understand what the entrepreneur is going through. As such, entrepreneurs should ensure that they maintain strong household relationships. While these can distract from business operations, as household members' problems and needs can sometimes require entrepreneurs to prioritise family over business, they are central to entrepreneurs' emotional health and motivation to continue innovating and growing their ventures.

Based on the findings regarding the impacts of embeddedness in communities, nascent entrepreneurs are encouraged to allocate time towards learning community norms and discovering the consequences that may result from infringing upon these conventions (e.g. gossip or ostracism). Once entrepreneurs are aware of extant rules within the community, they can employ strategic actions to conceal traits that are incompatible with expectations. Pretending, restraining and performing practices were all found to help entrepreneurs fit into different community contexts. Furthermore, the findings suggest that craft entrepreneurs should approach their peers for advice and help, as sharing appears common in many craft communities. In addition, nascent entrepreneurs are urged to draw upon the close-knit networks of people available to them through online craft communities, who can provide helpful support without expecting or requiring reciprocity.

While the majority of this study's practical implications benefit entrepreneurs, policy makers should also heed its findings. For example, the data showed that there is a need for more relevant training courses. Instituting more comprehensive business courses for entrepreneurs coming from non-business backgrounds and (re-)introducing courses that are currently lacking in the cultural sector would be an important first step towards aiding the expansion of craft entrepreneurship. Similarly, the provision of networking opportunities for entrepreneurs would do much to facilitate the development of valuable social capital. Bringing together members of cultural fields from across the supply chain would make such events especially rewarding. In addition, the current reduction of funding for cultural institutions threatens the abilities of craft entrepreneurs to pursue vital activities, such as displaying their work in museums and meeting other makers in person. Thus, it is recommended that further support be provided to facilitate such pursuits. Given the lack of award shows in certain industries, the development of new competitions and associated prizes for underrepresented crafts (e.g. print making and yarn dying) is also suggested, as it would provide craft entrepreneurs with further opportunities to establish and maintain their reputation.

Furthermore, the findings demonstrated that both entrepreneurs and their household members could benefit from outside support. Policy makers are encouraged to help offset the loss of benefits that entrepreneurs encounter when they become self-employed. For example, subsidised childcare programmes and business relocation assistance would do much to relieve some of the household pressures entrepreneurs face. In addition, policy makers are urged to recognise the role of the household in successful entrepreneurship and provide entrepreneurial households with support mechanisms, such as paid leave to care for ill family members. The public sector could also emulate the emotional support that cohabitants provide by offering services, such as therapeutic counselling, to allow entrepreneurs who lack family support to gain similar assistance.

Finally, policy makers should appreciate the significant impacts that community members have upon entrepreneurial processes and work to aid supportive community activities. If attention continues to be limited to individual entrepreneurs, the potential for community members to foster entrepreneurial activities will not be exploited. The role of craft entrepreneurs in perpetuating the arts should also be acknowledged if we wish to ensure that their efforts continue over time. By supporting the resource sharing practices of craft entrepreneurs, policy makers can help sustain a range of artistic expressions as well as valuable cultural goods for future generations to come.

## 10.6. Contributions of the Research

This thesis made a number of contributions, which are summarised in this section to demonstrate the relevance and impact of this research. Most importantly, it proposed

a theory of practice for craft entrepreneurs which incorporates multiple levels of context to explain how social, spatial and temporal embeddedness affects entrepreneurs' resource management practices. Future research will be able to compare the conceptual model presented in this study to the experiences of entrepreneurs in other contexts and adapt it accordingly. As such, by drawing on Bourdieu's work to present a context-sensitive theory of practice that can be tested in other settings, this study engages in accumulative theory development (Cope, 2005).

Regarding capital conversions, this study brought together a diverse range of largely disjointed literature on the various resources accessible to entrepreneurs. It produced empirical evidence of capital transformations which have previously received only conceptual attention (cf. Adler and Kwon, 2002; Nicolopoulou, 2014) and revealed hitherto unidentified capital conversion processes. By presenting a range of transformations which result in the accumulation of multiple forms of capital, this study highlighted the complex nature of capital conversion, which has largely been overlooked. The findings also showed that non-material capitals are temporal products (outcomes of entrepreneurs' personal experiences and histories) and that the values of different forms of capital vary with context. While prior research emphasises the importance of economic and social capital for entrepreneurial success (Jonsson and Lindbergh, 2013; Steier and Greenwood, 2000), this study revealed that cultural and symbolic capital hold particularly high conversion rates and value for craft entrepreneurs. In addition, the findings showed that craft entrepreneurs give no primacy to economic capital when undertaking capital conversions. Thus, this study demonstrated that economic capital does not play a dominant role in all forms of entrepreneurship (Rindova et al., 2009).

Concerning the impact of households on entrepreneurial practices, this study found that entrepreneurs benefit not only from access to household financial, human and social capital (Dyer et al., 2014; Rodriguez et al., 2009), but also from cohabitants' advice, creative inspiration and support in maintaining reputation. Furthermore, this study demonstrated that cohabitants provide entrepreneurs with psychological capital by helping them develop their self-efficacy, hope, optimism and resilience. In so doing, it added to the growing body of research which calls attention to the role of emotions in entrepreneurship (Baron, 2008; Cardon et al., 2012), generated new insights into the interpersonal, rather than the intrapersonal, aspects of emotions (Jennings et al., 2015) and highlighted the temporal foundations of household relationships and support. Importantly, this study also showed that the problems and needs of household members can impede entrepreneurial processes. As such, it highlighted the importance of investigating the interwoven relationship between business and family to better understand entrepreneurial practices (Aldrich and Cliff, 2003; Jennings and McDougald, 2007).

With regard to embeddedness in communities, this study revealed that craft entrepreneurs carry out a range of strategic actions to temporarily adapt their conduct and disguise practices that fail to meet the expectations of community members. Thus, it shed light onto the temporal nature of entrepreneurs' adaptive behaviours, as well as their covert practices, which are often disregarded when portraying entrepreneurs in a more favourable light. In addition, this study illustrated that embeddedness in craft communities leads entrepreneurs to experience a sense of fellowship and camaraderie, instead of a desire to gain advantage over others. As such, it challenged descriptions of entrepreneurs as lone heroes who are driven by profit and competition (Knight, 1921; Schumpeter, 1942). Rather, this study showed that craft entrepreneurs collaborate and share their various resources with community members in pursuit of social value creation. As these socially-oriented practices are embraced in response to community norms and expectations, this study supports the view that social entrepreneurial behaviour cannot be fully understood without accounting for the social, spatial and temporal contexts in which entrepreneurs are embedded (Lewis, 2013; Shaw and de Bruin, 2013). Contexts do indeed matter and must be explored by entrepreneurship scholars.
## REFERENCES

- Aarstad J, Haugland SA and Greve A (2010) Performance Spillover Effects in Entrepreneurial Networks: Assessing a Dyadic Theory of Social Capital. *Entrepreneurship Theory and Practice* 34 (5): 1003-1019.
- Abbas M, Raja U, Darr W and Bouckenooghe D (2014) Combined Effects of Perceived Politics and Psychological Capital on Job Satisfaction, Turnover Intentions, and Performance. *Journal of Management* 40 (7): 1813-1830.
- Abeysekera I (In Press) How Best to Communicate Intangible Resources on Websites to Inform Corporate-Growth Reputation of Small Entrepreneurial Businesses. *Journal of Small Business Management*. Epub ahead of print 24 February 2017. DOI: 10.1111/jsbm.12320.
- Adams GA, King LA and King DW (1996) Relationships of job and family involvement, family social support, and work–family conflict with job and life satisfaction. *Journal of Applied Psychology* 81 (4): 411-420.
- Adler PS and Kwon S-W (2002) Social Capital: Prospects for a new Concept. Academy of Management Review 27 (1): 17-40.
- Ahl H (2006) Why Research on Women Entrepreneurs Needs New Directions. Entrepreneurship Theory and Practice 30 (5): 595-621.
- Ahuja G (2000) Collaboration Networks, Structural Holes, and Innovation: A Longitudinal Study. *Administrative Science Quarterly* 45 (3): 425-455.
- Ahuja G, Soda G and Zaheer A (2012) The Genesis and Dynamics of Organizational Networks. *Organization Science* 23 (2): 434-448.
- Akgün AA, Baycan-Levent T, Nijkamp P and Poot J (2011) Roles of Local and Newcomer Entrepreneurs in Rural Development: A Comparative Metaanalytic Study. *Regional Studies* 45 (9): 1207-1223.
- Al Ariss A and Syed J (2011) Capital Mobilization of Skilled Migrants: A Relational Perspective. *British Journal of Management* 22 (2): 286-304.
- Al-Dajani H, Carter S, Shaw E and Marlow S (2015) Entrepreneurship among the Displaced and Dispossessed: Exploring the Limits of Emancipatory Entrepreneuring. *British Journal of Management* 26 (4): 713–730.
- Aldrich HE and Cliff JE (2003) The pervasive effects of family on entrepreneurship: toward a family embeddedness perspective. *Journal of Business Venturing* 18 (5): 573-596.
- Aldrich HE and Fiol CM (1994) Fools Rush in? The Institutional Context of Industry Creation. *Academy of Management Review* 19 (4): 645-670.
- Aldrich HE and Kim PH (2007) Small worlds, infinite possibilities? How social networks affect entrepreneurial team formation and search. *Strategic Entrepreneurship Journal* 1 (1-2): 147-165.
- Aldrich HE and Zimmer C (1986) Entrepreneurship through Social Networks. In: Sexton DL and Smilor RW (eds) *The Art and Science of Entrepreneurship*. Cambridge, MA: Ballinger Publishing Company, pp. 3-23.
- Alexander VD (1996) Pictures at an Exhibition: Conflicting Pressures in Museums and the Display of Art. *American Journal of Sociology* 101 (4): 797-839.
- Allison TH, Davis BC, Short JC and Webb JW (2015) Crowdfunding in a Prosocial Microlending Environment: Examining the Role of Intrinsic Versus Extrinsic Cues. *Entrepreneurship Theory and Practice* 39 (1): 53-73.

- Alsos GA and Carter S (2006) Multiple business ownership in the Norwegian farm sector: Resource transfer and performance consequences. *Journal of Rural Studies* 22 (3): 313-322.
- Alsos GA, Carter S and Ljunggren E (2014) Kinship and business: how entrepreneurial households facilitate business growth. *Entrepreneurship & Regional Development* 26 (1-2): 97-122.
- Alsos GA, Isaksen EJ and Ljunggren E (2006) New Venture Financing and Subsequent Business Growth in Men- and Women-Led Businesses. *Entrepreneurship Theory and Practice* 30 (5): 667-686.
- Alvarez JL, Mazza C, Pedersen JS and Svejenova S (2005) Shielding Idiosyncrasy from Isomorphic Pressures: Towards Optimal Distinctiveness in European Filmmaking. *Organization* 12 (6): 863-888.
- Alvarez SA and Barney JB (2007) Discovery and creation: alternative theories of entrepreneurial action. *Strategic Entrepreneurship Journal* 1 (1-2): 11-26.
- Alvarez SA and Busenitz LW (2001) The entrepreneurship of resource-based theory. Journal of Management 27 (6): 755-775.
- Alvesson M and Sandberg J (2013) Has Management Studies Lost Its Way? Ideas for More Imaginative and Innovative Research. *Journal of Management Studies* 50 (1): 128-152.
- Amit R, MacCrimmon KR, Zietsma C and Oesch JM (2001) Does money matter? Wealth attainment as the motive for initiating growth-oriented technology ventures. *Journal of Business Venturing* 16 (2): 119-143.
- Amit R and Schoemaker PJH (1993) Strategic assets and organizational rent. *Strategic Management Journal* 14 (1): 33-46.
- Anand N and Watson MR (2004) Tournament Rituals in the Evolution of Fields: the Case of the Grammy Awards. *Academy of Management Journal* 47 (1): 59-80.
- Anderson AR (2000) Paradox in the periphery: an entrepreneurial reconstruction? Entrepreneurship & Regional Development 12 (2): 91-109.
- Anderson AR, Drakopoulou Dodd S and Jack SL (2010) Network practices and entrepreneurial growth. *Scandinavian Journal of Management* 26 (2): 121-133.
- Anderson AR and Jack SL (2002) The articulation of social capital in entrepreneurial networks: a glue or a lubricant? *Entrepreneurship & Regional Development* 14 (3): 193-210.
- Anderson AR, Jack SL and Drakopoulou Dodd S (2005) The Role of Family Members In Entrepreneurial Networks: Beyond the Boundaries of the Family Firm. *Family Business Review* 18 (2): 135-154.
- Anderson AR and Miller CJ (2003) "Class matters": human and social capital in the entrepreneurial process. *The Journal of Socio-Economics* 32 (1): 17-36.
- Anderson AR, Park J and Jack SL (2007) Entrepreneurial Social Capital: Conceptualizing Social Capital in New High-tech Firms. *International Small Business Journal* 25 (3): 245-272.
- Anderson AR, Wallace C and Townsend L (2016) Great Expectations or Small Country Living? Enabling Small Rural Creative Businesses with ICT. *Sociologia Ruralis* 56 (3): 450–468.
- Anderson B (1983) Imagined communities: Reflections on the origin and spread of *nationalism*. London: Verso.
- Anglin AH, McKenny AF and Short JC (In Press) The Impact of Collective Optimism on New Venture Creation and Growth: A Social Contagion

Perspective. *Entrepreneurship Theory and Practice*. Epub ahead of print 11 October 2016. DOI: 10.1111/etap.12256.

- Anheier HK, Gerhards J and Romo FP (1995) Forms of Capital and Social Structure in Cultural Fields: Examining Bourdieu's Social Topography. *American Journal of Sociology* 100 (4): 859-903.
- Antcliff V, Saundry R and Stuart M (2007) Networks and social capital in the UK television industry: The weakness of weak ties. *Human Relations* 60 (2): 371-393.
- Anthopoulou T (2010) Rural women in local agrofood production: Between entrepreneurial initiatives and family strategies. A case study in Greece. *Journal of Rural Studies* 26 (4): 394-403.
- Aram JD and Salipante PF (2003) Bridging Scholarship in Management: Epistemological Reflections. *British Journal of Management* 14 (3): 189-205.
- Arregle J-L, Batjargal B, Hitt MA, Webb JW, Miller T and Tsui AS (2015) Family Ties in Entrepreneurs' Social Networks and New Venture Growth. *Entrepreneurship Theory and Practice* 39 (2): 313–344.
- Arregle J-L, Hitt MA, Sirmon DG and Very P (2007) The Development of Organizational Social Capital: Attributes of Family Firms. *Journal of Management Studies* 44 (1): 73-95.
- Åstebro T, Chen J and Thompson P (2011) Stars and Misfits: Self-Employment and Labor Market Frictions. *Management Science* 57 (11): 1999-2017.
- Atkinson W (2010) Phenomenological Additions to the Bourdieusian Toolbox: Two Problems for Bourdieu, Two Solutions from Schutz. *Sociological Theory* 28 (1): 1-19.
- Audretsch DB, Falck O, Feldman MP and Heblich S (2010) Local Entrepreneurship in Context. *Regional Studies* 46 (3): 379-389.
- Audretsch DB, Keilbach M and Lehmann EE (2006) *Entrepreneurship and Economic Growth*. New York, NY: Oxford University Press.
- Austin J, Stevenson H and Wei-Skillern J (2006) Social and Commercial Entrepreneurship: Same, Different, or Both? *Entrepreneurship Theory and Practice* 30 (1): 1-22.
- Autio E, Kenney M, Mustar P, Siegel D and Wright M (2014) Entrepreneurial innovation: The importance of context. *Research Policy* 43 (7): 1097-1108.
- Avey JB, Luthans F and Youssef CM (2010) The Additive Value of Positive Psychological Capital in Predicting Work Attitudes and Behaviors. *Journal* of Management 36 (2): 430-452.
- Azmat F and Fujimoto Y (2016) Family embeddedness and entrepreneurship experience: a study of Indian migrant women entrepreneurs in Australia. *Entrepreneurship & Regional Development* 28 (9-10): 630-656.
- Bacq S and Janssen F (2011) The multiple faces of social entrepreneurship: A review of definitional issues based on geographical and thematic criteria. *Entrepreneurship & Regional Development* 23 (5-6): 373-403.
- Baines S and Wheelock J (1998) Working for Each Other: Gender, the Household and Micro-business Survival and Growth. *International Small Business Journal* 17 (1): 16-35.
- Baines S, Wheelock J and Gelder U (2003) *Riding the Rollercoaster: Family Life* and Self-employment. Bristol: The Policy Press.

- Baines S, Wheelock J and Oughton E (2002) A household based approach to the small business family. In: Fletcher DE (ed.) *Understanding the Small Business Family*. Abingdon: Routledge, pp. 168-179.
- Baker T and Nelson RE (2005) Creating Something from Nothing: Resource Construction through Entrepreneurial Bricolage. *Administrative Science Quarterly* 50 (3): 329-366.
- Bansal P and Corley K (2012) Publishing in AMJ–Part 7: What's Different about Qualitative Research? *Academy of Management Journal* 55 (3): 509-513.
- Barlow MA, Verhaal JC and Hoskins JD (In Press) Guilty by Association: Product-Level Category Stigma and Audience Expectations in the U.S. Craft Beer Industry. *Journal of Management*. Epub ahead of print 25 July 2016. DOI: 10.1177/0149206316657593.
- Barney J (1991) Firm Resources and Sustained Competitive Advantage. *Journal of Management* 17 (1): 99-120.
- Barney JB, Ketchen DJ and Wright M (2011) The Future of Resource-Based Theory: Revitalization or Decline? *Journal of Management* 37 (5): 1299-1315.
- Baron RA (2008) The Role of Affect in the Entrepreneurial Process. *Academy of Management Review* 33 (2): 328-340.
- Baron RA, Franklin RJ and Hmieleski KM (2013) Why Entrepreneurs Often Experience Low, Not High, Levels of Stress: The Joint Effects of Selection and Psychological Capital. *Journal of Management* 42 (3): 742–768.
- Baron RA and Markman G (2003) Beyond social capital: the role of entrepreneurs' social competence in their financial success. *Journal of Business Venturing* 18 (1): 41-60.
- Barrett R and Vershinina N (2017) Intersectionality of Ethnic and Entrepreneurial Identities: A Study of Post-War Polish Entrepreneurs in an English City. *Journal of Small Business Management* 55 (3): 430-443.
- Barth S, Barraket J, Luke B and McLaughlin J (2015) Acquaintance or partner? Social economy organizations, institutional logics and regional development in Australia. *Entrepreneurship & Regional Development* 27 (3-4): 219-254.
- Baruch-Feldman C, Brondolo E, Ben-Dayan D and Schwartz J (2002) Sources of social support and burnout, job satisfaction, and productivity. *Journal of Occupational Health Psychology* 7 (1): 84-93.
- Batjargal B (2003) Social Capital and Entrepreneurial Performance in Russia: A Longitudinal Study. *Organization Studies* 24 (4): 535-556.
- Batjargal B (2010) The effects of network's structural holes: polycentric institutions, product portfolio, and new venture growth in China and Russia. *Strategic Entrepreneurship Journal* 4 (2): 146-163.
- Battilana J (2006) Agency and Institutions: The Enabling Role of Individuals' Social Position. *Organization* 13 (5): 653-676.
- Battilana J, Leca B and Boxenbaum E (2009) How Actors Change Institutions: Towards a Theory of Institutional Entrepreneurship. *Academy of Management Annals* 3 (1): 65-107.
- Battilana J and Lee M (2014) Advancing Research on Hybrid Organizing Insights from the Study of Social Enterprises. *Academy of Management Annals* 8 (1): 397-441.
- Baumgartner D, Schulz T and Seidl I (2013) Quantifying entrepreneurship and its impact on local economic performance: A spatial assessment in rural Switzerland. *Entrepreneurship & Regional Development* 25 (3-4): 222-250.

- Becerra M, Lunnan R and Huemer L (2008) Trustworthiness, Risk, and the Transfer of Tacit and Explicit Knowledge Between Alliance Partners. *Journal of Management Studies* 45 (4): 691-713.
- Becker GS (1964) *Human Capital: A Theoretical Analysis with Special Reference to Education.* New York, NY: National Bureau of Economic Research.
- Becker HS (1978) Arts and Crafts. American Journal of Sociology 83 (4): 862-889.
- Belenzon S, Patacconi A and Zarutskie R (2016) Married to the firm? A large-scale investigation of the social context of ownership. *Strategic Management Journal* 37 (13): 2611-2638.
- Bell D and Jayne M (2010) The creative countryside: Policy and practice in the UK rural cultural economy. *Journal of Rural Studies* 26 (3): 209-218.
- Bennett J, License A and Tuck F (2014) *Measuring the Craft Economy*. Newcastle: Crafts Council.
- Berg JM (2016) Balancing on the Creative Highwire: Forecasting the Success of Novel Ideas in Organizations. *Administrative Science Quarterly* 61 (3): 433-468.
- Berglund H (2007) Researching Entrepreneurship as lived experience. In: Neergaard H and Ulhøi JP (eds) *Handbook of qualitative research in entrepreneurship*. Cheltenham: Edward Elgar, pp. 75-93.
- Berglund H (2015) Between cognition and discourse: phenomenology and the study of entrepreneurship. *International Journal of Entrepreneurial Behavior & Research* 21 (3): 472-488.
- Bertacchini EE and Borrione P (2011) The Geography of the Italian Creative Economy: The Special Role of the Design and Craft-based Industries. *Regional Studies* 47 (2): 135-147.
- Besser TL, Miller N and Perkins RK (2006) For the greater good: business networks and business social responsibility to communities. *Entrepreneurship & Regional Development* 18 (4): 321-339.
- Beverland MB (2005) Crafting Brand Authenticity: The Case of Luxury Wines. Journal of Management Studies 42 (5): 1003-1029.
- Bhagavatula S, Elfring T, van Tilburg A and van de Bunt GG (2010) How social and human capital influence opportunity recognition and resource mobilization in India's handloom industry. *Journal of Business Venturing* 25 (3): 245-260.
- Bianchi AJ, Kang SM and Stewart A (2012) The Organizational Selection of Status Characteristics: Status Evaluations in an Open Source Community. *Organization Science* 23 (2): 341-354.
- Biniari MG (2012) The Emotional Embeddedness of Corporate Entrepreneurship: The Case of Envy. *Entrepreneurship Theory and Practice* 36 (1): 141-170.
- Biraglia A and Kadile V (2017) The Role of Entrepreneurial Passion and Creativity in Developing Entrepreneurial Intentions: Insights from American Homebrewers. *Journal of Small Business Management* 55 (1): 170–188.
- Bird B and Brush C (2002) A Gendered Perspective on Organizational Creation. *Entrepreneurship Theory and Practice* 26 (3): 41-65.
- Bird M and Wennberg K (2014) Regional influences on the prevalence of family versus non-family start-ups. *Journal of Business Venturing* 29 (3): 421-436.
- Bird M and Wennberg K (2016) Why family matters: The impact of family resources on immigrant entrepreneurs' exit from entrepreneurship. *Journal of Business Venturing* 31 (6): 687-704.

- Birley S (1985) The role of networks in the entrepreneurial process. *Journal of Business Venturing* 1 (1): 107-117.
- BIS (2016) *Business population estimate for the UK and regions: 2016 statistical release*. London: Department for Business, Innovation and Skills.
- Bitektine A (2011) Toward a Theory of Social Judgments of Organizations: The Case of Legitimacy, Reputation, and Status. *Academy of Management Review* 36 (1): 151-179.
- Bittman M, England P, Sayer L, Folbre N and Matheson G (2003) When Does Gender Trump Money? Bargaining and Time in Household Work. *American Journal of Sociology* 109 (1): 186-214.
- Block JH, De Vries G, Schumann JH and Sandner P (2014) Trademarks and venture capital valuation. *Journal of Business Venturing* 29 (4): 525-542.
- Blundel R (2002) Network evolution and the growth of artisanal firms: a tale of two regional cheese makers. *Entrepreneurship & Regional Development* 14 (1): 1-30.
- Boden RJ and Nucci AR (1997) Counting the Self-Employed Using Household and Business Sample Data. *Small Business Economics* 9 (5): 427-436.
- Boswell WR and Olson-Buchanan JB (2007) The Use of Communication Technologies After Hours: The Role of Work Attitudes and Work-Life Conflict. *Journal of Management* 33 (4): 592-610.
- Bourdieu P (1977) *Outline of a Theory of Practice* (Nice R, Trans.). Cambridge, MA: Cambridge University Press.
- Bourdieu P (1984) *Distinction: a social critique of the judgement of taste* (Nice R, Trans.). Cambridge, MA: Harvard University Press.
- Bourdieu P (1986) The Forms of Capital (Nice R, Trans.). In: Richardson JG (ed.) *The handbook of theory and research for the sociology of education*. New York, NY: Greenwood Press, pp. 241–258.
- Bourdieu P (1996) On the Family as a Realized Category. *Theory, Culture & Society* 13 (3): 19-26.
- Bourdieu P and Wacquant LJD (1992) *An invitation to reflexive sociology*. Cambridge: Polity Press.
- Bowey JL and Easton G (2007) Entrepreneurial Social Capital Unplugged: An Activitybased Analysis. *International Small Business Journal* 25 (3): 273-306.
- Bradley SW, McMullen JS, Artz K and Simiyu EM (2012) Capital Is Not Enough: Innovation in Developing Economies. *Journal of Management Studies* 49 (4): 684-717.
- Brannon DL, Wiklund J and Haynie JM (2013) The Varying Effects of Family Relationships in Entrepreneurial Teams. *Entrepreneurship Theory and Practice* 37 (1): 107-132.
- Brinckmann J, Salomo S and Gemuenden HG (2011) Financial Management Competence of Founding Teams and Growth of New Technology-Based Firms. *Entrepreneurship Theory and Practice* 35 (2): 217-243.
- Brüderl J and Preisendörfer P (1998) Network support and the success of newly founded businesses. *Small Business Economics* 10 (3): 213-225.
- Brünjes J and Diez JR (2013) 'Recession push' and 'prosperity pull' entrepreneurship in a rural developing context. *Entrepreneurship & Regional Development* 25 (3-4): 251-271.
- Brush CG (1992) Research on Women Business Owners: Past Trends, a New Perspective and Future Directions. *Entrepreneurship: Theory & Practice* 16 (4): 5-30.

- Brush CG, Greene PG and Hart MM (2001) From initial idea to unique advantage: The entrepreneurial challenge of constructing a resource base. *Academy of Management Executive* 15 (1): 64-78.
- Bruton GD, Ahlstrom D and Li H-L (2010) Institutional Theory and Entrepreneurship: Where Are We Now and Where Do We Need to Move in the Future? *Entrepreneurship Theory and Practice* 34 (3): 421-440.
- BSA (2017) *Bourdieu Study Group Selected Publications*. Available from: http://www.bsabourdieu.wordpress.com/selected-publications/ [Accessed on: 01 April 2017].
- Bullough A, Renko M and Abdelzaher D (In Press) Women's Business Ownership: Operating Within the Context of Institutional and In-Group Collectivism. *Journal of Management*. Epub ahead of print 4 December 2014. DOI: 10.1177/0149206314561302.
- Bullough A, Renko M and Myatt T (2014) Danger Zone Entrepreneurs: The Importance of Resilience and Self-Efficacy for Entrepreneurial Intentions. *Entrepreneurship Theory and Practice* 38 (3): 473-499.
- Burns J, Gibbon C, Rosemberg C and Yair K (2012) *Craft in an Age of Change.* London: Crafts Council, Creative Scotland, Arts Council of Wales and Craft Northern Ireland.
- Burrell G and Morgan G (1979) *Sociological Paradigms and Organizational Analysis*. London: Heinemann Educational Books.
- Burt RS (1992) *Structural holes: the social structure of competition*. Cambridge, MA: Harvard University Press.
- Burt RS and Merluzzi J (2016) Network Oscillation. *Academy of Management Discoveries* 2 (4): 368-391.
- Busenitz LW, West III GP, Shepherd D, Nelson T, Chandler GN and Zacharakis A (2003) Entrepreneurship Research in Emergence: Past Trends and Future Directions. *Journal of Management* 29 (3): 285-308.
- Byrne O and Shepherd DA (2015) Different Strokes for Different Folks: Entrepreneurial Narratives of Emotion, Cognition, and Making Sense of Business Failure. *Entrepreneurship Theory and Practice* 39 (2): 375–405.
- Calás MB, Smircich L and Bourne KA (2009) Extending the Boundaries: Reframing "Entrepreneurship as Social Change" Through Feminist Perspectives. *Academy of Management Review* 34 (3): 552-569.
- Caputo RK and Dolinsky A (1998) Women's choice to pursue self-employment: The Role of financial and human capital of household members. *Journal of Small Business Management* 36 (3): 8-17.
- Cardon MS, Foo M-D, Shepherd D and Wiklund J (2012) Exploring the Heart: Entrepreneurial Emotion Is a Hot Topic. *Entrepreneurship Theory and Practice* 36 (1): 1-10.
- Cardon MS and Kirk CP (2015) Entrepreneurial Passion as Mediator of the Self-Efficacy to Persistence Relationship. *Entrepreneurship Theory and Practice* 39 (5): 1027-1050.
- Cardon MS, Wincent J, Singh J and Drnovsek M (2009) The nature and experience of entrepreneurial passion. *Academy of Management Review* 34 (3): 511-532.
- Cardon MS, Zietsma C, Saparito P, Matherne BP and Davis C (2005) A tale of passion: New insights into entrepreneurship from a parenthood metaphor. *Journal of Business Venturing* 20 (1): 23-45.

- Carlson DS, Hunter EM, Ferguson M and Whitten D (2014) Work–Family Enrichment and Satisfaction: Mediating Processes and Relative Impact of Originating and Receiving Domains. *Journal of Management* 40 (3): 845-865.
- Carroll GR and Swaminathan A (2000) Why the Microbrewery Movement? Organizational Dynamics of Resource Partitioning in the U.S. Brewing Industry. *American Journal of Sociology* 106 (3): 715-762.
- Carter NM, Gartner WB and Reynolds PD (1996) Exploring start-up event sequences. *Journal of Business Venturing* 11 (3): 151-166.
- Carter S (1998) Portfolio entrepreneurship in the farm sector: indigenous growth in rural areas? *Entrepreneurship & Regional Development* 10 (1): 17-32.
- Carter S (2011) The Rewards of Entrepreneurship: Exploring the Incomes, Wealth, and Economic Well-Being of Entrepreneurial Households. *Entrepreneurship Theory and Practice* 35 (1): 39-55.
- Carter S, Kuhl A, Marlow S and Mwaura S (2017) Households as a site of entrepreneurial activity. *Foundations and Trends in Entrepreneurship* 13 (2): 81-190.
- Carter S and Ram M (2003) Reassessing Portfolio Entrepreneurship. *Small Business Economics* 21 (4): 371-380.
- Carter S, Shaw E, Lam W and Wilson F (2007) Gender, Entrepreneurship, and Bank Lending: The Criteria and Processes Used by Bank Loan Officers in Assessing Applications. *Entrepreneurship Theory and Practice* 31 (3): 427-444.
- Carter S, Tagg S and Dimitratos P (2004) Beyond portfolio entrepreneurship: multiple income sources in small firms. *Entrepreneurship & Regional Development* 16 (6): 481-499.
- Casson M and Della Giusta M (2007) Entrepreneurship and Social Capital: Analysing the Impact of Social Networks on Entrepreneurial Activity from a Rational Action Perspective. *International Small Business Journal* 25 (3): 220-244.
- Chalmers DM and Balan-Vnuk E (2013) Innovating not-for-profit social ventures: Exploring the microfoundations of internal and external absorptive capacity routines. *International Small Business Journal* 31 (7): 785-810.
- Chalmers DM and Shaw E (2017) The endogenous construction of entrepreneurial contexts: A practice-based perspective. *International Small Business Journal* 35 (1): 19–39.
- Chandler GN and Hanks SH (1994) Market attractiveness, resource-based capabilities, venture strategies, and venture performance. *Journal of Business Venturing* 9 (4): 331-349.
- Chandler GN and Hanks SH (1998) An examination of the substitutability of founders human and financial capital in emerging business ventures. *Journal of Business Venturing* 13 (5): 353-369.
- Chaston I and Sadler-Smith E (2012) Entrepreneurial Cognition, Entrepreneurial Orientation and Firm Capability in the Creative Industries. *British Journal of Management* 23 (3): 415-432.
- Chell E (2013) Review of skill and the entrepreneurial process. *International Journal* of Entrepreneurial Behavior & Research 19 (1): 6-31.
- Chell E and Baines S (2000) Networking, entrepreneurship and microbusiness behaviour. *Entrepreneurship & Regional Development* 12 (3): 195-215.

- Chell E, Nicolopoulou K and Karataş-Özkan M (2010) Social entrepreneurship and enterprise: International and innovation perspectives. *Entrepreneurship & Regional Development* 22 (6): 485-493.
- Chen X-P, Yao X and Kotha S (2009) Entrepreneur Passion and Preparedness in Business Plan Presentations: A Persuasion Analysis of Venture Capitalists' Funding Decisions. *Academy of Management Journal* 52 (1): 199-214.
- Chia R (2004) Strategy-as-practice: reflections on the research agenda. *European* Management Review 1 (1): 29-34.
- Chia R and Holt R (2006) Strategy as Practical Coping: A Heideggerian Perspective. *Organization Studies* 27 (5): 635-655.
- Choi DY and Gray ER (2008) Socially responsible entrepreneurs: What do they do to create and build their companies? *Business Horizons* 51 (4): 341-352.
- Cholakova M and Clarysse B (2015) Does the Possibility to Make Equity Investments in Crowdfunding Projects Crowd Out Reward-Based Investments? *Entrepreneurship Theory and Practice* 39 (1): 145-172.
- Chudzikowski K and Mayrhofer W (2011) In search of the blue flower? Grand social theories and career research: The case of Bourdieu's theory of practice. *Human Relations* 64 (1): 19-36.
- Clarke J (2011) Revitalizing Entrepreneurship: How Visual Symbols are Used in Entrepreneurial Performances. *Journal of Management Studies* 48 (6): 1365-1391.
- Clemons EK, Gao GG and Hitt LM (2006) When Online Reviews Meet Hyperdifferentiation: A Study of the Craft Beer Industry. *Journal of Management Information Systems* 23 (2): 149-171.
- Cliff JE (1998) Does one size fit all? Exploring the relationship between attitudes towards growth, gender, and business size. *Journal of Business Venturing* 13 (6): 523-542.
- Cobb S (1976) Social Support as a Moderator of Life Stress. *Psychosomatic Medicine* 38 (5): 300-314.
- Cohen JH (2001) Textile, tourism and community development. *Annals of Tourism Research* 28 (2): 378-398.
- Cohen S and Wills TA (1985) Stress, social support, and the buffering hypothesis. *Psychological Bulletin* 98 (2): 310-357.
- Coleman JS (1988) Social Capital in the Creation of Human Capital. *American Journal of Sociology* 94: S95-S120.
- Coleman S (2007) The Role of Human and Financial Capital in the Profitability and Growth of Women-Owned Small Firms. *Journal of Small Business Management* 45 (3): 303-319.
- Conklin TA (2007) Method or Madness: Phenomenology as Knowledge Creator. Journal of Management Inquiry 16 (3): 275-287.
- Conklin TA (2012) Work Worth Doing: A Phenomenological Study of the Experience of Discovering and Following One's Calling. *Journal of Management Inquiry* 21 (3): 298-317.
- Cooper AC, Gimeno-Gascon FJ and Woo CY (1994) Initial human and financial capital as predictors of new venture performance. *Journal of Business Venturing* 9 (5): 371-395.
- Cope J (2005) Researching Entrepreneurship through Phenomenological Inquiry: Philosophical and Methodological Issues. *International Small Business Journal* 23 (2): 163-189.

- Cope J (2011) Entrepreneurial learning from failure: An interpretative phenomenological analysis. *Journal of Business Venturing* 26 (6): 604-623.
- Cope J, Jack SL and Rose MB (2007) Social Capital and Entrepreneurship: An Introduction. *International Small Business Journal* 25 (3): 213-219.
- Cordes CL and Dougherty TW (1993) A Review and an Integration of Research on Job Burnout. *Academy of Management Review* 18 (4): 621-656.
- Corley KG and Gioia DA (2004) Identity Ambiguity and Change in the Wake of a Corporate Spin-off. *Administrative Science Quarterly* 49 (2): 173-208.
- Coulson S (2012) Collaborating in a competitive world: musicians' working lives and understandings of entrepreneurship. *Work, Employment & Society* 26 (2): 246-261.
- Cox Pahnke E, McDonald R, Wang D and Hallen B (2015) Exposed: Venture Capital, Competitor Ties, and Entrepreneurial Innovation. *Academy of Management Journal* 58 (5): 1334-1360.
- Crook TR, Ketchen DJ, Combs JG and Todd SY (2008) Strategic resources and performance: a meta-analysis. *Strategic Management Journal* 29 (11): 1141-1154.
- Crossland P and Smith F (2002) Value creation in fine arts: a system dynamics model of inverse demand and information cascades. *Strategic Management Journal* 23 (5): 417-434.
- Cruz C, Justo R and De Castro JO (2012) Does family employment enhance MSEs performance?: Integrating socioemotional wealth and family embeddedness perspectives. *Journal of Business Venturing* 27 (1): 62-76.
- Cumming D and Johan S (2010) The Differential Impact of the Internet on Spurring Regional Entrepreneurship. *Entrepreneurship Theory and Practice* 34 (5): 857-883.
- Dacin MT, Dacin PA and Tracey P (2011) Social Entrepreneurship: A Critique and Future Directions. *Organization Science* 22 (5): 1203-1213.
- Dacin MT, Ventresca MJ and Beal BD (1999) The Embeddedness of Organizations: Dialogue & Directions. *Journal of Management* 25 (3): 317-356.
- Dacin PA, Dacin MT and Matear M (2010) Social Entrepreneurship: Why We Don't Need a New Theory and How We Move Forward From Here. *Academy of Management Perspectives* 24 (3): 37-57.
- Dakhli M and De Clercq D (2004) Human capital, social capital, and innovation: a multi-country study. *Entrepreneurship & Regional Development* 16 (2): 107-128.
- Dalpiaz E, Rindova V and Ravasi D (2016) Combining Logics to Transform Organizational Agency: Blending Industry and Art at Alessi. *Administrative Science Quarterly* 61 (3): 347-392.
- Dana L-P and Light I (2011) Two forms of community entrepreneurship in Finland: Are there differences between Finnish and Sámi reindeer husbandry entrepreneurs? *Entrepreneurship & Regional Development* 23 (5-6): 331-352.
- Danes SM, Stafford K, Haynes G and Amarapurkar SS (2009) Family Capital of Family Firms: Bridging Human, Social, and Financial Capital. *Family Business Review* 22 (3): 199-215.
- Daskalaki M, Hjorth D and Mair J (2015) Are Entrepreneurship, Communities, and Social Transformation Related? *Journal of Management Inquiry* 24 (4): 419-423.
- Davidsson P (2015) Entrepreneurial opportunities and the entrepreneurship nexus: A re-conceptualization. *Journal of Business Venturing* 30 (5): 674-695.
- Davidsson P and Honig B (2003) The role of social and human capital among nascent entrepreneurs. *Journal of Business Venturing* 18 (3): 301-331.

- Davidsson P, Low MB and Wright M (2001) Editor's Introduction: Low and MacMillan Ten Years On: Achievements and Future Directions for Entrepreneurship Research. *Entrepreneurship Theory and Practice* 25 (4): 5-15.
- Dawkins S, Martin A, Scott J and Sanderson K (2015) Advancing conceptualization and measurement of psychological capital as a collective construct. *Human Relations* 68 (6): 925–949.
- DCMS (2016) *Creative Industries Economic Estimates January 2016*. London: Department for Culture, Media and Sport.
- De Bettignies J-E (2008) Financing the Entrepreneurial Venture. *Management Science* 54 (1): 151-166.
- De Carolis DM and Saparito P (2006) Social Capital, Cognition, and Entrepreneurial Opportunities: A Theoretical Framework. *Entrepreneurship Theory and Practice* 30 (1): 41-56.
- De Clercq D and Honig B (2011) Entrepreneurship as an integrating mechanism for disadvantaged persons. *Entrepreneurship & Regional Development* 23 (5-6): 353-372.
- De Clercq D, Lim DSK and Oh CH (2013) Individual-Level Resources and New Business Activity: The Contingent Role of Institutional Context. *Entrepreneurship Theory and Practice* 37 (2): 303-330.
- De Clercq D and Voronov M (2009a) The Role of Cultural and Symbolic Capital in Entrepreneurs' Ability to Meet Expectations about Conformity and Innovation. *Journal of Small Business Management* 47 (3): 398-420.
- De Clercq D and Voronov M (2009b) The Role of Domination in Newcomers' Legitimation as Entrepreneurs. *Organization* 16 (6): 799-827.
- De Clercq D and Voronov M (2009c) Toward a Practice Perspective of Entrepreneurship: Entrepreneurial Legitimacy as Habitus. *International Small Business Journal* 27 (4): 395-419.
- De Clercq D and Voronov M (2011) Sustainability in entrepreneurship: A tale of two logics. *International Small Business Journal* 29 (4): 322-344.
- Deephouse DL (1999) To be different, or to be the same? It's a question (and theory) of strategic balance. *Strategic Management Journal* 20 (2): 147-166.
- Deephouse DL (2000) Media Reputation as a Strategic Resource: An Integration of Mass Communication and Resource-Based Theories. *Journal of Management* 26 (6): 1091-1112.
- Deetz S (1996) Describing Differences in Approaches to Organization Science: Rethinking Burrell and Morgan and Their Legacy. *Organization Science* 7 (2): 191-207.
- DeFillippi R, Grabher G and Jones C (2007) Introduction to paradoxes of creativity: managerial and organizational challenges in the cultural economy. *Journal of Organizational Behavior* 28 (5): 511-521.
- Defra (2015) United Kingdom Rural Development Programme (Regional) -England. London: Department for Environment, Food & Rural Affairs.
- Defra (2016) *Defining rural areas*. London: Department for Environment, Food & Rural Affairs.
- DeMartino R and Barbato R (2003) Differences between women and men MBA entrepreneurs: exploring family flexibility and wealth creation as career motivators. *Journal of Business Venturing* 18 (6): 815-832.

- De Massis A, Frattini F, Kotlar J, Petruzzelli AM and Wright M (2016) Innovation through tradition: lessons from innovative family businesses and directions for future research. *Academy of Management Perspectives* 30 (1): 93-116.
- Denzin NK and Lincoln YS (1994) Introduction: Entering the Field of Qualitative Research. In: Denzin NK and Lincoln YS (eds) *Handbook of Qualitative Research*. London: Sage.
- Desan MH (2013) Bourdieu, Marx, and Capital: A Critique of the Extension Model. *Sociological Theory* 31 (4): 318-342.
- Di Domenico M, Tracey P and Haugh H (2009) The Dialectic of Social Exchange: Theorizing Corporate–Social Enterprise Collaboration. *Organization Studies* 30 (8): 887-907.
- DiMaggio PJ (1979) On Pierre Bourdieu. *American Journal of Sociology* 84 (6): 1460-1474.
- DiMaggio PJ (1982) Cultural entrepreneurship in nineteenth-century Boston: the creation of an organizational base for high culture in America. *Media, Culture & Society* 4 (1): 33-50.
- Dimov D (2010) Nascent Entrepreneurs and Venture Emergence: Opportunity Confidence, Human Capital, and Early Planning. *Journal of Management Studies* 47 (6): 1123-1153.
- Dimov D, Shepherd DA and Sutcliffe KM (2007) Requisite expertise, firm reputation, and status in venture capital investment allocation decisions. *Journal of Business Venturing* 22 (4): 481-502.
- Discua Cruz A, Howorth C and Hamilton E (2013) Intrafamily Entrepreneurship: The Formation and Membership of Family Entrepreneurial Teams. *Entrepreneurship Theory and Practice* 37 (1): 17-46.
- Distelberg B and Sorenson RL (2009) Updating Systems Concepts in Family Businesses. *Family Business Review* 22 (1): 65-81.
- Djupdal K and Westhead P (2015) Environmental certification as a buffer against the liabilities of newness and smallness: Firm performance benefits. *International Small Business Journal* 33 (2): 148-168.
- Donckels R and Lambrecht J (1995) Networks and small business growth: An explanatory model. *Small Business Economics* 7 (4): 273-289.
- Douglas EJ (2013) Reconstructing entrepreneurial intentions to identify predisposition for growth. *Journal of Business Venturing* 28 (5): 633-651.
- Downing S (2005) The Social Construction of Entrepreneurship: Narrative and Dramatic Processes in the Coproduction of Organizations and Identities. *Entrepreneurship Theory and Practice* 29 (2): 185-204.
- Drakopoulou Dodd S (2002) Metaphors and meaning: A grounded cultural model of us entrepreneurship. *Journal of Business Venturing* 17 (5): 519-535.
- Drakopoulou Dodd S (2014) Roots radical place, power and practice in punk entrepreneurship. *Entrepreneurship & Regional Development* 26 (1-2): 165-205.
- Drakopoulou Dodd S and Anderson AR (2007) Mumpsimus and the Mything of the Individualistic Entrepreneur. *International Small Business Journal* 25 (4): 341-360.
- Drakopoulou Dodd S, McDonald S, McElwee G and Smith R (2014) A Bourdieuan Analysis of Qualitative Authorship in Entrepreneurship Scholarship. *Journal* of Small Business Management 52 (4): 633–654.

- Drakopoulou Dodd S, Pret T and Shaw E (2016) Advancing Understanding of Entrepreneurial Embeddedness: Forms of Capital, Social Contexts and Time. In: Welter F and Gartner WB (eds) *A Research Agenda for Entrepreneurship and Context*. Cheltenham: Edward Elgar, pp. 120-133.
- Drori I, Honig B and Ginsberg A (2006) *Transnational entrepreneurship: Toward a unifying theoretical framework.* Paper presented at the Academy of Management, Atlanta, GA.
- Drori I, Honig B and Sheaffer Z (2009a) The Life Cycle of an Internet Firm: Scripts, Legitimacy, and Identity. *Entrepreneurship Theory and Practice* 33 (3): 715-738.
- Drori I, Honig B and Wright M (2009b) Transnational Entrepreneurship: An Emergent Field of Study. *Entrepreneurship Theory and Practice* 33 (5): 1001-1022.
- Dubini P and Aldrich H (1991) Personal and extended networks are central to the entrepreneurial process. *Journal of Business Venturing* 6 (5): 305-313.
- Durkheim E (1964) *The division of labor in society* (Halls WD, Trans.). New York, NY: The Free Press.
- Dushnitsky G (2010) Entrepreneurial Optimism in the Market for Technological Inventions. *Organization Science* 21 (1): 150-167.
- Dyer JH and Singh H (1998) The Relational View: Cooperative Strategy and Sources of Interorganizational Competitive Advantage. *Academy of Management Review* 23 (4): 660-679.
- Dyer WG (2003) The Family: The Missing Variable in Organizational Research. Entrepreneurship Theory and Practice 27 (4): 401-416.
- Dyer WG (2006) Examining the "Family Effect" on Firm Performance. *Family Business Review* 19 (4): 253-273.
- Dyer WG, Nenque E and Hill EJ (2014) Toward a Theory of Family Capital and Entrepreneurship: Antecedents and Outcomes. *Journal of Small Business Management* 52 (2): 266-285.
- Easterby-Smith M, Thorpe R and Jackson P (2012) *Management Research* (4th ed.). London: Sage.
- Ebben J and Johnson A (2006) Bootstrapping in small firms: An empirical analysis of change over time. *Journal of Business Venturing* 21 (6): 851-865.
- Eddleston KA and Powell GN (2012) Nurturing Entrepreneurs' Work–Family Balance: A Gendered Perspective. *Entrepreneurship Theory and Practice* 36 (3): 513-541.
- Edelman LF, Manolova T, Shirokova G and Tsukanova T (2016) The impact of family support on young entrepreneurs' start-up activities. *Journal of Business Venturing* 31 (4): 428-448.
- Edmondson AC and McManus SE (2007) Methodological Fit in Management Field Research. *Academy of Management Review* 32 (4): 1155-1179.
- Edwards P, Sengupta S and Tsai C-J (2010) The context-dependent nature of small firms' relations with support agencies: A three-sector study in the UK. *International Small Business Journal* 28 (6): 543-565.
- Eikhof DR and Haunschild A (2007) For art's sake! Artistic and economic logics in creative production. *Journal of Organizational Behavior* 28 (5): 523-538.
- Eisenhardt KM and Schoonhoven CB (1996) Resource-Based View of Strategic Alliance Formation: Strategic and Social Effects in Entrepreneurial Firms. *Organization Science* 7 (2): 136-150.
- Elam AB (2008) *Gender and entrepreneurship: A multilevel theory and analysis.* Cheltenham: Edward Elgar.

- Elfring T and Hulsink W (2007) Networking by Entrepreneurs: Patterns of Tie-Formation in Emerging Organizations. *Organization Studies* 28 (12): 1849-1872.
- Ellickson RC (2006) Unpacking the Household: Informal Property Rights around the Hearth. *The Yale Law Journal* 116 (2): 226-328.
- Elsbach KD (2009) Identity affirmation through `signature style': A study of toy car designers. *Human Relations* 62 (7): 1041-1072.
- Emirbayer M and Johnson V (2008) Bourdieu and Organizational Analysis. *Theory and Society* 37 (1): 1-44.
- Emirbayer M and Mische A (1998) What Is Agency? *American Journal of Sociology* 103 (4): 962-1023.
- Ertug G, Yogev T, Lee Y and Hedstrom P (2016) The Art of Representation: How Audience-Specific Reputations Affect Success in the Contemporary Art Field. *Academy of Management Journal* 59 (1): 113-134.
- Estrin S, Mickiewicz T and Stephan U (2016) Human capital in social and commercial entrepreneurship. *Journal of Business Venturing* 31 (4): 449-467.
- Evens TMS (1999) Bourdieu and the Logic of Practice: Is All Giving Indian-Giving or is "Generalized Materialism" Not Enough? *Sociological Theory* 17 (1): 3-31.
- Evert RE, Martin JA, McLeod MS and Payne GT (2015) Empirics in Family Business Research: Progress, Challenges, and the Path Ahead. *Family Business Review* 29 (1): 17-43.
- Feldman M and Worline M (2016) The Practicality of Practice Theory. *Academy of Management Learning & Education* 15 (2): 304-324.
- Feldman MS and Orlikowski WJ (2011) Theorizing Practice and Practicing Theory. Organization Science 22 (5): 1240-1253.
- Fiet JO, Norton Jr WI and Clouse VGH (2013) Search and discovery by repeatedly successful entrepreneurs. *International Small Business Journal* 31 (8): 890-913.
- Fillis I (2004) The Internationalizing Smaller Craft Firm: Insights from the Marketing/ Entrepreneurship Interface. *International Small Business Journal* 22 (1): 57-82.
- Fine C (2010) Delusions of Gender. London: Icon Books.
- Finlay L (2006) 'Rigour', 'Ethical Integrity' or 'Artistry'? Reflexively Reviewing Criteria for Evaluating Qualitative Research. *The British Journal of Occupational Therapy* 69 (7): 319-326.
- Finlay L (2009) Debating Phenomenological Research Methods. *Phenomenology & Practice* 3 (1): 6-25.
- Fiol CM (1991) Managing Culture as a Competitive Resource: An Identity-Based View of Sustainable Competitive Advantage. *Journal of Management* 17 (1): 191-211.
- Fischer E and Reuber AR (2014) Online entrepreneurial communication: Mitigating uncertainty and increasing differentiation via Twitter. *Journal of Business Venturing* 29 (4): 565-583.
- Fischer E and Reuber R (2007) The Good, the Bad, and the Unfamiliar: The Challenges of Reputation Formation Facing New Firms. *Entrepreneurship Theory and Practice* 31 (1): 53-75.
- Fisher G (2012) Effectuation, Causation, and Bricolage: A Behavioral Comparison of Emerging Theories in Entrepreneurship Research. *Entrepreneurship Theory and Practice* 36 (5): 1019-1051.
- Fisher G, Kotha S and Lahiri A (2016) Changing with the Times: An Integrated View of Identity, Legitimacy and New Venture Life Cycles. *Academy of Management Review* 41 (3): 383-409.

- Fletcher DE (2010) 'Life-making or risk taking'? Co-preneurship and family business start-ups. *International Small Business Journal* 28 (5): 452-469.
- Foley D and O'Connor AJ (2013) Social Capital and the Networking Practices of Indigenous Entrepreneurs. *Journal of Small Business Management* 51 (2): 276-296.
- Foo M-D (2011) Emotions and Entrepreneurial Opportunity Evaluation. Entrepreneurship Theory and Practice 35 (2): 375-393.
- Foo M-D, Uy MA and Baron RA (2009) How do feelings influence effort? An empirical study of entrepreneurs' affect and venture effort. *Journal of Applied Psychology* 94 (4): 1086-1094.
- Foo M-D, Wong PK and Ong A (2005) Do others think you have a viable business idea? Team diversity and judges' evaluation of ideas in a business plan competition. *Journal of Business Venturing* 20 (3): 385-402.
- Forbes DP (2005) The Effects of Strategic Decision Making on Entrepreneurial Self-Efficacy. *Entrepreneurship Theory and Practice* 29 (5): 599-626.
- Forson C (2013) Contextualising migrant black business women's work-life balance experiences. *International Journal of Entrepreneurial Behavior & Research* 19 (5): 460-477.
- Foss NJ, Klein PG, Kor YY and Mahoney JT (2008) Entrepreneurship, subjectivism, and the resource-based view: toward a new synthesis. *Strategic Entrepreneurship Journal* 2 (1): 73-94.
- Foster P, Borgatti SP and Jones C (2011) Gatekeeper search and selection strategies: Relational and network governance in a cultural market. *Poetics* 39 (4): 247-265.
- Fowler SW, Lawrence TB and Morse EA (2004) Virtually Embedded Ties. *Journal* of Management 30 (5): 647-666.
- Friedland R (2009) The Endless Fields of Pierre Bourdieu. Organization 16 (6): 887-917.
- Friedland R and Alford RR (1991) Bringing society back in: Symbols, practices, and institutional contradictions. In: Powell WW and DiMaggio PJ (eds) *The New Institutionalism in Organizational Analysis*. Chicago, IL: University of Chicago Press, pp. 232–263.
- Füller J, Matzler K and Hoppe M (2008) Brand Community Members as a Source of Innovation. *Journal of Product Innovation Management* 25 (6): 608-619.
- Fuller T and Tian Y (2006) Social and Symbolic Capital and Responsible Entrepreneurship: An Empirical Investigation of SME Narratives. *Journal of Business Ethics* 67 (3): 287-304.
- Gaddefors J and Anderson AR (2017) Entrepreneursheep and context: when entrepreneurship is greater than entrepreneurs. *International Journal of Entrepreneurial Behavior & Research* 23 (2): 267-278.
- Galloway L, Sanders J and Deakins D (2011) Rural small firms' use of the internet: From global to local. *Journal of Rural Studies* 27 (3): 254-262.
- Gargiulo M and Benassi M (2000) Trapped in Your Own Net? Network Cohesion, Structural Holes, and the Adaptation of Social Capital. *Organization Science* 11 (2): 183-196.
- Gartner WB (1985) A Conceptual Framework for Describing the Phenomenon of New Venture Creation. *Academy of Management Review* 10 (4): 696-706.
- Gartner WB (2001) Is There an Elephant in Entrepreneurship? Blind Assumptions in Theory Development. *Entrepreneurship Theory and Practice* 25 (4): 27-39.

- Gartner WB (2008) Variations in entrepreneurship. *Small Business Economics* 31 (4): 351-361.
- Gartner WB and Birley S (2002) Introduction to the special issue on qualitative methods in entrepreneurship research. *Journal of Business Venturing* 17 (5): 387-395.
- Garud R, Schildt HA and Lant TK (2014) Entrepreneurial Storytelling, Future Expectations, and the Paradox of Legitimacy. *Organization Science* 25 (5): 1479-1492.
- Gedajlovic E, Carney M, Chrisman JJ and Kellermanns FW (2012) The Adolescence of Family Firm Research. *Journal of Management* 38 (4): 1010-1037.
- Gedajlovic E, Honig B, Moore CB, Payne GT and Wright M (2013) Social Capital and Entrepreneurship: A Schema and Research Agenda. *Entrepreneurship Theory and Practice* 37 (3): 455-478.
- Geertz C (1973) The Interpretation of Cultures. New York, NY: Basic Books.
- Gehman J and Grimes M (In Press) Hidden Badge of Honor: How Contextual Distinctiveness Affects Category Promotion Among Certified B Corporations. *Academy of Management Journal*. Epub ahead of print 25 October 2016. DOI: 10.5465/amj.2015.0416.
- Gehman J and Soublière J-F (2017) Cultural entrepreneurship: from making culture to cultural making. *Innovation* 19 (1): 61-73.
- George G (2005) Slack Resources and the Performance of Privately Held Firms. *Academy of Management Journal* 48 (4): 661-676.
- Gersick CJG (1994) Pacing strategic change: The case of a new venture. Academy of Management Journal 37 (1): 9-45.
- Giddens A (1984) *Constitution of society: outline of the theory of structuration.* Cambridge: Polity Press.
- Gioia DA, Corley KG and Hamilton AL (2013) Seeking Qualitative Rigor in Inductive Research: Notes on the Gioia Methodology. *Organizational Research Methods* 16 (1): 15-31.
- Gioia DA, Schultz M and Corley KG (2000) Organizational Identity, Image, and Adaptive Instability. *Academy of Management Review* 25 (1): 63-81.
- Giorgi A (2010) Phenomenology and the Practice of Science. *Existential Analysis* 21 (1): 3-22.
- Giorgi S, Lockwood C and Glynn MA (2015) The Many Faces of Culture: Making Sense of 30 Years of Research on Culture in Organization Studies. *Academy* of Management Annals 9 (1): 1-54.
- Glover JL (2010) Capital Usage in Adverse Situations: Applying Bourdieu's Theory of Capital to Family Farm Businesses. *Journal of Family and Economic Issues* 31 (4): 485-497.
- Glynn MA (2000) When Cymbals Become Symbols: Conflict Over Organizational Identity Within a Symphony Orchestra. *Organization Science* 11 (3): 285-298.
- Glynn MA and Abzug R (2002) Institutionalizing Identity: Symbolic Isomorphism and Organizational Names. *Academy of Management Journal* 45 (1): 267-280.
- Glynn MA and Lounsbury M (2005) From the Critics' Corner: Logic Blending, Discursive Change and Authenticity in a Cultural Production System. *Journal of Management Studies* 42 (5): 1031-1055.
- Golsorkhi D, Leca B, Lounsbury M and Ramirez C (2009) Analysing, Accounting for and Unmasking Domination: On Our Role as Scholars of Practice, Practitioners of Social Science and Public Intellectuals. *Organization* 16 (6): 779-797.

- Gómez-Mejía LR, Haynes KT, Núñez-Nickel M, Jacobson KJL and Moyano-Fuentes J (2007) Socioemotional Wealth and Business Risks in Familycontrolled Firms: Evidence from Spanish Olive Oil Mills. *Administrative Science Quarterly* 52 (1): 106-137.
- Granovetter M (1973) The Strength of Weak Ties. *American Journal of Sociology* 78 (6): 1360-1380.
- Granovetter M (1985) Economic Action and Social Structure: The Problem of Embeddedness. *American Journal of Sociology* 91 (3): 481-510.
- Grant P and Perren L (2002) Small Business and Entrepreneurial Research: Metatheories, Paradigms and Prejudices. *International Small Business Journal* 20 (2): 185-211.
- Grant RM (1996) Toward a knowledge-based theory of the firm. *Strategic Management Journal* 17 (S2): 109-122.
- Gras D and Nason RS (2015) Bric by bric: The role of the family household in sustaining a venture in impoverished Indian slums. *Journal of Business Venturing* 30 (4): 546–563.
- Graves LM, Ohlott PJ and Ruderman MN (2007) Commitment to family roles: Effects on managers' attitudes and performance. *Journal of Applied Psychology* 92 (1): 44-56.
- Greenhaus JH and Beutell NJ (1985) Sources of Conflict Between Work and Family Roles. *Academy of Management Review* 10 (1): 76-88.
- Greenhaus JH and Powell GN (2006) When work and family are allies: a theory of work-family enrichment. *Academy of Management Review* 31 (1): 72-92.
- Greenman A (2012) Entrepreneurial activities and occupational boundary work during venture creation and development in the cultural industries. *International Small Business Journal* 30 (2): 115-137.
- Greenman A (2013) Everyday entrepreneurial action and cultural embeddedness: an institutional logics perspective. *Entrepreneurship & Regional Development* 25 (7-8): 631-653.
- Greenwood R and Suddaby R (2006) Institutional Entrepreneurship in Mature Fields: The Big Five Accounting Firms. *Academy of Management Journal* 49 (1): 27-48.
- Greenwood R, Suddaby R and Hinings CR (2002) Theorizing Change: The Role of Professional Associations in the Transformation of Institutionalized Fields. *Academy of Management Journal* 45 (1): 58-80.
- Greve A and Salaff JW (2003) Social Networks and Entrepreneurship. Entrepreneurship Theory and Practice 28 (1): 1-22.
- Grichnik D, Brinckmann J, Singh L and Manigart S (2014) Beyond environmental scarcity: Human and social capital as driving forces of bootstrapping activities. *Journal of Business Venturing* 29 (2): 310-326.
- Grimes MG, McMullen JS, Vogus TJ and Miller TL (2013) Studying the Origins of Social Entrepreneurship: Compassion and the Role of Embedded Agency. *Academy of Management Review* 38 (3): 460-463.
- Grossman EB, Yli-Renko H and Janakiraman R (2012) Resource Search, Interpersonal Similarity, and Network Tie Valuation in Nascent Entrepreneurs' Emerging Networks. *Journal of Management* 38 (6): 1760-1787.
- Grunow D, Schulz F and Blossfeld H-P (2012) What determines change in the division of housework over the course of marriage? *International Sociology* 27 (3): 289-307.

- Gudmunson CG, Danes SM, Werbel JD and Loy JT-C (2009) Spousal Support and Work–Family Balance in Launching a Family Business. *Journal of Family Issues* 30 (8): 1098-1121.
- Habbershon TG and Williams ML (1999) A Resource-Based Framework for Assessing the Strategic Advantages of Family Firms. *Family Business Review* 12 (1): 1-25.
- Haber S and Reichel A (2007) The cumulative nature of the entrepreneurial process: The contribution of human capital, planning and environment resources to small venture performance. *Journal of Business Venturing* 22 (1): 119-145.
- Hallen BL and Eisenhardt KM (2012) Catalyzing Strategies and Efficient Tie Formation: How Entrepreneurial Firms Obtain Investment Ties. Academy of Management Journal 55 (1): 35-70.
- Hampel CE and Tracey P (In Press) How Organizations Move From Stigma to Legitimacy: The Case of Cook's Travel Agency in Victorian Britain. *Academy of Management Journal*. Epub ahead of print 18 April 2016. DOI: 10.5465/amj.2015.0365.
- Hanna V and Walsh K (2008) Interfirm Cooperation among Small Manufacturing Firms. *International Small Business Journal* 26 (3): 299-321.
- Harrison RT and Mason CM (2007) Does Gender Matter? Women Business Angels and the Supply of Entrepreneurial Finance. *Entrepreneurship Theory and Practice* 31 (3): 445-472.
- Harrison RT, Mason CM and Girling P (2004) Financial bootstrapping and venture development in the software industry. *Entrepreneurship & Regional Development* 16 (4): 307-333.
- Harrison S and Rouse E (2015) An inductive study of feedback interactions over the course of creative projects. *Academy of Management Journal* 58 (2): 375–404.
- Harrison S and Wagner D (2016) Spilling Outside the Box: The Effects of Individuals' Creative Behaviors at Work on Time Spent with their Spouses at Home. *Academy of Management Journal* 59 (3): 841-859.
- Harvey C and Maclean M (2008) Capital theory and the dynamics of elite business networks in Britain and France. *Sociological Review* 56 (Supp. 1): 103-120.
- Harvey C, Maclean M, Gordon J and Shaw E (2011a) Andrew Carnegie and the foundations of contemporary entrepreneurial philanthropy. *Business History* 53 (3): 425-450.
- Harvey C, Press J and Maclean M (2011b) William Morris, Cultural Leadership, and the Dynamics of Taste. *Business History Review* 85 (2): 245-271.
- Harvey M and Evans R (1995) Forgotten Sources of Capital for the Family-Owned Business. *Family Business Review* 8 (3): 159-176.
- Harvey S (2014) Creative synthesis: Exploring the Process of Extraordinary Group Creativity. *Academy of Management Review* 39 (3): 324-343.
- Haugh H (2007) Community-Led Social Venture Creation. *Entrepreneurship Theory* and Practice 31 (2): 161-182.
- Hayton JC and Cholakova M (2012) The Role of Affect in the Creation and Intentional Pursuit of Entrepreneurial Ideas. *Entrepreneurship Theory and Practice* 36 (1): 41-68.
- Heidegger M (1962) *Being and Time* (Macquarrie J and Robinson E, Trans.). Oxford: Blackwell Publishers.
- Hernández-Carrión C, Camarero-Izquierdo C and Gutiérrez-Cillán J (2017) Entrepreneurs' Social Capital and the Economic Performance of Small

Businesses: The Moderating Role of Competitive Intensity and

- Entrepreneurs' Experience. *Strategic Entrepreneurship Journal* 11 (1): 61-89.
- Hess M (2004) 'Spatial' relationships? Towards a reconceptualization of embeddedness. *Progress in Human Geography* 28 (2): 165-186.
- Higgins CA and Duxbury LE (1992) Work–family conflict: A comparison of dualcareer and traditional-career men. *Journal of Organizational Behavior* 13 (4): 389-411.
- Hindle K (2010) How community context affects entrepreneurial process: A diagnostic framework. *Entrepreneurship & Regional Development* 22 (7-8): 599-647.
- Hirsch PM (1972) Processing Fads and Fashions: An Organization-Set Analysis of Cultural Industry Systems. *American Journal of Sociology* 77 (4): 639-659.
- Hirsch PM (2000) Cultural Industries Revisited. Organization Science 11 (3): 356-361.
- Hite JM (2005) Evolutionary Processes and Paths of Relationally Embedded Network Ties in Emerging Entrepreneurial Firms. *Entrepreneurship Theory and Practice* 29 (1): 113-144.
- Hite JM and Hesterly WS (2001) The evolution of firm networks: From emergence to early growth of the firm. *Strategic Management Journal* 22 (3): 275-286.
- Hjorth D (2005) Organizational Entrepreneurship: With de Certeau on Creating Heterotopias (or Spaces for Play). *Journal of Management Inquiry* 14 (4): 386-398.
- Hjorth D and Holt R (2016) It's entrepreneurship, not enterprise: Ai Weiwei as entrepreneur. *Journal of Business Venturing Insights* 5: 50-54.
- Hlady-Rispal M and Jouison-Laffitte E (2014) Qualitative Research Methods and Epistemological Frameworks: A Review of Publication Trends in Entrepreneurship. *Journal of Small Business Management* 52 (4): 594-614.
- Hmieleski KM and Baron RA (2009) Entrepreneurs' Optimism And New Venture Performance: A Social Cognitive Perspective. Academy of Management Journal 52 (3): 473-488.
- Hmieleski KM, Carr JC and Baron RA (2015) Integrating Discovery and Creation Perspectives of Entrepreneurial Action: The Relative Roles of Founding CEO Human Capital, Social Capital, and Psychological Capital in Contexts of Risk Versus Uncertainty. *Strategic Entrepreneurship Journal* 9 (4): 289–312.
- Hoang H and Antoncic B (2003) Network-based research in entrepreneurship: A critical review. *Journal of Business Venturing* 18 (2): 165-187.
- Hoegl M, Gibbert M and Mazursky D (2008) Financial constraints in innovation projects: When is less more? *Research Policy* 37 (8): 1382-1391.
- Hoffman J, Hoelscher M and Sorenson R (2006) Achieving Sustained Competitive Advantage: A Family Capital Theory. *Family Business Review* 19 (2): 135-145.
- Holmes H (2015) Transient craft: reclaiming the contemporary craft worker. *Work, Employment & Society* 29 (3): 479–495.
- Holt R and Popp A (2013) Emotion, succession, and the family firm: Josiah Wedgwood & Sons. *Business History* 55 (6): 892-909.
- Honig B, Lerner M and Raban Y (2006) Social Capital and the Linkages of High-Tech Companies to the Military Defense System: Is there a Signaling Mechanism? *Small Business Economics* 27 (4-5): 419-437.
- Howorth C, Tempest S and Coupland C (2005) Rethinking entrepreneurship methodology and definitions of the entrepreneur. *Journal of Small Business and Enterprise Development* 12 (1): 24-40.

- Huang L and Knight AP (2017) Resources and Relationships in Entrepreneurship: An Exchange Theory of the Development and Effects of the Entrepreneur-Investor Relationship. *Academy of Management Review* 42 (1): 80-102.
- Huang L and Pearce JL (2015) Managing the Unknowable: The Effectiveness of Early-stage Investor Gut Feel in Entrepreneurial Investment Decisions. *Administrative Science Quarterly* 60 (4): 634-670.
- Huggins R, Izushi H, Prokop D and Thompson P (2015) Network evolution and the spatiotemporal dynamics of knowledge sourcing. *Entrepreneurship & Regional Development* 27 (7-8): 474-499.
- Hughes M, Morgan RE, Ireland RD and Hughes P (2014) Social Capital and Learning Advantages: A Problem of Absorptive Capacity. *Strategic Entrepreneurship Journal* 8 (3): 214-233.
- Husserl E (1954) *The Crisis of European Sciences and Transcendental Phenomenology* (Carr D, Trans.). Evanston, IL: Northwestern University Press.
- Hycner RH (1985) Some Guidelines for the Phenomenological Analysis of Interview Data. *Human Studies* 8 (3): 279-303.
- Iacobucci D and Rosa P (2010) The Growth of Business Groups by Habitual Entrepreneurs: The Role of Entrepreneurial Teams. *Entrepreneurship Theory and Practice* 34 (2): 351-377.
- Imas JM, Wilson N and Weston A (2012) Barefoot entrepreneurs. *Organization* 19 (5): 563-585.
- Islam G, Endrissat N and Noppeney C (2016) Beyond 'the Eye' of the Beholder: Scent innovation through analogical reconfiguration. *Organization Studies* 37 (6): 769-795.
- Iyer GR (2004) Ethnic business families. *International Research in the Business* Disciplines 4: 243–260.
- Jack SL (2005) The Role, Use and Activation of Strong and Weak Network Ties: A Qualitative Analysis. *Journal of Management Studies* 42 (6): 1233-1259.
- Jack SL (2010) Approaches to studying networks: Implications and outcomes. Journal of Business Venturing 25 (1): 120-137.
- Jack SL and Anderson AR (2002) The effects of embeddedness on the entrepreneurial process. *Journal of Business Venturing* 17 (5): 467-487.
- Jack SL, Moult S, Anderson AR and Drakopoulou Dodd S (2010) An entrepreneurial network evolving: Patterns of change. *International Small Business Journal* 28 (4): 315-337.
- Jansen RJG, Curşeu PL, Vermeulen PAM, Geurts JLA and Gibcus P (2013) Information processing and strategic decision-making in small and mediumsized enterprises: The role of human and social capital in attaining decision effectiveness. *International Small Business Journal* 31 (2): 192-216.
- Jaskiewicz P, Combs JG, Shanine KK and Kacmar KM (2017) Introducing the Family: A Review of Family Science with Implications for Management Research. *Academy of Management Annals* 11 (1): 309-341.
- Jayawarna D, Jones O and Macpherson A (2011) New business creation and regional development: Enhancing resource acquisition in areas of social deprivation. *Entrepreneurship & Regional Development* 23 (9-10): 735-761.
- Jayawarna D, Jones O and Macpherson A (2014) Entrepreneurial potential: The role of human and cultural capitals. *International Small Business Journal* 32 (8): 918-943.

Jayawarna D, Rouse J and Kitching J (2013) Entrepreneur motivations and life course. *International Small Business Journal* 31 (1): 34-56.

- Jennings JE and Brush CG (2013) Research on Women Entrepreneurs: Challenges to (and from) the Broader Entrepreneurship Literature? *Academy of Management Annals* 7 (1): 663-715.
- Jennings JE, Edwards T, Jennings PD and Delbridge R (2015) Emotional arousal and entrepreneurial outcomes: Combining qualitative methods to elaborate theory. *Journal of Business Venturing* 30 (1): 113–130.
- Jennings JE, Hughes KD and Jennings PD (2010) The work–family interface strategies of male and female entrepreneurs: are there any differences. In: De Bruin A, Brush CG, Gatewood EJ and Henry C (eds) *Women Entrepreneurs and the Global Environment for Growth*. Cheltenham: Edward Elgar Publishing, pp. 163-186.
- Jennings JE and McDougald MS (2007) Work-family interface experiences and coping strategies: Implications for entrepreneurship research and practice. *Academy of Management Review* 32 (3): 747-760.
- Jennings PD, Greenwood R, Lounsbury MD and Suddaby R (2013) Institutions, entrepreneurs, and communities: A special issue on entrepreneurship. *Journal* of Business Venturing 28 (1): 1-9.
- Jennings PL, Perren L and Carter S (2005) Guest Editors' Introduction: Alternative Perspectives on Entrepreneurship Research. *Entrepreneurship Theory and Practice* 29 (2): 145-152.
- Johannisson B (1986) Network Strategies: Management Technology for Entrepreneurship and Change. *International Small Business Journal* 5 (1): 19-30.
- Johannisson B (1990) Community entrepreneurship cases and conceptualization. Entrepreneurship & Regional Development 2 (1): 71-88.
- Johannisson B (1998) Personal networks in emerging knowledge-based firms: spatial and functional patterns. *Entrepreneurship & Regional Development* 10 (4): 297-312.
- Johannisson B, Ramírez-Pasillas M and Karlsson G (2002) The institutional embeddedness of local inter-firm networks: A leverage for business creation. *Entrepreneurship & Regional Development* 14 (4): 297-315.
- Johnson P and Duberley J (2000) Understanding Management Research: An Introduction to Epistemology. London: Sage.
- Johnson V (2007) What Is Organizational Imprinting? Cultural Entrepreneurship in the Founding of the Paris Opera. *American Journal of Sociology* 113 (1): 97-127.
- Johnstone H and Lionais D (2004) Depleted communities and community business entrepreneurship: revaluing space through place. *Entrepreneurship & Regional Development* 16 (3): 217-233.
- Jones C (2010) Finding a place in history: Symbolic and social networks in creative careers and collective memory. *Journal of Organizational Behavior* 31 (5): 726-748.
- Jones C, Svejenova S, Pedersen JS and Townley B (2016) Misfits, Mavericks and Mainstreams: Drivers of Innovation in the Creative Industries. *Organization Studies* 37 (6): 751-768.
- Jones T, Ram M, Edwards P, Kiselinchev A and Muchenje L (2014) Mixed embeddedness and new migrant enterprise in the UK. *Entrepreneurship & Regional Development* 26 (5-6): 500-520.

- Jonsson S and Lindbergh J (2013) The Development of Social Capital and Financing of Entrepreneurial Firms: From Financial Bootstrapping to Bank Funding. *Entrepreneurship Theory and Practice* 37 (4): 661-686.
- Josefy M, Dean TJ, Albert LS and Fitza MA (2017) The Role of Community in Crowdfunding Success: Evidence on Cultural Attributes in Funding Campaigns to "Save the Local Theater". *Entrepreneurship Theory and Practice* 41 (2): 161-182.
- Jung H, Vissa B and Pich M (2017) How do entrepreneurial founding teams allocate task positions? *Academy of Management Journal* 60 (1): 264-294.
- Jurik NC (1998) Getting Away and Getting By. Work and Occupations 25 (1): 7-35.
- Kacperczyk A and Younkin P (In Press) The Paradox of Breadth. *Administrative Science Quarterly*. Epub ahead of print 13 March 2017. DOI: 10.1177/0001839217700352.
- Kahn RL, Wolfe DM, Quinn RP, Snoek JD and Rosenthal RA (1964) *Organizational stress: Studies in role conflict and ambiguity*. New York, NY: Wiley.
- Kalantaridis C and Bika Z (2006) In-migrant entrepreneurship in rural England: beyond local embeddedness. *Entrepreneurship & Regional Development* 18 (2): 109-131.
- Kale P, Singh H and Perlmutter H (2000) Learning and protection of proprietary assets in strategic alliances: building relational capital. *Strategic Management Journal* 21 (3): 217-237.
- Kalnins A and Williams M (2014) When do female-owned businesses out-survive male-owned businesses? A disaggregated approach by industry and geography. *Journal of Business Venturing* 29 (6): 822–835.
- Kandori M (1992) Social Norms and Community Enforcement. *The Review of Economic Studies* 59 (1): 63-80.
- Karataş-Özkan M (2011) Understanding relational qualities of entrepreneurial learning: Towards a multi-layered approach. *Entrepreneurship & Regional Development* 23 (9-10): 877-906.
- Karataş-Özkan M, Anderson AR, Fayolle A, Howells J and Condor R (2014)
  Understanding Entrepreneurship: Challenging Dominant Perspectives and Theorizing Entrepreneurship through New Postpositivist Epistemologies. *Journal of Small Business Management* 52 (4): 589-593.
- Karataş-Özkan M and Chell E (2010) *Nascent Entrepreneurship and Learning*. Cheltenham: Edward Elgar.
- Karataş-Özkan M and Chell E (2015) Gender Inequalities in Academic Innovation and Enterprise: A Bourdieuian Analysis. *British Journal of Management* 26 (1): 109–125.
- Karra N, Tracey P and Phillips N (2006) Altruism and Agency in the Family Firm: Exploring the Role of Family, Kinship, and Ethnicity. *Entrepreneurship Theory and Practice* 30 (6): 861-877.
- Keating A, Geiger S and McLoughlin D (2014) Riding the Practice Waves: Social Resourcing Practices During New Venture Development. *Entrepreneurship Theory and Practice* 38 (5): 1207–1235.
- Kellermanns F, Walter J, Crook TR, Kemmerer B and Narayanan V (2016) The Resource-Based View in Entrepreneurship: A Content-Analytical Comparison of Researchers' and Entrepreneurs' Views. *Journal of Small Business Management* 54 (1): 26–48.

Kellermanns FW, Eddleston KA and Zellweger TM (2012) Extending the Socioemotional Wealth Perspective: A Look at the Dark Side. *Entrepreneurship Theory and Practice* 36 (6): 1175-1182.

Kempster S and Cope J (2010) Learning to lead in the entrepreneurial context. International Journal of Entrepreneurial Behavior & Research 16 (1): 5-34.

- Kenworthy T and McMullan WE (2013) Finding Practical Knowledge in Entrepreneurship. *Entrepreneurship Theory and Practice* 37 (5): 983-997.
- Kerr R and Robinson S (2016) Architecture, symbolic capital and elite mobilisations: The case of the Royal Bank of Scotland corporate campus. *Organization* 23 (5): 699-721.
- Kets de Vries MFR (1977) The Entrepreneurial Personality: A Person at the Crossroads. *Journal of Management Studies* 14 (1): 34-57.
- Khaire M and Hall EV (2016) Medium and Message: Globalization and innovation in the production field of Indian fashion. *Organization Studies* 37 (6): 845-865.
- Khan FR, Munir KA and Willmott H (2007) A Dark Side of Institutional Entrepreneurship: Soccer Balls, Child Labour and Postcolonial Impoverishment. *Organization Studies* 28 (7): 1055-1077.
- Khavul S, Bruton GD and Wood E (2009) Informal Family Business in Africa. *Entrepreneurship Theory and Practice* 33 (6): 1219-1238.
- Khavul S, Chavez H and Bruton GD (2013) When institutional change outruns the change agent: The contested terrain of entrepreneurial microfinance for those in poverty. *Journal of Business Venturing* 28 (1): 30-50.
- Khayesi JNO, George G and Antonakis J (2014) Kinship in Entrepreneur Networks: Performance Effects of Resource Assembly in Africa. *Entrepreneurship Theory and Practice* 38 (6): 1323-1342.
- Kilduff M and Brass DJ (2010) Organizational Social Network Research: Core Ideas and Key Debates. *Academy of Management Annals* 4 (1): 317-357.
- Kim PH, Aldrich H and Keister L (2006) Access (Not) Denied: The Impact of Financial, Human, and Cultural Capital on Entrepreneurial Entry in the United States. *Small Business Economics* 27 (1): 5-22.
- Kim PH, Longest KC and Aldrich HE (2013) Can You Lend Me a Hand? Task-Role Alignment of Social Support for Aspiring Business Owners. *Work and Occupations* 40 (3): 213-249.
- Kim PH, Wennberg K and Croidieu G (2016) Untapped Riches of Meso-Level Applications in Multilevel Entrepreneurship Mechanisms. *Academy of Management Perspectives* 30 (3): 273-291.
- King A (2000) Thinking with Bourdieu Against Bourdieu: A 'Practical' Critique of the Habitus. *Sociological Theory* 18 (3): 417-433.
- King LA, Mattimore LK, King DW and Adams GA (1995) Family Support Inventory for Workers: A new measure of perceived social support from family members. *Journal of Organizational Behavior* 16 (3): 235-258.
- Kirzner IM (1973) Competition and entrepreneurship. Chicago, IL: University of Chicago Press.
- Kisfalvi V (2002) The entrepreneur's character, life issues, and strategy making: A field study. *Journal of Business Venturing* 17 (5): 489-518.
- Kisfalvi V and Maguire S (2011) On the Nature of Institutional Entrepreneurs: Insights From the Life of Rachel Carson. *Journal of Management Inquiry* 20 (2): 152-177.

- Kislov R, Hyde P and McDonald A (In Press) New Game, Old Rules? Mechanisms and Consequences of Legitimation in Boundary Spanning Activities. *Organization Studies*. Epub ahead of print 8 January 2017. DOI: 10.1177/0170840616679455.
- Kiss AN, Fernhaber S and McDougall-Covin PP (In Press) Slack, Innovation, and Export Intensity: Implications for Small- and Medium-Sized Enterprises. *Entrepreneurship Theory and Practice*. Epub ahead of print 20 February 2017. DOI: 10.1111/etap.12276.
- Kistruck GM and Beamish PW (2010) The Interplay of Form, Structure, and Embeddedness in Social Intrapreneurship. *Entrepreneurship Theory and Practice* 34 (4): 735-761.
- Kloosterman RC (2010) Matching opportunities with resources: A framework for analysing (migrant) entrepreneurship from a mixed embeddedness perspective. *Entrepreneurship & Regional Development* 22 (1): 25-45.
- Klyver K and Schenkel MT (2013) From Resource Access to Use: Exploring the Impact of Resource Combinations on Nascent Entrepreneurship. *Journal of Small Business Management* 51 (4): 539–556.
- Knight FH (1921) Risk, uncertainty, and profit. Boston, MA: Houghton Mifflin.
- Ko WW and Liu G (2015) Understanding the Process of Knowledge Spillovers: Learning to Become Social Enterprises. *Strategic Entrepreneurship Journal* 9 (3): 263-285.
- Koppman S (2016) Different Like Me: Why Cultural Omnivores Get Creative Jobs. Administrative Science Quarterly 61 (2): 291-331.
- Korsgaard S and Anderson AR (2011) Enacting entrepreneurship as social value creation. *International Small Business Journal* 29 (2): 135-151.
- Korsgaard S, Ferguson R and Gaddefors J (2015) The best of both worlds: how rural entrepreneurs use placial embeddedness and strategic networks to create opportunities. *Entrepreneurship & Regional Development* 27 (9-10): 574-598.
- Kostova T and Zaheer S (1999) Organizational Legitimacy Under Conditions of Complexity: The Case of the Multinational Enterprise. *Academy of Management Review* 24 (1): 64-81.
- Kotha R and George G (2012) Friends, family, or fools: Entrepreneur experience and its implications for equity distribution and resource mobilization. *Journal of Business Venturing* 27 (5): 525-543.
- Kovács B and Sharkey AJ (2014) The Paradox of Publicity: How Awards Can Negatively Affect the Evaluation of Quality. *Administrative Science Quarterly* 59 (1): 1-33.
- Kuhn KM and Galloway TL (2015) With a Little Help From My Competitors: Peer Networking Among Artisan Entrepreneurs. *Entrepreneurship Theory and Practice* 39 (3): 571–600.
- Kuhn T (1962) *The structure of scientific revolutions*. Chicago: Chicago University Press.
- Kvale S (1989) To validate is to question. In: Kvale S (ed.) *Issues of validity in qualitative research*. Lund, Sweden: Studentlitteratur, pp. 73-91.
- Kwon S-W and Adler PS (2014) Social Capital: Maturation of a Field of Research. Academy of Management Review 39 (4): 412-422.
- Kwon S-W, Heflin C and Ruef M (2013) Community Social Capital and Entrepreneurship. *American Sociological Review* 78 (6): 980-1008.

- Lado AA, Boyd NG, Wright P and Kroll M (2006) Paradox And Theorizing Within The Resource-Based View. *Academy of Management Review* 31 (1): 115-131.
- Lafuente E, Vaillant Y and Rialp J (2007) Regional Differences in the Influence of Role Models: Comparing the Entrepreneurial Process of Rural Catalonia. *Regional Studies* 41 (6): 779-796.
- Lam W and Harker MJ (2015) Marketing and entrepreneurship: An integrated view from the entrepreneur's perspective. *International Small Business Journal* 33 (3): 321-348.
- Lamb P, Sandberg J and Liesch PW (2011) Small firm internationalisation unveiled through phenomenography. *Journal of International Business Studies* 42 (5): 672-693.
- Lamertz K, Heugens PPMAR and Calmet L (2005) The Configuration of Organizational Images Among Firms in the Canadian Beer Brewing Industry. *Journal of Management Studies* 42 (4): 817-843.
- Lampel J, Lant T and Shamsie J (2000) Balancing Act: Learning from Organizing Practices in Cultural Industries. *Organization Science* 11 (3): 263-269.
- Lang R, Fink M and Kibler E (2014) Understanding place-based entrepreneurship in rural Central Europe: A comparative institutional analysis. *International Small Business Journal* 32 (2): 204-227.
- Lange D, Lee PM and Dai Y (2011) Organizational Reputation: A Review. *Journal* of Management 37 (1): 153-184.
- Lans T, Blok V and Gulikers J (2015) Show me your network and I'll tell you who you are: social competence and social capital of early-stage entrepreneurs. *Entrepreneurship & Regional Development* 27 (7-8): 458-473.
- Larkin M, Watts S and Clifton E (2006) Giving voice and making sense in interpretative phenomenological analysis. *Qualitative Research in Psychology* 3 (2): 102-120.
- Larson A (1991) Partner networks: Leveraging external ties to improve entrepreneurial performance. *Journal of Business Venturing* 6 (3): 173-188.
- Larson A and Starr JA (1993) A Network Model of Organization Formation. *Entrepreneurship Theory and Practice* 17 (2): 5-15.
- Laspita S, Breugst N, Heblich S and Patzelt H (2012) Intergenerational transmission of entrepreneurial intentions. *Journal of Business Venturing* 27 (4): 414-435.
- Lather P (1993) Fertile obsession: Validity after poststructuralism. *Sociological Quarterly* 34 (4): 673-693.
- Lave J and Wenger E (1991) *Situated Learning: Legitimate Peripheral Participation.* Cambridge: Cambridge University Press.
- Lawrence TB (2004) Rituals and Resistance: Membership Dynamics in Professional Fields. *Human Relations* 57 (2): 115-143.
- Lawrence TB and Phillips N (2002) Understanding cultural industries. *Journal of Management Inquiry* 11 (4): 430-441.
- Lazarus RS and Folkman S (1984) *Stress, Appraisal, and Coping.* New York, NY: Springer.
- Lazear EP (2004) Balanced Skills and Entrepreneurship. *The American Economic Review* 94 (2): 208-211.
- Le Breton-Miller I and Miller D (2015) The paradox of resource vulnerability: Considerations for organizational curatorship. *Strategic Management Journal* 36 (3): 397-415.

- Le NTB and Nguyen TV (2009) The Impact of Networking on Bank Financing: The Case of Small and Medium-Sized Enterprises in Vietnam. *Entrepreneurship Theory and Practice* 33 (4): 867-887.
- Leca B and Naccache P (2006) A Critical Realist Approach To Institutional Entrepreneurship. *Organization* 13 (5): 627-651.
- Lechner C and Dowling M (2003) Firm networks: external relationships as sources for the growth and competitiveness of entrepreneurial firms. *Entrepreneurship & Regional Development* 15 (1): 1-26.
- Lechner C and Leyronas C (2009) Small-Business Group Formation as an Entrepreneurial Development Model. *Entrepreneurship Theory and Practice* 33 (3): 645-667.
- Lechner C and Leyronas C (2012) The competitive advantage of cluster firms: the priority of regional network position over extra-regional networks a study of a French high-tech cluster. *Entrepreneurship & Regional Development* 24 (5-6): 457-473.
- Lee N and Cowling M (2015) Do Rural Firms Perceive Different Problems? Geography, Sorting, and Barriers to Growth in UK SMEs. *Environment and Planning C: Government and Policy* 33 (1): 25-42.
- Lee R and Jones O (2008) Networks, Communication and Learning during Business Start-up: The Creation of Cognitive Social Capital. *International Small Business Journal* 26 (5): 559-594.
- Lee R and Shaw E (2016) Bourdieu's non-material forms of capital: Implications for start-up policy. *Environment and Planning C: Government and Policy* 34 (8): 1734-1758.
- Lee R, Tüselmann H, Jayawarna D and Rouse J (2011) Investigating the Social Capital and Resource Acquisition of Entrepreneurs Residing in Deprived Areas of England. *Environment and Planning C: Government and Policy* 29 (6): 1054-1072.
- Lee RT and Ashforth BE (1996) A Meta-Analytic Examination of the Correlates of the Three Dimensions of Job Burnout. *Journal of Applied Psychology* 81 (2): 123-133.
- Lehner OM (2014) The formation and interplay of social capital in crowdfunded social ventures. *Entrepreneurship & Regional Development* 26 (5-6): 478-499.
- Leitch CM, Hill FM and Harrison RT (2010) The Philosophy and Practice of Interpretivist Research in Entrepreneurship: Quality, Validation, and Trust. *Organizational Research Methods* 13 (1): 67-84.
- Leitch CM, McMullan C and Harrison RT (2013) The Development of Entrepreneurial Leadership: The Role of Human, Social and Institutional Capital. *British Journal of Management* 24 (3): 347-366.
- Lester RH, Hillman A, Zardkoohi A and Cannella AA (2008) Former Government Officials as Outside Directors: The Role of Human and Social Capital. *Academy of Management Journal* 51 (5): 999-1013.
- Lewis KV (2013) The power of interaction rituals: The Student Volunteer Army and the Christchurch earthquakes. *International Small Business Journal* 31 (7): 811-831.
- Lewis KV (2015) Enacting Entrepreneurship and Leadership: A Longitudinal Exploration of Gendered Identity Work. *Journal of Small Business Management* 53 (3): 662-682.

- Liao J and Welsch H (2005) Roles of Social Capital in Venture Creation: Key Dimensions and Research Implications. *Journal of Small Business Management* 43 (4): 345-362.
- Lichtenstein BMB and Brush CG (2001) How Do "Resource Bundles" Develop and Change in New Ventures? A Dynamic Model and Longitudinal Exploration. *Entrepreneurship Theory and Practice* 25 (3): 37.
- Light I and Dana L-P (2013) Boundaries of Social Capital in Entrepreneurship. Entrepreneurship Theory and Practice 37 (3): 603-624.
- Lin N (1999) Social Networks and Status Attainment. *Annual Review of Sociology* 25 (1): 467-487.
- Lincoln YS and Guba EG (1985) Naturalistic inquiry. Newbury Park, CA: Sage.
- Lingo EL and O'Mahony S (2010) Nexus Work: Brokerage on Creative Projects. Administrative Science Quarterly 55 (1): 47-81.
- Lippmann S and Aldrich HE (2016a) A Rolling Stone Gathers Momentum: Generational Units, Collective Memory, and Entrepreneurship. *Academy of Management Review* 41 (4): 658-675.
- Lippmann S and Aldrich HE (2016b) The temporal dimension of context. In: Welter F and Gartner WB (eds) *A Research Agenda for Entrepreneurship and Context*. Cheltenham: Edward Elgar, pp. 54-64.
- Livingston BA (2014) Bargaining Behind the Scenes: Spousal Negotiation, Labor, and Work–Family Burnout. *Journal of Management* 40 (4): 949-977.
- Lockett A, Currie G, Finn R, Martin G and Waring J (2014) The Influence of Social Position on Sensemaking about Organizational Change. *Academy of Management Journal* 57 (4): 1102-1129.
- Lopez KA and Willis DG (2004) Descriptive Versus Interpretive Phenomenology: Their Contributions to Nursing Knowledge. *Qualitative Health Research* 14 (5): 726-735.
- Lounsbury M and Glynn MA (2001) Cultural Entrepreneurship: Stories, Legitimacy, and the Acquisition of Resources. *Strategic Management Journal* 22 (6/7): 545-564.
- Love EG and Kraatz M (2009) Character, Conformity, or the Bottom Line? How and Why Downsizing Affected Corporate Reputation. *Academy of Management Journal* 52 (2): 314-335.
- Low MB and MacMillan IC (1988) Entrepreneurship: Past Research and Future Challenges. *Journal of Management* 14 (2): 139-161.
- Luthans F (2002) The need for and meaning of positive organizational behavior. Journal of Organizational Behavior 23 (6): 695-706.
- Luthans F and Youssef CM (2004) Human, Social, and Now Positive Psychological Capital Management: Investing in People for Competitive Advantage. *Organizational Dynamics* 33 (2): 143-160.
- Luthans F, Youssef CM and Avolio BJ (2007) *Psychological capital*. New York, NY: Oxford University Press.
- Maclean M, Harvey C and Chia R (2010) Dominant Corporate Agents and the Power Elite in France and Britain. *Organization Studies* 31 (3): 327-348.
- Maclean M, Harvey C and Gordon J (2013) Social innovation, social entrepreneurship and the practice of contemporary entrepreneurial philanthropy. *International Small Business Journal* 31 (7): 747-763.

- Maclean M, Harvey C, Gordon J and Shaw E (2015) Identity, storytelling and the philanthropic journey. *Human Relations* 68 (10): 1623-1652.
- Maclean M, Harvey C and Kling G (2014) Pathways to Power: Class, Hyper-Agency and the French Corporate Elite. *Organization Studies* 35 (6): 825-855.
- Mair J and Martí I (2006) Social entrepreneurship research: A source of explanation, prediction, and delight. *Journal of World Business* 41 (1): 36-44.
- Mair J, Martí I and Ventresca MJ (2012) Building Inclusive Markets in Rural Bangladesh: How Intermediaries Work Institutional Voids. *Academy of Management Journal* 55 (4): 819-850.
- Manolova TS, Carter NM, Manev IM and Gyoshev BS (2007) The Differential Effect of Men and Women Entrepreneurs' Human Capital and Networking on Growth Expectancies in Bulgaria. *Entrepreneurship Theory and Practice* 31 (3): 407-426.
- Markantoni M and van Hoven B (2012) Bringing 'invisible' side activities to light. A case study of rural female entrepreneurs in the Veenkoloniën, the Netherlands. *Journal of Rural Studies* 28 (4): 507-516.
- Markman GD and Baron RA (2003) Person–entrepreneurship fit: why some people are more successful as entrepreneurs than others. *Human Resource Management Review* 13 (2): 281-301.
- Markman GD, Espina MI and Phan PH (2004) Patents as Surrogates for Inimitable and Non-Substitutable Resources. *Journal of Management* 30 (4): 529-544.
- Markman GD, Russo M, Lumpkin GT, Jennings PD and Mair J (2016) Entrepreneurship as a Platform for Pursuing Multiple Goals: A Special Issue on Sustainability, Ethics, and Entrepreneurship. *Journal of Management Studies* 53 (5): 673-694.
- Marlow S and McAdam M (2015) Incubation or Induction? Gendered Identity Work in the Context of Technology Business Incubation. *Entrepreneurship Theory and Practice* 39 (4): 791–816.
- Marquis C, Lounsbury M and Greenwood R (2011) Introduction: Community as an Institutional Order and a Type of Organizing. In: Marquis C, Lounsbury M and Greenwood R (eds) *Communities and Organizations*. Bingley: Emerald Group Publishing Limited, pp. ix-xxvii.
- Martens ML, Jennings JE and Jennings PD (2007) Do the stories they tell get them the money they need? The role of entrepreneurial narratives in resource acquisition. *Academy of Management Journal* 50 (5): 1107-1132.
- Martí I, Courpasson D and Barbosa SD (2013) "Living in the fishbowl". Generating an entrepreneurial culture in a local community in Argentina. *Journal of Business Venturing* 28 (1): 10-29.
- Martinez Dy A, Marlow S and Martin L (2017) A Web of opportunity or the same old story? Women digital entrepreneurs and intersectionality theory. *Human Relations* 70 (3): 286-311.
- Marvel MR (2013) Human Capital and Search-Based Discovery: A Study of High-Tech Entrepreneurship. *Entrepreneurship Theory and Practice* 37 (2): 403-419.
- Marvel MR, Davis JL and Sproul CR (2016) Human Capital and Entrepreneurship Research: A Critical Review and Future Directions. *Entrepreneurship Theory and Practice* 40 (3): 599–626.
- Marx K (1887) *Capital: A Critique of Political Economy, Volume 1* (Moore S and Aveling E, Trans.). New York, NY: International Publishers.

- Mason CM, Carter S and Tagg S (2010) Invisible Businesses: The Characteristics of Home-based Businesses in the United Kingdom. *Regional Studies* 45 (5): 625-639.
- Mathias B, Huyghe A, Frid C and Galloway T (In Press) An Identity Perspective on Coopetition in the Craft Beer Industry. *Strategic Management Journal*. Epub ahead of print expected August 2017.
- Matzek AE, Gudmunson CG and Danes SM (2010) Spousal Capital as a Resource for Couples Starting a Business. *Family Relations* 59 (1): 60-73.
- Maurer I and Ebers M (2006) Dynamics of Social Capital and Their Performance Implications: Lessons from Biotechnology Start-ups. *Administrative Science Quarterly* 51 (2): 262-292.
- McAdam M, McAdam R, Dunn A and McCall C (2014) Development of small and medium-sized enterprise horizontal innovation networks: UK agri-food sector study. *International Small Business Journal* 32 (7): 830-853.
- McAuley A and Fillis I (2002) *Crafts Businesses in Scotland: A Study*. Edinburgh: The Scottish Arts Council and Scottish Enterprise.
- McAuley A and Fillis I (2004) *Making it in the 21st Century: a socio-economic study of crafts activity in England and Wales, 2002-2003.* London: Crafts Council and Arts Councils of England and Wales.
- McAuley A and Fillis I (2005) The Orkney based craft entrepreneur: remote yet global? Journal of Small Business and Enterprise Development 12 (4): 498-509.
- McDonald S, Gan BC, Fraser SS, Oke A and Anderson AR (2015) A review of research methods in entrepreneurship 1985-2013. *International Journal of Entrepreneurial Behavior & Research* 21 (3): 291-315.
- McDonough P (2006) Habitus and the practice of public service. *Work, Employment & Society* 20 (4): 629-647.
- McIntosh NJ (1991) Identification and investigation of properties of social support. Journal of Organizational Behavior 12 (3): 201-217.
- McKeever E, Anderson AR and Jack SL (2014) Entrepreneurship and mutuality: social capital in processes and practices. *Entrepreneurship & Regional Development* 26 (5-6): 453-477.
- McKeever E, Jack SL and Anderson AR (2015) Embedded entrepreneurship in the creative re-construction of place. *Journal of Business Venturing* 30 (1): 50-65.
- McLeod C, O'Donohoe S and Townley B (2009) The elephant in the room? Class and creative careers in British advertising agencies. *Human Relations* 62 (7): 1011-1039.
- McLeod C, O'Donohoe S and Townley B (2011) Pot Noodles, Placements and Peer Regard: Creative Career Trajectories and Communities of Practice in the British Advertising Industry. *British Journal of Management* 22 (1): 114-131.
- McMullen JS and Dimov D (2013) Time and the Entrepreneurial Journey: The Problems and Promise of Studying Entrepreneurship as a Process. *Journal of Management Studies* 50 (8): 1481-1512.
- McMullen JS and Warnick BJ (2016) Should We Require Every New Venture to Be a Hybrid Organization? *Journal of Management Studies* 53 (4): 630-662.
- Meccheri N and Pelloni G (2006) Rural entrepreneurs and institutional assistance: an empirical study from mountainous Italy. *Entrepreneurship & Regional Development* 18 (5): 371-392.

Messer E (1990) Intra-Household Allocation of Resources: Perspectives from Anthropology. In: Regers BL and Schlossman NP (eds) *Intra-Household Resource Allocation*. Tokyo, Japan: United Nations University Press.

- Metiu A (2006) Owning the Code: Status Closure in Distributed Groups. Organization Science 17 (4): 418-435.
- Miettinen R, Samra-Fredericks D and Yanow D (2009) Re-Turn to Practice: An Introductory Essay. *Organization Studies* 30 (12): 1309-1327.
- Milanov H and Shepherd DA (2013) The importance of the first relationship: The ongoing influence of initial network on future status. *Strategic Management Journal* 34 (6): 727-750.
- Miles MB and Huberman AM (1994) *Qualitative Data Analysis: An Expanded Sourcebook* (2nd ed.). Thousand Oaks, CA: Sage.
- Miller D, Le Breton-Miller I and Lester RH (2011) Family and Lone Founder Ownership and Strategic Behaviour: Social Context, Identity, and Institutional Logics. *Journal of Management Studies* 48 (1): 1-25.
- Miller D, Le Breton-Miller I and Scholnick B (2008) Stewardship vs. Stagnation: An Empirical Comparison of Small Family and Non-Family Businesses. *Journal of Management Studies* 45 (1): 51-78.
- Miller D, Steier L and Le Breton-Miller I (2003) Lost in time: intergenerational succession, change, and failure in family business. *Journal of Business Venturing* 18 (4): 513-531.
- Miller NJ, Besser T and Malshe A (2007) Strategic Networking among Small Businesses in Small US Communities. *International Small Business Journal* 25 (6): 631-665.
- Miller NJ, Fitzgerald MA, Winter M and Paul J (1999) Exploring the Overlap of Family and Business Demands: Household and Family Business Managers' Adjustment Strategies. *Family Business Review* 12 (3): 253-268.
- Miller T, Grimes M, McMullen J and Vogus T (2012) Venturing for Others with Heart and Head: How Compassion Encourages Social Entrepreneurship. *Academy of Management Review* 37 (4): 616-640.
- Millward LJ (2006) The transition to motherhood in an organizational context: An interpretative phenomenological analysis. *Journal of Occupational and Organizational Psychology* 79 (3): 315-333.
- Mincer J (1958) Investment in Human Capital and Personal Income Distribution. Journal of Political Economy 66 (4): 281-302.
- Mitchell CJA (2013) Creative destruction or creative enhancement? Understanding the transformation of rural spaces. *Journal of Rural Studies* 32: 375-387.
- Moran D (2000) Introduction to phenomenology. London: Routledge.
- Morris G, Hargreaves J and McIntyre A. (2010a). Consuming Craft: the contemporary craft market in a changing economy. London: Crafts Council.
- Morris MH, Allen JA, Kuratko DF and Brannon D (2010b) Experiencing Family Business Creation: Differences Between Founders, Nonfamily Managers, and Founders of Nonfamily Firms. *Entrepreneurship Theory and Practice* 34 (6): 1057-1084.
- Morris MH, Kuratko DF, Schindehutte M and Spivack AJ (2012) Framing the Entrepreneurial Experience. *Entrepreneurship Theory and Practice* 36 (1): 11-40.

Morris MH, Schindehutte M, Edmonds V and Watters C (2011) Inner city engagement and the university: Mutuality, emergence and transformation. *Entrepreneurship & Regional Development* 23 (5-6): 287-315.

Morse EA, Fowler SW and Lawrence TB (2007) The Impact of Virtual Embeddedness on New Venture Survival: Overcoming the Liabilities of Newness. *Entrepreneurship Theory and Practice* 31 (2): 139-159.

Mosakowski E (1998) Entrepreneurial Resources, Organizational Choices, and Competitive Outcomes. *Organization Science* 9 (6): 625-643.

Mosey S and Wright M (2007) From Human Capital to Social Capital: A Longitudinal Study of Technology-Based Academic Entrepreneurs. *Entrepreneurship Theory and Practice* 31 (6): 909-935.

Mulholland K (1997) The Family Enterprise and Business Strategies. *Work, Employment & Society* 11 (4): 685-711.

Mutch A (2003) Communities of Practice and Habitus: A Critique. *Organization Studies* 24 (3): 383-401.

- Mwaura S and Carter S (2015) Entrepreneurship as the business of the household. In: Mason C, Reuschke D, Syrett S and van Ham M (eds) *Entrepreneurship in Cities: Neighbourhoods, Households and Homes*. Cheltenham: Edward Elgar, pp. 201-222.
- Nagy BG, Pollack JM, Rutherford MW and Lohrke FT (2012) The Influence of Entrepreneurs' Credentials and Impression Management Behaviors on Perceptions of New Venture Legitimacy. *Entrepreneurship Theory and Practice* 36 (5): 941-965.

Nahapiet J and Ghoshal S (1998) Social Capital, Intellectual Capital, and the Organizational Advantage. *Academy of Management Review* 23 (2): 242-266.

- Naldi L and Davidsson P (2014) Entrepreneurial growth: The role of international knowledge acquisition as moderated by firm age. *Journal of Business Venturing* 29 (5): 687–703.
- Nambisan S (In Press) Digital Entrepreneurship: Toward a Digital Technology Perspective of Entrepreneurship. *Entrepreneurship Theory and Practice*. Epub ahead of press 12 October 2016. DOI: 10.1111/etap.12254.
- Nason RS and Wiklund J (In Press) An Assessment of Resource-Based Theorizing on Firm Growth and Suggestions for the Future. *Journal of Management*. Epub ahead of print 27 October 2015. DOI: 10.1177/0149206315610635.
- Nathan M, Kemeny T, Pratt A and Spencer G (2016) *Creative Economy Employment in the US, Canada and the UK.* London: Nesta.
- Nathan M, Pratt A and Rincon–Aznar A (2015) *Creative economy employment in the EU and UK: A comparative analysis.* London: Nesta.
- Navis C and Glynn MA (2010) How New Market Categories Emerge: Temporal Dynamics of Legitimacy, Identity, and Entrepreneurship in Satellite Radio, 1990–2005. *Administrative Science Quarterly* 55 (3): 439-471.
- Navis C and Glynn MA (2011) Legitimate Distinctiveness and The Entrepreneurial Identity: Influence on Investor Judgments of New Venture Plausibility. *Academy of Management Review* 36 (3): 479-499.
- Navis C and Ozbek OV (2016) The right people in the wrong places: The paradox of entrepreneurial entry and the successful realization of venture opportunities. *Academy of Management Review* 41 (1): 109-129.

- Neergaard H and Ulhøi JP (2006) Government Agency and Trust in the Formation and Transformation of Interorganizational Entrepreneurial Networks. *Entrepreneurship Theory and Practice* 30 (4): 519-539.
- Neergaard H and Ulhøi JP (2007) Introduction: Methodological variety in entrepreneurship research. In: Neergaard H and Ulhøi JP (eds) *Handbook of qualitative research in entrepreneurship*. Cheltenham: Edward Elgar, pp. 1-14.
- Newbert SL (2007) Empirical research on the resource-based view of the firm: an assessment and suggestions for future research. *Strategic Management Journal* 28 (2): 121-146.
- Newbert SL and Tornikoski ET (2013) Resource Acquisition in the Emergence Phase: Considering the Effects of Embeddedness and Resource Dependence. *Entrepreneurship Theory and Practice* 37 (2): 249-280.
- Newbert SL, Tornikoski ET and Quigley NR (2013) Exploring the evolution of supporter networks in the creation of new organizations. *Journal of Business Venturing* 28 (2): 281-298.
- Newman A, Goulding A and Whitehead C (2013) How cultural capital, habitus and class influence the responses of older adults to the field of contemporary visual art. *Poetics* 41 (5): 456-480.
- Newman A, Schwarz S and Borgia D (2014a) How does microfinance enhance entrepreneurial outcomes in emerging economies? The mediating mechanisms of psychological and social capital. *International Small Business Journal* 32 (2): 158-179.
- Newman A, Ucbasaran D, Zhu F and Hirst G (2014b) Psychological capital: A review and synthesis. *Journal of Organizational Behavior* 35 (S1): S120-S138.
- Nicholls A (2010) The Legitimacy of Social Entrepreneurship: Reflexive Isomorphism in a Pre-Paradigmatic Field. *Entrepreneurship Theory and Practice* 34 (4): 611-633.
- Nicholson L and Anderson AR (2005) News and Nuances of the Entrepreneurial Myth and Metaphor: Linguistic Games in Entrepreneurial Sense-Making and Sense-Giving. *Entrepreneurship Theory and Practice* 29 (2): 153-172.
- Nicolopoulou K (2014) Social Entrepreneurship between Cross-Currents: Toward a Framework for Theoretical Restructuring of the Field. *Journal of Small Business Management* 52 (4): 678-702.
- Niehm LS, Swinney J and Miller NJ (2008) Community Social Responsibility and Its Consequences for Family Business Performance. *Journal of Small Business Management* 46 (3): 331-350.
- Nordqvist M and Melin L (2010) Entrepreneurial families and family firms. Entrepreneurship & Regional Development 22 (3-4): 211-239.
- O'Cass A and Sok P (2014) The role of intellectual resources, product innovation capability, reputational resources and marketing capability combinations in SME growth. *International Small Business Journal* 32 (8): 996-1018.
- O'Connor J (2015) Intermediaries and Imaginaries in the Cultural and Creative Industries. *Regional Studies* 49 (3): 374-387.
- O'Mahoney J (2007) Constructing habitus: the negotiation of moral encounters at Telekom. *Work, Employment & Society* 21 (3): 479-496.
- O'Mahony S (2003) Guarding the commons: how community managed software projects protect their work. *Research Policy* 32 (7): 1179-1198.

- O'Mahony S and Bechky BA (2008) Boundary Organizations: Enabling Collaboration among Unexpected Allies. *Administrative Science Quarterly* 53 (3): 422-459.
- O'Mahony S and Ferraro F (2007) The Emergence of Governance in an Open Source Community. *Academy of Management Journal* 50 (5): 1079-1106.
- Ocasio W, Mauskapf M and Steele C (2016) History, Society, and Institutions: The Role of Collective Memory in the Emergence and Evolution of Societal Logics. *Academy of Management Review* 41 (4): 676-699.
- OECD (2006) *The New Rural Paradigm: Policies and Governance*. Paris: Organisation for Economic Cooperation and Development.
- OECD (2008) OECD Rural Policy Reviews: Scotland. Paris: Organisation for Economic Cooperation and Development.
- OECD (2011) OECD Rural Policy Reviews: England, United Kingdom 2011. Paris: Organisation for Economic Cooperation and Development.
- OECD (2014a) *Innovation and Modernising the Rural Economy*. Paris: Organization for Economic Cooperation and Development.
- OECD (2014b) *Tourism and the Creative Economy*. Paris: Organisation for Economic Co-operation and Development.
- Ofstein L, Lepisto D and Flanagan D (2017) *From Pastime to Product in the Craft Brewing Industry: Bricolage and Co-opetition in Action.* Paper presented at the Babson College Entrepreneurship Research Conference, Normal, Oklahoma.
- Omidvar O and Kislov R (2014) The Evolution of the Communities of Practice Approach. *Journal of Management Inquiry* 23 (3): 266-275.
- Onkelinx J, Manolova TS and Edelman LF (2016) Human capital and SME internationalization: Empirical evidence from Belgium. *International Small Business Journal* 34 (6): 818-837.
- Oughton E and Wheelock J (2003) A capabilities approach to austainable household livelihoods. *Review of Social Economy* 61 (1): 1-22.
- Oughton E, Wheelock J and Baines S (2003) Micro-businesses and Social Inclusion in Rural Households: a Comparative Analysis. *Sociologia Ruralis* 43 (4): 331-348.
- Özbilgin M and Tatli A (2005) Book Review Essay: Understanding Bourdieu's Contribution to Organization and Management Studies. *Academy of Management Review* 30 (4): 855-869.
- Ozcan P and Eisenhardt KM (2009) Origin of alliance portfolios: entrepreneurs, network strategies, and firm performance. *Academy of Management Journal* 52 (2): 246-279.
- Ozdemir SZ, Moran P, Zhong X and Bliemel MJ (2016) Reaching and Acquiring Valuable Resources: The Entrepreneur's Use of Brokerage, Cohesion, and Embeddedness. *Entrepreneurship Theory and Practice* 40 (1): 49-79.
- Ozgen E and Baron RA (2007) Social sources of information in opportunity recognition: Effects of mentors, industry networks, and professional forums. *Journal of Business Venturing* 22 (2): 174-192.
- Pache A-C and Santos F (2013) Inside the Hybrid Organization: Selective Coupling as a Response to Competing Institutional Logics. *Academy of Management Journal* 56 (4): 972-1001.
- Packalen KA (2007) Complementing Capital: The Role of Status, Demographic Features, and Social Capital in Founding Teams' Abilities to Obtain Resources. *Entrepreneurship Theory and Practice* 31 (6): 873-891.

Paige RC and Littrell MA (2002) Craft Retailers' Criteria for Success and Associated Business Strategies. *Journal of Small Business Management* 40 (4): 314-331.

- Parasuraman S, Greenhaus JH and Granrose CS (1992) Role stressors, social support, and well-being among two-career couples. *Journal of Organizational Behavior* 13 (4): 339-356.
- Parasuraman S, Purohit YS, Godshalk VM and Beutell NJ (1996) Work and Family Variables, Entrepreneurial Career Success, and Psychological Well-Being. *Journal of Vocational Behavior* 48 (3): 275-300.
- Parhankangas A and Renko M (2017) Linguistic style and crowdfunding success among social and commercial entrepreneurs. *Journal of Business Venturing* 32 (2): 215-236.
- Parker SC (2008a) The economics of formal business networks. *Journal of Business Venturing* 23 (6): 627-640.
- Parker SC (2008b) Entrepreneurship among married couples in the United States: A simultaneous probit approach. *Labour Economics* 15 (3): 459-481.
- Parkinson C and Howorth C (2008) The language of social entrepreneurs. Entrepreneurship & Regional Development 20 (3): 285-309.
- Parkinson C, Howorth C and Southern A (2017) The crafting of an (un)enterprising community: Context and the social practice of talk. *International Small Business Journal* 35 (4): 385-404.
- Parrilli MD (2009) Collective efficiency, policy inducement and social embeddedness: Drivers for the development of industrial districts. *Entrepreneurship & Regional Development* 21 (1): 1-24.
- Parry S (2010) Smalltalk: Rhetoric of control as a barrier to growth in artisan microfirms. *International Small Business Journal* 28 (4): 378-397.
- Partanen J and Goel S (2017) Interplay between reputation and growth: the source, role and audience of reputation of rapid growth technology-based SMEs. *Entrepreneurship & Regional Development* 29 (3-4): 238-270.
- Patel PC and Conklin B (2009) The Balancing Act: The Role of Transnational Habitus and Social Networks in Balancing Transnational Entrepreneurial Activities. *Entrepreneurship Theory and Practice* 33 (5): 1045-1078.
- Patel PC and Terjesen S (2011) Complementary effects of network range and tie strength in enhancing transnational venture performance. *Strategic Entrepreneurship Journal* 5 (1): 58-80.
- Patriotta G and Hirsch PM (2016) Mainstreaming Innovation in Art Worlds: Cooperative links, conventions and amphibious artists. *Organization Studies* 37 (6): 867-887.
- Patton MQ (2002) *Qualitative research and evaluation methods* (3rd ed). Thousand Oaks, CA: Sage.
- Patzelt H and Shepherd DA (2011) Negative emotions of an entrepreneurial career: Self-employment and regulatory coping behaviors. *Journal of Business Venturing* 26 (2): 226-238.
- Payne GT, Moore CB, Griffis SE and Autry CW (2011) Multilevel Challenges and Opportunities in Social Capital Research. *Journal of Management* 37 (2): 491-520.
- Pearce JA and Doh JP (2005) The high impact of collaborative social initiatives. *MIT* sloan management review 46 (3): 29-39.

- Pearson AW, Carr JC and Shaw JC (2008) Toward a Theory of Familiness: A Social Capital Perspective. *Entrepreneurship Theory and Practice* 32 (6): 949-969.
- Peltoniemi M (2015) Cultural Industries: Product–Market Characteristics, Management Challenges and Industry Dynamics. *International Journal of* Management Reviews 17 (1): 41–68.
- Peng Y (2004) Kinship Networks and Entrepreneurs in China's Transitional Economy. *American Journal of Sociology* 109 (5): 1045-1074.
- Penrose E (1959) The Theory of the Growth of the Firm. New York, NY: John Wiley.
- Peredo AM and Chrisman JJ (2006) Toward a Theory of Community-Based Enterprise. *Academy of Management Review* 31 (2): 309-328.
- Peredo AM and McLean M (2006) Social entrepreneurship: A critical review of the concept. *Journal of World Business* 41 (1): 56-65.
- Perry-Rivers P (2016) Stratification, Economic Adversity, and Entrepreneurial Launch: The Effect of Resource Position on Entrepreneurial Strategy. *Entrepreneurship Theory and Practice* 40 (3): 685–712.
- Perry-Smith JE and Mannucci PV (2017) From creativity to innovation: The social network drivers of the four phases of the idea journey. *Academy of Management Review* 42 (1): 53-79.
- Petrova K (2012) Part-time entrepreneurship and financial constraints: evidence from the Panel Study of Entrepreneurial Dynamics. *Small Business Economics* 39 (2): 473-493.
- Phillips N, Tracey P and Karra N (2013) Building entrepreneurial tie portfolios through strategic homophily: The role of narrative identity work in venture creation and early growth. *Journal of Business Venturing* 28 (1): 134–150.
- Phillipson J, Gorton M, Raley M and Moxey A (2004) Treating Farms as Firms? the Evolution of Farm Business Support from Productionist to Entrepreneurial Models. *Environment and Planning C: Government and Policy* 22 (1): 31-54.
- Pittaway L (2005) Philosophies in entrepreneurship: a focus on economic theories. International Journal of Entrepreneurial Behavior & Research 11 (3): 201-221.
- Piva E, Rentocchini F and Rossi-Lamastra C (2012) Is Open Source Software about Innovation? Collaborations with the Open Source Community and Innovation Performance of Software Entrepreneurial Ventures. *Journal of Small Business Management* 50 (2): 340-364.
- Plate M, Schiede C and von Schlippe A (2010) Portfolio Entrepreneurship in the Context of Family Owned Businesses. In: Nordqvist M and Zellweger T (eds) *Transgenerational Entrepreneurship: Exploring Growth and Performance in Family Firms Across Generations*. Cheltenham: Edward Elgar, pp. 96–122.
- Ployhart RE, Nyberg AJ, Reilly G and Maltarich MA (2014) Human Capital Is Dead; Long Live Human Capital Resources! *Journal of Management* 40 (2): 371-398.
- Ployhart RE, Van Iddekinge CH and MacKenzie WI (2011) Acquiring and Developing Human Capital in Service Contexts: The Interconnectedness of Human Capital Resources. *Academy of Management Journal* 54 (2): 353-368.
- Podolny JM (1993) A Status-Based Model of Market Competition. *American Journal of Sociology* 98 (4): 829-872.
- Polanyi M (1944) *The great transformation: The political and economic origins of our time*. New York, NY: Farrar & Rinehart.

- Pollock TG, Chen G, Jackson EM and Hambrick DC (2010) How much prestige is enough? Assessing the value of multiple types of high-status affiliates for young firms. *Journal of Business Venturing* 25 (1): 6-23.
- Porac JF, Thomas H, Wilson F, Paton D and Kanfer A (1995) Rivalry and the Industry Model of Scottish Knitwear Producers. *Administrative Science Quarterly* 40 (2): 203-227.
- Porter ME (1996) What is strategy? Harvard Business Review 74 (6): 61-78.
- Portes A (1998) Social Capital: Its Origins and Applications in Modern Sociology. Annual Review of Sociology 24: 1-24.
- Powell EE and Baker T (2014) It's what you make of it: Founder identity and enacting strategic responses to adversity. *Academy of Management Journal* 57 (5): 1406–1433.
- Powell GN and Eddleston KA (2013) Linking family-to-business enrichment and support to entrepreneurial success: Do female and male entrepreneurs experience different outcomes? *Journal of Business Venturing* 28 (2): 261-280.
- Powell GN and Eddleston KA (In Press) Family Involvement in the Firm, Family-to-Business Support, and Entrepreneurial Outcomes: An Exploration. *Journal of Small Business Management*. Epub ahead of print 21 January 2017. DOI: 10.5465/amj.2015.0175.
- Pratt MG (2009) From the editors: for the lack of a boilerplate: tips on writing up (and reviewing) qualitative research. *Academy of Management Journal* 52 (5): 856-862.
- Pret T (2016) *Collaborating to perpetuate an industry: A social responsibility*. Paper presented at the Social Innovation and Entrepreneurship Research Conference, Auckland, New Zealand.
- Pret T and Carter S (2016) *Why context matters: The influence of embeddedness in multiple communities.* Paper presented at the Academy of Management Annual Meeting, Anaheim, CA.
- Pret T and Carter S (In Press) The importance of 'fitting in': collaboration and social value creation in response to community norms and expectations. *Entrepreneurship & Regional Development*. Epub ahead of print expected August 2017. DOI: 10.1080/08985626.2017.1328903.
- Pret T and Shaw E (2013) *Entrepreneurial capital: exploring the perspectives of craft entrepreneurs.* Paper presented at the Institute for Small Business and Entrepreneurship, Cardiff, UK.
- Pret T and Shaw E (2014a) *Exploring the Impact of Household and Family on Craft Entrepreneurs*. Paper presented at the Economic and Social Research Council Seminar, Glasgow, UK.
- Pret T and Shaw E (2014b) *Fields of power: Exploring the perspectives of women entrepreneurs in rural communities.* Paper presented at the Institute for Small Business and Entrepreneurship Conference, Manchester, UK.
- Pret T, Shaw E and Carter S (2014) *Entrepreneurial capital: The effects of household and family on women entrepreneurs.* Paper presented at the Babson College Entrepreneurship Research Conference, London, Canada.
- Pret T, Shaw E and Drakopoulou Dodd S (2016) Painting the Full Picture: The Conversion of Economic, Cultural, Social and Symbolic Capital. *International Small Business Journal* 34 (8): 1004–1027.
- Priem RL and Butler JE (2001) Is the Resource-Based "View" a Useful Perspective for Strategic Management Research? *Academy of Management Review* 26 (1): 22-40.
- Prieur A, Rosenlund L and Skjott-Larsen J (2008) Cultural capital today: A case study from Denmark. *Poetics* 36 (1): 45-71.
- Pyrko I, Dörfler V and Eden C (2017) Thinking together: What makes Communities of Practice work? *Human Relations* 70 (4): 389–409.
- Qian Y (2014) Counterfeiters: Foes or Friends? How Counterfeits Affect Sales by Product Quality Tier. *Management Science* 60 (10): 2381-2400.
- Ram M (1994) *Managing to survive: Working lives in small firms*. Oxford: Blackwell Business.
- Ram M (2001) Family Dynamics in a Small Consultancy Firm: A Case Study. *Human Relations* 54 (4): 395-418.
- Ram M, Abbas T, Sanghera B, Barlow G and Jones T (2001) Making the link: households and small business activity in a multi-ethnic context. *Community*, *Work & Family* 4 (3): 327-348.
- Ram M, Edwards P, Jones T, Kiselinchev A and Muchenje L (2015) Getting your hands dirty: critical action research in a state agency. *Work, Employment & Society* 29 (3): 462-478.
- Ram M, Theodorakopoulos N and Jones T (2008) Forms of capital, mixed embeddedness and Somali enterprise. *Work, Employment & Society* 22 (3): 427-446.
- Ramachandran J, Pant A and Pani SK (2012) Building the BoP Producer Ecosystem: The Evolving Engagement of Fabindia with Indian Handloom Artisans. *Journal of Product Innovation Management* 29 (1): 33-51.
- Randle K, Forson C and Calveley M (2015) Towards a Bourdieusian analysis of the social composition of the UK film and television workforce. *Work, Employment & Society* 29 (4): 590-606.
- Rantisi NM (2014) Gendering Fashion, Fashioning Fur: On the (re)production of a Gendered Labor Market within a Craft Industry in Transition. *Environment and Planning D: Society and Space* 32 (2): 223-239.
- Rasmussen E, Mosey S and Wright M (2015) The transformation of network ties to develop entrepreneurial competencies for university spin-offs. *Entrepreneurship & Regional Development* 27 (7-8): 430-457.
- Rauch A, Frese M and Utsch A (2005) Effects of Human Capital and Long-Term Human Resources Development and Utilization on Employment Growth of Small-Scale Businesses: A Causal Analysis. *Entrepreneurship Theory and Practice* 29 (6): 681-698.
- Rauch A and Rijsdijk SA (2013) The Effects of General and Specific Human Capital on Long-Term Growth and Failure of Newly Founded Businesses. *Entrepreneurship Theory and Practice* 37 (4): 923-941.
- Renzulli LA and Aldrich H (2005) Who Can You Turn To? Tie Activation within Core Business Discussion Networks. *Social Forces* 84 (1): 323-341.
- Renzulli LA, Aldrich H and Moody J (2000) Family matters: gender, networks, and entrepreneurial outcomes. *Social Forces* 79 (2): 523-546.
- Reuber AR and Fischer E (2005) The Company You Keep: How Young Firms in Different Competitive Contexts Signal Reputation through Their Customers. *Entrepreneurship Theory and Practice* 29 (1): 57-78.
- Reuber AR and Fischer E (2007) Don't rest on your laurels: Reputational change and young technology-based ventures. *Journal of Business Venturing* 22 (3): 363-387.

Reuber AR and Fischer E (2009) Signalling reputation in international online markets. *Strategic Entrepreneurship Journal* 3 (4): 369-386.

- Reuber AR and Fischer E (2011) International entrepreneurship in internet-enabled markets. *Journal of Business Venturing* 26 (6): 660-679.
- Revilla G and Dodd TH (2003) Authenticity Perceptions of Talavera Pottery. Journal of Travel Research 42 (1): 94-99.
- Reynolds PD, Carter NM, Gartner WB and Greene PG (2004) The Prevalence of Nascent Entrepreneurs in the United States: Evidence from the Panel Study of Entrepreneurial Dynamics. *Small Business Economics* 23 (4): 263-284.
- Riding AL and Haines G (2001) Loan guarantees: Costs of default and benefits to small firms. *Journal of Business Venturing* 16 (6): 595-612.
- Rindova V, Barry D and Ketchen JDJ (2009) Entrepreneuring as emancipation. *Academy of Management Review* 34 (3): 477-491.
- Rindova VP, Williamson IO, Petkova AP and Sever JM (2005) Being Good or Being Known: An Empirical Examination of the Dimensions, Antecedents, and Consequences of Organizational Reputation. *Academy of Management Journal* 48 (6): 1033-1049.
- Ring JK, Peredo AM and Chrisman JJ (2010) Business Networks and Economic Development in Rural Communities in the United States. *Entrepreneurship Theory and Practice* 34 (1): 171-195.
- Roberts P (1991) Anthropological Perspectives on the Household. *IDS Bulletin* 22 (1): 60-64.
- Rodriguez P, Tuggle CS and Hackett SM (2009) An Exploratory Study of How Potential "Family and Household Capital" Impacts New Venture Start-Up Rates. *Family Business Review* 22 (3): 259-272.
- Roodhouse S (2011) The creative industries definitional discourse. In: Henry C and De Bruin A (eds) *Entrepreneurship and the creative economy: process, practice and policy*. Cheltenham: Edward Elgar Publishing, pp. 7-29.
- Rooks G, Klyver K and Sserwanga A (2016) The Context of Social Capital: A Comparison of Rural and Urban Entrepreneurs in Uganda. *Entrepreneurship Theory and Practice* 40 (1): 111–130.
- Rosa P (1998) Entrepreneurial Processes of Business Cluster Formation and Growth by 'Habitual' Entrepreneurs. *Entrepreneurship Theory and Practice* 22 (4): 43-61.
- Rosenbusch N, Brinckmann J and Müller V (2013) Does acquiring venture capital pay off for the funded firms? A meta-analysis on the relationship between venture capital investment and funded firm financial performance. *Journal of Business Venturing* 28 (3): 335-353.
- Rouse J and Kitching J (2006) Do Enterprise Support Programmes Leave Women Holding the Baby? *Environment and Planning C: Government and Policy* 24 (1): 5-19.
- Ruef M, Aldrich HE and Carter NM (2003) The Structure of Founding Teams: Homophily, Strong Ties, and Isolation among U.S. Entrepreneurs. *American Sociological Review* 68 (2): 195-222.
- Sánchez-Medina P, Díaz-Pichardo R, Bautista-Cruz A and Toledo-López A (2015)
  Environmental Compliance and Economic and Environmental Performance:
  Evidence from Handicrafts Small Businesses in Mexico. *Journal of Business Ethics* 126 (3): 381-393.

- Sandberg J (2000) Understanding Human Competence at Work: An Interpretative Approach. *Academy of Management Journal* 43 (1): 9-25.
- Sandberg J (2005) How Do We Justify Knowledge Produced Within Interpretive Approaches? *Organizational Research Methods* 8 (1): 41-68.
- Sandberg J and Dall'Alba G (2009) Returning to Practice Anew: A Life-World Perspective. *Organization Studies* 30 (12): 1349-1368.
- Sanders JM and Nee V (1996) Immigrant Self-Employment: The Family as Social Capital and the Value of Human Capital. *American Sociological Review* 61 (2): 231-249.

Sanders P (1982) Phenomenology: A New Way of Viewing Organizational Research. *Academy of Management Review* 7 (3): 353-360.

Sarasvathy SD, Ramesh A and Forster WR (2015) The ordinary entrepreneur. In: Baker T and Welter F (eds) *The Routledge Companion to Entrepreneurship*. Abingdon: Routledge, pp. 227-244.

Sauder M, Lynn F and Podolny JM (2012) Status: Insights from Organizational Sociology. *Annual Review of Sociology* 38: 267–283.

- Sayce S, Ackers P and Greene A-M (2007) Work restructuring and changing craft identity: the Tale of the Disaffected Weavers (or what happens when the rug is pulled from under your feet). *Work, Employment & Society* 21 (1): 85-101.
- Schatzki TR (1997) Practices and Actions A Wittgensteinian Critique of Bourdieu and Giddens. *Philosophy of the Social Sciences* 27 (3): 283-308.
- Schatzki TR (2002) *The site of the social: A philosophical account of the constitution of social life and change*. University Park, PA: Pennsylvania State University Press.
- Schultz PL, Marin A and Boal KB (2014) The impact of media on the legitimacy of new market categories: The case of broadband internet. *Journal of Business Venturing* 29 (1): 34-54.
- Schultz TW (1961) Investment in Human Capital. *The American Economic Review* 51 (1): 1-17.
- Schulze WS, Lubatkin MH, Dino RN and Buchholtz AK (2001) Agency Relationships in Family Firms: Theory and Evidence. *Organization Science* 12 (2): 99-116.
- Schumpeter JA (1942) *Capitalism, socialism and democracy*. New York, NY: Harper & Row.
- Schwarz M and Yair K. (2010). *Making Value: craft and the economic and social contribution of makers*. London: Crafts Council.
- Scott AJ (2006) Entrepreneurship, Innovation and Industrial Development: Geography and the Creative Field Revisited. *Small Business Economics* 26 (1): 1-24.
- Scott M (2012) Cultural entrepreneurs, cultural entrepreneurship: Music producers mobilising and converting Bourdieu's alternative capitals. *Poetics* 40 (3): 237-255.
- Scottish Government (2015a) *Rural Scotland Key Facts*. Edinburgh: Rural and Environmental Research and Analysis Directorate.

Scottish Government (2015b) United Kingdom - Rural Development Programme (Regional) - Scotland. Edinburgh: Agriculture and Rural Development Division.

- Seghers A, Manigart S and Vanacker T (2012) The Impact of Human and Social Capital on Entrepreneurs' Knowledge of Finance Alternatives. *Journal of Small Business Management* 50 (1): 63-86.
- Seligman MEP and Csikszentmihalyi M (2000) Positive psychology: An introduction. *American Psychologist* 55 (1): 5-14.

Semrau T and Hopp C (2016) Complementary or compensatory? A contingency perspective on how entrepreneurs' human and social capital interact in shaping start-up progress. *Small Business Economics* 46 (3): 407-423.

Semrau T and Werner A (2014) How Exactly Do Network Relationships Pay Off? The Effects of Network Size and Relationship Quality on Access to Start-Up Resources. *Entrepreneurship Theory and Practice* 38 (3): 501–525.

Sennett R (2008) The Craftsman. New Haven, CT: Yale University Press.

Senyard J, Baker T, Steffens P and Davidsson P (2014) Bricolage as a Path to Innovativeness for Resource-Constrained New Firms. *Journal of Product Innovation Management* 31 (2): 211-230.

- Seymour RG (2006) Hermeneutic phenomenology and international entrepreneurship research. *Journal of International Entrepreneurship* 4 (4): 137-155.
- Shalley CE and Perry-Smith JE (2008) The emergence of team creative cognition: the role of diverse outside ties, sociocognitive network centrality, and team evolution. *Strategic Entrepreneurship Journal* 2 (1): 23-41.

Shane S (2000) Prior Knowledge and the Discovery of Entrepreneurial Opportunities. *Organization Science* 11 (4): 448-469.

- Shane S (2003) *A General Theory of Entrepreneurship. The Individual-opportunity Nexus.* Northampton, MA: Edward Elgar.
- Shane S (2012) Reflections on the 2010 AMR Decade Award: Delivering on the Promise of Entrepreneurship as a Field of Research. *Academy of Management Review* 37 (1): 10-20.
- Shane S and Cable D (2002) Network Ties, Reputation, and the Financing of New Ventures. *Management Science* 48 (3): 364-381.
- Shane S, Locke EA and Collins CJ (2003) Entrepreneurial motivation. *Human Resource Management Review* 13 (2): 257-279.
- Shane S and Stuart T (2002) Organizational Endowments and the Performance of University Start-ups. *Management Science* 48 (1): 154-170.
- Shane S and Venkataraman S (2000) The Promise of Entrepreneurship as a Field of Research. *Academy of Management Review* 25 (1): 217-226.
- Sharma P (2008) Familiness: Capital Stocks and Flows Between Family and Business. *Entrepreneurship Theory and Practice* 32 (6): 971-977.
- Sharma P, Chrisman JJ and Gersick KE (2012) 25 Years of Family Business Review: Reflections on the Past and Perspectives for the Future. *Family Business Review* 25 (1): 5-15.
- Shaw E (2006) Small Firm Networking: An Insight into Contents and Motivating Factors. *International Small Business Journal* 24 (1): 5-29.
- Shaw E and Carter S (2007) Social entrepreneurship: Theoretical antecedents and empirical analysis of entrepreneurial processes and outcomes. *Journal of Small Business and Enterprise Development* 14 (3): 418-434.
- Shaw E and de Bruin A (2013) Reconsidering capitalism: the promise of social innovation and social entrepreneurship? *International Small Business Journal* 31 (7): 737-746.
- Shaw E, Gordon J, Harvey C and Maclean M (2013) Exploring contemporary entrepreneurial philanthropy. *International Small Business Journal* 31 (5): 580-599.
- Shaw E, Lam W and Carter S (2008) The role of entrepreneurial capital in building service reputation. *The Service Industries Journal* 28 (7): 899-917.

- Shaw E, Marlow S, Lam W and Carter S (2009) Gender and entrepreneurial capital: implications for firm performance. *International Journal of Gender and Entrepreneurship* 1 (1): 25-41.
- Shaw E, Wilson J and Pret T (2017) The process of embedding a small firm in its industrial context. *International Small Business Journal* 35 (3): 219–243.
- Shelton LM (2006) Female Entrepreneurs, Work–Family Conflict, and Venture Performance: New Insights into the Work–Family Interface. *Journal of Small Business Management* 44 (2): 285-297.
- Shepherd DA (2003) Learning from Business Failure: Propositions of Grief Recovery for the Self-Employed. *Academy of Management Review* 28 (2): 318-328.
- Shepherd DA (2009) Grief recovery from the loss of a family business: A multi- and meso-level theory. *Journal of Business Venturing* 24 (1): 81-97.
- Shepherd DA (2015) Party On! A call for entrepreneurship research that is more interactive, activity based, cognitively hot, compassionate, and prosocial. *Journal of Business Venturing* 30 (4): 489–507.
- Shepherd DA and Zacharakis A (2003) A New Venture's Cognitive Legitimacy: An Assessment by Customers. *Journal of Small Business Management* 41 (2): 148-167.
- Short JC, Ketchen DJ, McKenny AF, Allison TH and Ireland RD (2017) Research on Crowdfunding: Reviewing the (Very Recent) Past and Celebrating the Present. *Entrepreneurship Theory and Practice* 41 (2): 149-160.
- Short JC, Moss TW and Lumpkin GT (2009) Research in social entrepreneurship: past contributions and future opportunities. *Strategic Entrepreneurship Journal* 3 (2): 161-194.
- Shultz B (2015) The Work Behind the Scenes: The New Intermediaries of the Indie Crafts Business. *Regional Studies* 49 (3): 451-460.
- Sieger P, Zellweger T, Nason RS and Clinton E (2011) Portfolio entrepreneurship in family firms: a resource-based perspective. *Strategic Entrepreneurship Journal* 5 (4): 327-351.
- Sieweke J (2014) Pierre Bourdieu in management and organization studies–A citation context analysis and discussion of contributions. *Scandinavian Journal of Management* 30 (4): 532-543.
- Silverman D (2013) Doing Qualitative Research (4th ed.). London: Sage.
- Simmel G (1957) Fashion. American Journal of Sociology 62 (6): 541-558.
- Simpson R, Irvine K, Balta M and Dickson K (2015) Emotions, performance and entrepreneurship in the context of fringe theatre. *Organization* 22 (1): 100-118.
- Simsek Z, Lubatkin MH and Floyd SW (2003) Inter-Firm Networks and Entrepreneurial Behavior: A Structural Embeddedness Perspective. *Journal* of Management 29 (3): 427-442.
- Sirmon DG and Hitt MA (2003) Managing Resources: Linking Unique Resources, Management, and Wealth Creation in Family Firms. *Entrepreneurship Theory and Practice* 27 (4): 339-358.
- Sirmon DG, Hitt MA and Ireland RD (2007) Managing Firm Resources in Dynamic Environments to Create Value: Looking Inside the Black Box. *Academy of Management Review* 32 (1): 273-292.
- Slavich B and Castellucci F (2016) Wishing Upon a Star: How apprentice-master similarity, status and career stage affect critics' evaluations of former apprentices in the haute cuisine industry. *Organization Studies* 37 (6): 823-843.

- Sleuwaegen L and Boiardi P (2014) Creativity and regional innovation: Evidence from EU regions. *Research Policy* 43 (9): 1508-1522.
- Slotte-Kock S and Coviello N (2010) Entrepreneurship Research on Network Processes: A Review and Ways Forward. *Entrepreneurship Theory and Practice* 34 (1): 31-57.
- Smallbone D (2009) Fostering entrepreneurship in rural areas. In: Potter J and Hofer AR (eds) Strengthening entrepreneurship and economic development in East Germany: lessons from local approaches. Paris: Organisation for Economic Co-operation and Development, pp. 161-187.
- Smallbone D, Baldock R and North D (2003) Policy Support for Small Firms in Rural Areas: The English Experience. *Environment and Planning C: Government and Policy* 21 (6): 825-841.
- Smallbone D, North D and Kalantaridis C (1999) Adapting to peripherality: a study of small rural manufacturing firms in northern England. *Entrepreneurship & Regional Development* 11 (2): 109-127.
- Smith BR and Stevens CE (2010) Different types of social entrepreneurship: The role of geography and embeddedness on the measurement and scaling of social value. *Entrepreneurship & Regional Development* 22 (6): 575-598.
- Smith C, Smith JB and Shaw E (2017) Embracing digital networks: Entrepreneurs' social capital online. *Journal of Business Venturing* 32 (1): 18-34.
- Smith JA (1996) Beyond the divide between cognition and discourse: Using interpretative phenomenological analysis in health psychology. *Psychology & Health* 11 (2): 261-271.
- Smith JA (2004) Reflecting on the development of interpretative phenomenological analysis and its contribution to qualitative research in psychology. *Qualitative Research in Psychology* 1 (1): 39-54.
- Smith JA (2010) Interpretative Phenomenological Analysis: A Reply to Amedeo Giorgi. *Existential Analysis* 21 (2): 186-192.
- Smith JA (2011) Evaluating the contribution of interpretative phenomenological analysis. *Health Psychology Review* 5 (1): 9-27.
- Smith JA, Flowers P and Larkin M (2009) *Interpretative Phenomenological Analysis: Theory Method and Research*. London: Sage.
- Smith JA and Osborn M (2008) Interpretative phenomenological analysis. In: Smith JA (ed.) *Qualitative Psychology: A practical guide to research methods* (2nd ed). London: Sage, pp. 53-80.
- Smith R, McElwee G, McDonald S and Drakopoulou Dodd S (2013) Qualitative entrepreneurship authorship: antecedents, processes and consequences. *International Journal of Entrepreneurial Behavior & Research* 19 (4): 364-386.
- Snyder CR, Sympson S, Ybasco F, Borders TF, Babyak MA and Higgins RL (1996) Development and validation of the State Hope Scale. *Journal of Personality and Social Psychology* 70 (2): 321-335.
- Solebello N, Tschirhart M and Leiter J (2016) The paradox of inclusion and exclusion in membership associations. *Human Relations* 69 (2): 439–460.
- Somaya D (2012) Patent Strategy and Management. *Journal of Management* 38 (4): 1084-1114.
- Sonnenberg SJ (2008) Household financial organisation and discursive practice: Managing money and identity. *The Journal of Socio-Economics* 37 (2): 533-551.

- Sorenson O and Stuart TE (2008) Entrepreneurship: A Field of Dreams? Academy of Management Annals 2 (1): 517-543.
- Sorenson RL and Bierman L (2009) Family Capital, Family Business, and Free Enterprise. *Family Business Review* 22 (3): 193-195.
- Sorenson RL, Folker CA and Brigham KH (2008) The Collaborative Network Orientation: Achieving Business Success through Collaborative Relationships. *Entrepreneurship Theory and Practice* 32 (4): 615-634.
- Spence C and Carter C (2014) An exploration of the professional habitus in the Big 4 accounting firms. *Work, Employment & Society* 28 (6): 946-962.
- Spence C, Carter C, Belal A, Husillos J, Dambrin C and Archel P (2016) Tracking habitus across a transnational professional field. *Work, Employment & Society* 30 (1): 3-20.
- Spence C, Carter C, Husillos J and Archel P (2017) Taste matters: Cultural capital and elites in proximate Strategic Action Fields. *Human Relations* 70 (2): 211-236.
- Spigel B (2013) Bourdieuian approaches to the geography of entrepreneurial cultures. *Entrepreneurship & Regional Development* 25 (9-10): 804-818.
- Spivack AJ, McKelvie A and Haynie JM (2014) Habitual entrepreneurs: Possible cases of entrepreneurship addiction? *Journal of Business Venturing* 29 (3): 651-667.
- Stam W (2010) Industry Event Participation and Network Brokerage among Entrepreneurial Ventures. *Journal of Management Studies* 47 (4): 625-653.
- Stam W, Arzlanian S and Elfring T (2014) Social capital of entrepreneurs and small firm performance: A meta-analysis of contextual and methodological moderators. *Journal of Business Venturing* 29 (1): 152–173.
- Starr JA and MacMillan IC (1990) Resource Cooptation Via Social Contracting: Resource Acquisition Strategies for New Ventures. *Strategic Management Journal* 11: 79-92.
- Steier L (2003) Variants of agency contracts in family-financed ventures as a continuum of familial altruistic and market rationalities. *Journal of Business Venturing* 18 (5): 597-618.
- Steier L (2009) Where Do New Firms Come From?: Households, Family Capital, Ethnicity, and the Welfare Mix. *Family Business Review* 22 (3): 273-278.
- Steier L and Greenwood R (2000) Entrepreneurship and the Evolution of Angel Financial Networks. *Organization Studies* 21 (1): 163-192.
- Stenholm P and Renko M (2016) Passionate bricoleurs and new venture survival. Journal of Business Venturing 31 (5): 595-611.
- Stewart A (2003) Help One Another, Use One Another: Toward an Anthropology of Family Business. *Entrepreneurship Theory and Practice* 27 (4): 383-396.
- Steyaert C (1997) A Qualitative Methodology for Process Studies of Entrepreneurship. International Studies of Management & Organization 27 (3): 13-33.
- Steyaert C (2007) 'Entrepreneuring' as a conceptual attractor? A review of process theories in 20 years of entrepreneurship studies. *Entrepreneurship & Regional Development* 19 (6): 453-477.
- Steyaert C and Katz J (2004) Reclaiming the space of entrepreneurship in society: geographical, discursive and social dimensions. *Entrepreneurship & Regional Development* 16 (3): 179-196.
- Stinchfield BT, Nelson RE and Wood MS (2013) Learning From Levi-Strauss' Legacy: Art, Craft, Engineering, Bricolage, and Brokerage in Entrepreneurship. *Entrepreneurship Theory and Practice* 37 (4): 889-921.

- Stringfellow L, Shaw E and Maclean M (2014) Apostasy versus legitimacy: Relational dynamics and routes to resource acquisition in entrepreneurial ventures. *International Small Business Journal* 32 (5): 571-592.
- Stuart TE, Hoang H and Hybels RC (1999) Interorganizational Endorsements and the Performance of Entrepreneurial Ventures. *Administrative Science Quarterly* 44 (2): 315-349.
- Suchman MC (1995) Managing Legitimacy: Strategic and Institutional Approaches. Academy of Management Review 20 (3): 571-610.
- Suddaby R (2006) From the Editors: What Grounded Theory is Not. Academy of Management Journal 49 (4): 633-642.
- Suddaby R, Bitektine A and Haack P (2017) Legitimacy. *Academy of Management* Annals 11 (1): 451-478.
- Suddaby R, Bruton GD and Si SX (2015) Entrepreneurship through a qualitative lens: Insights on the construction and/or discovery of entrepreneurial opportunity. *Journal of Business Venturing* 30 (1): 1-10.
- Suddaby R, Foster WM and Trank CQ (2010) Rhetorical history as a source of competitive advantage. In: Baum JAC and Lampel J (eds.) *The Globalization of Strategy Research*. Bingley: Emerald Group Publishing Limited, pp. 147-173.
- Suddaby R and Young R (2015) The Art of the Form: A Configurational Perspective. *Entrepreneurship Theory and Practice* 39 (6): 1371-1377.
- Sullivan DM and Ford CM (2014) How Entrepreneurs Use Networks to Address Changing Resource Requirements During Early Venture Development. *Entrepreneurship Theory and Practice* 38 (3): 551–574.
- Sullivan DM and Marvel MR (2011) Knowledge Acquisition, Network Reliance, and Early-Stage Technology Venture Outcomes. *Journal of Management Studies* 48 (6): 1169-1193.
- Sund L-G and Smyrnios KX (2005) Striving for Happiness and Its Impact on Family Stability: An Exploration of the Aristotelian Conception of Happiness. *Family Business Review* 18 (2): 155-170.
- Svejenova S (2005) 'The Path with the Heart': Creating the Authentic Career. Journal of Management Studies 42 (5): 947-974.
- Svejenova S, Mazza C and Planellas M (2007) Cooking up change in haute cuisine: Ferran Adrià as an institutional entrepreneur. *Journal of Organizational Behavior* 28 (5): 539-561.
- Swartz DL (2008) Bringing Bourdieu's Master Concepts into Organizational Analysis. *Theory and Society* 37 (1): 45-52.
- Swedberg R (2006) The cultural entrepreneur and the creative industries: beginning in Vienna. *Journal of Cultural Economics* 30 (4): 243-261.
- Tan J, Shao Y and Li W (2013) To be different, or to be the same? An exploratory study of isomorphism in the cluster. *Journal of Business Venturing* 28 (1): 83-97.
- Tapsell P and Woods C (2010) Social entrepreneurship and innovation: Selforganization in an indigenous context. *Entrepreneurship & Regional Development* 22 (6): 535-556.
- Tatli A, Vassilopoulou J, Özbilgin M, Forson C and Slutskaya N (2014) A Bourdieuan Relational Perspective for Entrepreneurship Research. *Journal of Small Business Management* 52 (4): 615–632.
- Taylor SE (2011) Social Support: A Review. In: Friedman HS (ed.) *The Handbook* of *Health Psychology*. New York, NY: Oxford University Press, pp. 189–214.

Terjesen S and Elam AB (2009) Transnational Entrepreneurs' Venture Internationalization Strategies: A Practice Theory Approach. *Entrepreneurship Theory and Practice* 33 (5): 1093-1120.

- Thomas NJ, Harvey DC and Hawkins H (2013) Crafting the Region: Creative Industries and Practices of Regional Space. *Regional Studies* 47 (1): 75-88.
- Thompson CJ, Locander WB and Pollio HR (1989) Putting Consumer Experience Back into Consumer Research: The Philosophy and Method of Existential-Phenomenology. *Journal of Consumer Research* 16 (2): 133-146.
- Thompson P, Jones M and Warhurst C (2007) From conception to consumption: creativity and the missing managerial link. *Journal of Organizational Behavior* 28 (5): 625-640.
- Thompson P, Jones-Evans D and Kwong C (2009) Women and Home-based Entrepreneurship: Evidence from the United Kingdom. *International Small Business Journal* 27 (2): 227-239.
- Thornton PH (2002) The Rise of the Corporation in a Craft Industry: Conflict and Conformity in Institutional Logics. *Academy of Management Journal* 45 (1): 81-101.
- Thornton PH, Ocasio W and Lounsbury M (2012) *The Institutional Logics Perspective: A New Approach to Culture, Structure, and Process.* Oxford: Oxford University Press.
- Throop CJ and Murphy KM (2002) Bourdieu and phenomenology: A critical assessment. *Anthropological Theory* 2 (2): 185-207.
- Townley B, Beech N and McKinlay A (2009) Managing in the creative industries: Managing the motley crew. *Human Relations* 62 (7): 939-962.
- Tracey P, Phillips N and Jarvis O (2011) Bridging Institutional Entrepreneurship and the Creation of New Organizational Forms: A Multilevel Model. *Organization Science* 22 (1): 60-80.
- Tregear A (2005) Lifestyle, growth, or community involvement? The balance of goals of UK artisan food producers. *Entrepreneurship & Regional Development* 17 (1): 1-15.
- Tweedie D and Holley S (2016) The subversive craft worker: Challenging 'disutility' theories of management control. *Human Relations* 69 (9): 1877-1900.
- Überbacher F (2014) Legitimation of New Ventures: A Review and Research Programme. *Journal of Management Studies* 51 (4): 667-698.
- Überbacher F, Jacobs CD and Cornelissen JP (2015) How Entrepreneurs Become Skilled Cultural Operators. *Organization Studies* 36 (7): 925-951.
- Ucbasaran D, Westhead P, Wright M and Flores M (2010) The nature of entrepreneurial experience, business failure and comparative optimism. *Journal of Business Venturing* 25 (6): 541-555.
- Uchino BN (2004) Social Support and Physical Health: Understanding the Health Consequences of Relationships. New Haven, CT: Yale University Press.
- UNESCO (2013) Creative Economy Report 2013 Special Edition. Paris and New York, NY: United Nations Educational, Scientific and Cultural Organization.
- Unger JM, Rauch A, Frese M and Rosenbusch N (2011) Human capital and entrepreneurial success: A meta-analytical review. *Journal of Business Venturing* 26 (3): 341-358.

- Uzzi B (1996) The Sources and Consequences of Embeddedness for the Economic Performance of Organizations: The Network Effect. *American Sociological Review* 61 (4): 674-698.
- Uzzi B (1997) Social Structure and Competition in Interfirm Networks: The Paradox of Embeddedness. *Administrative Science Quarterly* 42 (1): 35-67.
- Uzzi B (1999) Embeddedness in the Making of Financial Capital: How Social Relations and Networks Benefit Firms Seeking Financing. *American Sociological Review* 64 (4): 481-505.
- Uzzi B and Gillespie JJ (2002) Knowledge spillover in corporate financing networks: embeddedness and the firm's debt performance. *Strategic Management Journal* 23 (7): 595-618.
- Uzzi B and Spiro J (2005) Collaboration and Creativity: The Small World Problem. *American Journal of Sociology* 111 (2): 447-504.
- Valdez Z (2016) Intersectionality, the household economy, and ethnic entrepreneurship. *Ethnic and Racial Studies* 39 (9): 1618-1636.
- van Burg E, Berends H and van Raaij EM (2014) Framing and Interorganizational Knowledge Transfer: A Process Study of Collaborative Innovation in the Aircraft Industry. *Journal of Management Studies* 51 (3): 349-378.
- van Werven R, Bouwmeester O and Cornelissen JP (2015) The power of arguments: How entrepreneurs convince stakeholders of the legitimate distinctiveness of their ventures. *Journal of Business Venturing* 30 (4): 616-631.
- Vanacker T, Manigart S, Meuleman M and Sels L (2011) A longitudinal study on the relationship between financial bootstrapping and new venture growth. *Entrepreneurship & Regional Development* 23 (9-10): 681-705.
- Vardaman JM, Amis JM, Dyson BP, Wright PM and Van de Graaff Randolph R (2012) Interpreting change as controllable: The role of network centrality and self-efficacy. *Human Relations* 65 (7): 835-859.
- Venkataraman S, Sarasvathy SD, Dew N and Forster WR (2012) Reflections on the 2010 AMR Decade Award: Whither the Promise? Moving forward with Entrepreneurship as a Science of the Artificial. *Academy of Management Review* 37 (1): 21-33.
- Vera D and Crossan M (2005) Improvisation and Innovative Performance in Teams. *Organization Science* 16 (3): 203-224.
- Verhaal JC, Khessina OM and Dobrev SD (2015) Oppositional Product Names, Organizational Identities, and Product Appeal. *Organization Science* 26 (5): 1466-1484.
- Vershinina N, Barrett R and Meyer M (2011) Forms of capital, intra-ethnic variation and Polish entrepreneurs in Leicester. *Work, Employment & Society* 25 (1): 101-117.
- Verver M and Koning J (In Press) Toward a Kinship Perspective on Entrepreneurship. *Entrepreneurship Theory and Practice*. Epub ahead of print 6 February 2017. DOI: 10.1111/etap.12274.
- Vesala HT and Vesala KM (2010) Entrepreneurs and producers: Identities of Finnish farmers in 2001 and 2006. *Journal of Rural Studies* 26 (1): 21-30.
- Vestrum I (2014) The embedding process of community ventures: creating a music festival in a rural community. *Entrepreneurship & Regional Development* 26 (7-8): 619-644.

- Vestrum I, Rasmussen E and Carter S (In Press) How nascent community enterprises build legitimacy in internal and external environments. *Regional Studies*. Epub ahead of print 20 September 2016. DOI: 10.1080/00343404.2016.1220675.
- Vincent S (2016) Bourdieu and the gendered social structure of working time: A study of self-employed human resources professionals. *Human Relations* 69 (5): 1163-1184.
- Vohora A, Wright M and Lockett A (2004) Critical junctures in spin-outs from universities. *Research Policy* 33 (7): 147–175.
- von Krogh G, Spaeth S and Lakhani KR (2003) Community, joining, and specialization in open source software innovation: a case study. *Research Policy* 32 (7): 1217-1241.
- Vorley T and Rodgers P (2014) Home is where the business is: Incidents in everyday life and the formation of home-based businesses. *International Small Business Journal* 32 (4): 428-448.
- Voronov M, De Clercq D and Hinings CR (2013) Conformity and Distinctiveness in a Global Institutional Framework: The Legitimation of Ontario Fine Wine. *Journal of Management Studies* 50 (4): 607-645.
- Ward M and Rhodes C (2014) *Small businesses and the UK economy*. London: The House of Commons Library.
- Watkins-Mathys L and Lowe S (2005) Small Business and Entrepreneurship Research: The Way Through Paradigm Incommensurability. *International Small Business Journal* 23 (6): 657-677.
- Webber D, Curry N and Plumridge A (2009) Business Productivity and Area Productivity in Rural England. *Regional Studies* 43 (5): 661-675.
- Weber M (1968) *Economy and society: An outline of interpretive sociology* (Roth G and Wittich C, Trans.). New York, NY: Bedminster Press.
- Welpe IM, Spörrle M, Grichnik D, Michl T and Audretsch DB (2012) Emotions and Opportunities: The Interplay of Opportunity Evaluation, Fear, Joy, and Anger as Antecedent of Entrepreneurial Exploitation. *Entrepreneurship Theory and Practice* 36 (1): 69-96.
- Welter F (2011) Contextualizing Entrepreneurship-Conceptual Challenges and Ways Forward. *Entrepreneurship Theory and Practice* 35 (1): 165-184.
- Welter F, Baker T, Audretsch DB and Gartner WB (2017) Everyday Entrepreneurship – A Call for Entrepreneurship Research to Embrace Entrepreneurial Diversity. *Entrepreneurship Theory and Practice* 41 (3): 311–321.
- Welter F and Gartner WB (2016) Advancing our research agenda for entrepreneurship and contexts. In: Welter F and Gartner WB (eds) *A Research Agenda for Entrepreneurship and Context*. Cheltenham: Edward Elgar, pp. 156-160.
- Welter F and Smallbone D (2006) Exploring the Role of Trust in Entrepreneurial Activity. *Entrepreneurship Theory and Practice* 30 (4): 465-475.
- Wernerfelt B (1984) A resource-based view of the firm. *Strategic Management Journal* 5 (2): 171-180.
- Wheelock J and Oughton E (1996) The Household as a Focus for Research. *Journal* of Economic Issues 30 (1): 143-159.
- Wiklund J, Davidsson P, Audretsch DB and Karlsson C (2011) The Future of Entrepreneurship Research. *Entrepreneurship Theory and Practice* 35 (1): 1-9.

Wiklund J, Nordqvist M, Hellerstedt K and Bird M (2013) Internal Versus External Ownership Transition in Family Firms: An Embeddedness Perspective. *Entrepreneurship Theory and Practice* 37 (6): 1319-1340.

Wiklund J and Shepherd DA (2009) The Effectiveness of Alliances and Acquisitions: The Role of Resource Combination Activities. *Entrepreneurship Theory and Practice* 33 (1): 193-212.

- Williams TA and Shepherd DA (2016) Victim entrepreneurs doing well by doing good: Venture creation and well-being in the aftermath of a resource shock. *Journal of Business Venturing* 31 (4): 365-387.
- Wilson J, Arshed N, Shaw E and Pret T (2017) Expanding the Domain of Festival Research: A Review and Research Agenda. *International Journal of Management Reviews* 19 (2): 195–213.
- Winborg J and Landström H (2001) Financial bootstrapping in small businesses: Examining small business managers' resource acquisition behaviors. *Journal* of Business Venturing 16 (3): 235-254.
- Winter M, Puspitawati H, Heck RKZ and Stafford K (1993) Time-management strategies used by households with home-based work. *Journal of Family and Economic Issues* 14 (1): 69-92.
- Witman Y, Smid GAC, Meurs PL and Willems DL (2011) Doctor in the lead: balancing between two worlds. *Organization* 18 (4): 477-495.
- Witt P (2004) Entrepreneurs' networks and the success of start-ups. Entrepreneurship & Regional Development 16 (5): 391-412.
- Wong S-S and Boh WF (2010) Leveraging the Ties of Others to Build a Reputation for Trustworthiness among their Peers. *Academy of Management Journal* 53 (1): 129-148.
- Wood G and Peel D (eds). (2015). Association of Business Schools Academic Journal Guide 2015. London: Association of Business Schools.
- Wright AL and Zammuto RF (2013) Creating opportunities for institutional entrepreneurship: The Colonel and the Cup in English County Cricket. *Journal of Business Venturing* 28 (1): 51-68.
- Wright M, Chrisman JJ, Chua JH and Steier LP (2014) Family Enterprise and Context. *Entrepreneurship Theory and Practice* 38 (6): 1247-1260.
- Wry T and York J (2017) An Identity Based Approach to Social Enterprise. Academy of Management Review 42 (3): 437-460.
- Yanow D (2006) Neither rigorous nor objective? In: Yanow D and Schwartz-Shea P (eds) *Interpretation and method: Empirical research methods and the interpretive turn*. Armonk, NY: M.E. Sharpe, pp. 67-88.
- Yli-Renko H, Autio E and Sapienza HJ (2001) Social capital, knowledge acquisition, and knowledge exploitation in young technology-based firms. *Strategic Management Journal* 22 (6-7): 587-613.
- Yu J, Zhou JX, Wang Y and Xi Y (2013) Rural Entrepreneurship in an Emerging Economy: Reading Institutional Perspectives from Entrepreneur Stories. *Journal of Small Business Management* 51 (2): 183-195.
- Zaheer A and Soda G (2009) Network Evolution: The Origins of Structural Holes. *Administrative Science Quarterly* 54 (1): 1-31.
- Zahra SA (2007) Contextualizing theory building in entrepreneurship research. Journal of Business Venturing 22 (3): 443-452.

- Zahra SA, Gedajlovic E, Neubaum DO and Shulman JM (2009) A typology of social entrepreneurs: Motives, search processes and ethical challenges. *Journal of Business Venturing* 24 (5): 519-532.
- Zahra SA, Hayton JC and Salvato C (2004) Entrepreneurship in Family vs. Non-Family Firms: A Resource-Based Analysis of the Effect of Organizational Culture. *Entrepreneurship Theory and Practice* 28 (4): 363-381.
- Zahra SA, Sapienza H and Davidsson P (2006) Entrepreneurship and dynamic capabilities: A review, model and research agenda. *Journal of Management Studies* 43 (4): 917–955.
- Zahra SA and Wright M (2016) Understanding the Social Role of Entrepreneurship. Journal of Management Studies 53 (4): 610-629.
- Zahra SA, Wright M and Abdelgawad SG (2014) Contextualization and the advancement of entrepreneurship research. *International Small Business Journal* 32 (5): 479–500.
- Zhang J, MacKenzie NG, Jones-Evans D and Huggins R (2016) Leveraging knowledge as a competitive asset? The intensity, performance and structure of universities' entrepreneurial knowledge exchange activities at a regional level. *Small Business Economics* 47 (3): 657–675.
- Zhang Y (2015) The contingent value of social resources: Entrepreneurs' use of debtfinancing sources in Western China. *Journal of Business Venturing* 30 (3): 390-406.
- Zhao EY, Fisher G, Lounsbury M and Miller D (2017) Optimal distinctiveness: Broadening the interface between institutional theory and strategic management. *Strategic Management Journal* 38 (1): 93-113.
- Zhao EY, Ishihara M and Lounsbury M (2013) Overcoming the Illegitimacy Discount: Cultural Entrepreneurship in the US Feature Film Industry. *Organization Studies* 34 (12): 1747-1776.
- Zietsma C, Groenewegen P, Logue DM and Hinings CR (2017) Field or Fields? Building the Scaffolding for Cumulation of Research on Institutional Fields. *Academy of Management Annals* 11 (1): 391-450.
- Zimmerman MA and Zeitz GJ (2002) Beyond Survival: Achieving New Venture Growth by Building Legitimacy. *Academy of Management Review* 27 (3): 414-431.
- Zott C and Huy QN (2007) How Entrepreneurs Use Symbolic Management to Acquire Resources. *Administrative Science Quarterly* 52 (1): 70-105.

Zou H, Chen X, Lam LWR and Liu X (2016) Psychological capital and conflict management in the entrepreneur–venture capitalist relationship in China: The entrepreneur perspective. *International Small Business Journal* 34 (4): 446-467.

Zukin S and DiMaggio P (1990) *Structures of capital: The social organization of the economy*. Cambridge: Cambridge University Press.

# **APPENDIX 1.** Interview Topic Guides

# Interview 1 Topic Guide

- 1. Tell me about your background.
- 2. Tell me about your business.

#### Interview 2 Topic Guide

- How have family members helped you to get where you are with your business?
  How have you been able to benefit from any of their contacts?
  - What other types of resources have you been able to access through your family?
- 2. Why do you choose to live and work in rural area?
  - What advantages do you gain from working out here?
  - How are you disadvantaged compared to makers in the city?
  - What type of business support groups do you access out here?
- 3. What type of connection do you feel with other crafts makers?- How do you help other makers?- How have you benefited from the help of others?
- 4. What happens if you are sick or unable to work?
- 5. What drives you to be self-employed?- How do you motivate yourself to work when you get into the studio?
- 6. What value do you place on the quality of photographs?- How do you take photographs of your products?

# Interview 3 Topic Guide

- 1. When does your family affect your work?
  - How have family members supported you when you have experienced difficulties?
  - Why do you value their emotional and moral support?
  - When do you need your family members' emotional support?
  - When have your family members held you back?
  - Emotionally, how have your family members affected you negatively?
  - How do you deal with family demands while at work?
- 2. What differences are there between your household and other family members? - In your household, who normally does the cooking, cleaning and washing?
- 3. What events in your life have had the biggest impact on your work?

- 4. How do you handle stress?
  - What do you do to relieve yourself from stress?
- 5. How do locals treat you and your business?- How do people treat you knowing that you were [not] born here?
- 6. Have you experienced differences in social class between you and your customers?
  - Do people make assumptions about you based on your appearance?
  - Have you experienced that people treat you differently because of your age?
- 7. Which groups of people do you interact with most?- How do you change your behaviour when you interact with different people?
- 8. What expectations do your customers have regarding your communications?
  - Do you share personal details with them?
  - How important is it for you to get on with your customers?
  - When do you highlight your experience or reputation to them?
- 9. Why do you do [Craft]?
  - What are your goals with your business?
  - How do your business goals change over time?
  - What are your long-term goals in life?
  - How does your business help you achieve these goals?
  - Do you grow tired of doing the same thing?
  - How do you tackle this issue?
  - What are your biggest worries with your business?
- 10. What do you think makes you successful?
  - How do you spot new business opportunities?
  - When and why do you change your craft practice over time?
- 11. How important is ethical practice and being environmentally friendly to you?
- 12. How are you planning to finance your retirement?- How does not having a pension scheme affect you?
- 13. How difficult would it be for others to acquire the skills to do your job?
- 14. Have you tried applying for government funding in the past?
  - What has prevented you from accessing government support?
  - What government support mechanisms do you feel are missing at the moment?

#### Interview 4 Topic Guide

- 1. To what degree do local traders accept you as one of theirs?
  - How do your relationships with them affect your business practices?
  - How does gossip around town impact you?
  - How about gossip within your creative community?

- How easy or difficult is it for you to fit in with other creatives in your field?
  How do you adapt your behaviour to fit in?
  - What are the unwritten rules of behaviours you have to adhere to?
  - How do you establish the legitimacy of your work?
- 3. What motivates you to tutor or train others?
- 4. When are you most likely to ask other people for help?
  - Why do you value the help of friends and other makers?
  - When have friends helped you increase your reputation?
  - What impact does the loss of other people's support have on you?
- 5. How, do you think, does your work affect your family?When does your business have a negative impact on your personal life?When does it have a positive impact?
- 6. How do you deal with having to reply to customers during the day while also crafting products?
- 7. How important is the long-term sustainability of your business to you?
- 8. How did your past education prepare you for running this craft business?
  - What type of further education do you think you could benefit from?
  - What training courses are currently missing in this area?
  - What type of new training course could benefit younger craft makers?
- 9. Tell me about creativity.
  - What drives your creativity?
  - How do you tap into your creativity?
  - When are you most successful at accessing your creativity?
- 10. How would you describe your surroundings?
- 11. How important is honesty to you?
- 12. Roughly how much of your work do you not to declare to the tax collectors?Why do you choose not to declare some of it?
- 13. How important is confidence to you?
  - How would you describe your confidence levels?
  - How do they change over time?
- 14. How do the emotions you feel affect your creative practice?
  - Which emotions affect your work the most?
  - How does being happy impact your work?
  - How do negative emotions, such as anger or grief, affect your work?
  - How do you deal with emotional problems at work?
  - When do emotional issues affect your work the most?
  - To what degree do you compartmentalise your emotions when you are at work?

#### APPENDIX 2. Interpretive Summary for Interview 1 with Julie

After finishing her bachelor's degree in English and Fine Art, Julie completed a postgraduate certificate in Education. This enabled her to teach English and Drama in her hometown, but upon divorcing her first husband, she moved and began working in setting up community initiatives. These experiences seem to have been critical in framing Julie's mentorship of nascent craftspeople and her current practice of running classes and workshops alongside making and selling her craft products.

Another aspect of her background, which appears to have figured largely in driving her business, was being taught how to sew and embroider by her maternal grandmother. Additionally, Julie was inspired by her mother, who taught her how to draw and whom Julie greatly adored and admired. She sees her formative years as the beginning of her path towards creative entrepreneurship, remembering that she was always making things and trying to sell them to her friends and relatives.

Julie went back to teaching when she had children, but decided to start her own business when she came close to retirement. Drawing on her contacts from working at the University, she attended free creative business classes which helped guide her in building up her business. Her years in the creative sector also enabled Julie to make many friends who help her out. For example, she currently leases her studio from a creative friend who provides the property in exchange for Julie and her husband fixing up the space. This shows how influential friends and relatives are in Julie's life.

Perhaps as a result of her academic background, Julie prefers utilising published media and workshops more than shows, exhibits or fairs to promote her business and products. In addition to writing articles for craft magazines, she has published two books on her craft, which not only yield profits themselves, but also increase demand for her kits and craft products. In contrast, her experiences with fairs have been largely negative, leading her to be cautious about which ones she attends. She seems to appreciate the fairs she does attend more for the exposure she receives and contacts she makes than for direct sales.

Julie has a strong independent streak that, along with being a main motivator for self-employment, also drives her desire to be self-sufficient: she takes no grants and borrows no loans, because she does not want to jump through anyone's hoops. When publishing her books, she received advice that she should approach a big publisher, but chose to self-publish because she did not want to dance to their tune. In other words, being in control is of high importance to Julie. This desire for control also prevents her from making use of friends' offers of help with production, even during busy periods.

The only people she admits to relying upon are her husband, son and daughter. Her son in particular is critical in helping her with her website and publishing, her daughter helps her with shows and exhibitions and her husband helps with fixing up her studio, managing her accounts, shipping out her products and running the household, so she can work on her craft. She divulged that her family members have had and continue to have many personal issues. Crafting products is thus not only a creative outlet for her, but also helps her deal with the related difficulties. In addition to helping herself, it is clear that Julie believes that another great benefit of her business is that it provides a job for her son.

Julie's passion for spreading the wealth through her craft is visible in the degree to which she shares her skills, experiences and advice with fellow creatives. Drawing on her contacts from teaching at University, she founded a network of creatives that is geared towards learning both craft technique and idea generation. Many practicing craftspeople and artists participate in running seminars, where they share lessons learnt from their experiences. While Julie passed on the responsibility of running this collective, she still provides help whenever asked because she feels that her experience is valuable and she wants to share it with others. She also believes that crafts need to be perpetuated for generations to come and wants to do her part for that cause.

Reputation is highly important for Julie and she considers her reputation to be synonymous with the quality of her products. This is another place where her need for control comes through. She recounts, for example, that when she began making dolls, people suggested she use bright colours. Julie followed this suggestion at first, but then scrapped these colours and started over because she did not like them. She believes her personal style to be essential to the integrity and quality of her designs and what draws customers to her products. She takes much pride in being complemented on her products and being considered the best at what she does. Living up to other people's expectations is therefore very important to Julie.

Despite receiving positive feedback for her work, Julie suffers from lack of confidence, which can hinder her. When discussing the possibility of acquiring certificates to prove her reputation to customers, she explained that she decided not to apply to be a master craftsman, because she did not want to be rejected. This fear of failure thus impedes Julie's ability to communicate her expertise to potential clients, especially online customers who are not able to see the quality of her work first-hand.

Success for Julie entails making enough money to be sustainable, so that she knows she can carry on without worrying too much about it. Importantly, Julie does not want her business to become so big that she loses control over it. She wants to maintain a high quality of live for her family. Here again, Julie's reveals how closely her business is tied to her family: they not only motivate her to pursue her business and facilitate her business processes, but they also guide her business aims.